1111 Broadway, Suite 800, Oakland, CA 94607



Alameda CTC Commission Agenda Thursday, May 27, 2021, 2:00 p.m.

510.208.7400

www.AlamedaCTC.ora

Due to the statewide stay at home order and the Alameda County Shelter in Place Order, and pursuant to the Executive Order issued by Governor Gavin Newsom (Executive Order N-29-20), the Commission will not be convening at its Commission Room but will instead move to a remote meeting.

Members of the public wishing to submit a public comment may do so by emailing the Clerk of the Commission at <u>vlee@alamedactc.org</u> by 5:00 p.m. the day before the scheduled meeting. Submitted comments will be read aloud to the Commission and those listening telephonically or electronically; if the comments are more than three minutes in length the comments will be summarized. Members of the public may also make comments during the meeting by using Zoom's "Raise Hand" feature on their phone, tablet or other device during the relevant agenda item, and waiting to be recognized by the Chair. If calling into the meeting from a telephone, you can use "Star (*) 9" to raise/ lower your hand. Comments will generally be limited to three minutes in length, or at the discretion of the Chair.

Chair:	Pauline Russo Cutter, Mayor City of San Leandro	Executive Director:	Tess Lengyel
Vice Chair:	John Bauters, Councilmember City of Emeryville	Clerk of the Commission:	<u>Vanessa Lee</u>

Location Information:

Virtual Meeting Information:	https://zoom.us/j/99173286551?pwd=cU9PbDdYSTVKV0tlcHdaU1ptb1lBUT09 Webinar ID: 991 7328 6551 Password: 693080
For Public Access Dial-in Information:	1 (669) 900 6833 Webinar ID: 991 7328 6551 Password: 693080

To request accommodation or assistance to participate in this meeting, please contact Vanessa Lee, the Clerk of the Commission, at least 48 hours prior to the meeting date at: <u>vlee@alamedactc.org</u>

Meeting Agenda

1. Call to Order

2. Roll Call

3. Public Comment

4. Chair and Vice Chair Report

5. Executive Director Report

6.	Con	sent Calendar	Page/	Action
		neda CTC standing committees approved all action items on the sent calendar, except Item 6.1 and 6.17.		
	6.1.	Approve April 25, 2021 Commission Meeting Minutes	1	А
	6.2.	FY2020-21 Third Quarter Report of Claims Acted Upon Under the Government Claims Act	9	I
	6.3.	2020 Alameda CTC Annual Report Update	11	Ι
	6.4.	Approve an update to Independent Watchdog Committee Bylaws	13	А
	6.5.	Approve the Alameda CTC FY2020-21 Third Quarter Consolidated Financial Report	25	A
	6.6.	Approve the Alameda CTC FY2020-21 Third Quarter Investment Report	31	А
	6.7.	Approve an Update to the Alameda CTC Investment Policy	47	А
	6.8.	Approve and adopt restatements of the Alameda CTC Cafeteria Plan	65	А
	6.9.	Approve Cost Reimbursement Agreement with the Tri-Valley San Joaquin Valley Regional Rail Authority	167	A
	6.10	. <u>Metropolitan Transportation Commission's Safe and Seamless Mobility</u> <u>Quick-Strike Program Update</u>	169	Ι
	6.11	Approve Alameda County 2021 Mid-Cycle Augmentation of the 2020 State Transportation Improvement Program	193	A
	6.12	Approve actions associated with allocation of Regional Measure 2 funds for Livermore Amador Valley Transit Authority's Rapid Bus Stop Improvement project	207	A
	6.13	Approve actions necessary to initiate and complete the preparation of Plans, Specifications, and Estimate (PS&E) and Right of Way Phases for Ready to List package for the Oakland Alameda Access Project	243	A
	6.14	Congestion Management Program (CMP): Summary of the Alameda CTC's Review and Comments on Environmental Documents and General Plan Amendments	249	Ι
	6.15	. <u>Approve Release of a Request for Proposals (RFP) for the I-580</u> <u>Sustainable Corridor Strategy</u>	251	A
	6.16	Alameda County Safe Routes to Schools Program Update and Approve Contract Amendments and Funding Allocation for the Safe Routes to Schools Program	257	Ι
	6.17	. <u>Community Advisory Committee Appointments</u>	263	А

7. Regular Matters

The following items were approved at Alameda CTC Committee meetings in May, unless otherwise noted in the recommendations.

7.1.	<u>Plan Bay Area 2050 Update</u>	269	I
7.2.	Approve Measure B and Measure BB Sales Tax Budget Update for FY2020-21	271	А
7.3.	Approve the Alameda CTC FY2021-22 Proposed Budget	273	А
7.4.	Federal, state, regional, and local legislative activities update	281	A/I
Com	nmission Member Reports		

9. Adjournment

Next Meeting: June 24, 2021

Notes:

8.

- All items on the agenda are subject to action and/or change by the Commission.
- To comment on an item not on the agenda, submit an email to the clerk or use the Raise Hand feature or if you are calling by telephone press *9 prior to or during the Public Comment section of the agenda. Generally public comments will be limited to 3 minutes.
- Call 510.208.7450 (Voice) or 1.800.855.7100 (TTY) five days in advance to request a sign-language interpreter.
- If information is needed in another language, contact 510.208.7400.
- Call 510.208.7400 48 hours in advance to request accommodation or assistance at this meeting.
- Meeting agendas and staff reports are available on the website calendar.



1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

Alameda CTC Schedule of Upcoming Meetings June 2021

Commission and Committee Meetings

Time	Description	Date	
10:00 a.m.	Programs and Projects Committee (PPC)		
11:30 a.m.	Planning, Policy and Legislation Committee (PPLC)	June 14, 2021	
1:00 p.m.	Alameda CTC Audit Committee		
2:00 p.m.	Alameda CTC Commission Meeting	June 24, 2021	

Advisory Committee Meetings

1:30 p.m.	Alameda County Technical Advisory Committee (ACTAC)	June 10, 2021
1:30 p.m.	Paratransit Advisory and Planning Committee	June 28, 2021

Due to the statewide stay at home order and the Alameda County Shelter in Place Order, and pursuant to the Executive Order issued by Governor Gavin Newsom (Executive Order N-29-20), the Commission will not be convening at its Commission Room but will instead move to a remote meeting.

Meeting materials, directions and parking information are all available on the <u>Alameda CTC website</u>. Meetings subject to change.

Commission Chair Mayor Pauline Russo Cutter City of San Leandro

Commission Vice Chair Councilmember John Bauters City of Emeryville

AC Transit Board President Elsa Ortiz

Alameda County

Supervisor David Haubert, District 1 Supervisor Richard Valle, District 2 Supervisor Wilma Chan, District 3 Supervisor Nate Miley, District 4 Supervisor Keith Carson, District 5

BART Vice President Rebecca Saltzman

City of Alameda Mayor Marilyn Ezzy Ashcraft

City of Albany Councilmember Rochelle Nason

City of Berkeley Councilmember Lori Droste

City of Dublin Mayor Melissa Hernandez

City of Fremont Mayor Lily Mei

City of Hayward Mayor Barbara Halliday

City of Livermore Mayor Bob Woerner

City of Newark Councilmember Luis Freitas

City of Oakland Councilmember At-Large Rebecca Kaplan Councilmember Sheng Thao

City of Piedmont Councilmember Jen Cavenaugh

City of Pleasanton Mayor Karla Brown

City of Union City Mayor Carol Dutra-Vernaci

Executive Director Tess Lengyel



1111 Broadway, Suite 800, Oakland, CA 94607 •

PH: (510) 208-7400

1. Call to Order

2. Roll Call

A roll call was conducted. All members were present with the exception of Commissioners Mei, Miley, Thao, and Valle.

Commissioner Duncan attended as an alternate for Commissioner Dutra-Vernaci. Commissioner McQuaid attended as an alternate for Commissioner Carson.

Subsequent to the roll call:

Commissioner Thao arrived during item 4. Commissioner Mei and Miley arrived during item 5.

3. Public Comment

A public comment was made by Robert Daulton, from Niles for Environmentally Safe Trains, regarding items 6.3 and 6.4 on the Consent Calendar. He requested that the Commission pause the current Environmental Impact Report (EIR) process for the South Bay Connect project, evaluate alternatives for the project, and rethink the scoping process to include the concerns of his community.

4. Chair and Vice Chair Report

Chair Cutter stated that Alameda CTC continues to deliver projects and implement programs despite the pandemic. She noted that the Commission will continue to do its part in the economic recovery by getting projects into construction and keeping a continued focus on project development and program delivery for ongoing investments throughout the County. Chair Cutter announced that Alameda CTC is looking forward to celebrating Bike Month in May and stated that as part of Alameda CTC's Bike Month celebration, the Alameda County Safe Routes to Schools Program (SR2S) is planning a remote encouragement event entitled Bike to the Moon Week.

Chair Cutter noted that earlier in the month, both she and Vice Chair Bauters, along with Alameda CTC Executive Director, Tess Lengyel, and staff held legislative meetings with the State Transportation Legislative Policy Committee Chairs to discuss legislative priorities for Alameda CTC. Alameda CTC priorities were shared with Senator Lena A. Gonzales, Chair of the Senate Transportation Committee, and Assemblywoman Laura Friedman, Chair of the Assembly Transportation Committee who is the author of AB 43. The group also met with leaders of the West Oakland Environmental Indicators Project, Ms. Margaret Gordon and Brian Beveridge, to discuss the 7th Street Grade Separation project. Commissioner Cutter shared a video of herself, Vice Chair Bauters, and Ms. Lengyel riding their bikes on the Doyle Street Bike & Ped path in the City of Emeryville and she noted that this is one of the projects funded by the COVID-19 Rapid Response Bicycle and Pedestrian Grant Program that was implemented in March 2021.

Vice Chair Bauters noted that the City of Emeryville deliberately chose Doyle Street as the project to be implemented through the quick-build grant program, which allowed Emeryville to fulfill a permanent connector to the greenway. He thanked Bike East Bay for a story in their newsletter about the Doyle Street Bike Path. Vice Chair Bauters proposed a friendly challenge to support the SR2S Bike to the Moon event sponsored by Alameda CTC by initiating a Commissioner Bike to the Moon challenge that encourages the Commissioners to also log moon miles during the week. He concluded by providing instructions regarding technology procedures including instructions on administering public comments during the meeting.

5. Executive Director Report

Tess Lengyel congratulated Chair Cutter and Vice Chair Bauters for the Doyle Street quick-build project. She noted that it is both a safe project and a beautiful corridor to ride.

Ms. Lengyel noted that Alameda CTC is fully committed to continuing to support the promise to the voters for high-quality planning and project delivery and for helping with economic recovery and access. Ms. Lengyel highlighted the progress and key efforts made by staff on various projects and programs and stated that the key initiatives set last year included safety, equity, clean and sustainable transportation, and accountability. Ms. Lengyel noted that Alameda CTC invested \$900,000 in the guick-build program and noted that both City and County partners matched that amount, which resulted in approximately \$1.8 million worth of bicycle and pedestrian investments. She announced that on May 20, 2021, Alameda CTC and Caltrans are jointly sponsoring a virtual groundbreaking celebration to announce the start of construction on the I-80/Gilman Street Interchange Improvement Project. Ms. Lengyel also noted that Alameda CTC staff has participated in diversity, equity and inclusion training and a draft statement was developed that she will bring before the Commission. She stated that Alameda CTC submitted a request to Congressman Swalwell to support a study that integrates AC Transit, LAVTA, and Contra Costa County Transit to look at hydrogen fuel uses and workforce development. Lastly, Ms. Lengyel noted that Assemblymember Friedman stated that she is interested in visiting the Bay Area and riding all transit systems in the area including in Alameda County.

Commissioner Saltzman stated that she would be happy to work with Alameda CTC to coordinate a ride of the County's public transit systems at Assemblymember Friedman's request.

Commissioner Kaplan stated that she appreciates the outreach to Assemblymember Friedman and noted that AB 43 will allow local jurisdictions the flexibility in setting their speed limits. She noted that the Commission should support this bill. Commissioner Ezzy Ashcraft stated that she wants to extend an invitation to Assemblymember Friedman to ride from the Seaplane Lagoon Ferry terminal in Alameda.

Commissioner Halliday commented on the quick-build bicycle and pedestrian grant program, noting that she is working with Bike East Bay to make the City of Hayward more bicycle-friendly. She stated that Hayward recently implemented a bicycle lane on a busy street that was not received favorably by the community so she wanted to know if there are available resources to help Hayward in this effort.

Commissioners Bauters stated that he is willing to speak with the City of Hayward staff to assist with creating an enhanced bicycle community and he suggested that the Commission receive a technical presentation regarding best practices for advancing bicycle and pedestrian initiatives.

Commissioner Mei commented she attended the bike safety class at Bike East Bay. She noted how cycling has increased during the time of COVID and it would be helpful to have additional talking points around active transportation to share with her community.

A public comment was made by Dave Campbell, Bike East Bay, who stated that the Emeryville quick build project was a good project. He noted that Hayward is working to make Patrick Avenue in South Hayward a safer bicycle street for a protected bike lane. He noted that it is not uncommon for a city implementing its first bicycle project to struggle and it will become easier with the second facility.

Ms. Lengyel stated that in May Alameda CTC will give a technical presentation to the Alameda County Technical Advisory Committee on a safe systems approach, which looks at how jurisdictions can incorporate safety in all aspects of street designs for bicycle, pedestrian, transit and motorist needs. She noted that if there is an interest in having safety information shared at the full Commission meeting, staff will do that as well.

6. Consent Calendar

- 6.1. Approve March 25, 2021, Commission Meeting Minutes
- 6.2. I-580 Express Lanes Operations Update
- 6.3. Southern Alameda County Rail Study (SoCo Rail) Update
- 6.4. South Bay Connect Project Update
- 6.5. Approve Measure B, Measure BB and Vehicle Registration Fee Programs Update and Interim Policy Updates
- 6.6. Approve Contract Amendment for E. 14th St./Mission Blvd. and Fremont Blvd. Multimodal Corridor Project
- 6.7. Approve actions associated with the Construction Phase of the I-80 Gilman Interchange Improvements Project
- 6.8. Congestion Management Program (CMP): Summary of the Alameda CTC's Review and Comments on Environmental Documents and General Plan Amendments

- **6.9.** Approve the Professional Services Agreement (A21-0025) with Nelson\Nygaard Consulting Associates for Paratransit Coordination Services
- 6.10. Approve administrative amendment to Alameda CTC agreement (A16-0027) in support of the Alameda CTC Affordable Student Transit Pass Program (STPP)
- 6.11. Approve the State Transit Assistance (STA) County Block Grant Program Distribution Formula for Fiscal Years 2021-22 and 2022-23

Commissioner Halliday moved to approve all items on the consent calendar with the exception of Item 6.4, which she pulled for further consideration. Commissioner Ortiz seconded the motion. The motion passed with the following roll call votes:

Yes:	Bauters, Brown, Cavenaugh, Cutter, Droste, Duncan, Ezzy Ashcraft,
	Freitas, Halliday, Haubert, Hernandez, Kaplan, McQuaid, Mei, Miley,
	Nason, Ortiz, Saltzman, Thao, Woerner
No:	None
Abstain:	None
Absent:	Chan, Valle

In regard to Item 6.4, Commissioner Halliday stated that she is concerned about the South Bay Connect Project, which has a lot of opposition from the public. She noted that the letter from the Mayors in that corridor is attached to the staff report and documents the cities concerns regarding the EIR. She noted that the full Commission has not had an opportunity to discuss the pros and cons of this project.

Commissioner Bauters noted that Alameda CTC does not have a role in the determination of the EIR; however, the agency has funding earmarked for the project. He suggested that several members of Alameda CTC could meet to address questions raised related to the project and Alameda CTC's role.

Commissioner Mei stated that this item was presented to the Multi-Modal Committee at the April 12, 2021 meeting. At the meeting, she encouraged Capital Corridor to have more interaction and outreach to the cities that are directly impacted by the project.

Commissioner Saltzman suggested staff bring updates for item 6.4 to the full commission for discussion. Ms. Lengyel stated that we will be sure to bring it to the full Commission at a future time.

Neal Parish, legal counsel at Wendel Rosen, noted that Item 6.4 was an informational item, so a separate vote on the item was not needed.

7. Community Advisory Committee Written Reports

7.1. Paratransit Advisory and Planning Committee Summary Minutes

Tess Lengyel stated that the written report was included in the packet.

8. Planning, Policy and Legislation Committee

8.1. Affordable Student Transit Pass Program Update

Tess Lengyel stated that this presentation is an update on how the agency will move forward with the Affordable Student Transit Pass Program (STPP) in the coming year, and in particular recognizing the impact that COVID-19 has had on Alameda CTC's transit partners. Ms. Lengyel introduced Kate Lefkowitz, who provided the update.

Ms. Lefkowitz noted that the report included an update on the STPP 2019-2020 evaluation report, program implementation for the current 2020-2021 school year, and STPP program plans for fall 2021. Given the uncertainties for school districts in Alameda County during the COVID-19 pandemic, the STPP has been actively working with schools to support students and families, modify program offerings, and proactively prepare for the evolving developments of the school year. The program now serves 14 school districts and 84 schools within Alameda County. Key programmatic changes that have been implemented due to COVID-19 impacts include the transition to an online application for the STPP. Ms. Lefkowitz noted that as a result of significant COVID-19 impacts on school districts and transit agencies in Alameda County, the STPP team recommends continuing the program as previously approved and not expanding further in fall 2021. The STPP team will continue to implement the Commission-approved program and look to expand to the remaining ~56 eligible schools in the county by the 2023-2024 school year. An update on any future expansions would come before the Commission in 2022.

Commissioner Brown asked if there is a photo of each student on the STPP card. Ms. Lefkowitz stated that the STPP card is a clipper card. If the student loses the card it will be deactivated and replaced.

Commissioner Brown asked what is the process to prevent a person from selling their card or allowing someone else to use it. Commissioner Cutter stated that there have not been reports of fraud or misuse in the program.

Commissioner Ortiz stated that the STPP clipper card is also used by students to go to work, the library, attend afterschool activities, and weekend jobs.

Commissioner Ezzy Ashcraft stated that the continuation school students in Alameda participate in the program, which is a great way for students to get to afterschool programs and jobs and cuts down on the need for parents to drop off students.

8.2. Federal, state, regional, and local legislative activities update

Tess Lengyel stated that staff will provide an update on the federal and state activities and focus on several bill positions. Ms. Lengyel introduced Carolyn Clevenger and Maisha Everhart to provide the updates. Ms. Clevenger gave an overview on the Climate Action Plan for Transportation Infrastructure (CAPTI), which will influence future state transportation funding, including funding programs that are derived from Senate Bill (SB) 1 funds. She stated that this effort is led by the California State Transportation Agency (CaISTA). Ms. Clevenger noted that CAPTI is a statewide high-level policy document that is being developed by CalSTA to establish a framework detailing how the state can advance the Governor's executive orders related to climate change. The comment period was extended to May 19, 2021, with the final plan submitted to the governor and legislature this summer.

Maisha Everhart stated that staff is in the process of reviewing over 2,000 bills that were submitted and evaluating them to see if they are in alignment with Alameda CTC's Legislative Program. The following bills will support the Commission's goal of enhancing multimodal transportation and safety:

AB 43 (Friedman) - Support	AB 550 (Chiu) - Support
AB 455 (Bonta) - Support and seek	AB 917 (Bloom) - Support
amendment	SB 18 (Skinner) - Support

Commissioner Kaplan thanked staff for support of AB 43 and AB 550 and she noted that the City of Oakland endorsed AB 550.

Commissioner Halliday expressed her disappointment in AB 550 and asked if Alameda CTC could get results from the original pilot program. She asked if Alameda CTC can lobby for a stronger bill and eliminate some of the ways AB 550 has been amended to weaken the effects.

Ms. Lengyel explained the distinction between AB 43 and AB 550. She noted that AB 43 is Assemblymember Friedman's bill, which will enable both Caltrans and local jurisdictions to have more authority to reduce speed limits. Ms. Lengyel stated that AB 550 is the pilot program and the jurisdictions that are listed in the bill currently are Los Angeles, Oakland, San Jose, the city and county of San Francisco, and two other cities in Southern California. She noted that the bill addresses automated speed enforcement and the pilot program is until 2027 and a report will be generated once the pilot is complete.

Commissioner Ortiz noted that AC Transit is a co-sponsor of AB 917 and it was voted on by their Privacy Committee on April 22, 2021.

Commissioner Bauters shared that he spoke with Assemblymember Friedman and asked her what are the opposition points to AB 550. She noted that there is opposition to this bill for various reasons, which is why it was amended. Assemblymember Friedman noted that if the Commissioners or cities are in support of AB 550 to call the Transportation Committee by April 26, 2021, and express their concerns. He noted that the American Civil Liberties Union is opposed to this bill because of language regarding data privacy.

Commissioner Cutter requested Alameda CTC staff to provide the Commissioners with the Assembly Transportation Committee contact information so they may reach out stating their support position. Commissioner Ortiz moved to approve the recommended bill positions. Commissioner Kaplan seconded the motion. The motion passed with the following roll call votes:

Yes:	Bauters, Brown, Cavenaugh, Cutter, Droste, Duncan, Ezzy Ashcraft,
	Freitas, Halliday, Haubert, Hernandez, Kaplan, McQuaid, Mei, Miley,
	Nason, Ortiz, Saltzman, Thao, Woerner
No:	None
Abstain:	None
Absent:	Chan, Valle

9. Zero-Emission Drayage Truck and Infrastructure Pilot Program

9.1. Approve Grant Matching Funds for Hydrogen Fuel Drayage Trucks and Fueling Pilot Program

Ms. Lengyel stated that Vivek Bhat will present this item and Jaimie Levin from the Center for Transportation and the Environment (CTE) is available to answer questions. Mr. Bhat recommended that the Commission approve the following actions associated with the Hydrogen Fuel Drayage Trucks and Fueling Pilot Program:

- 1. Approve allocation of \$3.64 million Measure BB Countywide Freight Corridors funds (TEP-27) as grant matching funds, and
- 2. Authorize Executive Director or designee to execute all necessary agreements.

Mr. Bhat stated that over the past several months, Alameda CTC staff has been working closely with the CTE on a Hydrogen Fuel Drayage Trucks and Fueling Pilot Program grant application that involves deploying 30 fuel cell trucks in Northern California. The project will also install a high-capacity hydrogen fueling station near the Port of Oakland. This project will provide Alameda County, the City of Oakland, and the West Oakland neighborhood the opportunity to take on a leadership role in advancing zero-emission truck technology to reduce emissions that are impacting disadvantaged communities. In February 2021, the CTE applied for a \$21.8 million statewide competitive grant administered jointly by the California Air Resources Board (CARB) and the California Energy Commission. The application included a local match of \$7 million with the Bay Area Air Quality Management District (BAAQMD) and Alameda CTC as funding partners.

Commissioner Kaplan expressed her appreciation for the work done on this project. She noted that it will benefit the region to cut the emissions from trucks.

Commissioner Cutter commented that BAAQMD talked about this item at their Mobile Source Committee meeting, and she requested Mr. Levin to provide comments on this effort.

Mr. Levin thanked the Commission and Alameda CTC staff for supporting this effort. He noted that having the agency's support was critical to CTE winning a much larger grant from CARB. He noted that this is the first commitment of state funds at this scale in Northern California and it puts Alameda County and Oakland at the forefront of this developing technology. He stated that another aspect is workforce training. CTE is partnering with NorCal Kenworth, which is a major truck service repair center in San Leandro for workforce training for local technicians and mechanics.

Commissioner Kaplan moved to approve the item. Commissioner Halliday seconded the motion. The motion passed with the following roll call votes:

 Yes: Bauters, Brown, Cavenaugh, Cutter, Droste, Duncan, Ezzy Ashcraft, Freitas, Halliday, Haubert, Hernandez, Kaplan, McQuaid, Mei, Miley, Nason, Ortiz, Saltzman, Woerner
No: None
Abstain: None
Absent: Chan, Thao, Valle

10. Commission Member Reports

Commissioner Ortiz invited Commissioners to join AC Transit on Transit Talks. April 26 through May 31, 2021.

11. Adjournment

The next meeting is Thursday, May 27, 2021, at 2:00 p.m.



Memorandum

510.208.7400

1111 Broadway, Suite 800, Oakland, CA 94607

DATE: Mary 20, 2021 TO: Alameda County Transportation Commission FROM Patricia Reavey, Deputy Executive Director of Finance and Administration SUBJECT: FY2020-21 Third Quarter Report of Claims Acted Upon Under the Government Claims Act

Recommendation

This item is to provide the Commission with an update on the FY2020-21 Third Quarter Report of Claims Acted upon under the Government Claims Act. This item is for information only.

Summary

There were no actions taken by staff under the Government Claims Act during the third quarter of FY2020-21.

Background

Tort claims against Alameda CTC and other California government entities are governed by the Government Claims Act (Act). The Act allows the Commission to delegate authority to an agency employee to review, reject, allow, settle, or compromise tort claims pursuant to a resolution adopted by the Commission. If the authority is delegated to an employee, that employee can only reject claims or allow, settle, or compromise claims \$50,000 or less. The decision to allow, settle, or compromise claims over \$50,000 must go before the Commission for review and approval.

California Government Code section 935.4 states:

"A charter provision, or a local public entity by ordinance or resolution, may authorize an employee of the local public entity to perform those functions of the governing body of the public entity under this part that are prescribed by the local public entity, but only a charter provision may authorize that employee to allow, compromise, or settle a claim against the local public entity if the amount to be paid pursuant to the allowance, compromise or

settlement exceeds fifty thousand dollars (\$50,000). A Charter provision, ordinance, or resolution may provide that, upon the written order of that employee, the auditor or other fiscal officer of the local public entity shall cause a warrant to be issued upon the treasury of the local public entity in the amount for which a claim has been allowed, compromised, or settled."

On June 30, 2016, the Commission adopted a resolution which authorized the Executive Director to reject claims or allow, settle, or compromise claims up to and including \$50,000.

There have only been a handful of small claims filed against Alameda CTC and its predecessors over the years, and many of these claims were erroneously filed, and should have been filed with other agencies. As staff moves forward with the implementation of Measure BB, Alameda CTC may experience an increase in claims against the agency as Alameda CTC puts more projects on the streets and highways of Alameda County and as Alameda CTC's name is recognized as a funding agency on these projects. Staff works directly with the agency's insurance provider, the Special District Risk Management Authority (SDRMA), when claims are received so that responsibility may be determined promptly and they might be resolved expediently or referred to the appropriate agency. This saves Alameda CTC money because when working with the SDRMA directly, much of the legal costs to address these claims are covered by insurance.

Fiscal Impact: There is no fiscal impact. This is an information item only.



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Patricia Reavey, Deputy Executive Director of Finance and Administration Carolyn Clevenger, Deputy Executive Director of Planning and Policy
SUBJECT:	2020 Alameda CTC Annual Report Update

Recommendation

This item is to inform the Commission that the 2020 Alameda CTC Annual Report has been prepared and completed for distribution. This item is for information only.

Summary

Alameda CTC prepares an annual report each year, as required in the Public Utilities Code section 180111, on progress made to achieve the objective of improving transportation in Alameda County. The 2020 Annual Report highlights key transportation programs and projects that Alameda CTC plans, funds, and delivers and includes financial information for FY 2019-20.

Many of these transportation investments are funded largely through local, voter-approved Measure B and Measure BB sales tax dollars and local, voter-approved Vehicle Registration Fee (VRF) funds. The annual report includes financial information related to Measure B and Measure BB revenues and expenditures for the year ended June 30, 2020, as well as information related to the VRF Program, including the total net VRF revenue from the start of the program, and revenues and expenditures through June 30, 2020.

Fiscal Impact: There is no fiscal impact. This is an information item only.

Attachment:

A. Draft 2020 Alameda CTC Annual Report

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:Mary 20, 2021TO:Alameda County Transportation CommissionFROMPatricia Reavey, Deputy Executive Director of Finance
and AdministrationSUBJECT:Approve an update to Independent Watchdog Committee Bylaws

Recommendation

Staff is recommending approval of an update to the Independent Watchdog Committee (IWC) bylaws, attached in redline, as proposed by the IWC.

Summary

Alameda CTC's Independent Watchdog Committee (IWC) reviewed the bylaws of their committee during their March 8, 2021 meeting and have proposed some minor edits. Staff has modified the currently adopted IWC bylaws in the attached redlined version to incorporate the edits suggested by the IWC.

Per Article 5.1 of the Commission's Administrative Code, the Commission is responsible for adopting and amending the bylaws for the IWC, as deemed necessary. Staff does not foresee any issues with the edits proposed by the IWC, which are administrative in nature, and recommends approval by the Commission of the updates to the IWC bylaws as outlined in Attachment A.

Background

The Independent Watchdog Committee (IWC) as defined in the 2014 Transportation Expenditure Plan (TEP), is the same committee as the Citizens Watchdog Committee, as defined in the 2000 TEP. The required composition of the IWC is defined in the 2000 and 2014 TEPs. The IWC, is a 17-member committee that reports directly to the public and is charged with reviewing all Measure B expenditures and reviewing Measure BB expenditures and performance measures, as appropriate. The members are Alameda County residents who are not elected officials at any level of government, nor individuals in a position to benefit personally in any way from the sales taxes. Fiscal Impact: There is no fiscal impact related to the approval of this item.

Attachment

A. Independent Watchdog Committee Bylaws (redlined)

www.AlamedaCTC.org

510.208.7400



1111 Broadway, Suite 800, Oakland, CA 94607

Independent Watchdog Committee Bylaws

Article 1: Definitions

1.1 2000 Transportation Expenditure Plan. The plan for expending transportation sales tax (Measure B) funds, presented to the voters in 2000, and implemented in 2002.

1.2 2014 Transportation Expenditure Plan. The plan for expending transportation sales tax (Measure BB) funds, presented to the voters in 2014, and implemented in 2015.

1.3 Agency. A business or government organization established to provide a particular service.

1.4 Alameda County Transportation Commission (Alameda CTC). Alameda CTC is a joint powers authority resulting from the merger of the Alameda County Congestion Management Agency ("ACCMA") and the Alameda County Transportation Improvement Authority ("ACTIA"). The 22-member Alameda CTC Commission ("Commission") is comprised of the following representatives:

- 1.4.1 All five Alameda County Supervisors.
- 1.4.2 Two City of Oakland representatives.

1.4.3 One representative from each of the other 13 incorporated cities in Alameda County.

1.4.4 A representative from Alameda-Contra Costa Transit District ("AC Transit").

1.4.5 A representative from San Francisco Bay Area Rapid Transit District

("BART").

1.5 Alameda County Transportation Improvement Authority (ACTIA). The governmental agency previously responsible for the implementation of the Measure B half-cent transportation sales tax in Alameda County, as approved by voters in 2000 and implemented in 2002. Alameda CTC has now assumed responsibility for administration of the sales tax.

1.6 Appointing Party. A person or group designated to appoint committee members.

1.7 At-Large Member. One of the 10 Independent Watchdog Committee (IWC) members representing supervisorial districts as described in Section 3.1.1 below.

1.8 Bicycle and Pedestrian Advisory Committee (BPAC). The Alameda CTC Committee that involves interested community members in the Alameda CTC's policy, planning, and implementation efforts related to bicycling and walking.

1.9 Brown Act. California's open meeting law, the Ralph M. Brown Act, California Government Code, Sections 54950 et seq.

1.10 Expenditures. Costs incurred and paid for with funds generated from the Measure B and Measure BB sales taxes.

1.11 Fiscal Year. July 1 through June 30.

1.12 Independent Watchdog Committee (IWC or "Committee"). The Alameda CTC Committee of individuals created by the Commission as required by Measure BB. This Committee was originally created by the ACTIA Board and called the Citizens Watchdog Committee as required by Measure B, and was continued by the Commission subsequent to the passage of Measure BB as the Independent Watchdog Committee. The Committee has the same composition as the Citizens Watchdog Committee required by Measure B. The Committee reports directly to the public and has the responsibility of reviewing all Measure B expenditures and reviewing and overseeing all Measure BB expenditures and performance measures of the agency, as appropriate. IWC members are Alameda County residents who are not elected officials at any level of government, nor individuals in a position to benefit personally in any way from the sales tax.

1.13 Local Newspapers. Periodical publications typically published weekly or daily that serves a city, cities or unincorporated communities within Alameda County, whereby the contents are reasonably accessible to the public. On-line publications of these periodicals are included in this definition.

1.14 Measure B. The measure approved by the voters authorizing the half-cent sales tax for transportation services now collected and administered by the Alameda CTC and governed by the 2000 Transportation Expenditure Plan. Collections for the sales tax authorized by Measure B began on April 1, 2002 and extends through March 31, 2022.

1.15 Measure BB. The measure approved by the voters authorizing the sales tax for transportation services collected and administered by the Alameda CTC and governed by the 2014 Transportation Expenditure Plan. Measure BB augments the half-cent Measure B sales tax by a half cent, beginning April 1, 2015 through March 31, 2022. The full one-cent sales tax authorized by Measure BB will begin April 1, 2022 and will extend through March 31, 2045.

1.16 Measure B Program. Transportation or transportation-related program specified in the 2000 Transportation Expenditure Plan for funding transportation programs and projects on a percentage-of-revenues or grant allocation basis.

1.17 Measure BB Program. Transportation or transportation-related program specified in the 2014 Transportation Expenditure Plan for funding transportation programs and projects on a percentage-of-revenues or grant allocation basis.

1.18 Measure B Project. Transportation and transportation-related capital projects specified in the 2000 Transportation Expenditure Plan for funding in the amounts allocated in the 2000 Transportation Expenditure Plan.

1.19 Measure BB Project. Transportation and transportation-related capital projects specified in the 2014 Transportation Expenditure Plan for funding in the amounts allocated in the 2014 Transportation Expenditure Plan.

1.20 Monitor. To observe, track, or keep a record of Measure projects, programs, and expenditures.

1.21 Organizational Meeting. An organizational meeting of the IWC will be held in July to elect officers and adopt the annual calendar/work plan and review the Alameda CTC budget related to IWC.

1.22 Organizational Member. One of the seven IWC members representing organizations as described in Section 3.1.2 below.

1.23 Oversee. To watch over Measure BB expenditures and performance measures.

1.24 Paratransit Advisory and Planning Committee (PAPCO). The Alameda CTC Committee that meets to address funding, planning, and coordination issues regarding paratransit services in Alameda County. Members must be Alameda County residents and eligible users of any transportation service available to seniors and people with disabilities in Alameda County. PAPCO is supported by a Paratransit Technical Advisory Committee comprised of Measure B and Measure BB-funded paratransit providers in Alameda County.

1.25 Performance Measures. Quantifiable methods, <u>adopted by the Commission</u>, used to assess how well the Alameda CTC is achieving its adopted objectives for Measure BB projects and programs.

1.26 Planning Area. Geographic groupings of cities and Alameda County for planning and funding purposes. North County: Alameda, Albany, Berkeley, Emeryville, Oakland, Piedmont; Central County: Hayward, San Leandro, unincorporated county (near Hayward); South County: Fremont, Newark, Union City; East County: Dublin, Livermore, Pleasanton, the unincorporated area of Sunol.

1.27 Subcommittee. A subset of the IWC, less than a quorum, usually organized for a certain purpose.

Article 2: Purpose and Responsibilities

2.1 Committee Purpose. The Committee is appointed pursuant to Measure B and Measure BB: 1) <u>1</u>To review all expenditures of the Measure B transportation sales tax; and 2) to review and oversee all expenditures and performance measures, as appropriate, of the Measure BB transportation sales tax; <u>3</u> to monitor <u>Measure B and Measure BB funded</u> projects and programs; and <u>4</u> to report directly to the public.

2.2 Committee Roles and Responsibilities from Expenditure Plan. As defined by the Measure B and Measure BB Transportation Expenditure Plans, the roles and responsibilities of the Committee include:

2.2.1 Hold public hearings and issue reports, on at least an annual basis, to inform Alameda County residents about how the sales tax funds are being spent. The hearings will be open to the public and must be held in compliance with the Brown Act, California's open meeting law, with information announcing the hearings well-publicized and posted in advance.

2.2.2 Have full access to Alameda CTC's independent auditor and have the authority to request and review specific information regarding use of the sales tax funds and to comment on the auditor's reports.

2.2.3 Publish an independent annual report, including any concerns the committee has about audits it reviews. The report will be published in local newspapers and will be made available to the public in a variety of forums to ensure access to this information.

2.2.4 Provide a balance of viewpoints, geography, age, gender, ethnicity and income status, to represent the different perspectives of the residents of the county.

2.3 Additional Responsibilities. Additional IWC member responsibilities are to:

2.3.1 Communicate from time to time to the Alameda CTC by resolution, suggestions and concerns pertinent to the administration and expenditure of Measure B and Measure BB funds.

2.3.2 Communicate as necessary to recommend that an appointing party appoint a new member when there is a vacancy or upcoming end of term.

Article 3: Members

3.1 Number of Members. The IWC will consist of 17 members.

3.1.1 Ten members shall be at-large, two each representing the five supervisorial districts in Alameda County, one of the two nominated by a member of the Board of Supervisors and one of the two selected by the Alameda County Mayors' Conference.

3.1.2 Seven of the members shall be nominated by the seven organizations specified in the 2014 Transportation Expenditure Plan: East Bay Economic Development Alliance; Alameda County Labor Council; Alameda County Taxpayers' Association; Alameda County Paratransit Advisory and Planning Committee; Bike East Bay, formerly known as East Bay Bicycle Coalition; League of Women Voters; and Sierra Club.

3.2 Appointment. The Commission will make appointments in the following manner:

3.2.1 Each member of the Alameda County Board of Supervisors shall select one At-Large Member to represent his or her supervisorial district.

3.2.2 The Alameda County Mayors' Conference shall select one At-Large Member to represent each of the five supervisorial districts.

3.2.3 Each organization listed in Section 3.1.2 above shall, subject to approval by the Commission, select one organizational member.

3.3 Membership Qualification. Each IWC member shall be an Alameda County resident. An IWC member shall not be an elected official at any level of government; or be a public employee of any agency that oversees or benefits from the proceeds of Measure B and Measure BB transportation sales taxes; or have any economic interest in any project or program.

3.4 Membership Term. Appointments for at-large members shall be for two-year terms. There is no maximum number of terms a member may serve. Members may serve until the Commission appoints their successor.

3.5 Attendance. Members will regularly attend meetings. Accordingly, more than three consecutive absences is cause for removal from the Committee.

3.6 Termination. A member's term shall terminate on the occurrence of any of the following:

3.6.1 The member voluntarily resigns by written notice to the chair or Alameda CTC staff.

3.6.2 The member fails to continue to meet the qualifications for membership, including attendance requirements.

3.6.3 The member becomes incapable of continuing to serve.

3.6.4 The appointing party or the Commission removes the member from the Committee.

3.7 Vacancies. An appointing party shall have the right to appoint (subject to approval by the Commission) a person to fill the vacant member position. Alameda CTC shall be responsible for notifying an appointing party of such vacancy and for urging expeditious appointment of a new member, as appropriate.

Article 4: Officers

4.1 Officers. The IWC shall annually elect a chair and vice chair. Each officer must be a duly appointed member of the IWC.

4.1.1 Duties. The chair shall preside at all meetings and will represent the IWC before the Commission to report on IWC activities. The chair shall serve as a voting ex-officio

member of all subcommittees except a nominating subcommittee (when the IWC discusses the chair position). The vice chair shall assume all duties of the chair in the absence of, or on the request of the chair.

4.2 Office Elections. Officers shall be elected by the members annually at the Organizational Meeting or as necessary to fill a vacancy. An individual receiving a majority of votes by a quorum shall be deemed to have been elected and will assume office at thate meeting following the election. In the event of multiple nominations, the vote shall be by ballot. Officers shall be eligible for re-election indefinitely.

Article 5: Meetings

5.1 Open and Public Meetings. All IWC meetings shall be open and public and governed by the Brown Act. Public comment shall be allowed at all IWC meetings. The time allotted for comments by a member of the public in the general public comment period or on any agenda item shall be up to 3 minutes per speaker at the discretion of the chair. Written comments may be submitted prior to the meeting. The number of IWC meetings, including regular meetings, sub-committee meetings, special meetings and public hearings, will be limited to the number of meetings approved in Alameda CTC's annual overall work program and budget, as approved by the Commission.

5.2 Regular Meetings. The IWC shall have a regular meeting at least once per quarter. Prior to each Organizational Meeting, the outgoing chair shall cause all members to be canvassed as to their available meeting times and shall recommend the day and time that best accommodates the schedules of all members, giving due regard to accommodating the schedule of any continuing member who has missed meetings due to a conflict in the prior year. Annually, at the Organizational Meeting, IWC shall establish the schedule of regular meetings for the ensuing year. Meeting dates and times may be changed and additional regular meetings scheduled during the year by action of the IWC.

5.3 Quorum. For purposes of decision making, a quorum shall consist of at least half (50 percent) plus one of the total number of members appointed at the time a decision is made. Members will not take actions at meetings with less than 50 percent plus one members present. Items may be discussed and information may be distributed on any item even if a quorum is not present; however, no action can be taken, until the Committee achieves a quorum.

5.4 Special Meetings. Special meetings may be called by the chair or by a majority of the members requesting the same in writing given to the chair, with copies to the vice chair and the Executive Director, specifying the matters to be considered at the special meeting. The chair or vice chair shall cause notice of a special meeting stating the matters to be considered to be given to all IWC members and posted and published in accordance with the Brown Act.

5.5 Public Hearing. At least annually, prior to publication of IWC's annual report, IWC shall conduct a public hearing on a draft of the IWC annual report. Each public hearing shall be conducted as part of a regular meeting.

5.6 Agenda. All meetings shall have a published agenda. Items for a regular meeting agenda may be submitted by any member to the chair and Alameda CTC staff. The Commission and/or Alameda CTC staff may also submit items for the agenda. Agenda planning meetings are held approximately three weeks prior to each IWC meeting. Alameda CTC staff will notify all IWC members when this meeting is established and remind members to submit any agenda item requests to the chair at least one day prior to the agenda planning meeting date. At the agenda planning meeting, the chair and Alameda CTC staff will discuss any agenda items submitted to the chair. Every agenda shall include a provision for members of the public to address the Committee. The chair and the vice chair shall review the agenda in advance of distribution. Copies of the agenda, with supporting material and the past meeting minutes, shall be mailed to members and any other interested parties who request it. The agenda shall be posted on the Alameda CTC website and in the Alameda CTC office and provided at the meeting, all in accordance with the Brown Act.

5.7 Roberts Rules of Order. The rules contained in the latest edition of "Roberts Rules of Order Newly Revised" shall govern the proceedings of the IWC and any subcommittees thereof to the extent that the person presiding over the proceeding determines that such formality is required to maintain order and make process, and to the extent that these actions are consistent with these bylaws.

5.8 Place of Meetings. IWC meetings shall be held at the Alameda CTC offices, unless otherwise designated by the Committee. Meeting locations shall be within Alameda County, accessible in compliance with the Americans with Disabilities Act of 1990 (41 U.S.C., Section 12132) or regulations promulgated thereunder, shall be accessible by public transportation, and shall not be in any facility that prohibits the admittance of any person, or persons, on the base of race, religious creed, color, national origin, ancestry, or sex, or where members of the public may not be present without making a payment or purchase.

5.9 Meeting Conduct. IWC members shall conduct themselves during meetings in a manner that encourages respectful behavior and provides a welcoming and safe environment for each member and staff member characterized by an atmosphere of mutual trust and respect. Members shall work with each other and staff to respectfully, fairly, and courteously deal with conflicts if they arise.

Article 6: Subcommittees

6.1 Establishment. The IWC may establish subcommittees when advisable and as necessary subject to the approved Alameda CTC overall work program and budget as approved by the Commission to conduct an investigation or to draft a report or other document within the authority of the IWC or for other purposes within the IWC's authority.

6.2 Membership. IWC members will be appointed to subcommittees by the IWC or by the chair. No subcommittee shall have fewer than three members, nor will a subcommittee have sufficient members to constitute a quorum of the IWC.

Article 7: Records and Notices

7.1 Minutes. Minutes of all meetings, including actions and the time and place of holding each meeting, shall be kept on file at the Alameda CTC office. Alameda CTC staff will prepare and include full minutes in meeting packets prior to each regular IWC meeting.

7.2 Attendance Roster. A member roster and a record of member attendance shall be kept on file at the Alameda CTC office.

7.3 Brown Act. All meetings of the IWC will comply with the requirements of the Brown Act. Notice of meetings and agendas will be given to all members and any member of the public requesting such notice in writing and shall be posted at the Alameda CTC office at least 72 hours prior to each meeting. Members of the public may address the IWC on any matter not on the agenda and on each matter listed on the agenda, in compliance with the Brown Act and time limits, up to three minutes per speaker, set at the discretion of the chair.

7.4 Meeting Notices. Meeting notices shall be in writing and shall be issued via U.S. Postal Service, Alameda CTC website, personal delivery, and/or email. Any other notice required or permitted to be given under these bylaws may be given by any of these means.

Article 8: General Matters

8.1 Per Diems. Committee members shall be entitled to a per diem stipend for meetings attended in amounts and in accordance with policies established by the Alameda CTC.

8.2 Conflicts of Interest. A conflict of interest exists when any Committee member has, or represents, a financial interest in the matter before the Committee. Such direct interest must be significant or personal. In the event of a conflict of interest, the Committee member shall declare the conflict, recuse himself or herself from the discussion, and shall not vote on that item. Failure to comply with these provisions shall be grounds for removal from the Committee.

8.3 Amendments to Bylaws. These bylaws will be reviewed annually, and may be amended, repealed, or altered, in whole or in part, by a vote taken at a duly constituted Committee meeting at which a quorum is present, as a recommendation to the Commission for approval.

8.4 Public Statements. No member of the Committee may make public statements on behalf of the Committee without authorization by affirmative vote of the Committee, except the chair, or in his or her place the vice chair, when making a regular report of the Committee activities and concerns to the Alameda CTC. This does not include presentations about the Committee to city councils, which all Committee members have a responsibility to make.

8.5 Conflict with Governing Documents. In the event of any conflict between these bylaws and the July 2000 Transportation Expenditure Plan, the January 2014 Transportation Expenditure Plan, California state law, or any action lawfully taken by ACTIA or the Alameda CTC, the Transportation Expenditure Plans, state law or the lawful action of ACTIA or the Alameda CTC shall prevail.

8.6 Staffing. Alameda CTC will provide staffing to the Committee including preparation and distribution of meeting agendas, packets, and minutes; tracking of attendance; and stipend administration.

8.7 Economic Interest. Each Committee member shall, no later than March 15 of every year, prepare and file with Alameda CTC a statement of economic interest in the form required by law, currently Form 700 which can be found on the California Fair Political Practices Commission website, <u>http://www.fppc.ca.gov/index.php?id=500</u>.

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Patricia Reavey, Deputy Executive Director of Finance & Administration Yoana Navarro, Accounting Manager
SUBJECT:	Approve the Alameda CTC FY2020-21 Third Quarter Consolidated Financial Report

Recommendation

It is recommended that the Commission approve the Alameda CTC FY2020-21 Third Quarter Consolidated Financial Report.

Summary

Alameda CTC's expenditures through March 31, 2021 are within year-to-date budget authority per the currently adopted budget. The agency remains in a strong financial position compared to budget through the third quarter of FY2020-21.

The attached FY2020-21 Third Quarter Financial Report has been prepared on a consolidated basis and is compared to the currently adopted budget on a year-to-date basis. This report provides a summary of FY2020-21 actual revenues and expenditures through March 31, 2021. Variances from the year-to-date budget are demonstrated as a percentage of the budget used by line item as well as stating either a favorable or unfavorable variance in dollars. Percentages over 100 percent indicate that actual revenue or expenditure items are more than 75 percent of the total annual budget through the third quarter of the fiscal year, and percentages under 100 percent indicate that actual budget through the third quarter of the fiscal year. As of March 31, 2021, Alameda CTC activity for the fiscal year results in a net increase in fund balance in the amount of \$35.3 million. While various funds saw an increase in their fund balances, the most significant contributors were the 2000 Measure B Capital Projects Fund and the 2014 Measure BB Special Revenue Fund which both collected sales tax revenues that outpaced expenditures during the fiscal year.

Background

The following are highlights of actual revenues and expenditures compared to budget as of March 31, 2021 by major category:

Revenues

Sales tax revenues are over budget by \$23.8 million, or 11.0 percent, and investment income is over budget by \$1.9 million or 61.2 percent primarily due to an adjustment to the budget to account for a decrease in market returns in the second half of the fiscal year. Grant revenues are under budget by \$51.9 million mostly related to timing on capital projects. Grant revenues are recognized on a reimbursement basis and, therefore, correlate directly with related expenditures. Consequently, capital and other project expenditures are also under budget.

Salaries and Benefits

Salaries and benefits are under budget by \$0.1 million, or 1.8 percent, as of March 31, 2021.

Administration

Costs for overall administration are over budget by \$4.2 million, or 15.6 percent, mainly due to debt service costs which incurred 100 percent of the annual costs by March 31, 2021 compare to 75 percent of the total annual budget amount. Debt service costs are required to be recorded when incurred per government accounting standards. No additional debt service costs will be incurred in the fourth quarter, and actual expenditures in the debt service fund will equal 100% of the budget by the end of the fiscal year.

Freeway Operations

Freeway Operations expenditures are under budget by \$2.0 million, or 38.4 percent, primarily related to operations and maintenance costs.

Planning

Planning expenditures are slightly over budget by \$0.03 million, or 4.3 percent, related to salaries and benefits.

Programs

Programs expenditures are over budget by \$5.0 million or 3.7 percent, mostly due to an increase in expenditures for Measure B and Measure BB direct local distributions (DLD) which is directly related to sales tax revenues coming in higher than projected.

Capital Projects

Capital Projects expenditures are under budget by \$225.6 million, or 76.4 percent. This variance is due, in part, to prolonged right-of-way acquisition negotiations resulting in project construction delays. Alameda CTC utilizes a rolling capital budget system in which any unused approved budget from prior years is available to pay for costs in subsequent fiscal years. Additional budget authority is requested by project only as needed in accordance with the budget process. The year-to-date budget amount used for comparisons is a straight-line amortization of the total approved project budget including unspent budget authority rolled over from the prior year. Expenditures planned

through March 31, 2021 in the budget process generally will differ from the straight-line budgeted amount used for this financial statement comparison. However, presenting the information with this comparison helps financial report users, project managers, and the project control team review year-to-date expenditures to give them an idea of how projects are progressing as compared to the approved budget. There are currently no real budget issues on capital projects.

Limitations Calculations

Staff has completed the limitation calculations required in both the 2000 Measure B and 2014 Measure BB Transportation Expenditure Plans related to salary and benefits and administration costs, and Alameda CTC is compliant with all limitation requirements.

Fiscal Impact: There is no fiscal impact associated with the requested action.

Attachment:

A. Alameda CTC Consolidated Revenues/Expenditures as of March 31, 2021

This page intentionally left blank

6.5A

ALAMEDA COUNTY TRANSPORTATION COMMISSION Consolidated Revenues/Expenditures March 31, 2021

	YTD <u>Actuals</u>	YTD <u>Budget</u>	% Used	<u>Favorable</u> (Unfavorable)/ Variance
REVENUES	Actuals	buget	<u>/// Useu</u>	variance
Sales Tax Revenue	\$ 241,330,090	\$ 217,500,000	110.96	\$ 23,830,090
Investment Income	4,890,417	3,033,750	161.20	1,856,667
Member Agency Fees	1,143,117	1,143,117	100.00	-
VRF Funds	9,804,631	9,000,000	108.94	804,631
TFCA Funds	1,545,481	1,558,892	99.14	(13,411)
Toll Revenues	6,186,250	5,250,000	117.83	936,250
Toll Violation and Penalty Revenues	1,999,661	1,125,000	177.75	874,661
Other Revenues	1,073	-	-	1,073
Regional/State/Federal Grants	9,796,249	54,707,402	17.91	(44,911,153)
Local and Other Grants	2,853,875	9,890,040	28.86	(7,036,165)
Total Revenues	\$ 279,550,844	\$ 303,208,201	92.20	\$ (23,657,357)
EXPENDITURES				
Administration				
Salaries and Benefits ⁽¹⁾	\$ 2,017,121	\$ 2,129,225	94.73	\$ 112,104
General Office Expenses	1,278,100	1,683,892	75.90	405,792
Travel Expense	4,806	7,500	64.08	2,694
Debt Service ⁽²⁾	26,470,200	19,852,650	133.33	(6,617,550)
Professional Services	1,282,004	2,758,376	46.48	1,476,372
Commission and Community Support	139,382	171,956	81.06	32,574
Contingency	-	375,000	-	375,000
Subtotal	31,191,613	26,978,599	115.62	(4,213,014)
Freeway Operations				
Salaries and Benefits ⁽¹⁾	162,038	222,528	72.82	60,490
Operating Expenditures	3,005,203	4,640,655	64.76	1,635,452
Special Project Expenditures	-	281,250	-	281,250
Subtotal	3,167,241	5,144,433	61.57	1,977,192
<u>Planning</u>				
Salaries and Benefits ⁽¹⁾	780,173	747,700	104.34	(32,473)
Subtotal	780,173	747,700	104.34	(32,473)
Programs				
Salaries and Benefits ⁽¹⁾	1,824,943	1,591,042	114.70	(233,901)
Programs Management and Support	865,534	2,332,379	37.11	1,466,845
Safe Routes to School Program	1,268,905	2,265,794	56.00	996,889
VRF Programming	6,838,650	7,656,000	89.32	817,350
Measure B/BB Direct Local Distribution	124,692,009	112,386,808	110.95	(12,305,201)
Grant Awards	3,305,334	5,785,500	57.13	2,480,166
TFCA Programming	474,164	1,833,255	25.86	1,359,091
Exchange Fund Programming	16,663	415,275	4.01	398,612
Subtotal	139,286,202	134,266,053	103.74	(5,020,149)
Capital Projects	072.025	4 074 645	04 55	407.000
Salaries and Benefits ⁽¹⁾	873,925	1,071,615	81.55	197,690
Capital Project Expenditures Subtotal	<u>68,953,648</u> 69,827,573	294,318,862 295,390,477	23.43	225,365,214 225,562,904
Total Expenditures	\$ 244,252,802	\$ 462,527,262	52.81	\$ 218,274,460
et Change in Fund Balance	\$ 35,298,042	\$ (159,319,061)		
-				
eginning Fund Balance	578,707,927	578,707,927		

(1) Salaries and benefits are under budget by \$103,910 or 1.8% as of March 31, 2021.

(2) Debt service cost are required to be recorded when incurred per government accounting standards and will equal budget by year end.

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Patricia Reavey, Deputy Executive Director of Finance and Administration Lily Balinton, Principal Financial Analyst
SUBJECT:	Approve the Alameda CTC FY2020-21 Third Quarter Investment Report

Recommendation

It is recommended that the Commission approve the Alameda CTC FY2020-21 Third Quarter Investment Report.

Summary

Alameda CTC's investments for the third quarter were in compliance with the Agency's investment policy, and the Agency has sufficient cash flow to meet expenditure requirements over the next six months.

The Consolidated Investment Report as of March 31, 2021 (Attachment A) provides balance and average return on investment information for all investments held by Alameda CTC at the end of the third quarter of fiscal year 2020-21. The report also shows balances as of June 30, 2020 for comparison purposes. The *Portfolio Review for the Quarter Ending March 31, 2021* (Attachment B), prepared by Public Trust Advisors, provides a review and outlook of market conditions and information regarding investment strategy, portfolio allocation, compliance, and returns by portfolio compared to the benchmarks.

Background

The following are highlights of key investment balance information as of March 31, 2021 compared to prior year-end balances:

- The 1986 Measure B investment balance increased by \$1.8 million or 1.3 percent related to investment earnings.
- The 2000 Measure B investment balance increased \$3.3 million or 1.7 percent due to 2000 Measure B sales tax collections outpacing expenditures in the

capital projects fund which was offset by a decrease in the bond fund balances as principal and interest debt service payments were made on March 1.

- The 2014 Measure BB investment balance decreased \$12.8 million or 8.0 percent due to payments for Measure BB capital project expenditures outpacing sales tax revenues through the end of the third quarter of the fiscal year as progress on Measure BB projects moves forward.
- The Non-Sales Tax investment balance increased \$11.1 million or 9.3 percent mostly related to deferred expenditures and deferred revenues received for projects and programs.

Investment yields have decreased from last fiscal year with an approximate average return on investments of 1.1 percent through March 31, 2021 compared to the prior fiscal year's average return of 2.1 percent. Projected return on investments for most funds was adjusted in the mid-year budget update for the FY2020-21 budget year to approximately 0.7 percent as interest rates continue to decline.

Fiscal Impact: There is no fiscal impact associated with the requested action.

Attachments:

- A. Consolidated Investment Report as of March 31, 2021
- B. Portfolio Review for Quarter Ending March 31, 2021 (provided by Public Trust Advisors)
- C. Holdings by Security Type as of March 31, 2021
| | | | | | | | | | | | _ | .0/ \ |
|---|----------|-----------------|-----------------|----------------|-------------------|-----------|---------|------------|--------|----------------------------|------|--------------------|
| | | | | Alame | da CTC | | | | | | | |
| | | | С | onsolidated In | vestment Report | | | | | | | |
| | | | - | | ch 31, 2021 | | | | | | | |
| | l | Un-Audited | Interest Earned | | | | | | FY 201 | 9-20 |)20 | |
| 1986 Measure B | | | | | As of March 31, 2 | 2021 | | | Inves | tment Balance | | Interest earned |
| | Inves | stment Balance | Int | terest earned | Approx. ROI | Budget | | Difference | Ju | ne 30, 2020 | | FY 2019-2020 |
| Bank Accounts | \$ | 1,472,386 | \$ | 15 | 0.00% | | | | \$ | 711,039 | | 633 |
| State Treasurer Pool (LAIF) ⁽¹⁾ | | 16,920,809 | | 83,711 | 0.66% | | | | | 13,308,410 | | 186,619 |
| Investment Advisor ⁽¹⁾⁽²⁾ | | 125,293,791 | | 1,819,949 | 1.94% | | | | | 127,883,958 | | 3,236,530 |
| 1986 Measure B Total | \$ | 143,686,986 | \$ | 1,903,675 | 1.77% \$ | 1,050,000 | \$ | 853,675 | \$ | 141,903,407 | \$ | 3,423,782 |
| | | | | | | | | | | Approx. ROI | | 2.41% |
| | l | Un-Audited | | | Interest Earne | ed | | | | FY 201 | 9-20 |)20 |
| 2000 Measure B | | | | | As of March 31, 2 | 2021 | | | Inves | tment Balance | | Interest earned |
| | Inves | stment Balance | Int | terest earned | Approx. ROI | Budget | | Difference | Ju | ne 30, 2020 | | FY 2019-2020 |
| Bank Accounts | \$ | 7,395,215 | \$ | 330 | 0.01% | | | | \$ | 2,130,652 | \$ | 16,495 |
| State Treasurer Pool (LAIF) ⁽¹⁾ | | 59,407,623 | | 267,452 | 0.60% | | | | | 48,329,778 | | 628,781 |
| Investment Advisor ⁽¹⁾⁽²⁾ | | 121,565,622 | | 1,749,072 | 1.92% | | | | | 127,831,715 | | 3,370,317 |
| 2014 Series A Bond Revenue Fund (1) | | 838 | | - | 0.00% | | | | | 838 | | 10 |
| 2014 Series A Bond Interest Fund (1) (2) | | 424,976 | | 427 | 0.07% | | | | | 1,083,059 | | 16,614 |
| 2014 Series A Bond Principal Fund (1) (2) | | 2,755,162 | | 13,328 | 0.12% | | | | | 8,708,557 | | 212,053 |
| Project Deferred Revenue ⁽¹⁾⁽³⁾ | | 192,510 | | 1,757 | 1.22% | | | | | 402,273 | | 9,764 |
| 2000 Measure B Total | \$ | 191,741,946 | \$ | 2,032,366 | 1.41% \$ | 1,218,750 | \$ | 813,616 | \$ | 188,486,872 | \$ | 4,254,034 |
| | | - , , | Ţ | , , | | , -, | , | , | Ţ | Approx. ROI | • | 2.26% |
| | | Un-Audited | | | Interest Earne | h | | | | FY 201 | 0-20 | 120 |
| 2014 Measure BB | ` | On Addited | | | As of March 31, 2 | | | | Inves | tment Balance | - | Interest earned |
| | Inves | streamt Deleman | اسا | torest some d | | | | | | | | |
| Dank Assounts | s s | stment Balance | | terest earned | Approx. ROI | Budget | | Difference | S Ju | ne 30, 2020 | ¢ | FY 2019-2020 |
| Bank Accounts
State Treasurer Pool (LAIF) ⁽¹⁾ | Ъ | 12,426,315 | \$ | 267 | 0.00% | | | | \$ | 4,653,766 | \$ | 15,538 |
| Investment Advisor ^{(1) (2)} | | 71,158,439 | | 376,878 | 0.71% | | | | | 60,913,897 | | 1,212,667 |
| | | 62,508,151 | | 279,026 | 0.60% | | | | | 94,604,658 | | 2,163,805 |
| Project Deferred Revenue ^{(1) (3)} | <u>_</u> | 1,571,765 | ¢ | 4,029 | 0.34% | 404.050 | | 100.050 | ¢ | 268,357 | ¢ | 28,103 |
| 2014 Measure BB Total | \$ | 147,664,670 | \$ | 660,200 | 0.60% \$ | 491,250 | \$ | 168,950 | \$ | 160,440,678
Approx. ROI | \$ | 3,420,113
2.13% |
| | | | | | | | | | | Αρριολ. Νοι | | 2.1370 |
| | l | Un-Audited | | | Interest Earne | | | | | FY 201 | | |
| Non-Sales Tax | | | | | As of March 31, 2 | | | | | tment Balance | | Interest earned |
| | | stment Balance | | terest earned | Approx. ROI | Budget | | Difference | | ne 30, 2020 | | FY 2019-2020 |
| Bank Accounts | \$ | 11,513,034 | \$ | 559 | 0.01% | | | | \$ | 3,934,443 | \$ | 16,668 |
| State Treasurer Pool (LAIF) ⁽¹⁾ | | 48,018,687 | | 218,003 | 0.61% | | | | | 45,626,235 | | 764,931 |
| California Asset Management Program (CAMP) | | 57,659,403 | | 81,401 | 0.19% | | | | | 57,578,002 | | 975,153 |
| Project Deferred Revenue ⁽¹⁾⁽³⁾ | | 12,443,476 | | 54,171 | 0.58% | | | | | 11,421,015 | | 207,639 |
| Non-Sales Tax Total | \$ | 129,634,600 | \$ | 354,134 | 0.36% \$ | 273,750 | \$ | 80,384 | \$ | 118,559,695 | \$ | 1,964,391 |
| | | | | | | | | | | Approx. ROI | | 1.66% |
| Alameda CTC TOTAL | \$ | 612,728,202 | \$ | 4,950,375 | 1.08% \$ | 3,033,750 | \$ | 1,916,625 | \$ | 609,390,652 | \$ | 13,062,320 |

Notes:

(1) All investments are marked to market on the financial statements at the end of the fiscal year per GASB 31 requirements.

(2) See attachments for detail of investment holdings managed by Investment Advisor.

(3) Project funds in deferred revenue are invested in LAIF with interest accruing back to the respective projects, as required per individual funding contracts.

6.6A

This page intentionally left blank



Alameda County Transportation Commission Portfolio Review for the Quarter Ending March 31, 2021

Fixed Income Market Review and Outlook

The U.S. economy continues to accelerate as the vaccination count advances and business restrictions ease. According to the Centers for Disease Control and Prevention, 32.6% of the U.S. population has received at least one dose of the COVID-19 vaccine, while 19% has been fully vaccinated. The swift rollout of vaccinations underpins the rise in consumer sentiment which registered a pandemic-high reading of 109.7 points in March as measured by the Conference Board Consumer Confidence Index. Federal stimulus distributions and increasing confidence in the recovery are expected to fuel a surge in household spending amongst a consumer base already flush with cash.

On the labor market front, nonfarm payrolls for the month of March shattered expectations as the 916,000 increase in employment came in well above the 660,000 survey estimate. The jobs report showed hiring rose across most industries, led by a 280,000 gain in the leisure and hospitality category. Despite growing improvement in the labor market, the Federal Reserve continues to maintain its dovish, or more accommodating, policies when assessing employment conditions. With the participation rate currently standing at just 61.5%, it would take approximately 5.2 million additional individuals reentering the labor force before returning to the pre-pandemic level. Reintroducing these workers as unemployed would translate to an unemployment rate close to 9.4% compared to March's reading of just 6.0%.

On March 31, the Biden administration unveiled its much anticipated \$2.3 trillion infrastructure investment plan. The proposal encompasses roughly \$620 billion related to transportation spending, \$300 billion to bolster domestic manufacturing, \$250 billion for research and development, \$215 billion allocated to affordable housing, and the residual spending spread across community-based care and education. To fund this ambitious spending plan, President Biden is calling for \$2 trillion in corporate tax increases over the next 15 years by proposing a raise in the corporate tax rate to 28% from the current rate of 21%.

The International Monetary Fund now projects the U.S. economy to expand by 6.4% this year which would mark the fastest annual pace of growth since 1984. Fed Chairman Powell, however, continues to reiterate that the Federal Reserve is in no hurry to change its easy monetary policy and that effects on inflation from stimulus distributions are expected to be transitory.

Short-term interest rates remain anchored by the Federal Reserve's near-zero interest rate policy and were generally unchanged over the quarter. By contrast, intermediate and long-term interest rates trended higher over the period as the combination of historically accommodative monetary and fiscal policies combined with the improving public health outlook and reopening optimism fueled a rise in inflation expectations. For the quarter, two-, five-, and ten-year U.S. Treasury yields rose 4 basis points (0.04%), 58 basis points (0.58%), and 83 basis points (0.83%), respectively. Amidst this backdrop and reflecting growing economic optimism and rising inflation expectations, the yield spread between two-and ten-year Treasuries rose to 158 basis points (1.58%) from 79 basis points (0.79%) last quarter.



Investment Strategy Update

Alameda CTC's liquidity portfolios remain invested in short-term securities to match anticipated expenditure dates to provide necessary liquidity for ongoing project costs. The long-term core portfolios remain invested in eligible and permitted securities as set forth in Alameda CTC's investment policy and California state code. Over the quarter, the core portfolios' durations were modestly shorter than their benchmark which benefited portfolio performance as the yield curve steepened over the period. In addition, the portfolios' continued allocation to high-quality corporate bonds served to enhance overall portfolio yield and income.

Portfolio Allocation

Provided below is a summary of the Alameda CTC consolidated portfolio as of the quarter ended March 31, 2021.





Compliance with Investment Policy Statement

As of the quarter ending March 31, 2021, the Alameda CTC portfolios were in compliance with the adopted investment policy.

Core Portfolios

The performance for the core 1986 and 2000 Measure B portfolios (the Portfolios) is reported on a total return basis. This method includes the coupon interest, amortization of discounts and premiums, capital gains and losses and price changes (i.e., unrealized gains and losses), but does not include the deduction of management fees. Performance Portfolios for the quarter ending March 31, 2021 is summarized in the table below. The Portfolios outperformed their respective benchmarks over the quarter and benefited from their shorter duration profile as the yield curve steepened. In addition, the Portfolios' continued allocation to high-quality corporate bonds served to enhance overall core portfolio yield and income.

Core Portfolio & Benchmark Total Return ¹	
1986 Measure B Portfolio	2000 Measure B Portfolio
Portfolio Return: 0.02 %	Portfolio Return: -0.03 %
Benchmark Return: -0.04%	Benchmark Return: -0.04 %

¹Note: Past performance is not an indication of future results. Performance is presented prior to the deduction of investment management fees.

1986 Measure B benchmark is the BofAML 1-3 Year AAA-AA US Corporate & Government Index. 2000 Measure B benchmark is the BofAML 1-3 Year AAA-AA US Corporate & Government Index.

Over the quarter, duration was generally shorter with values of 1.64 in the core 1986 Measure B portfolio and 1.65 in the core 2000 Measure B portfolio, compared to the benchmark duration of 1.83 as of March 31, 2021.

The Portfolios' yield to maturity, representing the return the Portfolios will earn in the future if all securities are held to maturity, is also reported below. This calculation is based on the current market value of the Portfolios including unrealized gains and losses. Portfolio yield to maturity for the quarter ending March 31, 2021 is summarized below:

2000 Measure B Portfolio
Portfolio YTM: 0.18%
Benchmark YTM: 0.19%



Liquidity and Bond Portfolios

The liquidity portions of the 1986 and 2000 Measure B portfolios (Liquidity portfolios), as well as the 2014 Measure BB and the Bond Interest and Principal Fund portfolios, remain invested in either short-term cash equivalents or permitted high-grade fixed income securities with maturity dates matched to appropriate anticipated expenditure and debt service payment dates.

One way to measure the anticipated return of the Liquidity and Bond portfolios is their yield to maturity. This is the return the portfolios will earn in the future if all securities are held to maturity. This calculation is based on the current market value of the portfolios. The yield to maturity and weighted average maturity (WAM) for the Liquidity and Bond portfolios and the comparable maturity of U.S. Treasury securities as of the quarter ending March 31, 2021 are summarized below:

Liquidity Portfolio & Comparable U	.S. Treasury Security Yield to Mat	turity
1986 Measure B Portfolio	2000 Measure B Portfolio	2014 Measure BB Portfolio
Portfolio YTM: 0.12%	Portfolio YTM: 0.08%	Portfolio YTM: 0.03%
Comparable TSY YTM: 0.07%	Comparable TSY YTM: 0.04%	Comparable TSY YTM: -0.02%
Portfolio WAM: 1.0 Years	Portfolio WAM: 0.5 Years	Portfolio WAM: 0.0 Years

Note: WAM is the weighted average amount of time until the securities in the portfolio mature.

Bond Portfolio & Comparable U.S. Treasury Security Yield	to Maturity
Interest Fund Portfolio	Principal Fund Portfolio
Portfolio YTM: 0.03%	Portfolio YTM: 0.09%
Comparable TSY YTM: -0.02%	Comparable TSY YTM: 0.07%
Portfolio WAM: 0.0 Years	Portfolio WAM: 0.87 Years

Note: WAM is the weighted average amount of time until the securities in the portfolio mature.

For the quarter ending March 31, 2021, the Alameda CTC Series 2014 Bonds Interest Fund and Principal Fund portfolios were invested in compliance with Section 5.11 of the Bond Indenture dated February 1, 2014.

VISORS D

Dated: 04/15/2021

ACTC 1986 Measure B (159781)

Holdings by Security Type ACTC

Base Currency: USD As of 03/31/2021

AGCY BOND

Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Marke Yield Value	t S&P Rating	Moody's Rating
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G04Q3	05/22/2023	2,480,000.00	100.1216	2,483,016.10	2,474,544.00	2,476,036.15	0.325 1.955%	AA+	Aaa
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G05G4	07/10/2023	1,850,000.00	100.0719	1,851,330.24	1,849,790.95	1,849,832.36	0.254 1.458%	AA+	Aaa
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G0S38	01/05/2022	5,800,000.00	101.4361	5,883,293.28	5,705,283.80	5,775,621.94	2.575 4.632%	AA+	Aaa
FEDERAL HOME LOAN MORTGAGE CORP	3137EAER6	05/05/2023	2,500,000.00	100.4257	2,510,643.75	2,503,150.00	2,502,230.40	0.332 1.977%	AA+	Aaa
FEDERAL HOME LOAN MORTGAGE CORP	3137EAES4	06/26/2023	1,850,000.00	100.0789	1,851,459.26	1,850,111.00	1,850,088.71	0.248 1.458%	AA+	Aaa
FEDERAL HOME LOAN MORTGAGE CORP	3137EAEV7	08/24/2023	1,850,000.00	100.0296	1,850,547.64	1,850,162.80	1,850,131.90	0.247 1.457%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130AFE78	12/09/2022	5,300,000.00	104.7465	5,551,562.86	5,367,787.00	5,329,412.50	2.651 4.371%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130ADRG9	03/10/2023	4,600,000.00	104.9084	4,825,786.77	4,613,018.00	4,606,091.70	2.677 3.800%	AA+	Aaa
EDERAL HOME LOAN BANKS	313381BR5	12/09/2022	2,285,000.00	102.8642	2,350,446.15	2,313,242.60	2,301,691.27	1.432 1.851%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3133834G3	06/09/2023	2,480,000.00	104.2063	2,584,315.87	2,612,010.40	2,576,501.66	0.337 2.035%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130AKDH6	10/21/2022	1,720,000.00	99.9357	1,718,894.64	1,717,729.60	1,718,231.02	0.191 1.353%	AA+	Aaa
FEDERAL HOME LOAN BANKS	313376C94	12/10/2021	2,285,000.00	101.7657	2,325,346.40	2,333,053.55	2,302,878.75	1.475 1.831%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130AFFN2	12/10/2021	3,300,000.00	102.0176	3,366,581.49	3,335,475.00	3,308,483.15	2.611 2.651%	AA+	Aaa
FEDERAL FARM CREDIT BANKS FUNDING	3133ELGN8	10/13/2022	2,285,000.00	102.2232	2,335,800.30	2,294,962.60	2,290,652.54	1.435 1.839%	AA+	Aaa
EDERAL FARM CREDIT BANKS FUNDING CORP	3133ELWD2	04/08/2022	2,500,000.00	100.2756	2,506,890.05	2,505,500.00	2,502,978.17	0.258 1.974%	AA+	Aaa
EDERAL FARM CREDIT BANKS FUNDING CORP	3133EMGX4	11/23/2022	1,450,000.00	99.9908	1,449,867.32	1,447,941.00	1,448,304.85	0.196 1.142%	AA+	Aaa
		10/26/2022	44,535,000.00	102.0816	45,445,782.13	44,773,762.30	44,689,167.06	1.459 35.784%	AA+	Aaa
CASH Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Marke Yield Value	t S&P Rating	Moody's Rating
Receivable	CCYUSD	03/31/2021	138.87	1.0000	138.87	138.87	138.87	0.000 0.000%	AA+	Aaa
Receivable	CCYUSD	03/31/2021	138.87	1.0000	138.87	138.87	138.87	0.000 0.000%	AA+	Aaa
CORP	ldentifier	Final Maturity	Current Units	Market	Market Value	Original Cost	Book Value	Book % of Marke	t S&P	Moody's
Description	luentiner	i inal maturity	Current Onits	Price	Warket Value	Unginar Cost	DOOK Value	Yield Value	Rating	Rating
TOYOTA MOTOR CREDIT CORP	89233P5T9	01/12/2022	1,300,000.00	102.2668	1,329,467.93	1,316,588.00	1,304,518.26	2.834 1.047%	A+	A1
PFIZER INC	717081DZ3	12/15/2021	1,300,000.00	101.3265	1,317,244.64	1,301,768.00	1,300,496.89	2.144 1.037%	A+	A2
PEPSICO INC	713448BW7	08/25/2021	1,300,000.00	101.0824	1,314,071.47	1,323,959.00	1,304,339.97	2.139 1.035%	A+	A1
ORACLE CORP	68389XBA2	07/08/2021	1,300,000.00	100.6575	1,308,547.12	1,300,949.00	1,300,107.89	2.767 1.030%	А	Baa2
MICROSOFT CORP	594918BH6	11/03/2022	1,000,000.00	103.5282	1,035,282.02	1,023,660.00	1,012,802.50	1.726 0.815%	AAA	Aaa
BERKSHIRE HATHAWAY INC	084670BC1	08/15/2021	1,125,000.00	101.3089	1,139,724.66	1,154,621.25	1,139,975.80	0.182 0.897%	AA	Aa2
APPLE INC	037833DC1	09/12/2022	2,000,000.00	102.5926	2,051,851.54	2,028,106.00	2,015,347.36	1.525 1.616%	AA+	Aa1
		01/29/2022	9,325,000.00	101.8441	9,496,189.39	9,449,651.25	9,377,588.65	1.910 7.477%	AA-	A1
MMFUND										
Description	Identifier	Final Maturity	Current Units	Market	Market Value	Original Cost	Book Value	Book % of Marke	t S&P	Moody's



Holdings by Security Type ACTC

ACTC 1986 Measure B (159781)

Base Currency: USD As of 03/31/2021

		08/10/2022	125,074,374.40	96.8291	127.001.611.02	125.356.090.07	125,293,929.83	1.377 100.000%	AA+	Aa1
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
Summary										
UNITED STATES TREASURY		09/18/2022	62,795,000.00	101.3574	63,640,290.67	62,714,271.83	62,807,919.36	1.417 50.110%	AA+	Aaa
UNITED STATES TREASURY	912828J76	03/31/2022	2,000,000.00	101.6562	2,033,124.00	2,006,015.62	2,002,622.38	1.615 1.601%	AA+	Aaa
UNITED STATES TREASURY	91282CAR2	10/31/2022	2,000,000.00	99.9883	1,999,766.00	1,999,453.12	1,999,483.50	0.141 1.575%	AA+	Aaa
UNITED STATES TREASURY	91282CAK7	09/15/2023	2,750,000.00	99.7148	2,742,157.00	2,741,943.35	2,742,237.58	0.240 2.159%	AA+	Aaa
UNITED STATES TREASURY	912828J43	02/28/2022	2,000,000.00	101.5156	2,030,312.00	2,005,390.62	2,002,232.68	1.625 1.599%	AA+	Aaa
JNITED STATES TREASURY	91282CBA8	12/15/2023	2,750,000.00	99.5430	2,737,432.50	2,737,646.48	2,738,057.46	0.286 2.155%	AA+	Aaa
JNITED STATES TREASURY	91282CAW1	11/15/2023	2,750,000.00	99.9375	2,748,281.25	2,748,388.67	2,748,443.90	0.272 2.164%	AA+	Aaa
JNITED STATES TREASURY	91282CAP6	10/15/2023	2,750,000.00	99.6602	2,740,655.50	2,740,654.29	2,740,984.94	0.255 2.158%	AA+	Aaa
JNITED STATES TREASURY	9128286C9	02/15/2022	1,400,000.00	102.1133	1,429,586.20	1,408,203.13	1,402,500.00	2.288 1.126%	AA+	Aaa
JNITED STATES TREASURY	9128283C2	10/31/2022	1,450,000.00	102.9336	1,492,537.20	1,501,429.69	1,491,691.95	0.180 1.175%	AA+	Aaa
JNITED STATES TREASURY	912828F96	10/31/2021	1,150,000.00	101.1289	1,162,982.35	1,170,484.38	1,162,537.85	0.129 0.916%	AA+	Aaa
JNITED STATES TREASURY	9128285A4	09/15/2021	3,900,000.00	101.2148	3,947,377.20	3,923,765.63	3,904,083.19	2.511 3.108%	AA+	Aaa
JNITED STATES TREASURY	912828ZH6	04/15/2023	1,850,000.00	100.1406	1,852,601.10	1,854,769.53	1,853,739.23	0.151 1.459%	AA+	Aaa
UNITED STATES TREASURY	912828Z29	01/15/2023	1,840,000.00	102.3984	1,884,130.56	1,903,034.37	1,882,543.32	0.204 1.484%	AA+	Aaa
JNITED STATES TREASURY	912828Z86	02/15/2023	1,840,000.00	102.2734	1,881,830.56	1,898,937.50	1,880,372.19	0.201 1.482%	AA+	Aaa
JNITED STATES TREASURY	912828ZD5	03/15/2023	1,840,000.00	100.6445	1,851,858.80	1,854,878.13	1,850,319.17	0.212 1.458%	AA+	Aaa
JNITED STATES TREASURY	9128284P2	05/15/2021	4,000,000.00	100.3104	4,012,416.00	3,997,031.24	3,999,879.16	2.651 3.159%	AA+	Aaa
JNITED STATES TREASURY	912828TY6	11/15/2022	2,600,000.00	102.3984	2,662,358.40	2,599,492.19	2,599,716.98	1.632 2.096%	AA+	Aaa
JNITED STATES TREASURY	912828YK0	10/15/2022	2,600,000.00	101.8984	2,649,358.40	2,581,414.06	2,589,888.38	1.634 2.086%	AA+	Aaa
INITED STATES TREASURY	9128282S8	08/31/2022	2,100,000.00	102.1133	2,144,379.30	2,099,015.63	2,099,524.37	1.641 1.688%	AA+	Aaa
JNITED STATES TREASURY	9128287C8	07/15/2022	2,100,000.00	102.1055	2,144,215.50	2,105,906.25	2,102,713.53	1.647 1.688%	AA+	Aaa
JNITED STATES TREASURY	912828XW5	06/30/2022	2,200,000.00	102.0352	2,244,774.40	2,196,992.18	2,198,738.66	1.797 1.768%	AA+	Aaa
JNITED STATES TREASURY	9128286Y1	06/15/2022	2,200,000.00	101.9766	2,243,485.20	2,197,765.61	2,199,081.19	1.786 1.767%	AA+	Aaa
UNITED STATES TREASURY	912828XD7	05/31/2022	1,725,000.00	102.0586	1,760,510.85	1,727,425.78	1,725,949.32	1.826 1.386%	AA+	Aaa
UNITED STATES TREASURY	912828L57	09/30/2022	5,300,000.00	102.4180	5,428,154.00	5,156,734.38	5,242,036.77	2.519 4.274%	AA+	Aaa
UNITED STATES TREASURY	912828XW5	06/30/2022	5,700,000.00	102.0352	5,816,006.40	5,557,500.00	5,648,541.66	2.510 4.579%	AA+	Aaa
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
JS GOV										
UNITED STATES TREASURY	912796D97	04/20/2021	2,325,000.00	99.9989	2,324,974.42	2,324,030.28	2,324,880.36	0.099 1.831%	A-1+	P-1
UNITED STATES TREASURY	912796D97	04/20/2021	2,325,000.00	99.9989	2,324,974.42	2,324,030.28	2,324,880.36	0.099 1.831%	A-1+	P-1
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody' Rating
-BILL										

* Grouped by: Security Type. * Groups Sorted by: Security Type. * Weighted by: Market Value + Accrued, except Book Yield by Base Book Value + Accrued. * Holdings Displayed by: Lot.



Holdings by Security Type ACTC

ACTC 2000 Measure B (159783)

Base Currency: USD As of 03/31/2021

AGCY BOND

Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G04Q3	05/22/2023	1,370,000.00	100.1216	1,371,666.15	1,366,986.00	1,367,810.29	0.325	1.119%	AA+	Aaa
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G05G4	07/10/2023	1,425,000.00	100.0719	1,426,024.65	1,424,838.98	1,424,870.87	0.254	1.163%	AA+	Aaa
FEDERAL HOME LOAN MORTGAGE CORP	3137EAER6	05/05/2023	2,300,000.00	100.4257	2,309,792.25	2,302,898.00	2,302,051.97	0.332	1.884%	AA+	Aaa
FEDERAL HOME LOAN MORTGAGE CORP	3137EAES4	06/26/2023	1,425,000.00	100.0789	1,426,124.03	1,425,085.50	1,425,068.33	0.248	1.163%	AA+	Aaa
FEDERAL HOME LOAN MORTGAGE CORP	3137EAEV7	08/24/2023	1,425,000.00	100.0296	1,425,421.83	1,425,125.40	1,425,101.60	0.247	1.163%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130AFE78	12/09/2022	3,500,000.00	104.7465	3,666,126.42	3,544,765.00	3,519,423.35	2.651	2.991%	AA+	Aaa
FEDERAL HOME LOAN BANKS	313381BR5	12/09/2022	2,200,000.00	102.8642	2,263,011.61	2,227,192.00	2,216,070.37	1.432	1.846%	AA+	Aaa
FEDERAL HOME LOAN BANKS	313379Q69	06/10/2022	2,225,000.00	102.3775	2,277,900.47	2,310,818.25	2,274,774.58	0.240	1.858%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3133834G3	06/09/2023	1,370,000.00	104.2063	1,427,626.10	1,442,925.10	1,423,309.38	0.337	1.165%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130AKDH6	10/21/2022	2,205,000.00	99.9357	2,203,582.96	2,202,089.40	2,202,732.21	0.191	1.798%	AA+	Aaa
FEDERAL HOME LOAN BANKS	3130AFFN2	12/10/2021	6,500,000.00	102.0176	6,631,145.37	6,569,875.00	6,516,709.24	2.611	5.410%	AA+	Aaa
FEDERAL FARM CREDIT BANKS FUNDING CORP	3133ELGN8	10/13/2022	2,200,000.00	102.2232	2,248,910.58	2,209,592.00	2,205,442.27	1.435	1.835%	AA+	Aaa
FEDERAL FARM CREDIT BANKS FUNDING CORP	3133ELWD2	04/08/2022	2,300,000.00	100.2756	2,306,338.85	2,305,060.00	2,302,739.91	0.258	1.882%	AA+	Aaa
FEDERAL FARM CREDIT BANKS FUNDING CORP	3133EMGX4	11/23/2022	2,045,000.00	99.9908	2,044,812.88	2,042,096.10	2,042,609.25	0.196	1.668%	AA+	Aaa
		10/13/2022	32,490,000.00	101.6871	33,028,484.12	32,799,346.73	32,648,713.63	1.149	26.946%	AA+	Aaa
	Identifier	Final Maturity	Current Units	Market	Market Value	Original Cost	Book Value			S&P	Moody's
Description		,		Price		Ŭ		Yield	Value	Rating	Moody's Rating
Description Receivable	CCYUSD	03/31/2021	352.40	<i>Price</i> 1.0000	352.40	352.40	352.40	<i>Yield</i> 0.000	Value 0.000%	Rating AA+	Rating Aaa
CASH Description Receivable Receivable		,		Price		Ŭ		<i>Yield</i> 0.000	Value	Rating	Rating
Description Receivable	CCYUSD	03/31/2021	352.40	<i>Price</i> 1.0000	352.40	352.40	352.40	<i>Yield</i> 0.000	Value 0.000%	Rating AA+	Rating Aaa
Description Receivable Receivable	CCYUSD	03/31/2021	352.40	<i>Price</i> 1.0000	352.40	352.40	352.40	Yield 0.000 0.000 Book	Value 0.000% 0.000%	Rating AA+	Rating Aaa
Description Receivable Receivable CORP Description	CCYUSD CCYUSD	03/31/2021 03/31/2021	352.40 352.40	Price 1.0000 1.0000 Market	352.40 352.40	352.40 352.40	352.40 352.40	Yield 0.000 0.000 Book Yield	Value 0.000% 0.000% % of Market	Rating AA+ AA+ S&P	Rating Aaa Aaa Moody's
Description Receivable Receivable CORP Description TOYOTA MOTOR CREDIT CORP	CCYUSD CCYUSD Identifier	03/31/2021 03/31/2021 Final Maturity	352.40 352.40 <i>Current Units</i>	Price 1.0000 1.0000 Market Price	352.40 352.40 Market Value	352.40 352.40 Original Cost	352.40 352.40 Book Value	Yield 0.000 0.000 Book Yield 2.834	Value 0.000% 0.000% % of Market Value	Rating AA+ AA+ S&P Rating	Rating Aaa Aaa Moody's Rating
Description Receivable Receivable CORP Description TOYOTA MOTOR CREDIT CORP ORACLE CORP	CCYUSD CCYUSD Identifier 89233P5T9	03/31/2021 03/31/2021 Final Maturity 01/12/2022	352.40 352.40 <i>Current Units</i> 1,350,000.00	Price 1.0000 1.0000 Market Price 102.2668	352.40 352.40 Market Value 1,380,601.31	352.40 352.40 Original Cost 1,367,226.00	352.40 352.40 <i>Book Value</i> 1,354,692.03	Yield 0.000 0.000 Book Yield 2.834 2.767	Value 0.000% 0.000% % of Market Value 1.126%	Rating AA+ AA+ S&P Rating A+	Rating Aaa Aaa Moody's Rating A1
Description Receivable Receivable	CCYUSD CCYUSD Identifier 89233P5T9 68389XBA2	03/31/2021 03/31/2021 Final Maturity 01/12/2022 07/08/2021	352.40 352.40 <i>Current Units</i> 1,350,000.00 1,350,000.00	Price 1.0000 1.0000 Market Price 102.2668 100.6575	352.40 352.40 <i>Market Value</i> 1,380,601.31 1,358,875.86	352.40 352.40 Original Cost 1,367,226.00 1,350,985.50	352.40 352.40 <i>Book Value</i> 1,354,692.03 1,350,112.04	Yield 0.000 0.000 Book Yield 2.834 2.767 1.726	Value 0.000% 0.000% % of Market Value 1.126% 1.109%	Rating AA+ AA+ S&P Rating A+ A	Rating Aaa Aaa Moody's Rating A1 Baa2
Description Receivable Receivable CORP Description TOYOTA MOTOR CREDIT CORP ORACLE CORP MICROSOFT CORP	CCYUSD CCYUSD Identifier 89233P5T9 68389XBA2 594918BH6	03/31/2021 03/31/2021 Final Maturity 01/12/2022 07/08/2021 11/03/2022	352.40 352.40 <i>Current Units</i> 1,350,000.00 1,350,000.00 1,675,000.00	Price 1.0000 1.0000 Market Price 102.2668 100.6575 103.5282	352.40 352.40 Market Value 1,380,601.31 1,358,875.86 1,734,097.38	352.40 352.40 0riginal Cost 1,367,226.00 1,350,985.50 1,714,630.50	352.40 352.40 <i>Book Value</i> 1,354,692.03 1,350,112.04 1,696,444.18	Yield 0.000 0.000 Book Yield 2.834 2.767 1.726 0.182	Value 0.000% 0.000% % of Market Value 1.126% 1.109% 1.415%	Rating AA+ AA+ S&P Rating A+ A AAA	Rating Aaa Aaa Moody's Rating A1 Baa2 Aaa

MMFUND

UNITED STATES TREASURY

912796D97

04/20/2021

0.15										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
MORG STAN I LQ:GV I	61747C707	03/31/2021	22,463,024.33	1.0000	22,463,024.33	22,463,024.33	22,463,024.33	0.030 18.326%	AAAm	Aaa
MORG STAN I LQ:GV I	61747C707	03/31/2021	22,463,024.33	1.0000	22,463,024.33	22,463,024.33	22,463,024.33	0.030 18.326%	AAAm	Aaa
T-BILL										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
UNITED STATES TREASURY	912796D97	04/20/2021	3,075,000.00	99.9989	3,074,966.17	3,073,717.47	3,074,841.77	0.099 2.509%	A-1+	P-1

3,074,966.17

3,073,717.47

3,074,841.77

99.9989

3,075,000.00

A-1+

P-1

1

0.099 2.509%



ACTC 2000 Measure B (159783)

Holdings by Security Type ACTC

Base Currency: USD As of 03/31/2021

US GOV

Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
UNITED STATES TREASURY	912828XW5	06/30/2022	4,000,000.00	102.0352	4,081,408.00	3,900,000.00	3,963,888.89	2.510 3.330%	AA+	Aaa
UNITED STATES TREASURY	912828S35	06/30/2023	500,000.00	102.6211	513,105.50	476,250.00	487,873.60	2.522 0.419%	AA+	Aaa
UNITED STATES TREASURY	9128286Y1	06/15/2022	1,950,000.00	101.9766	1,988,543.70	1,948,019.52	1,949,185.60	1.786 1.622%	AA+	Aaa
UNITED STATES TREASURY	912828XW5	06/30/2022	1,950,000.00	102.0352	1,989,686.40	1,947,333.98	1,948,881.99	1.797 1.623%	AA+	Aaa
UNITED STATES TREASURY	9128282S8	08/31/2022	2,000,000.00	102.1133	2,042,266.00	2,000,234.38	2,000,122.65	1.620 1.666%	AA+	Aaa
UNITED STATES TREASURY	912828XD7	05/31/2022	1,900,000.00	102.0586	1,939,113.40	1,911,949.22	1,905,667.88	1.612 1.582%	AA+	Aaa
UNITED STATES TREASURY	912828WZ9	04/30/2022	1,900,000.00	101.7695	1,933,620.50	1,905,789.06	1,902,636.87	1.618 1.578%	AA+	Aaa
UNITED STATES TREASURY	912828YF1	09/15/2022	2,000,000.00	101.9883	2,039,766.00	1,993,906.25	1,996,767.82	1.614 1.664%	AA+	Aaa
UNITED STATES TREASURY	912828TY6	11/15/2022	2,000,000.00	102.3984	2,047,968.00	1,999,609.38	1,999,782.30	1.632 1.671%	AA+	Aaa
UNITED STATES TREASURY	912828YK0	10/15/2022	2,000,000.00	101.8984	2,037,968.00	1,985,703.12	1,992,221.83	1.634 1.663%	AA+	Aaa
UNITED STATES TREASURY	912828Z29	01/15/2023	1,900,000.00	102.3984	1,945,569.60	1,965,089.84	1,943,930.60	0.204 1.587%	AA+	Aaa
UNITED STATES TREASURY	912828ZD5	03/15/2023	1,900,000.00	100.6445	1,912,245.50	1,915,363.29	1,910,655.67	0.212 1.560%	AA+	Aaa
UNITED STATES TREASURY	912828Z86	02/15/2023	1,900,000.00	102.2734	1,943,194.60	1,960,859.38	1,941,688.68	0.201 1.585%	AA+	Aaa
UNITED STATES TREASURY	912828WR7	06/30/2021	7,500,000.00	100.5107	7,538,302.50	7,430,566.43	7,493,017.85	2.516 6.150%	AA+	Aaa
UNITED STATES TREASURY	912828WR7	06/30/2021	2,300,000.00	100.5107	2,311,746.10	2,284,457.04	2,298,320.69	2.431 1.886%	AA+	Aaa
UNITED STATES TREASURY	912828ZH6	04/15/2023	1,425,000.00	100.1406	1,427,003.55	1,428,673.83	1,427,880.22	0.151 1.164%	AA+	Aaa
UNITED STATES TREASURY	9128283C2	10/31/2022	1,875,000.00	102.9336	1,930,005.00	1,941,503.91	1,928,912.01	0.180 1.575%	AA+	Aaa
UNITED STATES TREASURY	912828F96	10/31/2021	1,500,000.00	101.1289	1,516,933.50	1,526,718.75	1,516,353.72	0.129 1.238%	AA+	Aaa
UNITED STATES TREASURY	91282CAW1	11/15/2023	2,550,000.00	99.9375	2,548,406.25	2,556,873.04	2,556,448.94	0.153 2.079%	AA+	Aaa
UNITED STATES TREASURY	91282CBA8	12/15/2023	2,550,000.00	99.5430	2,538,346.50	2,547,011.73	2,547,190.86	0.166 2.071%	AA+	Aaa
UNITED STATES TREASURY	91282CAP6	10/15/2023	2,550,000.00	99.6602	2,541,335.10	2,548,007.81	2,548,134.59	0.154 2.073%	AA+	Aaa
UNITED STATES TREASURY	91282CAK7	09/15/2023	2,550,000.00	99.7148	2,542,727.40	2,548,107.42	2,548,231.62	0.153 2.074%	AA+	Aaa
UNITED STATES TREASURY	91282CAR2	10/31/2022	1,850,000.00	99.9883	1,849,783.55	1,850,361.32	1,850,325.81	0.114 1.509%	AA+	Aaa
UNITED STATES TREASURY	912828ZP8	05/15/2023	1,550,000.00	99.8633	1,547,881.15	1,549,697.27	1,549,720.06	0.134 1.263%	AA+	Aaa
UNITED STATES TREASURY	912828J76	03/31/2022	1,900,000.00	101.6562	1,931,467.80	1,905,714.84	1,902,491.26	1.615 1.576%	AA+	Aaa
UNITED STATES TREASURY		09/13/2022	56,000,000.00	101.1521	56,638,393.60	56,027,800.81	56,110,331.97	1.227 46.208%	AA+	Aaa

Summary

Description	ldentifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
		05/20/2022	121,253,376.73	83.0223	122,572,036.95	121,678,033.23	121,565,974.69	1.003	100.000%	AA+	Aa1

* Grouped by: Security Type. * Groups Sorted by: Security Type. * Weighted by: Market Value + Accrued, except Book Yield by Base Book Value + Accrued. * Holdings Displayed by: Lot.



Holdings by Security Type ACTC

ACTC 2014 Measure BB (159782)

Base Currency: USD As of 03/31/2021

AGCY BOND										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G0U27	04/13/2021	3,000,000.00	100.0819	3,002,457.42	3,028,710.00	3,002,376.00	0.106 4.803%	AA+	Aaa
FEDERAL NATIONAL MORTGAGE ASSOCIATION	3135G0U27	04/13/2021	3,000,000.00	100.0819	3,002,457.42	3,028,710.00	3,002,376.00	0.106 4.803%	AA+	Aaa
CASH										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
Receivable	CCYUSD	03/31/2021	1,114.42	1.0000	1,114.42	1,114.42	1,114.42	0.000 0.002%	AA+	Aaa
Receivable	CCYUSD	03/31/2021	1,114.42	1.0000	1,114.42	1,114.42	1,114.42	0.000 0.002%	AA+	Aaa
MMFUND										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
MORG STAN I LQ:GV I	61747C707	03/31/2021	52,506,086.93	1.0000	52,506,086.93	52,506,086.93	52,506,086.93	0.030 83.997%	AAAm	Aaa
MORG STAN I LQ:GV I	61747C707	03/31/2021	52,506,086.93	1.0000	52,506,086.93	52,506,086.93	52,506,086.93	0.030 83.997%	AAAm	Aaa
T-BILL										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
UNITED STATES TREASURY	912796D97	04/20/2021	2,500,000.00	99.9989	2,499,972.50	2,499,082.72	2,499,885.34	0.088 3.999%	A-1+	P-1
UNITED STATES TREASURY	912796D89	04/13/2021	2,500,000.00	99.9998	2,499,995.00	2,499,134.03	2,499,928.33	0.087 3.999%	A-1+	P-1
UNITED STATES TREASURY	912796F20	04/27/2021	2,000,000.00	99.9987	1,999,974.00	1,999,255.68	1,999,874.34	0.088 3.199%	A-1+	P-1
UNITED STATES TREASURY		04/19/2021	7,000,000.00	99.9992	6,999,941.50	6,997,472.43	6,999,688.01	0.088 11.198%	A-1+	P-1
Summary										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
		04/03/2021	62,507,201.35	16.8918	62,509,600.27	62,533,383.78	62,509,265.36	0.040 100.000%	AAA	Aaa

* Grouped by: Security Type. * Groups Sorted by: Security Type. * Weighted by: Market Value + Accrued, except Book Yield by Base Book Value + Accrued. * Holdings Displayed by: Lot.



Holdings by Security Type ACTC

ACTC Series 2014-Interest Fd (159784)

Base Currency: USD As of 03/31/2021

CASH											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
Receivable	CCYUSD	03/31/2021	10.06	1.0000	10.06	10.06	10.06	0.000	0.002%	AA+	Aaa
Receivable	CCYUSD	03/31/2021	10.06	1.0000	10.06	10.06	10.06	0.000	0.002%	AA+	Aaa
MMFUND											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
MORG STAN I LQ:GV I	61747C707	03/31/2021	424,975.70	1.0000	424,975.70	424,975.70	424,975.70	0.030	99.998%	AAAm	Aaa
MORG STAN I LQ:GV I	61747C707	03/31/2021	424,975.70	1.0000	424,975.70	424,975.70	424,975.70	0.030	99.998%	AAAm	Aaa
Summary											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
		03/31/2021	424,985.76	1.0000	424,985.76	424,985.76	424,985.76	0.030	100.000%	AAA	Aaa

* Grouped by: Security Type. * Groups Sorted by: Security Type. * Weighted by: Market Value + Accrued, except Book Yield by Base Book Value + Accrued. * Holdings Displayed by: Lot.



Holdings by Security Type ACTC

ACTC Series 2014-Principal Fd (159786)

Dated: 04/15/2021

Base Currency: USD As of 03/31/2021

CASH											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
Receivable	CCYUSD	03/31/2021	55.84	1.0000	55.84	55.84	55.84	0.000 0	0.002%	AA+	Aaa
Receivable	CCYUSD	03/31/2021	55.84	1.0000	55.84	55.84	55.84	0.000 0	0.002%	AA+	Aaa
MMFUND											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book 9 Yield N	% of Market Value	S&P Rating	Moody's Rating
MORG STAN I LQ:GV I	61747C707	03/31/2021	23,121.28	1.0000	23,121.28	23,121.28	23,121.28	0.030 0	0.839%	AAAm	Aaa
MORG STAN I LQ:GV I	61747C707	03/31/2021	23,121.28	1.0000	23,121.28	23,121.28	23,121.28	0.030 0	0.839%	AAAm	Aaa
US GOV											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value		% of Market Value	S&P Rating	Moody's Rating
UNITED STATES TREASURY	9128286C9	02/15/2022	2,675,000.00	102.1133	2,731,530.77	2,732,575.20	2,732,040.45	0.064 9	99.159%	AA+	Aaa
UNITED STATES TREASURY	9128286C9	02/15/2022	2,675,000.00	102.1133	2,731,530.77	2,732,575.20	2,732,040.45	0.064 9	99.159%	AA+	Aaa
Summary											
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book 9 Yield N	% of Market Value	S&P Rating	Moody's Rating
-		02/12/2022	2,698,177.12	101.2651	2,754,707.89	2,755,752.32	2,755,217.57	0.064 1	100.000%	AA+	Aaa

* Grouped by: Security Type. * Groups Sorted by: Security Type. * Weighted by: Market Value + Accrued, except Book Yield by Base Book Value + Accrued. * Holdings Displayed by: Lot.



Holdings by Security Type ACTC

ACTC Series 2014-Revenue Fd (159787)

Dated: 04/15/2021

Base Currency: USD As of 03/31/2021

CASH										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
Receivable	CCYUSD	03/31/2021	0.02	1.0000	0.02	0.02	0.02	0.000 0.002%	AA+	Aaa
Receivable	CCYUSD	03/31/2021	0.02	1.0000	0.02	0.02	0.02	0.000 0.002%	AA+	Aaa
MMFUND										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
MORG STAN I LQ:GV I	61747C707	03/31/2021	838.31	1.0000	838.31	838.31	838.31	0.030 99.998%	AAAm	Aaa
MORG STAN I LQ:GV I	61747C707	03/31/2021	838.31	1.0000	838.31	838.31	838.31	0.030 99.998%	AAAm	Aaa
Summary										
Description	Identifier	Final Maturity	Current Units	Market Price	Market Value	Original Cost	Book Value	Book % of Market Yield Value	S&P Rating	Moody's Rating
		03/31/2021	838.33	1.0000	838.33	838.33	838.33	0.030 100.000%	AAA	Aaa

* Grouped by: Security Type. * Groups Sorted by: Security Type. * Weighted by: Market Value + Accrued, except Book Yield by Base Book Value + Accrued. * Holdings Displayed by: Lot.



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:May 20, 2021TO:Alameda County Transportation CommissionFROM:Patricia Reavey, Deputy Executive Director of Finance
and Administration
Lily Balinton, Principal Financial AnalystSUBJECT:Approve an Update to the Alameda CTC Investment Policy

Recommendation

It is recommended that the Commission approve an update to the Alameda CTC investment policy that was adopted in May 2020.

Summary

An update to the Alameda CTC investment policy is attached as a red line version to show recommended changes since the investment policy was adopted in May 2020. The recommended changes to the policy include administrative edits to clarify ambiguous language and incorporate state statutory updates effective January 1, 2021 under Senate Bill (SB) 998 which updates, clarifies, and strengthens the California Government Code (the Code) with amendments to Sections 53601(h), 53601(k), and 53601.6. Under SB 998, and since Alameda CTC has more than \$100 million of investments, the agency is authorized to augment the investment policy to include the following changes:

- Increase the allowable limit for investments in commercial paper from 25% to 40% (per the Code Section 53601(h));
- Prohibit the investment of more than 10% of total investment assets in the commercial paper and medium-term notes of any single issuer (per the Code Sections 53601(h) and 53601(k)); and
- Allow for the investment in securities issued by, or backed by, the United States Government that could result in zero or negative interest accrual if held to maturity, in the event of, and for the duration of, a period of negative market interest rates (per the Code Section 53601.6(b)(2)).

Due to the sunset provision, on and after January 1, 2026, unless a subsequent bill amends Sections 53601 and 53601.6 of the Code, the provisions authorizing certain local agencies with investable assets in excess of \$100 million to invest 40% of their moneys in eligible commercial paper will expire reverting back to a limit of 25% and all local agencies will again be prohibited from investing funds in securities that could result in zero interest accrual if held to maturity even during a period of negative market interest rates. However, the 10% single issuer limitation for commercial paper and medium-term notes will remain in effect. Staff will continue to bring the investment policy to the Commission for review annually, as is best practice, and monitor changes to the Code. If there is a rescission of these changes to the Code on January 1, 2026, Alameda CTC's investment policy will be updated at that time.

Background

The California Government Code (the Code) Section 53600.5 states, "... the primary objective of a trustee shall be to safeguard the principal of the funds under its control. The secondary objective shall be to meet the liquidity needs of the depositor. The third objective shall be to achieve a return on the funds under its control." These objectives also are reflected in Alameda CTC's investment policy, in the order of priority demonstrated in the Code. Staff has reviewed the investment policy in consultation with investment advisors and is recommending changes to the currently adopted investment policy to incorporate the recent state statutory updates that became effective January 1, 2021 under Senate Bill 998 and to make administrative edits to remove ambiguity in the policy. The current investment policy was adopted by the Commission in May 2020.

The attached investment policy (Attachment A) was developed in accordance with the Code in order to define parameters and guide staff and investment advisors in managing Alameda CTC's investment portfolio. The policy formalizes the framework for Alameda CTC's investment activities that must be exercised to ensure effective and prudent fiscal management of Alameda CTC's funds. The guidelines are intended to be broad enough to allow staff and the investment advisors to function properly within the parameters of fiscal responsibility and authority, yet specific enough to adequately safeguard the investment assets.

The primary objectives of the investment activities within the policy safeguard Alameda CTC assets by mitigating credit and interest rate risk, provide adequate liquidity to meet all operating requirements of Alameda CTC, and attain a market rate of return on investments taking into account the investment risk constraints of safety and liquidity needs.

Through the proposed investment policy, the Commission appoints the Executive Director and the Deputy Executive Director of Finance and Administration as Investment Officers who are responsible for the investment program of the Alameda CTC and will act responsibly as custodians of the public trust. The policy requires the Investment Officers to design internal controls around investments that would prevent the loss of public funds from fraud, employee error, misrepresentation by third parties, unanticipated changes in financial markets or imprudent actions by employees and officers of Alameda CTC. It also allows the Investment Officers to periodically reset performance benchmarks to reflect changing investment objectives and constraints. Fiscal Impact: There is no fiscal impact to the approval of this investment policy.

Attachment:

A. Draft Alameda CTC Investment Policy May 2021

This page intentionally left blank

Alameda County Transportation Commission Investment Policy

May 202<u>1</u>0

I. Introduction

The intent of the Investment Policy of the Alameda County Transportation Commission (Alameda CTC) is to define the parameters within which funds are to be managed. The policy formalizes the framework for Alameda CTC's investment activities that must be exercised to ensure effective and prudent fiscal and investment management of Alameda CTC's funds. The guidelines are intended to be broad enough to allow Alameda CTC's Investment Officers (as defined below) to function properly within the parameters of responsibility and authority, yet specific enough to adequately safeguard the investment assets.

II. Governing Authority

The investment program shall be operated in conformance with federal, state, and other legal requirements, including the California Government Code.

III. Scope

This policy applies to activities of Alameda CTC with regard to investing the financial assets of all funds (except bond funds and retirement funds). In addition, any funds held by trustees or fiscal agents are excluded from these rules; however, all such funds are subject to regulations established by the State of California.

Note that any excluded funds such as employee retirement funds, proceeds from certain bond issuances and Other Postemployment Benefits (OPEB) trust assets are covered by separate policies.

IV. General Objectives

The primary objectives, in order of priority, of investment activities shall be:

1. Safety

Safety of principal is the foremost objective of the investment program. Investments shall be undertaken in a manner that seeks to ensure the preservation of capital in the overall portfolio. The goal will be to mitigate credit and interest rate risk.

2. Liquidity

The investment portfolio shall remain sufficiently liquid to meet all operating requirements that may be reasonably anticipated.

3. Return

The investment portfolio shall be designed with the objective of attaining a market rate of return throughout budgetary and economic cycles, taking into account the investment risk contraints of safety and liquidity needs.

V. Standard of Care

1. Prudence

The standard of prudence to be used by investment officials shall be the "prudent investor" standard (California Government Code Section 53600.3) and shall be applied in the context of managing an overall portfolio. Investment Officers acting in accordance with written procedures and the investment policy and exercising due diligence shall be relieved of personal responsibility for an individual security's credit risk or market price changes, provided deviations from expectations are reported in a timely fashion and appropriate action is taken to control adverse developments.

"When investing, reinvesting, purchasing, acquiring, exchanging, selling, or managing public funds, a trustee shall act with care, skill, prudence, and diligence under the circumstances then prevailing, including, but not limited to, the general economic conditions and the anticipated needs of the agency, that a prudent person acting in a like capacity and familiarity with those matters would use in the conduct of funds of a like character and with like aims, to safeguard the principal and maintain the liquidity needs of the agency. Within the limitations of this section and considering individual investments as part of an overall strategy, investments may be acquired as authorized by law."

2. Delegation of Authority and Responsibilities

<u>Responsibilities of the Commission</u> - The Commission, in its role as Alameda CTC's governing body, will retain ultimate fiduciary responsibility for the portfolios. They will receive quarterly reports for review, designate Investment Officers and annually review and adopt the investment policy.

The Commission hereby designates the Executive Director and the Deputy Executive Director of Finance and Administration, as Treasurer, as the Investment Officers.

<u>Responsibilities of the Investment Officers</u> - The Investment Officers are jointly responsible for the operation of the investment program. The Investment Officers shall act in accordance with written procedures and internal controls for the operation of the investment program consistent with the Investment Policy. All participants in the investment process shall seek to act responsibly as custodians of the public trust. No officer may engage in an investment transaction except as provided under the terms of this policy and supporting procedures.

<u>Responsibilities of the Investment Advisor</u> - Alameda CTC may engage the services of one or more external investment advisors to assist in the management of the investment portfolio in a manner consistent with Alameda CTC's objectives. Investment advisors may be granted discretion to purchase and sell investment securities in accordance with this Investment Policy and the California Government Code and must be registered under the Investment Advisors Act of 1940 or be a bank, regulated by the Office of the Comptroller of the Currency (OCC) or Federal Reserve operating under the fiduciary exemption from the Security and Exchange Commission. Any investment advisor shall be required to prepare and provide comprehensive reports on Alameda CTC's investments on a monthly and quarterly basis, and as requested by Alameda CTC's Investment Officers. At no time shall the investment advisor maintain custody of Alameda CTC cash or assets. Responsibilities of the Custodian - A third party bank custodian shall hold Alameda CTC cash and assets under management by any investment advisor in the name of Alameda CTC. The custodian shall receive direction from the investment advisor on settlement of investment transactions.

VI. Selection of Financial Institutions and Broker/Dealers

Alameda CTC's procedures are designed to encourage competitive bidding on transactions from an approved list of broker/dealers in order to provide for the best execution on transactions.

The Investment Officer, or the investment advisors, shall maintain a list of authorized broker/dealers and financial institutions that are approved for investment purposes. This list will be developed after a process of due diligence confirming that the firms qualify under the Securities and Exchange Commission Rule 15C3-1 (uniform net capital rule). Alameda CTC shall purchase securities only from authorized institutions or firms.

The Investment Officer, or the investment advisor, shall obtain competitive offers on all purchases of investment instruments purchased on the secondary market whenever possible. A competitive bid can be executed through a bidding process involving at least three separate brokers/financial institutions or through the use of a nationally recognized trading platform.

VII. Safekeeping and Custody

1. Delivery vs. Payment

All trades of marketable securities will be executed on a delivery vs. payment (DVP) basis to ensure that securities are deposited in Alameda CTC's safekeeping institution prior to the release of funds.

2. Third-Party Safekeeping

Securities will be held by an independent third-party safekeeping institution selected by Alameda CTC's Investment Officers. All securities will be evidenced by safekeeping receipts in Alameda CTC's name. The safekeeping institution shall annually provide a copy of its most recent report on internal controls – Service Organization Control Reports (formerly SAS 70) prepared in accordance with the Statement on Standards for Attestation Engagements (SSAE) No. 16 (effective June 15, 2011.)

3. Internal Controls

The Investment Officers are responsible for establishing, maintaining and documenting an internal control structure designed to ensure that the assets of Alameda CTC are protected from loss, theft or misuse. The controls shall be designed to prevent the loss of public funds arising from fraud, employee error, misrepresentation by third parties, unanticipated changes in financial markets, or imprudent actions by employees and officers of Alameda CTC.

VIII. Authorized Investments

The following investments will be permitted by this policy and are those authorized in the California Government Code.

- 1. United States Treasury notes, bonds, bills, or certificates of indebtedness, or those for which the faith and credit of the United States are pledged for the payment of principal and interest.
 - a. Maximum maturity: 5 years
 - b. Maximum percent of portfolio: 100%
- 2. Federal agency or United States government-sponsored enterprise obligations, participations, or other instruments, including those issued by or fully guaranteed as to principal and interest by federal agencies or United States government-sponsored enterprises.
 - a. Maximum maturity: 5 years
 - b. Maximum percent of portfolio: 100%
 - c. Type: Senior and fully guaranteed debt obligations
 - d. Maximum per issuer: 35%
- 3. Repurchase Agreements used solely as short-term investments.

The following collateral restrictions will be observed: Only U.S. Treasury securities or Federal Agency securities, as described in VIII 1 and 2 above, will be acceptable collateral. All securities underlying Repurchase Agreements must be delivered to Alameda CTC's custodian bank versus payment or be handled under a tri-party repurchase agreement. The total of all collateral for each Repurchase Agreement must equal or exceed, on the basis of market value plus accrued interest, 102 percent of the total dollar value of the money invested by Alameda CTC for the term of the investment. Since the market value of the underlying securities is subject to daily market fluctuations, the investments in repurchase agreements shall be in compliance if the value of the underlying securities is brought back up to 102 percent no later than the next business day. For any Repurchase Agreement with a term of more than one day, the value of the underlying securities must be reviewed on a regular basis.

Market value must be calculated each time there is a substitution of collateral.

Alameda CTC or its trustee shall have a perfected first security interest under the Uniform Commercial Code in all securities subject to Repurchase Agreement.

Alameda CTC may enter into Repurchase Agreements with (1) primary dealers in U.S. Government securities who are eligible to transact business with, and who report to, the Federal Reserve Bank of New York, and (2) California and non-California banking institutions having assets in excess of \$25 billion and having debt rated in the highest short-term rating category as provided by a nationally recognized statistical rating organization.

Alameda CTC will enter into a Master Repurchase Agreement, substantially in the form approved by the Securities Industry and Financial Markets Association (SIFMA) and by Alameda CTC's counsel, with each firm with which it enters into Repurchase Agreements.

- a. Maximum maturity: 90 days
- b. Maximum percent of portfolio: 20%
- 4. Obligations of the State of California or any local agency within the state, including bonds payable solely out of revenues from a revenue-producing property owned, controlled or operated by the state or any local agency or by a department, board, agency or authority of the state or any local agency.
 - a. Maximum maturity: 5 years
 - b. Maximum percent of portfolio: 10%
 - c. Minimum credit quality: A (S&P); or A2 (Moody's); or A (Fitch)
 - d. Maximum per issuer: 5%
- 5. Registered treasury notes or bonds of any of the other 49 states in addition to California, including bonds payable solely out of revenues from a revenue-producing property owned, controlled or operated by the state or by a department, board, agency or authority of any of the other 49 states, in addition to California.
 - a. Maximum maturity: 5 years
 - b. Maximum percent of portfolio: 10%
 - c. Minimum credit quality: A (S&P); or A2 (Moody's): or A (Fitch)
 - d. Maximum per issuer: 5%
- 6. Bankers' Acceptances, otherwise known as bills of exchange or time drafts which are drawn on and accepted by a commercial bank.
 - a. Maximum maturity: 180 days
 - b. Maximum percent of portfolio: 40%
 - c. Minimum credit quality: A_1 (S&P); or P_1 (Moody's); or F_1 (Fitch)
 - d. Maximum per issuer: 5%
- 7. Commercial paper rated in the highest two short-term rating categories, as provided by a nationally recognized statistical rating organization. The entity that issues the commercial paper shall meet all of the following conditions: (a) is organized and operating in the United States as a general corporation; (b) has total assets in excess of five hundred million dollars (\$500,000,000); and (c) has debt other than commercial paper, if any, that is rated "A" or higher by a nationally recognized statistical-rating organization.
 - a. Maximum maturity: 270 days
 - b. Maximum percent of portfolio: <u>2540</u>%
 - c. Minimum credit quality: A₁ (S&P); or P₁ (Moody's); or F₁ (Fitch)
 - d. Maximum per issuer: 5%

No more than 40% of the total portfolio may be invested cumulatively in commercial paper or asset-backed commercial paper as defined in Section 8 below. No more than 10% of the outstanding commercial paper of any single issuer may be purchased. No more than 10% of the total portfolio may be invested in the commercial paper and medium-term notes of a single issuer.

- 8. Asset-backed commercial paper of "prime" quality and issued by an entity organized within the United States as a special purpose corporation, trust, or limited liability company. The entity that issues the asset-backed commercial paper must meet all of the following conditions: (a) is rated "A-1" (or the equivalent) or higher by at least one nationally recognized statistical rating organization; and (b) has programwide credit enhancements including, but not limited to, overcollateralization, letters of credit, or a surety bond.
 - a. Maximum maturity: 270 days
 - b. Maximum percent of portfolio: 40%
 - c. Minimum credit quality: A-1 (S&P); or P-1 (Moody's); or F-1 (Fitch)
 - d. Maximum per issuer: 5%

No more than 40% of the total portfolio may be invested cumulatively in assetbacked commercial paper or commercial paper as defined in Section 7 above. No more than 10% of the outstanding commercial paper of any single issuer may be purchased. No more than 10% of the total portfolio may be invested in the commercial paper and medium-term notes of a single issuer.

- 8.9. Medium-term notes, defined as all corporate and depository institution debt securities with a maximum remaining maturity of five years or less, issued by corporations organized and operating within the United States or by depository institutions licensed by the U.S. or any state and operating within the U.S. Medium-term corporate notes shall be rated a minimum of "A" or its equivalent by a nationally recognized statistical rating organization.
 - a. Maximum maturity: 5 years
 - b. Maximum percent of portfolio: 30%
 - c. Minimum credit quality: A (S&P); or A2 (Moody's); or A (Fitch)
 - d. Maximum per issuer: 5%

No more than 10% of the total portfolio may be invested in the medium-term notes and commercial paper of a single issuer.

- 9:10. Asset-backed securities, including any consumer receivable pass-through certificate, equipment lease-backed certificate, consumer receivable backed bond, or other pay-through bond with a maximum maturity of five years or less. Asset-backed securities shall be rated "AAA" or its equivalent or better by a nationally recognized statistical rating organization.
 - a. Maximum Maturity: 5 years
 - b. Maximum percent of portfolio: 20%

- c. Minimum credit quality: AAA (S&P); or Aaa (Moody's); or AAA (Fitch)
- d. Maximum per issuer: 5%
- <u>10.11.</u> FDIC insured or fully collateralized time certificates of deposit in financial institutions located in California.
 - a. Maximum maturity: 1 year
 - b. Maximum percent of portfolio: 10%
 - c. Maximum per issuer: 5%
- 11.12. Negotiable certificates of deposit or deposit notes issued by a nationally or state-chartered bank, a savings association or a federal association, a state or federal credit union, or by a federally licensed or state-licensed branch of a foreign bank.
 - a. Maximum maturity: 3 years
 - b. Maximum percent of portfolio: 30%
 - c. Minimum credit quality: A (S&P); or A2 (Moody's); or A (Fitch)
 - d. Maximum per issuer: 5%

12.13. State of California Local Agency Investment Fund (LAIF)

Although LAIF may invest in securities not permitted in the Alameda CTC's Investment Policy, such investments shall not exclude LAIF from the Alameda CTC's list of eligible investments, provided that LAIF's periodic reports allow the Investment Officer to adequately assess the risk inherent in LAIF's portfolio. Funds invested in LAIF will follow LAIF policies and procedures.

a. Maximum dollar limit: as determined by LAIF

The LAIF portfolio shall be reviewed annually in order to monitor its continuing suitability as an investment option for the Alameda CTC.

13.14. The California Asset Management Program (CAMP)

a. Maximum dollar limit: double the LAIF limit

The CAMP shall be reviewed annually in order to monitor its continuing suitability as an investment option for Alameda CTC. Funds invested in CAMP will follow CAMP policies and procedures.

14.15. Shares of beneficial interest issued by diversified management companies that are money market funds registered with the Securities and Exchange Commission under the Investment Company Act of 1940 (15 U.S.C. Sec. 80a-1, et seq.). To be eligible for investment pursuant to this subdivision, these companies shall either: (1) attain the highest ranking or the highest letter and numerical rating provided by not less than two nationally recognized statistical rating organizations; or (2) retain an investment advisor registered or exempt from

registration with the Securities and Exchange Commission with not less than five years experience managing money market mutual funds with assets under management in excess of five hundred million dollars (\$500,000,000).

- a. Maximum percent of portfolio: 20%
- b. Maximum per Prime Money Market Fund: 5%
- c. Maximum per Government Money Market Fund: 10%
- d. Minimum credit quality: AAAm (S&P); or Aaa-mf (Moody's); AAAmmf (Fitch)
- 15.16. United States dollar denominated senior unsecured unsubordinated obligations issued or unconditionally guaranteed by the International Bank for Reconstruction and Development, International Finance Corporation, or Inter-American Development Bank and eligible for purchase and sale within the United States.
 - a. Maximum maturity: 5 years
 - b. Maximum percent of portfolio: 10%
 - c. Minimum credit quality: AA (S&P); or Aa (Moody's); or AA (Fitch)

Important Notes:

- a) The percentage limitation for all categories of investments <u>and individual issuers</u> refers to the percentage in the overall Alameda CTC portfolio on the date the security or shares are purchased<u>as measured by the settlement date</u>.
- b) The credit rating requirements of this Investment Policy shall apply at the time of purchase. If the credit rating of a security is downgraded below the minimum required rating level for a new investment of that security type subsequent to its purchase, the investment advisor shall promptly notify the Investment Officer. The Investment Officer shall evaluate the downgrade on a case-by-case basis in order to determine if the security should be held or sold. The Investment Officer will apply the general objectives of safety, liquidity, yield and legality to make the decision.

IX. Ineligible Investments

Any security type or structure not specifically approved by this policy is hereby specifically prohibited. Security types which are thereby prohibited include, but are not limited to:

- 1. "Complex" derivative structures such as range notes, dual index notes, inverse floaters, leveraged or de-leveraged floating-rate notes, or any other complex variable-rate or structured note;
- 2. Interest-only strips that are derived from a pool of mortgages, or any security that could result in zero interest accrual if held to maturity except for the purchase of securities issued or backed by the United States government in the event of, and for the duration of, a period of negative market interest rates;
- 3. Non-agency mortgage-backed pass-through securities;

- 4. Other non-agency mortgage-backed securities; and
- 5. Non-agency collateralized mortgage obligations.

X. Investment Parameters

- 1. Credit Risk Credit risk is the risk that a security or a portfolio will lose some or all of its value due to a real or perceived change in the ability of the issuer to repay its debt. The diversification requirements included in Section VIII are designed to mitigate credit risk. Alameda CTC shall additionally mitigate credit risk by adopting the following diversification strategies:
 - a. Avoiding overconcentration in any one issuer or business sector;
 - b. Limiting investments in securities with higher credit risks; and
 - c. Maintaining a portion of the portfolio in a highly liquid investment such as LAIF
- 2. Market Risk Market risk is the risk that the portfolio will fluctuate due to changes in the general level of interest rates. Alameda CTC recognizes that, over time, longer-term portfolios have the potential to achieve higher returns. On the other hand, longer-term portfolios have higher volatility of return. Alameda CTC shall mitigate market risk by providing adequate liquidity for short-term cash needs, and by making some longer-term investments only with funds that are not needed for current cash flow purposes. Alameda CTC further recognizes that certain types of securities, including variable rate securities, securities with principal paydowns prior to maturity, and securities with embedded options, will affect the market risk profile of the portfolio differently in different interest rate environments. Alameda CTC, therefore, adopts the following strategies to control and mitigate its exposure to market risk:
 - a. Alameda CTC shall invest in securities with varying maturities, maintaining a minimum of three months of budgeted operating expenditures in short term investments to provide sufficient liquidity for expected disbursements;
 - b. The maximum percent of callable securities in the portfolio shall be 25%;
 - c. The maximum stated final maturity of individual securities in the portfolio shall be five years, except as otherwise stated in this policy;
 - d. Liquidity funds will be held in LAIF, CAMP or in money market instruments maturing within one year or less or held in securities with maturities matched to anticipated expenditures;
 - e. Longer term/Core funds will be defined as the funds in excess of liquidity requirements. The investments in this portion of the porfolio will have maturities between 1 day and 5 years and will only be invested in higher quality and liquid securities; and

- f. The duration of the <u>Core or benchmarked portion of the portfolio shall at all</u> times be approximately equal to the duration of a Market Benchmark Index selected by Alameda CTC based on Alameda CTC's investment objectives, constraints and risk tolerances, plus or minus 25%. <u>Duration flexibility is</u> necessary because of the short-term benchmarks utilized on the portfolio due to capital project cashflow demands.
- 3. Maximum percentages for a particular issuer or investment type may be exceeded at a point in time subsequent to the purchase of a particular issuer or investment type. Securities need not be liquidated to realign the portfolio; however, consideration should be given to this matter when future purchases are made to ensure that appropriate diversification is maintained.

XI. Performance and Program Evaluation

The investment portfolio will be managed in accordance with the parameters specified within this policy. The portfolio should obtain a market average rate of return during a market/economic environment of stable interest rates. A series of appropriate benchmarks shall be established against which portfolio performance shall be compared on a regular basis. The benchmarks shall be reflective of the actual securities being purchased and risks undertaken and the benchmarks shall have a similar weighted average maturity and credit profile commensurate with investment risk constraints and liquidity needs of Alameda CTC.

Alameda CTC may periodically update the performance benchmarks to reflect current investment objectives and constraints and shall communicate such changes to the investment advisor.

Appendix I

AUTHORIZED INVESTMENTS SUMMARY TABLE

INVESTMENT	% OF PORTFOLIO		PURCHASE RESTRICTIONS		IMUM URITY	MINIMUM CREDIT QUALITY		
	Per Cal. Gov't Code	Alameda CTC Policy	Alameda CTC Policy	Per Cal. Gov't Code	Alameda CTC Policy	Per Cal. Gov't Code	Alameda CTC Policy	
U.S. Treasury Notes, Bonds, Bills or Certificates of Indebtedness	100%	100%	None	5 years	5 years	NA	NA	
Federal or U.S. Sponsored Obligations fully guaranteed by Federal Agencies or U.S. Government Sponsored Enterprises	100%	100%	Max 35% per issuer	5 years	5 years	NA	Senior <u>and Fully</u> <u>Guaranteed</u> Debt	
Repurchase Agreements	NA	20%	Strict collateral requirements; Master Repurchase Agreement	1 year	90 days	NA	NA	
State of California and California Local Agency Bonds	NA	10%	Max 5% per issuer	5 years	5 years	NA	A (S&P) or A2 (Moody's) or A (Fitch)	
Bonds of any of the other 49 states in addition to California	NA	10%	Max 5% per issuer	5 years	5 years	NA	A (S&P) or A2 (Moody's) or A (Fitch)	
Bankers' Acceptances	40%	40%	Max 5% per issuer	180 days	180 days	NA	A_1 (S&P) or P_1 (Moody's) or F_1 (Fitch)	

INVESTMENT	% OF PORTFOLIO		PURCHASE RESTRICTIONS		IMUM URITY	MINIMUM CREDIT QUALITY		
	Per Cal. Gov't Code	Alameda CTC Policy	Alameda CTC Policy	Per Cal. Gov't Code	Alameda CTC Policy	Per Cal. Gov't Code	Alameda CTC Policy	
Commercial paper of U <u>.S.</u> corporations with total assets exceeding \$500,000,000	25<u>40</u>%	25<u>40</u>%	Max 510% of outstanding paper of any single issuer & max 5% of portfolio of any one issuer	270 days	270 days	A <u>-</u> 1 or P <u>-</u> 1 or F <u>-</u> 1	A_1 (S&P) or P_1 (Moody's) or F_1 (Fitch)	
Asset-backed commercial paper issued by entities organized in the U.S.	<u>40%</u>	<u>40%</u>	Max 10% of outstanding paper of any single issuer & max 5% of portfolio of any one issuer	<u>270 days</u>	<u>270 days</u>	<u>A-1 or P-1 or</u> <u>F-1</u>	A-1 (S&P) or P-1 (Moody's) or F-1 (Fitch	
Medium Term Corporate Notes of U.S. Corporations	30%	30%	Max 5% per issuer	5 years	5 years	А	A (S&P) or A2 (Moody's) or A (Fitch)	
Asset-Backed Securities	20%	20%	Max 5% per issuer	5 years	5 years	AA	AAA (S&P) or Aaa (Moody's) or AAA (Fitch)	
California Collateralized Time Deposits	NA	10%	Max 5% per issuer	NA	1 year	NA	NA	
Negotiable Certificate of Deposits	30%	30%	Max 5% per issuer	5 years	3 years	NA	A (S&P) or A2 (Moody's) or A (Fitch)	
State of California- Local Agency Investment Fund (LAIF)	NA	NA	As limited by LAIF	NA	NA	NA	NA	
California Asset Management Program	NA	NA	Double the LAIF limit	NA	NA	NA	NA	
Shares of Beneficial Interests (Money Market Funds)	20%	20%	Max 5% per Prime fund, Max 10% per Government fund	NA	N/A	AAA	AAAm (S&P) or Aaa-mf (Moody's) or AAAmmf (Fitch)	

Alameda CTC Investment Policy May 202<u>1</u>0

INVESTMENT	% OF PORTFOLIO		PURCHASE RESTRICTIONS		IMUM URITY	MINIMUM CREDIT QUALITY	
	Per Cal. Gov't Code	Alameda CTC Policy	Alameda CTC Policy	Per Cal. Gov't Code	Alameda CTC Policy	Per Cal. Gov't Code	Alameda CTC Policy
Obligations issued or unconditionally guaranteed by the International Bank for Reconstruction and Development, International Finance Corporation, or Inter-American Development Bank	30%	10%	NA	5 years	5 years	АА	AA (S&P) or Aa (Moody's) or AA (Fitch)

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Patricia Reavey, Deputy Executive Director of Finance and Administration
SUBJECT:	Approve and adopt restatements of the Alameda CTC Cafeteria Plan

Recommendation

It is recommended that the Commission approve and adopt restatements of the Alameda CTC Cafeteria Plan (the Plan), one each for calendar year 2020 and calendar year 2021. If approved by the Commission, the restated Cafeteria Plan for 2020 would technically be effective January 1, 2020, and the restated Cafeteria Plan for 2021 would technically be effective January 1, 2021. These changes are proposed in response to COVID-19 related relief in accordance with the Consolidated Appropriations Act, 2021 IRS Notice 2021-15 and the American Rescue Plan of 2021 which were enacted to provide economic relief to families.

Summary

The Plan was established by Alameda CTC in 2012 to permit eligible employees to pay for their share of contributions for benefits on a pre-tax basis, and to contribute on a pre-tax salary reduction basis to an account for reimbursement of eligible medical care expenses, referred to as a Flexible Spending Account or and FSA, and/or an account for reimbursement of certain dependent care expenses, referred to as a Dependent Care Assistance Program or a DCAP. This Cafeteria Plan is a qualified plan under Code Section 125 and is administered as such.

The updates outlined in Attachment A are specific to the 2020 plan year, and those in Attachment B are specific the 2021 plan year. These changes are proposed in response to COVID-19 related relief in accordance with the Consolidated Appropriations Act, 2021 IRS Notice 2021-15 and the American Rescue Plan of 2021 which were enacted to provide economic relief to families.

In summary, the updates include the following, as outlined in Attachment A, for the 2020 plan year:

- An extended grace period during which employees and employees terminated during 2020 can make claims for reimbursement of expenses incurred through December 31, 2021 if submitted by April 30, 2022, up to the amount the employee has set aside through payroll deductions for medical expenses and/or dependent care expenses through the end of the plan year or their termination date,
- Allows for a tax dependent of the participant who attains age thirteen during 2020 to be considered a qualifying individual as it pertains to the DCAP, and
- Other administrative clean-up of plan language.

In summary, the updates include the following, as outlined in Attachment B, for the 2021 plan year:

- An extended grace period during which employees and employees terminated during 2021 can make claims for reimbursement of expenses incurred through December 31, 2022 if submitted by April 30, 2023, up to the amount the employee has set aside through payroll deductions for medical expenses and/or dependent care expenses through the end of the plan year or their terminations date,
- Allows for a tax dependent of the participant who attains age thirteen during 2021 to be considered a qualifying individual as it pertains to the DCAP,
- Allows for limited mid-year changes to participant's salary reduction elections, and
- Doubles the maximum election amount allowed on the FSA, going from \$2,500 to \$5,000, and on the DCAP, going from \$5,250 to \$10,500.

Alameda CTC recommends these updates to the Plan to allow staff to take advantage of the relief intended by the Consolidated Appropriations Act and the American Rescue Plan Act of 2021.

Background

Alameda CTC adopted the Plan in 2012 to permit eligible employees to pay for their share of contributions for benefits on a pre-tax basis, and to contribute on a pre-tax salary reduction basis to an account for reimbursement of eligible medical care expenses and/or an account for reimbursement of certain dependent care expenses. The Plan has not been amended or restated since that time.

In response to the COVID-19 pandemic with the goal of providing economic relief to families, the Consolidated Appropriations Act 2021 was enacted in December 2020, and the American Rescue Plan Act of 2021 was enacted in January 2021 (the Acts), for which the Internal Revenue Service issued Notice 2021-15 in February to clarify guidance. These Acts allow for the updates recommended in Attachment A and Attachment B. Because the Acts apply different rules to the 2020 plan year versus the 2021 plan year, the changes have been shown in two separate documents for which the Plan restated as of January 1, 2021

(Attachment B) will replace the Plan restated as of January 1, 2020 (Attachment A) automatically when appropriate.

The recommend updates to the Plan make the benefit to employees more generous due to its pretax nature, but since all reimbursements are made from funds that have been set aside through payroll deductions from participants, it does not increase the cost to the agency.

Fiscal Impact: There is no fiscal budget impact related to the approval and adoption of the restated Cafeteria plans.

Attachments:

- A. Alameda County Transportation Commission Cafeteria Plan As Restated Effective January 1, 2020 (Draft Restatement in Redline)
- B. Alameda County Transportation Commission Cafeteria Plan As Restated Effective January 1, 2021 (Draft Restatement in Redline)
- C. Resolution Approving and Adopting Restated Cafeteria Plans for Calendar Years 2020 and 2021

This page intentionally left blank
ALAMEDA COUNTY TRANSPORTATION COMMISSION CAFETERIA PLAN

As Adopted Restated Effective February January 1, 20122020

2962648.2



TABLE OF CONTENTS

1.1 Establishment of Plan. 1 1.2 Legal Status 1 ARTICLE III. Eligibility and Participation 6 3.4 Eligibility to Participate 6 3.2 Use of Contributions 6 3.4 Participation Following Termination of Employment or Loss of Eligibility 7 3.6 FMLA Leaves of Absence 9 ARTICLE IV. Method and Timing of Elections 9 4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections Puring Open Enrolment Pariad 9 4.4 Elections Puring Open Enrolment Pariad 9 4.5 Elections Puring Open Enrolment Pariad 9 4.6 Irrovecability of Elections 10 6.7 Benefite Offered 10 6.4 Benefitie Offered 10 6.5.2 Source of Benefit Funding 10 6.3 Employer Contributions for Cost of Coverage 11 6.4 Insurance Panefit: CogRA 12 7.4 Health FSA Component 12 7.5 Establish	ARTICL	E I. Introduction	1
4.2 Legal Statue 1 ARTICLE II. Eligibility and Participation 6 3.4 Eligibility to Participation 6 3.2 Use of Contributione 6 3.3 Termination of Participation 6 3.4 Participation Following Termination of Employment or Loss of Eligibility 7 3.5 FMLA Leaves of Absence 9 4.1 Election to Participate 9 4.2 Election to Participate 9 4.3 Elections During Open Enrollment Period 9 4.4 Elections During Open Enrollment Period 9 4.5 Elections During Open Enrollment Period 9 4.6 Irrevocability of Elections 10 5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 6.4 Benefits Offered and Method of Funding 10 5.3 Employer Contributions 11 6.4 Isouras of Elections 11 6.5 Benefits Provided Under the Insurance Plane 11 6.4 <th>1 1</th> <th>Establishment of Plan</th> <th>1</th>	1 1	Establishment of Plan	1
ARTICLE II. Eligibility and Participation 6 3.1 Eligibility to Participate 6 3.2 Use of Contributions 6 3.4 Participation of Participation 6 3.4 Participation Following Termination of Employment or Less of Eligibility 7 3.5 FMLA Leaves of Absence 7 3.6 Non-FMLA Leaves of Absence 9 4.1 Election to Participate 9 4.2 Elections Puring Open Encollment Period 9 4.3 Elections During Open Encollment Period 9 4.4 Faiture of Eligible 9 4.5 Irrevocability of Elections 10 5.4 Benefits Offered and Method of Funding 10 5.4 Benefits Offered and Method of Funding 10 5.3 Employeer Contributions 11 6.4 Benefits College 11 6.5 Source of Benefits College 11 6.4 Insurance Benefits (COBRA 12 7.4 Health FSA Component 12 7.4 Health FSA Component 12 7.4	1.1		
ARTICLE III. Eligibility and Participation 6 3.1 Eligibility to Participate 6 3.2 Use of Contributions 6 3.3 Termination of Participation 6 3.4 Participation Following Termination of Employment or Loss of Eligibility 7 3.6 Non-FMLA Leaves of Absence 9 4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections When First Eligible 9 4.4 Fallure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Iscians Offered and Method of Funding 10 5.4 Benefits Offered and Method of Funding 10 5.2 Source of Benefits Funding 10 5.3 Employer Contributions 11 6.4 Benefits Offered 10 5.2 Source of Benefits Funding 10 5.3 Employer Contributions 11 6.4 Benefits Contributions 11 6.5 Benefits Provided Under the Insurance Plans 11 6.4 Benefits Provided Under the I		с. С	
3.1 Eligibility to Participate 6 3.2 Use of Contributions 6 3.3 Termination of Participation 6 3.4 Participation Following Termination of Employment or Loss of Eligibility 7 3.5 FMLA Leaves of Absence 9 3.6 Non-FMLA Leaves of Absence 9 4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections When First Eligible 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irivocability of Elections 10 5.4 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 6.4 Benefits Offered 10 5.3 Employer Contributions for Cost of Coverage 11 6.4 Benefits Cotra 6.4 Insurance Plane 12 7.4 Health FSA Component 12 7.4 Health FSA Accounts for Cost of Coverage 12	ARTICL	E II. Definitions	<u> </u>
3.1 Eligibility to Participate 6 3.2 Use of Contributions 6 3.3 Termination of Participation 6 3.4 Participation Following Termination of Employment or Loss of Eligibility 7 3.5 FMLA Leaves of Absence 9 3.6 Non-FMLA Leaves of Absence 9 4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections When First Eligible 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irivocability of Elections 10 5.4 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 6.4 Benefits Offered 10 5.3 Employer Contributions for Cost of Coverage 11 6.4 Benefits Cotra 6.4 Insurance Plane 12 7.4 Health FSA Component 12 7.4 Health FSA Accounts for Cost of Coverage 12		E III. Eligibility and Participation	6
3.2 Ues of Contributions 6 3.3 Termination of Participation 6 3.4 Participation Following Termination of Employment or Lose of Eligibility 7 3.5 FMLA Leaves of Absence 7 3.6 Non-FMLA Leaves of Absence 9 ARTICLE IV. Method and Timing of Elections 9 4.1 Election to Participate 9 4.2 Election During Open Enrollment Period 9 4.3 Election Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered 10 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 6.4 Benefits Offered 10 5.3 Employer Contributions for Cost of Coverage 11 6.4 Insurance Benefits (COBRA 12 7.4 Health FSA Component 12 7.5 Eligible Medical Care Expences for Health FSA 12 7.4 Health FSA Accounts: Use			
3.3 Termination of Participation 6 3.4 Participation Following Termination of Employment or Lose of Eligibility 7 3.5 FMLA Leaves of Absence 7 3.6 Non FMLA Leaves of Absence 9 ARTICLE IV. Method and Timing of Elections 9 4.1 Election to Participate 9 4.2 Elections During Open Enrollment Period 9 4.3 Elections During Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Hirevocability of Elections 10 5.4 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions for Cost of Coverage 11 6.4 Insurance Benefits: COBRA 11 6.3 Benefits 11 6.4 Insurance Benefits 12 7.4 Health FSA Component 12 7.5 Participant Contributions for Cost of Coverage of Health FSA 12 7.4 Health FSA Component 12 7.5			6
3.4 Participation Following Tormination of Employment or Loss of Eligibility 7 3.5 FMLA Leaves of Absence 7 3.6 Non-FMLA Leaves of Absence 9 ARTICLE IV. Method and Timing of Elections 9 4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections Proving Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevecability of Elections 10 5.1 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions for Cost of Coverage 11 6.4 Benefits 11 6.5 Participant Contributions for Cost of Coverage 11 6.4 Insurance Benefits: COBRA 12 7.4 Health FSA Component 12 7.4 Health FSA Accounts; Use or			••••••
3.5 FMLA Leaves of Absence			6
3.6 Non-FMLA Leaves of Absence. 9 ARTICLE IV. Method and Timing of Elections. 9 4.1 Election to Participate. 9 4.2 Elections When First Eligible 9 4.3 Elections Uning Open Enrollment Period. 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 5.1 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 6.4 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits: COBRA 12 ARTICLE VII. Health FSA Component 12 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA 12 7.3 Eligible Medical Care Exponses for Health FSA 12 7.4 Health FSA Account; Use or Lose Rule 13 7.5 Forbilutro of Healt		—Participation Following Termination of Employment or Loss of Eligibility	7
ARTICLE IV. Method and Timing of Elections 9 4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections During Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE V. Premium Payment Component 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits: COBRA 12 ARTICLE VI. Health FSA Component 12 12 ARTICLE VII. Health FSA Component 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.4 Health FSA Account 15 7.5 Establishment of Health FSA Account 15	0.0		
4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections During Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits; COBRA 12 7.4 Health FSA Component 12 7.5 Establishment of Health FSA 12 7.4 Health FSA Account; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.6 Forfeiture of Health FSA Account; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Coverage for DCAP Benefits 17	3.6 —	Non-FMLA Leaves of Absence	9
4.1 Election to Participate 9 4.2 Elections When First Eligible 9 4.3 Elections During Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits; COBRA 12 7.4 Health FSA Component 12 7.5 Establishment of Health FSA 12 7.4 Health FSA Account; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.6 Forfeiture of Health FSA Account; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Coverage for DCAP Benefits 17	ARTICL	E IV. Method and Timing of Elections	9
4.2 Elections When First Eligible 9 4.3 Elections During Open Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrovocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered and Method of Funding 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component. 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits 12 7.4 Health FSA Component 12 7.5 Establishment of Health FSA 12 7.4 Health FSA Accounts; Use-or-Lose Rule 15 7.5 Establishment of Health FSA Accounts; Use-or-Lose Rule 16 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule 16 7.7 Reimbursement S From Health FSA After Termination of Participation; COBRA 17 7.6 Establishment of Health FSA Accounts; Use-or-Lose Rule 16		-	
4.3 Elections During Open Enrollment Period 9 4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Plans 11 6.3 Benefits; COBRA 12 7.4 Health FSA Component 12 7.5 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.4 Hoalth FSA Account 12 7.5 Establishment of Health FSA Account: Use or Lose Rule 15 7.6 Forfuture of Health FSA Account: Use or Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 15 7.6 Forfuture of Health FSA Ac			
4.4 Failure of Eligible Émployee to File an Election Form/Salary Reduction Agreement 10 4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits 12 7.4 Health FSA Component 12 7.4 Health FSA Component 12 7.5 Eligible Medical Care Expenses for Health FSA 13 7.5 Establishment of Health FSA Accounts; Use or-Lose Rule 16 7.7 Reimburscement Claime Procedure for Health FSA 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 17 7.6 Forficiture of Health FSA Accounts; Use or-Lose Rule 16 7.7 Reimburscement C			
4.5 Irrevocability of Elections 10 ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component 11 6.1 Bonefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits Provided Under the Insurance Plans 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.4 Health FSA Account 13 7.5 Establishment of Health FSA Account 15 7.6 Forfeiture of Health FSA Account 15 7.7 Reimbursement From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits	4.3	Elections During Open Enrollment Period	
ARTICLE V. Benefits Offered and Method of Funding 10 5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component. 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Health FSA Account 12 7.5 Establishment of Health FSA Account 13 7.6 Forfeiture of Health FSA Account; Use or Lose Rule 16 7.7 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits			
5.1 Benefits Offered 10 5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component 11 6.1 Benefits 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 12 7.1 Health FSA Component 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Modical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account; Use or Lose Rule 16 7.6 Forfoiture of Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Cove			
5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component. 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits; COBRA 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component. 12 7.4 Health FSA Component. 12 7.4 Health FSA Benefits 12 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.4 Health FSA Account. 12 7.5 Eligible Medical Care Exponses for Health FSA 13 7.6 Eorfeiture of Health FSA Account. 15 7.6 Forfeiture of Health FSA Account. 15 7.6 Forfeiture of Health FSA Account. 16 7.7 Reimbursement Claims Procedure for Health FSA. 16 7.7 Reimbursement S From Health FSA After Termination of Participation; COBRA 17 7.9 Co	ARTICL	E V. Benefits Offered and Method of Funding	10
5.2 Source of Benefit Funding 10 5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component. 11 6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits 11 6.4 Insurance Benefits; COBRA 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component. 12 7.4 Health FSA Component. 12 7.4 Health FSA Benefits 12 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.4 Health FSA Account. 12 7.5 Eligible Medical Care Exponses for Health FSA 13 7.6 Eorfeiture of Health FSA Account. 15 7.6 Forfeiture of Health FSA Account. 15 7.6 Forfeiture of Health FSA Account. 16 7.7 Reimbursement Claims Procedure for Health FSA. 16 7.7 Reimbursement S From Health FSA After Termination of Participation; COBRA 17 7.9 Co	E 4	Demofile Offered	10
5.3 Employer Contributions 11 ARTICLE VI. Premium Payment Component. 11 6.1 Benofite 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benofite Provided Under the Insurance Plans 11 6.4 Insurance Benefite; COBRA 12 ARTICLE VII. Health FSA Component 12 7.4 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.4 Health FSA Accounts 12 7.4 Health FSA Accounts; Use-or-Lose Rule 13 7.5 Establishment of Health FSA Accounts; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.7 Reimbursement Short Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 </td <td>-</td> <td></td> <td></td>	-		
ARTICLE VI. Premium Payment Component	· · -		
6.1 Benefits 11 6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits Provided Under the Insurance Plans 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Accounts; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Tormination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Accounts; Use-It-or-Lose-It Rule 22			
6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits Provided Under the Insurance Plans 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account 15 7.6 Forfeiture of Health FSA Accounts; Use or Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Accounts 22 8.6 <td< td=""><td>ARTICL</td><td>E VI. Premium Payment Component</td><td>11</td></td<>	ARTICL	E VI. Premium Payment Component	11
6.2 Participant Contributions for Cost of Coverage 11 6.3 Benefits Provided Under the Insurance Plans 11 6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account 15 7.6 Forfeiture of Health FSA Accounts; Use or Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Accounts 22 8.6 <td< td=""><td>61</td><td>Banafite</td><td>11</td></td<>	61	Banafite	11
6.3 Benefits Provided Under the Insurance Planc. 11 6.4 Insurance Benefits; COBRA. 12 ARTICLE VII. Health FSA Component. 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Accounts; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Tormination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Component 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Accounts; Use-It-or-Lose-It Rule 2	0.11	-Participant Contributions for Cost of Coverage	11
6.4 Insurance Benefits; COBRA 12 ARTICLE VII. Health FSA Component 12 7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Accounts; Use-or-Lose Rule 15 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 17 7.9 Coordination of Benefits 18 ARTICLE VIII. DCAP Component 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedur			
ARTICLE VII. Health FSA Component	0.0		
7.1 Health FSA Benefits 12 7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Accounts 15 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 22 8.7 Reimbursement Claims Procedure for DCAP 23	0.1		
7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account. 15 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule. 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	ARTICL	E VII. Health FSA Component	12
7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits 12 7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account. 15 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule. 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	71-	Health FSA Benefits	12
7.3 Eligible Medical Care Expenses for Health FSA 12 7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account. 15 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule. 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Account 22 8.7 Reimbursement Claims Procedure for DCAP 23	7.2		
7.4 Maximum and Minimum Benefits for Health FSA 13 7.5 Establishment of Health FSA Account. 15 7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule. 16 7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23		-Eligible Medical Care Expenses for Health FSA	12
7.5 Establishment of Health FSA Account			13
7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 ARTICLE VIII. DCAP Component 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23			
7.7 Reimbursement Claims Procedure for Health FSA 16 7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 ARTICLE VIII. DCAP Component 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	7.6	Forfeiture of Health FSA Accounts: Use-or-Lose Rule	
7.8 Reimbursements From Health FSA After Termination of Participation; COBRA 17 7.9 Coordination of Benefits 18 ARTICLE VIII. DCAP Component 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23			
ARTICLE VIII. DCAP Component 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	7.8		
ARTICLE VIII. DCAP Component 18 8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	7.9 —	Coordination of Benefits	18
8.1 DCAP Benefits 18 8.2 Participant Contributions for Cost of Coverage for DCAP Benefits 18 8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23			
8.2 Participant Contributions for Cost of Coverage for DCAP Benefits	ARTICL	E VIII. DCAP Component	
8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	8.1-		
8.3 Eligible Dependent Care Expenses 19 8.4 Maximum and Minimum Benefits for DCAP 20 8.5 Establishment of DCAP Account 22 8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule 22 8.7 Reimbursement Claims Procedure for DCAP 23	<u>8.2</u>		
8.5 Establishment of DCAP Account	8.3	Eligible Dependent Care Expenses	
8.5 Establishment of DCAP Account	8.4—		20
8.7—Reimbursement Claims Procedure for DCAP	8.5 —		22
8.7—Reimbursement Claims Procedure for DCAP	8.6 —		
8.8 Reimbursements From DCAP After Termination of Participation	8.7 —		23
	8.8 —		24

ARTICL	E IX. HIPAA PROVISIONS FOR HEALTH FSA	24
9.1	-Provision of Protected Health Information to Employer	24
9.2	-Permitted Disclosure of Enrollment/Disenrollment Information	24
9.3 —	-Permitted Uses and Disclosure of Summary Health Information	24
9.4 —		25
9.5	-Conditions of Disclosure for Plan Administration Purposes	
9.6 —	Adequate Separation Between Plan and Employer	26
9.7	-Certification of Plan Sponsor	
9.8	-Privacy Official	
9.9 —	-Interpretation and Limited Applicability	
9.10	-Service Performed for the Employer	
ARTICL	E X. Irrevocability of Elections; Exceptions	27
10.1	Irrevocability of Elections	27
10.2 -		28
10.3	-Events Permitting Exception to Irrevocability Rule for All Benefits.	28
10.4 —	Election Modifications Required by Plan Administrator	
ARTICL	E XI. Appeals Procedure	34
11.1-	-Procedure If Benefits Are Denied Under This Plan	34
11.2	-Claims Procedures for Insurance Benefits	
ARTICL	E XII. Recordkeeping and Administration	35
12.1-	– Plan Administrator	35
12.1	Powers of the Plan Administrator	
12.2	Reliance on Participant, Tables, etc	
12.5	Provision for Third-Party Plan Service Providers	
12.4	-Fiduciary Liability	
12.6	Compensation of Plan Administrator	
12.0	Insurance Contracts	
12.2	-Inability to Locate Payee	
12.0 12.9	Effect of Mistake	
	E XIII. General Provisions	
13.1-		37
13.1 13.2	- Expenses	
13.2	Amendment and Termination	37 37
10.0	Governing Law	-
	Compliance With Code and Other Applicable Laws	،ر <u></u> 22
13.5 12.6	No Guarantee of Tax Consequences	37 20
	- No Guarantee of Flax Consequences	
	Non-Assignability of Rights	38 38
10.9 10.10	Headings Plan Provisions Controlling	
	Severability	
	E I. Introduction	
<u>1.1</u> 1.2	ESTABLISHMENT OF PLAN	
<u></u>	E II. Definitions	
<u>4K HCL</u> 2.1		
	E III. Eligibility and Participation	
<u>3.1</u>	ELIGIBILITY TO PARTICIPATE	<u></u> 6

<u>3.2</u>	USE OF CONTRIBUTIONS	
<u>3.3</u>	TERMINATION OF PARTICIPATION	<u></u> 6
<u>3.4</u>	PARTICIPATION FOLLOWING TERMINATION OF EMPLOYMENT OR LOSS OF ELIGIBILITY	
<u>3.5</u>	FMLA LEAVES OF ABSENCE	
<u>3.6</u>	NON-FMLA LEAVES OF ABSENCE	<u></u> 9
ARTICL	E IV. Method and Timing of Elections	9
<u>4.1</u>	ELECTION TO PARTICIPATE	
4.2	ELECTIONS WHEN FIRST ELIGIBLE	<u></u> 9
<u>4.3</u>	ELECTIONS DURING OPEN ENROLLMENT PERIOD	<u></u> 9
4.4	FAILURE OF ELIGIBLE EMPLOYEE TO FILE AN ELECTION FORM/SALARY REDUCTION AGREEMENT	<u></u> 10
<u>4.5</u>	IRREVOCABILITY OF ELECTIONS	
	E V. Benefits Offered and Method of Funding	
<u>5.1</u>	BENEFITS OFFERED	<u></u> 10
<u>5.2</u>	SOURCE OF BENEFIT FUNDING	<u></u> 10
<u>5.3</u>	EMPLOYER CONTRIBUTIONS	<u></u> 11
	E VI. Premium Payment Component	11
ANTICL		
<u>6.1</u>	Benefits	<u></u> 11
<u>6.2</u>	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE	
<u>6.3</u>	BENEFITS PROVIDED UNDER THE INSURANCE PLANS	
<u>6.4</u>	INSURANCE BENEFITS; COBRA	<u></u> 12
ARTICL	E VII. Health FSA Component	12
7.1	HEALTH FSA BENEFITS	12
7.2	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE OF HEALTH FSA BENEFITS	
7.3	ELIGIBLE MEDICAL CARE EXPENSES FOR HEALTH FSA	12
7.4	MAXIMUM AND MINIMUM BENEFITS FOR HEALTH FSA	
7.5	ESTABLISHMENT OF HEALTH FSA ACCOUNT	15
7.6	FORFEITURE OF HEALTH FSA ACCOUNTS; USE-OR-LOSE RULE	16
7.7	REIMBURSEMENT CLAIMS PROCEDURE FOR HEALTH FSA	16
7.8	REIMBURSEMENTS FROM HEALTH FSA AFTER TERMINATION OF PARTICIPATION; COBRA	
<u>7.9</u>	COORDINATION OF BENEFITS	<u></u> 18
ARTICL	E VIII. DCAP Component	<u></u> 18
8.1	DCAP BENEFITS	18
8.2	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE FOR DCAP BENEFITS	
8.3	ELIGIBLE DEPENDENT CARE EXPENSES	
8.4	MAXIMUM AND MINIMUM BENEFITS FOR DCAP	
8.5	ESTABLISHMENT OF DCAP ACCOUNT	
8.6	FORFEITURE OF DCAP ACCOUNTS; USE-IT-OR-LOSE-IT RULE	
8.7	REIMBURSEMENT CLAIMS PROCEDURE FOR DCAP	
8.8	REIMBURSEMENT CLAIMS PROCEDURE FOR DCAP REIMBURSEMENTS FROM DCAP AFTER TERMINATION OF PARTICIPATION	24
ARTICL	E IX. HIPAA PROVISIONS FOR HEALTH FSA	
	PROVISION OF PROTECTED HEALTH INFORMATION TO EMPLOYER	
<u>9.1</u> 9.2	PROVISION OF PROTECTED HEALTH INFORMATION TO EMPLOYER PERMITTED DISCLOSURE OF ENROLLMENT/DISENROLLMENT INFORMATION	
<u>9.2</u> 9.3	PERMITTED DISCLOSURE OF ENROLLMENT/DISENROLLMENT INFORMATION	
<u>9.3</u> 9.4	PERMITTED OSES AND DISCLOSURE OF SUMMARY HEALTH INFORMATION PERMITTED AND REQUIRED USES AND DISCLOSURE OF PHI FOR PLAN ADMINISTRATION PURPOS	<u></u> 24
3.4	25	153
9.5		25
<u>9.5</u> 9.6	ADEQUATE SEPARATION BETWEEN PLAN AND EMPLOYER	
<u>9.0</u> 9.7	CERTIFICATION OF PLAN SPONSOR	
<u>9.7</u> 9.8	PRIVACY OFFICIAL	
<u>9.9</u>	INTERPRETATION AND LIMITED APPLICABILITY	

<u>9.10</u>	SERVICE PERFORMED FOR THE EMPLOYER	27	
ARTICLE X. Irrevocability of Elections; Exceptions			
10.1	IRREVOCABILITY OF ELECTIONS		
10.2	PROCEDURE FOR MAKING NEW ELECTION IF EXCEPTION TO IRREVOCABILITY APPLIES		
10.3	EVENTS PERMITTING EXCEPTION TO IRREVOCABILITY RULE FOR ALL BENEFITS		
10.4	ELECTION MODIFICATIONS REQUIRED BY PLAN ADMINISTRATOR	34	
ARTICLE	XI. Appeals Procedure	34	
<u>11.1</u>	PROCEDURE IF BENEFITS ARE DENIED UNDER THIS PLAN	34	
<u>11.2</u>	CLAIMS PROCEDURES FOR INSURANCE BENEFITS	<u></u> 35	
	XII. Recordkeeping and Administration	<u></u> 35	
<u>12.1</u>	PLAN ADMINISTRATOR		
12.2	POWERS OF THE PLAN ADMINISTRATOR		
12.3	RELIANCE ON PARTICIPANT, TABLES, ETC.		
<u>12.4</u>	PROVISION FOR THIRD-PARTY PLAN SERVICE PROVIDERS	<u></u> 36	
12.5	FIDUCIARY LIABILITY		
12.6	FIDUCIARY LIABILITY	<u></u> 36	
<u>12.7</u>	INSURANCE CONTRACTS		
<u>12.8</u>	INABILITY TO LOCATE PAYEE.	<u></u> 36	
<u>12.9</u>	EFFECT OF MISTAKE	<u></u> 37	
ARTICLE XIII. General Provisions			
13.1	EXPENSES		
13.2	NO CONTRACT OF EMPLOYMENT		
13.3	AMENDMENT AND TERMINATION		
13.4	GOVERNING LAW		
13.5	COMPLIANCE WITH CODE AND OTHER APPLICABLE LAWS		
13.6	No GUARANTEE OF TAX CONSEQUENCES.	<u></u> 38	
13.7	INDEMNIFICATION OF EMPLOYER	<u></u> 38	
13.8	NON-ASSIGNABILITY OF RIGHTS	<u></u> 38	
<u>13.9</u>	HEADINGS	<u></u> 38	
<u>13.10</u>	PLAN PROVISIONS CONTROLLING		
13.11	SEVERABILITY		

ALAMEDA COUNTY TRANSPORTATION COMMISSION CAFETERIA PLAN

As AdoptedRestated Effective FebruaryJanuary 1, 20122020

ARTICLE I. Introduction

1.1 Establishment of Plan

Alameda County Transportation Commission (the Employer) hereby establishes the<u>restates</u> Alameda County Transportation Commission Cafeteria Plan (the Plan) effective FebruaryJanuary 1, 20122020 (the Effective Date)..) to incorporate applicable COVID-19 related relief in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15. The Plan was established effective as of February 1, 2012. Capitalized terms used in this Plan that are not otherwise defined have the meanings set forth in Article II.

This Plan is designed to permit an Eligible Employee to pay for his or her share of contributions for the Benefit Plan Options in Appendix A on a pre-tax Salary Reduction basis, and to contribute on a pre-tax Salary Reduction basis to an Employee's account for reimbursement of certain Medical Care Expenses (Health FSA Account) and/or to an account for reimbursement of certain Dependent Care Expenses (DCAP Account).

1.2 Legal Status

This Plan is intended to qualify as a cafeteria plan under Code Section 125 and will be interpreted and administered consistent with the requirements of Code Section 125 and the regulations issued thereunder.

The Health FSA Component is intended to qualify as a self-insured medical reimbursement plan under Code Section 105, and the Medical Care Expenses reimbursed thereunder are intended to be eligible for exclusion from participating Employees' gross income under Code Section 105(b). The DCAP Component is intended to qualify as a dependent care assistance program under Code Section 129, and the Dependent Care Expenses reimbursed thereunder are intended to be eligible for exclusion from participating Employees' gross income under Code Section 129(a).

Although reprinted within this document, the Health FSA Component and the DCAP Component are separate plans for purposes of administration and all reporting and nondiscrimination requirements imposed by Code Sections 105 and 129. The Health FSA Component is also a separate plan for purposes of applicable provisions of HIPAA and COBRA. In the event that the Health FSA Component is determined not to be a separate plan, the Plan will be designated as a hybrid entity for purposes of HIPAA, such that it will be a covered entity only with respect to the Health FSA Component.

ARTICLE II. Definitions

2.1 Definitions

Account(s) means the Health FSA Accounts and the DCAP Accounts described in Sections 7.5 and 8.5.

Benefit Plan Option means a qualified benefit under Code Section 125(f) that is available to a Participant under this Plan as set forth in Appendix A, as amended from time to time. The Employer may substitute, add, subtract, or revise at any time the menu of such Benefit Plan Options and/or the benefits, terms, and conditions of any such options or plans. Any such substitution, addition, subtraction, or revision will be communicated to Participants and will automatically be incorporated by reference under this Plan.

Board of Directors means the Board of Commissioners of the Alameda County Transportation Commission.

Change in Status means any of the events described below, as well as any other events included under subsequent changes to Code Section 125 or regulations issued thereunder, which the Plan Administrator, in its sole discretion and on a uniform and consistent basis, determines are permitted under IRS regulations and under this Plan:

- (a) *Legal Marital Status.* A change in a Participant's legal marital status, including marriage, death of a Spouse, divorce, legal separation, or annulment;
- (b) *Number of Dependents.* Events that change a Participant's number of Dependents, including birth, death, adoption, and placement for adoption;
- (c) Employment Status. Any of the following events that change the employment status of the Participant or his or her Spouse or Dependents: (1) a termination or commencement of employment; (2) a strike or lockout; (3) a commencement of or return from an unpaid leave of absence; (4) a change in worksite; and (5) the eligibility conditions of this Plan or other employee benefits plan of the Participant or his or her Spouse or Dependents depend on the employment status of that individual and there is a change in that individual's status with the consequence that the individual becomes (or ceases to be) eligible under this Plan or other employee benefits plan;
- (d) *Dependent Eligibility Requirements.* An event that causes a Dependent to satisfy or cease to satisfy the Dependent eligibility requirements for a particular benefit, such as attaining a specified age, student status, or any similar circumstance; and
- (e) *Change in Residence.* A change in the place of residence of the Participant or his or her Spouse or Dependents.

COBRA means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.

Code means the Internal Revenue Code of 1986 and the Treasury Regulations issued thereunder, as amended.

Compensation means the cash wages or salary paid to an Employee by the Employer.

DCAP means dependent care assistance program.

DCAP Account means the account described in Section 8.5.

DCAP Benefits has the meaning described in Section 8.1.

DCAP Component means the component of this Plan described in Article VIII.

Dependent means: (a) for purposes of accident or health coverage, (1) a dependent as defined in Code Section 152, determined without regard to subsections (b)(1), (b)(2), and (d)(1)(B) thereof, (2) any child (as defined in Code Section 152(f)(1) of the Participant who as of the end of the taxable year has not attained age 27, and (3) any child of the Participant to whom IRS Revenue Procedure 2008-48 applies (regarding certain children of divorced or separated parents who receive more than half of their support for the calendar year from one or both parents and are in the custody of one or both parents for more than half of the calendar year); and (b) for purposes of the DCAP Component, a Qualifying Individual. Notwithstanding the foregoing, the Health FSA Component will provide benefits in accordance with the applicable requirements of any National Medical Support Order, even if the child does not meet the definition of Dependent.

Dependent Care Expenses has the meaning described in Section 8.3.

Earned Income will have the meaning given such term in Code Section 129(e)(2).

Effective Date of this Plan means FebruaryJanuary 1, 20122020.

Election Form/Salary Reduction Agreement means the form provided by the Plan Administrator for the purpose of allowing an Eligible Employee to participate in this Plan by electing Benefit Plan Options(s) and authorizing Salary Reductions to pay for any of the Benefit Plan Options.

Eligible Employee means an Employee eligible to participate in this Plan, as provided in Section 3.1.

Employee means an individual that the Employer classifies as a common-law employee and who is on the Employer's W-2 payroll, but does not include the following: (a) any leased employee (including but not limited to those individuals defined as leased employees in Code Section 414(n) or an individual classified by the Employer as a contract worker, independent contractor, temporary employee, or casual employee for the period during which such individual is so classified, whether or not any such individual is on the Employer's W-2 payroll or is determined by the IRS or others to be a common-law employee of the Employer; (b) any individual who performs services for the Employer but who is paid by a temporary or other employment or staffing agency for the period during which such individual is paid by such agency, whether or not such individual is determined by the IRS or others to be a common-law employee of the Employer; and (c) any employee covered under a collective bargaining agreement, unless that agreement provides for the employee's participation in the Plan. The term Employee does include former Employees for the limited purpose of allowing continued eligibility for benefits under the Plan for a limited duration following termination of employment provided any required contributions are made and only to the extent specifically provided under this Plan.

Employer means the Alameda County Transportation Commission.

FMLA means the Family and Medical Leave Act of 1993, as amended.

Grace Period means the period that begins immediately following the close of a Plan Year and ends on the day that is two months plus 15 days following the close of that Plan Year. Notwithstanding the foregoing, for purposes of the Plan Year ending December 31, 2020, the Grace Period is the period from January 1, 2021 through December 31, 2021. (Note: For purposes of the DCAP, as the Grace Period is available pursuant to the relief provided in the Consolidated Appropriations Act, 2021, the Grace Period is available for amounts in DCAP Accounts that remain as of December 31, 2020 only. The Grace Period is available under the Health FSA irrespective of the relief.)

Health FSA means the health flexible spending arrangement.

Health FSA Account means the account described in Section 7.5.

Health FSA Benefits has the meaning described in Section 7.1.

Health FSA Component means the component of this Plan described in Article VII.

HIPAA means the Health Insurance Portability and Accountability Act of 1996, as amended.

Insurance Plan(s) means the plan(s) that the Employer maintains for its Employees (and for their Spouses, same-sex spouses, domestic partners, and Dependents that may be eligible under the terms of such plan), which provide benefits through a group insurance policy or policies (e.g., medical, dental and vision insurance). The Employer may substitute, add, subtract, or revise at any time the menu of such plans and/or the benefits, terms, and conditions of any such plans. Any such substitution, addition, subtraction, or revision will be communicated to Participants and will automatically be incorporated by reference under this Plan.

Medical Care Expenses has the meaning described in Section 7.3.

Nonelective Contribution(s) means any amount that the Employer, in its sole discretion, may contribute under the Plan to provide benefits for individual Participants and their Spouses, Dependents, domestic partners, and same-sex spouses, as applicable, under one or more of the Benefit Plan Options offered under the Plan.

Open Enrollment Period means the period during the Plan Year during which Eligible Employees may elect to participate in the Plan or make changes to their elections for the next Plan Year. The Employer will determine this period each Plan Year, which the Plan Administrator will make known in the Plan's open enrollment materials.

Participant means a person who is an Eligible Employee and who is participating in this Plan in accordance with the provisions of Article III. Participants include those who elect one or more Benefit Plan Options under the Plan.

Period of Coverage means the Plan Year, with the following exceptions: (a) for Employees who first become eligible to participate, it will mean the portion of the Plan Year following the date on which participation commences, as described in Section 4.2; and (b) for Employees who terminate participation, it will mean the portion of the Plan Year prior to the date on which participation terminates, as described in Section 3.3.

Plan means the Alameda County Transportation Commission Cafeteria Plan as set forth herein, as amended from time to time.

Plan Administrator means the person(s), entity, or committee as may be appointed from time to time by the Board of Directors (or its authorized designee) to administer the Plan. If no such person, entity, or committee is appointed, the Plan Administrator is the Employer.

Plan Sponsor means the Employer.

Plan Year means the calendar year (i.e., the 12-month period commencing January 1 and ending on December 31), except in the case of a short plan year representing the initial Plan Year or where the Plan Year is being changed, in which case the Plan Year will be the entire short plan year.

Premium Payment Benefits means the Premium Payment Benefits described in Section 6.1.

Premium Payment Component means the component of this Plan described in Article VI.

Qualifying Dependent Care Services has the meaning described in Section 8.3.

Qualifying Individual means (a) a tax dependent of the Participant as defined in Code Section 152 who is under the age of 13 and who is the Participant's qualifying child as defined in Code Section 152(a)(1); (b) a tax dependent of the Participant as defined in Code Section 152, but determined without regard to subsections (b)(1), (b)(2), and (d)(1)(B) thereof, who is physically or mentally incapable of self-care and who has the same principal place of abode as the Participant for more than half of the year; or (c) a Participant's Spouse who is physically or mentally incapable of self-care, and who has the same principal place of abode as the Participant for more than half of the year. Notwithstanding the foregoing, in the case of divorced or separated parents, a Qualifying Individual who is a child will, as provided in Code Section 21(e)(5), be treated as a Qualifying Individual of the custodial parent (within the meaning of Code Section 152(e)) and will not be treated as a Qualifying Individual with respect to the noncustodial parent. Furthermore, notwithstanding the foregoing, in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15 and subject to Section 8.6, for purposes of the 2020 Plan Year, the term "Qualifying Individual" shall also include a tax dependent of the Participant as defined in Code Section 152(a)(1) who attains the age of thirteen (13) at any time during the 2020 calendar year, who has the same principal place of abode as the Participant for more than one-half of the Plan Year, and who has not provided over one-half of his or her own support for the Plan Year.

Salary Reduction means the amount by which the Participant's Compensation is reduced and applied by the Employer under this Plan to pay for one or more of the benefits, as permitted for the applicable component, before any applicable state and/or federal taxes have been deducted from the Participant's Compensation (i.e., on a pre-tax basis).

Spouse means an individual who is legally married to a Participant as determined under applicable state law and who is treated as a spouse under the Code. A domestic partner or same-sex spouse is not treated as a spouse under the Code. Notwithstanding the above, for purposes of the DCAP Component the term Spouse does not include (a) an individual legally separated from the Participant under a divorce or separate maintenance decree; or (b) an individual who, although married to the Participant, files a separate federal income tax return, maintains a principal residence separate from the Participant during the last six months of the taxable year, and does not furnish more than half of the cost of maintaining the principal place of abode of the Participant.

Student means an individual who, during each of five or more calendar months during the Plan Year, is a full-time student at any educational organization that normally maintains a regular faculty and curriculum and normally has an enrolled student body in attendance at the location where its educational activities are regularly carried on.

ARTICLE III. Eligibility and Participation

3.1 Eligibility to Participate

All full-time Employees are eligible to participate in the Plan. To become a Participant, an Eligible Employee must make a timely election to participate in accordance with Article IV. Eligibility for any Benefit Plan Option will be subject to the requirements specified in the governing plan documents of the applicable Benefit Plan Option. The provisions of this Article are not intended to override any eligibility requirement or waiting period specified in the applicable Benefit Plan Options and the terms of eligibility and participation for any Benefit Plan Option offered under the Plan are subject to the requirements specified in the Benefit Plan Option's governing documents.

3.2 Use of Contributions

As a Participant, an Employee will be permitted to (1) elect Benefit Plan Options for which he or she is eligible, (2) receive available Nonelective Contributions for which he or she is eligible in the manner set forth in the enrollment materials, (3) pay his or her share of the cost of his or her elected benefits with Salary Reduction contributions, and (4) if permitted under the terms of the Benefit Plan Options and uniform rules adopted by the Plan Administrator, pay his or her share of the costs of the elected benefits with after-tax dollars (e.g., if Salary Reduction contributions are not available or are insufficient to pay his or her share of the cost of the Benefit Plan Option). In addition, as a Participant, an Employee may be permitted to elect health coverage for an individual who is not the employee's Spouse or Dependent if permitted under the terms of the Benefit Plan Options and in accordance with uniform rules adopted by the Plan Administrator; provided, however, that the fair market value of such coverage will be included in the Employee's gross income to the extent required by applicable law, and the Employee will be treated as having purchased the coverage with after-tax dollars.

3.3 Termination of Participation

A Participant will cease to be a Participant in this Plan upon the earlier of:

(a) the date the Participant makes a permitted election not to participate in the Plan;

- (b) the date that the Participant no longer satisfies the eligibility requirements of this Plan or all of the Benefit Plan Options. Notwithstanding the foregoing, for purposes of pre-tax COBRA coverage, certain Employees may continue eligibility for certain periods subject to the restrictions and terms otherwise described in this Plan; or
- (c) The date that the Plan is either terminated or amended to exclude the Participant or the class of employees to which the Participant belongs.

Termination of participation in this Plan will automatically revoke the Participant's elections. Benefits under any Insurance Plan will terminate as of the date(s) specified in the Insurance Plan. Reimbursements from the Health FSA and DCAP Accounts after termination of participation will be made pursuant to Section 7.8 for Health FSA Benefits and Section 8.8 for DCAP Benefits. If revocation occurs under this Section 3.3, no new election may be made by such Participant during the remainder of the Plan Year except as set forth in Section 3.4.

3.4 Participation Following Termination of Employment or Loss of Eligibility

If a Participant terminates his or her employment for any reason, including (but not limited to) disability, retirement, layoff, or voluntary resignation, or otherwise loses eligibility and then is rehired or becomes eligible once again within 30 days or less after the date of a termination of employment or loss of eligibility, then the Employee will be reinstated with the same elections that such individual had before termination or other loss of eligibility. If a former Participant is rehired more than 30 days following termination of employment or becomes eligible after 30 days following a loss of eligibility and is otherwise eligible to participate in the Plan, then the individual may make new elections as a new hire as described in Section 4.2. Notwithstanding the above, an election to participate in the Premium Payment Component will be reinstated only to the extent that coverage under the applicable Insurance Plan is reinstated.

3.5 FMLA Leaves of Absence

(a) *Health Insurance Benefits.* Notwithstanding any provision to the contrary in this Plan, if a Participant goes on a qualifying leave under the FMLA, then to the extent required by the FMLA, the Employer will continue to maintain the Participant's health insurance benefits and Health FSA Benefits on the same terms and conditions as if the Participant were still an active Employee. That is, if the Participant elects to continue his or her coverage while on leave, the Employer will continue to pay its share of the contributions for those benefits under this Plan.

An Employer may require participants to continue all health insurance benefits and Health FSA Benefits coverage for Participants while they are on paid leave (provided that Participants on non-FMLA paid leave are required to continue coverage). If so, the Participant's share of the contributions will be paid by the method normally used during any paid leave (e.g., on a pre-tax Salary Reduction basis).

In the event of unpaid FMLA leave (or paid FMLA leave where coverage is not required to be continued), a Participant may elect to continue his or her health insurance benefits and Health FSA Benefits during the leave. If the Participant elects to continue coverage while on FMLA leave, then the Participant may pay his or her share of the contributions in one of the following ways:

- with after-tax dollars, by sending monthly payments to the Employer by the due date established by the Employer;
- with pre-tax dollars, by having such amounts withheld from the Participant's ongoing Compensation (if any), including unused sick days and vacation days, or pre-paying all or a portion of the contributions for the expected duration of the leave on a pre-tax Salary Reduction basis out of pre-leave Compensation. To pre-pay the contributions, the Participant must make a special election to that effect prior to the date that such Compensation would normally be made available (pre-tax dollars may not be used to fund coverage during the next Plan Year); or
- under another arrangement agreed upon between the Participant and the Plan Administrator (e.g., the Plan Administrator may fund coverage during the leave and withhold "catch-up" amounts from the Participant's Compensation on a pretax or after-tax basis) upon the Participant's return.

If the Employer requires all Participants to continue health insurance benefits and Health FSA Benefits during an unpaid FMLA leave, then the Participant may elect to discontinue payment of the Participant's required contributions until the Participant returns from leave. Upon returning from leave, the Participant will be required to repay the contributions not paid by the Participant during the leave. Payment will be withheld from the Participant's Compensation either on a pre-tax or after-tax basis, as agreed to by the Plan Administrator and the Participant.

If a Participant's health insurance benefits or Health FSA Benefits coverage ceases while on FMLA leave (e.g., for non-payment of required contributions), then the Participant is permitted to re-enter the Premium Payment Component or Health FSA Component as applicable, upon return from such leave on the same basis as when the Participant was participating in the Plan prior to the leave, or as otherwise required by the FMLA. In addition, the Plan may require Participants whose health insurance benefits or Health FSA Benefits coverage terminated during the leave to be reinstated in such coverage upon return from a period of unpaid leave, provided that Participants who return from a period of unpaid, non-FMLA leave are required to be reinstated in such coverage. Notwithstanding the preceding sentence, with regard to Health FSA Benefits, a Participant whose coverage ceased will be permitted to elect whether to be reinstated in the Health FSA Benefits at the same coverage level as was in effect before the FMLA leave (with increased contributions for the remaining period of coverage) or at a coverage level that is reduced pro rata for the period of FMLA leave during which the Participant did not pay contributions. If a Participant elects a coverage level that is reduced pro rata for the period of FMLA leave, then the amount withheld from a Participant's Compensation on a pay-period-by-pay-period basis for the purpose of paying for reinstated Health FSA Benefits will be equal to the amount withheld prior to the period of FMLA leave.

(b)-____Non-Health Benefits. If a Participant goes on a qualifying leave under the FMLA, then entitlement to non-health benefits (such as DCAP Benefits) is to be determined by the Employer's policy for providing such benefits when the Participant is on non-FMLA leave, as described in Section 3.6. If such policy permits a Participant to discontinue contributions while on leave, then the Participant will, upon returning from leave, be required to repay the contributions not paid by the Participant during the leave. Payment will be withheld from the Participant's Compensation either on a pre-tax or after-tax basis, as may be agreed upon by the Plan Administrator and the Participant or as the Plan Administrator otherwise deems appropriate.

3.6 Non-FMLA Leaves of Absence

If a Participant goes on an unpaid leave of absence that does not affect eligibility, then the Participant will continue to participate and the contributions due for the Participant will be paid by pre-payment before going on leave, by after-tax contributions while on leave, or with catchup contributions after the leave ends, as may be determined by the Plan Administrator. If a Participant goes on an unpaid leave that affects eligibility, then the applicable election change rules in Section 10.3 will apply.

ARTICLE IV. Method and Timing of Elections

4.1 Election to Participate

To become a Participant, an Eligible Employee must submit a completed and signed Election Form/Salary Reduction Agreement to the Plan Administrator in the time and in the manner required by the Plan Administrator.

4.2 Elections When First Eligible

- (a) *Currently Eligible Employees.* An Employee who is eligible to participate in this Plan as of the Effective Date must complete, sign, and file an Election Form/Salary Reduction Agreement with the Plan Administrator during the election period (as specified by the Plan Administrator) immediately preceding the Effective Date of the Plan to become a Participant on the Effective Date. The elections made by the Eligible Employee on this initial Election Form/Salary Reduction Agreement will be effective for the Plan Year beginning on the Effective Date.
- (b) New Employees or Newly Eligible Employees. An Employee who first becomes eligible to participate in the Plan mid-year (and after the Effective Date) may elect to commence participation in the Plan after the eligibility requirements of Section 3.1 have been satisfied by completing, signing, and filing an Election Form/Salary Reduction Agreement with the Plan Administrator in the time and in the manner required by the Plan Administrator. Participation in the Plan will commence on the first day of the month following the Plan Administrator's receipt of a properly completed and signed Election Form/Salary Reduction Agreement. An Employee who does not elect benefits when first eligible may not enroll until the next Open Enrollment Period, unless an event occurs that would justify a mid-year election change, as described under Section 10.3. Eligibility for Premium Payment Benefits will be subject to the additional requirements, if any, specified in the applicable Insurance Plans.

4.3 Elections During Open Enrollment Period

During each Open Enrollment Period with respect to a Plan Year, the Plan Administrator will provide an Election Form/Salary Reduction Agreement to each Employee who is eligible to

participate in this Plan. The Election Form/Salary Reduction Agreement will enable the Employee to elect to participate in the various components of this Plan for the next Plan Year and to authorize the necessary Salary Reductions to pay for the benefits elected. The Election Form/Salary Reduction Agreement must be returned to the Plan Administrator on or before the last day of the Open Enrollment Period, and it will become effective on the first day of the next Plan Year. If an Eligible Employee fails to return the Election Form/Salary Reduction Agreement during the Open Enrollment Period, then the Employee may not elect any benefits under this Plan until the next Open Enrollment Period, unless an event occurs that would justify a mid-year election change, as described under Section 10.3.

4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement

If an Eligible Employee fails to file an Election Form/Salary Reduction Agreement within the time period described in Sections 4.2 and 4.3, then the Employee may not elect any benefits under the Plan (a) until the next Open Enrollment Period; or (b) until an event occurs that would justify a mid-year election change, as described under Section 10.3. Notwithstanding any contrary provision in the Plan, if an Employee who fails to file an Election Form/Salary Reduction Agreement is eligible for benefits under an Insurance Plan and has made an effective election for such benefits outside the Plan, then the Employee's share of the contributions for such benefits will automatically be paid with pre-tax dollars and will be deemed a "default election" under the Plan. Such default elections cannot be changed until such time as the Employee files, during a subsequent Open Enrollment Period (or after an event occurs that would justify a mid-year election change as described under Section 10.3), a timely Election Form/Salary Reduction Form/Salary Reduction Agreement to elect Premium Payment Benefits. No default elections are permitted for Health FSA or DCAP Benefits.

4.5 Irrevocability of Elections

Unless an exception applies (as described in Article X), a Participant's election under the Plan is irrevocable for the duration of the Period of Coverage to which it relates.

ARTICLE V. Benefits Offered and Method of Funding

5.1 Benefits Offered

When first eligible or during the Open Enrollment Period as described under Article IV, Participants will be given the opportunity to elect one or more of the following benefits:

- (a) Premium Payment Benefits, as described in Article VI;
- (b) Health FSA Benefits, as described in Article VII.
- (c) DCAP Benefits, as described in Article VIII.

In no event will benefits under the Plan be provided in the form of deferred compensation. Notwithstanding the foregoing, amounts remaining in a Participant's Health FSA Account at the end of a Plan Year can be used to reimburse the Participant for Medical Care Expenses that are incurred during the Grace Period immediately following the close of that Plan Year as provided in Article VII. No Grace Period is available for DCAP Benefits.

5.2 Source of Benefit Funding

The cost of coverage under the component Benefit Plan Options will be funded by a Participant's Salary Reductions, Nonelective Contributions provided by the Employer, or a combination of the foregoing. The required contributions for each of the Benefit Plan Options offered under the Plan will be made known to employees in annual enrollment materials. Salary Reduction Contributions that are allocated to any Benefit Plan Option will equal the contributions required from the Participant less any available Nonelective Contributions allocated to that option. A Participant may elect to receive Nonelective Contributions in the form of cash to the extent described in the applicable annual enrollment materials. The maximum amount of employee contributions, plus any Nonelective Contributions made available by the Employer, will not exceed the aggregate cost of the Benefit Plan Options elected.

5.3 Employer Contributions

The Employer may, in its sole discretion, make Nonelective Contributions on behalf of a Participant toward the cost of one or more Benefit Plan Options. The amount of Nonelective Contributions that may be applied towards the cost of each of the Benefit Plan Option(s) for any Participant will be subject to the sole discretion of the Employer and may be adjusted upward or downward at any time in the Employer's sole discretion. The amount will be calculated for each Plan Year in a uniform and nondiscriminatory manner and may be based upon the Participant's dependent status, commencement or termination date of the Participant's employment during the Plan Year, and such other factors as the Employer may prescribe.

No provision of this Plan will be construed to require the Employer or Plan Administrator to maintain any fund or segregate any amount for the benefit of any Participant, and no Participant or other person will have any claim against, right to, or security or other interest in, any fund, account or asset of the Employer from which any payment under the Plan may be made. The Plan does not create a trust in favor of a Participant or any person claiming on a Participant's behalf.

ARTICLE VI. Premium Payment Component

6.1 Benefits

An Eligible Employee can elect to participate in the Premium Payment Component by electing (a) to receive benefits under the Insurance Plans described in Appendix A; and (b) to pay for his or her share of the contributions for those benefits on a pre-tax Salary Reduction basis. Unless an exception applies (as described in Article X), such election is irrevocable for the duration of the Period of Coverage to which it relates. Notwithstanding any other provision in this Plan, insurance benefits under the Insurance Plans are subject to the terms and conditions of the Insurance Plans, and no changes can be made with respect to such plans (such as mid-year changes in election) if such changes are not permitted under the applicable Insurance Plan.

6.2 Participant Contributions for Cost of Coverage

The annual contribution for a Participant's portion of the Premium Payment Benefits is equal to the amount as set by the Employer in the annual enrollment materials.

6.3 Benefits Provided Under the Insurance Plans

Insurance benefits will be provided by the Insurance Plans in accordance with their governing documents, and not this Plan. The types and amounts of insurance benefits, the requirements for participating in the Insurance Plans, and the other terms and conditions of coverage and benefits of such plans are set forth in their governing documents. All claims to receive benefits under the Insurance Plans will be subject to and governed by the terms and conditions of the Insurance Plans and the rules, regulations, policies, and procedures adopted in accordance with those plans, as may be amended from time to time.

6.4 Insurance Benefits; COBRA

Notwithstanding any provision to the contrary in this Plan, to the extent required by COBRA, a Participant and his or her Spouse and Dependents, as applicable, whose health coverage terminates under an Insurance Plan because of a COBRA qualifying event (and who is a qualified beneficiary as defined under COBRA), will be given the opportunity to continue on a self-pay basis the same health coverage that he or she had under the applicable Insurance Plan the day before the qualifying event for the periods prescribed by COBRA. Such continuation coverage will be subject to all conditions and limitations under COBRA.

Contributions for COBRA coverage under an Insurance Plan may be paid on a pre-tax basis for current Employees receiving taxable compensation (as may be permitted by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year) where COBRA coverage arises either (a) because the Employee ceases to be eligible because of a reduction in hours; or (b) because the Employee's Dependent ceases to satisfy the eligibility requirements for coverage. For all other individuals (e.g., Employees who cease to be eligible because of retirement, termination of employment, or layoff), contributions for COBRA coverage for Insurance Plan benefits will be paid on an after-tax basis (unless as may be otherwise permitted by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year).

ARTICLE VII. Health FSA Component

7.1 Health FSA Benefits

An Eligible Employee can elect to participate in the Health FSA Component by electing (a) to receive benefits in the form of reimbursements for Medical Care Expenses from the Health FSA (Health FSA Benefits); and (b) to pay his or her contribution for such Health FSA Benefits on a pre-tax Salary Reduction basis. Unless an exception applies (as described in Article X), any such election is irrevocable for the duration of the Period of Coverage to which it relates.

7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits

The annual contribution for a Participant's portion of the Health FSA Benefits is equal to the annual benefit amount elected by the Participant, subject to the dollar limits set forth in the annual enrollment materials.

7.3 Eligible Medical Care Expenses for Health FSA

Under the Health FSA Component, a Participant may receive reimbursement for Medical Care Expenses incurred during the Period of Coverage for which an election is in force. In addition,

certain individuals may receive reimbursement for Medical Care Expenses incurred during the Grace Period immediately following the close of a Plan Year from amounts remaining in their Health FSA Accounts for that Plan Year in accordance with Section 7.4(e).

- (a) *Incurred*. A Medical Care Expense is incurred at the time the medical care or service giving rise to the expense is furnished and not when the Participant is formally billed for, is charged for, or pays for the medical care.
- (b) Medical Care Expenses. "Medical Care Expenses" means expenses incurred by a Participant or his or her Spouse or Dependents for medical care, as defined in Code Section 213(d), but only to the extent that the expense has not been reimbursed through insurance or otherwise. If only a portion of a Medical Care Expense has been reimbursed elsewhere, then the Health FSA can reimburse the remaining portion of such Medical Care Expense if it otherwise meets the requirements of this Article VII. Notwithstanding the foregoing, the term Medical Care Expenses does not include:
 - premium payments for other health coverage, including but not limited to health insurance premiums for any other plan (whether or not sponsored by the Employer);
 - medicines or drugs, unless the medicine or drug is a prescribed drug (determined without regard to whether the medicine or drug is available without a prescription) or is insulin (for this purpose, the Plan Administrator will have sole discretion to determine, on a uniform and consistent basis, whether a particular item is a medicine or drug and whether the requirement of a prescription has been satisfied);
 - cosmetic surgery or other similar procedures, unless the surgery or procedure is necessary to ameliorate a deformity arising from, or directly related to, a congenital abnormality, a personal injury resulting from an accident or trauma, or a disfiguring disease (for this purpose, "cosmetic surgery" means any procedure that is directed at improving the patient's appearance and does not meaningfully promote the proper function of the body or prevent or treat illness or disease); or
 - any other expense excluded under Appendix B or otherwise under the terms of this Plan.

The Plan Administrator may promulgate procedures regarding the eligibility of various expenses for reimbursement as Medical Care Expenses and may limit reimbursement of expenses described in such procedures.

7.4 Maximum and Minimum Benefits for Health FSA

(a) Maximum Reimbursement Available; Uniform Coverage. The maximum dollar amount elected by the Participant for reimbursement of Medical Care Expenses incurred during a Period of Coverage (reduced by prior reimbursements during the Period of Coverage) will be available at all times during the Period of Coverage, regardless of the actual amounts credited to the Participant's Health FSA Account pursuant to Section 7.5. Notwithstanding the foregoing, no reimbursements will be available for Medical Care Expenses incurred after coverage under this Plan has terminated, unless the Participant has elected COBRA as provided in Section 7.8 or is entitled to submit expenses incurred during a Grace Period as provided in Section 7.4(e). Payment will be made to the Participant in cash as reimbursement for Medical Care Expenses incurred during the Period of Coverage for which the Participant's election is effective (or during a Grace Period, if applicable under Section 7.4(e)), provided that the other requirements of this Article VII have been satisfied. Notwithstanding the foregoing, in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15, a Participant whose participation in the Plan terminates in 2020 may continue to receive reimbursements from any contributions that remain in his or her Health FSA as of the date of such termination through the end of the 2020 Plan Year, including any Grace Period

- (b) Maximum and Minimum Dollar Limits. The maximum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Medical Care Expenses incurred in any Period of Coverage will be set forth in the enrollment materials-, but shall be no more than the contribution limit announced by the IRS for the applicable year. The minimum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Medical Care Expenses incurred in any Period of Coverage is \$0. Reimbursements due for Medical Care Expenses incurred by the Participant's Spouse or Dependents will be charged against the Participant's Health FSA Account. In no event will<u>Unless otherwise</u> permitted under applicable law, the maximum annual benefit may not exceed the maximum limit provided under federalsuch law.
- (c) *Changes; No Proration.* For each Plan Year, the maximum and minimum dollar limit may be changed by the Plan Administrator and will be communicated to Employees through the Election Form/Salary Reduction Agreement or other enrollment materials. If a Participant enters the Health FSA Component mid-year or wishes to increase his or her election mid-year as permitted under Section 10.3, then there will be no proration rule i.e., the Participant may elect coverage up to the maximum dollar limit or may increase coverage to the maximum dollar limit, as applicable.
- (d) Effect on Maximum Benefits If Election Change Permitted. Any change in an election under Article X (other than under Section 10.3(c) for FMLA leave) that increases contributions to the Health FSA Component also will change the maximum reimbursement benefits for the balance of the Period of Coverage commencing with the election change. Such maximum reimbursement benefits for the balance of the Period of Coverage will be calculated by adding (1) the contributions (if any) made by the Participant as of the end of the portion of the Period of Coverage immediately preceding the change in election, to (2) the total contributions scheduled to be made by the Participant during the remainder of such Period of Coverage to the Health FSA Account, reduced by (3) all reimbursements made during the entire Period of Coverage. Any change in an election under Section 10.3(c) for FMLA leave will change the maximum reimbursement benefits in accordance with the regulations governing the effect of the FMLA on the operation of cafeteria plans.
- (e) Grace Periods; Special Rules for Claims Incurred During a Grace Period. Notwithstanding any contrary provision in this Plan and subject to the conditions of this Section 7.4(e), an individual may be reimbursed for Medical Care Expenses incurred during a Grace Period from amounts remaining in his or her Health FSA Account at the end of the Plan Year to which that Grace Period relates ("Prior Plan Year Health FSA Amounts") if he or she is either: (1) a Participant with Health FSA coverage that is in effect on the last day of that Plan Year; or (2) a qualified beneficiary (as defined under COBRA) who has COBRA coverage under the Health FSA Component on the last day of that Plan Year.

- Prior Plan Year Health FSA Amounts may not be cashed out or converted to any other taxable or non-taxable benefit. For example, Prior Plan Year Health FSA Amounts may not be used to reimburse Dependent Care Expenses.
- Medical Care Expenses incurred during a Grace Period and approved for reimbursement in accordance with Section 7.7 will be reimbursed first from any available Prior Plan Year Health FSA Amounts and then from any amounts that are available to reimburse expenses that are incurred during the current Plan Year, except that if the Health FSA is accessible by an electronic payment card (e.g., debit card, credit card, or similar arrangement), Medical Care Expenses incurred during the Grace Period may need to be submitted manually in order to be reimbursed from Prior Plan Year Health FSA Amounts if the card is unavailable for such reimbursement. An individual's Prior Plan Year Health FSA Amounts will be debited for any reimbursement of Medical Care Expenses incurred during the Grace Period that is made from such Prior Plan Year Health FSA Amounts.
- Claims for reimbursement of Medical Care Expenses incurred during a Grace Period must be submitted no later than the April 30 following the close of the Plan Year to which the Grace Period relates in order to be reimbursed from Prior Plan Year Health FSA Amounts. For purposes of the Grace Period for amounts relating to the 2020 Plan Year, claims for reimbursement incurred during the Grace Period from January 1, 2021 through December 31, 2021, must be submitted no later than April 30, 2022. Any Prior Plan Year Health FSA Amounts that remain after all reimbursements have been made for the Plan Year and its related Grace Period will not be carried over to reimburse the Participant for expenses incurred in any subsequent period. The Participant will forfeit all rights with respect to these amounts, which will be subject to the Plan's provisions regarding forfeitures in Section 7.6(b).

7.5 Establishment of Health FSA Account

The Plan Administrator will establish and maintain a Health FSA Account with respect to each Participant for each Plan Year or other Period of Coverage for which the Participant elects to participate in the Health FSA Component, but it will not create a separate fund or otherwise segregate assets for this purpose. The Account so established will merely be a recordkeeping account with the purpose of keeping track of contributions and determining forfeitures under Section 7.6.

- (a) *Crediting of Accounts*. A Participant's Health FSA Account for a Plan Year or other Period of Coverage will be credited periodically during such period with an amount equal to the Participant's Salary Reductions elected to be allocated to such Account.
- (b) *Debiting of Accounts*. A Participant's Health FSA Account for a Plan Year or other Period of Coverage will be debited for any reimbursement of Medical Care Expenses incurred during such period (or for reimbursement of Medical Care Expenses incurred during any Grace Period to which he or she is entitled as provided in Section 7.4(e)).
- (c) Available Amount Not Based on Credited Amount. As described in Section 7.4, the amount available for reimbursement of Medical Care Expenses is the Participant's

annual benefit amount, reduced by prior reimbursements for Medical Care Expenses incurred during the Plan Year or other Period of Coverage (or during the Grace Period, if applicable); it is not based on the amount credited to the Health FSA Account at a particular point in time. Thus, a Participant's Health FSA Account may have a negative balance during a Plan Year or other Period of Coverage, but the aggregate amount of reimbursement will in no event exceed the maximum dollar amount elected by the Participant under this Plan.

7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule

- (a) Use-or-Lose Rule. Except as otherwise provided in Section 7.4(e) (regarding certain individuals who may be reimbursed from Prior Plan Year Health FSA Amounts for expenses incurred during a Grace Period), if any balance remains in the Participant's Health FSA Account for a Period of Coverage after all reimbursements have been made for the Period of Coverage, then such balance will not be carried over to reimburse the Participant for Medical Care Expenses incurred during a subsequent Plan Year. The Participant will forfeit all rights with respect to such balance.
- (b) Use of Forfeitures. All forfeitures under this Plan will be used as follows: (1) first, to offset any losses experienced by the Employer during the Plan Year as a result of making reimbursements (i.e., providing Health FSA Benefits) with respect to all Participants in excess of the contributions paid by such Participants through Salary Reductions; (2) second, to reduce the cost of administering the Health FSA Component during the Plan Year or the subsequent Plan Year (all such administrative costs will be documented by the Plan Administrator); and (3) third, to provide increased benefits or compensation to Participants in subsequent years in any weighted or uniform fashion that the Plan Administrator deems appropriate, consistent with applicable regulations. In addition, any Health FSA Account benefit payments that are unclaimed (e.g., uncashed benefit checks) by the close of the Plan Year following the Period of Coverage in which the Medical Care Expense was incurred will be forfeited and applied as described above.

7.7 Reimbursement Claims Procedure for Health FSA

- (a) Timing. Within 30 days after receipt by the Plan Administrator of a reimbursement claim from a Participant, the Employer will reimburse the Participant for the Participant's Medical Care Expenses (if the Plan Administrator approves the claim), or the Plan Administrator will notify the Participant that his or her claim has been denied. This time period may be extended by an additional 15 days for matters beyond the control of the Plan Administrator, including in cases where a reimbursement claim is incomplete. The Plan Administrator will provide written notice of any extension, including the reasons for the extension, and will allow the Participant 45 days in which to complete the previously incomplete reimbursement claim.
- (b) Claims Substantiation. A Participant who has elected to receive Health FSA Benefits for a Period of Coverage may apply for reimbursement by submitting a request in writing to the Plan Administrator in such form as the Plan Administrator may prescribe, by no later than the April 30 following the close of the Plan Year in which the Medical Care Expense was incurred (except that for a Participant who ceases to be eligible to participate, this must be done no later than 90 days after the date that eligibility ceases, as described in Section 7.8) setting forth:
 - the person(s) on whose behalf Medical Care Expenses have been incurred;

- the nature and date of the expenses so incurred;
- the amount of the requested reimbursement;
- a statement that such expenses have not otherwise been reimbursed and that the Participant will not seek reimbursement through any other source; and
- other such details about the expenses that may be requested by the Plan Administrator in the reimbursement request form or otherwise (e.g., a statement from a medical practitioner that the expense is to treat a specific medical condition, documentation that a medicine or drug was prescribed, or a more detailed certification from the Participant).

The application must be accompanied by bills, invoices, or other statements from an independent third party showing that the Medical Care Expenses have been incurred and showing the amounts of such expenses, along with any additional documentation that the Plan Administrator may request. Except for the final reimbursement claim for a Participant's Health FSA Account for a Plan Year or other Period of Coverage, no claim for reimbursement may be made unless and until the aggregate claim for reimbursement is at least \$25. If the Health FSA is accessible by an electronic payment card (e.g., debit card, credit card, or similar arrangement), the Participant will be required to comply with substantiation procedures established by the Plan Administrator in accordance with Rev. Rul. 2003-43, IRS Notice 2006-69, or other IRS guidance.

- (c) *Claims Denied*. For reimbursement claims that are denied, see the appeals procedure in Article XI.
- (d) Claims Ordering; No Reprocessing. All claims for reimbursement under the Health FSA Component will be paid in the order in which they are approved. Once paid, a claim will not be reprocessed or otherwise recharacterized solely for the purpose of paying it (or treating it as paid) from amounts attributable to a different Plan Year or Period of Coverage.

7.8 Reimbursements From Health FSA After Termination of Participation; COBRA

When a Participant ceases to be a Participant under Section 3.3, the Participant's Salary Reductions and election to participate will terminate. Except as otherwise provided in Section 7.4(e) (regarding certain individuals who may be reimbursed from Prior Plan Year Health FSA Amounts for expenses incurred during a Grace Period), the Participant will not be able to receive reimbursements for Medical Care Expenses incurred after the end of the day on which the Participant's employment terminates or the Participant otherwise ceases to be eligible. However, such Participant (or the Participant's estate) may claim reimbursement for any Medical Care Expenses incurred during the Period of Coverage prior to the date that the Participant ceases to be eligible (or during any Grace Period to which he or she is entitled as provided in Section 7.4(e)), provided that the Participant (or the Participant's estate) files a claim within 90 days after the date that the Participant ceases to be a Participant.

Notwithstanding any provision to the contrary in this Plan, to the extent required by COBRA, a Participant and his or her Spouse and Dependents, as applicable, whose coverage terminates under the Health FSA Component because of a COBRA qualifying event (and who is a qualified beneficiary as defined under COBRA) will be given the opportunity to continue on a self-pay basis the same coverage that he or she had under the Health FSA Component the day before the qualifying event for the periods prescribed by COBRA. Specifically, such individuals will be

eligible for COBRA continuation coverage only if, under Section 7.5, they have a positive Health FSA Account balance at the time of a COBRA qualifying event (taking into account all claims submitted before the date of the qualifying event). Such individuals will be notified if they are eligible for COBRA continuation coverage. If COBRA is elected, it will be available only for the remainder of the Plan Year in which the qualifying event occurs; such COBRA coverage for the Health FSA Component will cease at the end of the Plan Year and cannot be continued for the next Plan Year. Such continuation coverage will be subject to all conditions and limitations under COBRA. Notwithstanding the foregoing, a qualified beneficiary (as defined under COBRA) who has COBRA coverage under the Health FSA Component on the last day of a Plan Year may be entitled to reimbursement of Medical Care Expenses incurred during the Grace Period following that Plan Year in accordance with the provisions of Section 7.4(e).

Contributions for coverage for Health FSA Benefits may be paid on a pre-tax basis for current Employees receiving taxable compensation (as may be permitted by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year) where COBRA coverage arises either (a) because the Employee ceases to be eligible because of a reduction of hours or (b) because the Employee's Dependent ceases to satisfy the eligibility requirements for coverage. For all other individuals (e.g., Employees who cease to be eligible because of retirement, termination of employment, or layoff), contributions for COBRA coverage for Health FSA Benefits must be paid on an after-tax basis (unless permitted otherwise by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year).

Notwithstanding the foregoing, a Participant whose participation in the Plan terminates in 2020 may continue to receive reimbursements from any contributions that remain in his or her Health FSA as of the date of such termination through the end of the 2020 Plan Year and any related Grace Period.

7.9 Coordination of Benefits

Health FSA Benefits are intended to pay benefits solely for Medical Care Expenses for which Participants have not been previously reimbursed and will not seek reimbursement elsewhere. Accordingly, the Health FSA will not be considered to be a group health plan for coordination of benefits purposes, and Health FSA Benefits will not be taken into account when determining benefits payable under any other plan.

ARTICLE VIII. DCAP Component

8.1 DCAP Benefits

An Eligible Employee can elect to participate in the DCAP Component by electing (a) to receive benefits in the form of reimbursements for Dependent Care Expenses from the DCAP Component (DCAP Benefits), and (b) to pay his or her contribution for such DCAP Benefits on a pre-tax Salary Reduction basis. Unless an exception applies (as described in Article X), such election is irrevocable for the duration of the Period of Coverage to which it relates.

8.2 Participant Contributions for Cost of Coverage for DCAP Benefits

The annual Contribution for a Participant's portion of the DCAP Benefits is equal to the annual benefit amount elected by the Participant, subject to the dollar limits set forth in Section 8.4(b). (For example, if the maximum \$5,000 annual benefit amount is elected, then the annual contribution amount is also \$5,000.)

8.3 Eligible Dependent Care Expenses

Under the DCAP Component, a Participant may receive reimbursement for Dependent Care Expenses incurred during the Period of Coverage for which an election is in force.

- (a) *Incurred.* A Dependent Care Expense is incurred at the time the Qualifying Dependent Care Services giving rise to the expense is furnished, not when the Participant is formally billed for, is charged for, or pays for the Qualifying Dependent Care Services (e.g., services rendered for the month of June are not fully incurred until June 30 and cannot be reimbursed in full until then).
- (b) Dependent Care Expenses. "Dependent Care Expenses" are expenses that are considered to be employment-related expenses under Code Section 21(b)(2) (relating to expenses for the care of a Qualifying Individual necessary for gainful employment of the Employee and Spouse, if any, and expenses for incidental household services), if paid for by the Eligible Employee to obtain Qualifying Dependent Care Services; provided, however, that this term will not include any expenses for which the Participant or other person incurring the expense is reimbursed for the expense through insurance or any other plan. If only a portion of a Dependent Care Expense has been reimbursed elsewhere (e.g., because the Spouse's DCAP imposes maximum benefit limitations), the DCAP can reimburse the remaining portion of such Expense if it otherwise meets the requirements of this Article VIII.
- (c) *Qualifying Dependent Care Services.* "Qualifying Dependent Care Services" means services that: (1) relate to the care of a Qualifying Individual that enable the Participant and his or her Spouse to remain gainfully employed after the date of participation in the DCAP Component and during the Period of Coverage; and (2) are performed—
 - in the Participant's home; or
 - outside the Participant's home for (1) the care of a Participant's qualifying child who is under age 13; or (2) the care of any other Qualifying Individual who regularly spends at least eight hours per day in the Participant's household. In addition, if the expenses are incurred for services provided by a dependent care center (i.e., a facility (including a day camp) that provides care for more than six individuals (other than individuals residing at the facility) on a regular basis and receives a fee, payment, or grant for such services), then the center must comply with all applicable state and local laws and regulations.
- (d) *Exclusion*. Dependent Care Expenses do not include amounts paid to:
 - an individual with respect to whom a personal exemption is allowable under Code Section 151(c) to a Participant or his or her Spouse;
 - a Participant's Spouse;
 - a Participant's child (as defined in Code Section 152(f)(1)) who is under 19 years of age at the end of the year in which the expenses were incurred; or

• a parent of a Participant's under age 13 qualifying child as defined in Code Section 152(a)(1) (e.g., a former spouse who is the child's noncustodial parent).

8.4 Maximum and Minimum Benefits for DCAP

- (a) Maximum Reimbursement Available. The maximum dollar amount elected by the Participant for reimbursement of Dependent Care Expenses incurred during a Period of Coverage (reduced by prior reimbursements during the Period of Coverage) will only be available during the Period of Coverage to the extent of the actual amounts credited to the Participant's DCAP Account pursuant to Section 8.5. (No reimbursement will be made to the extent that such reimbursement would exceed the balance in the Participant's Account (that is, the year-to-date amount that has been withheld from the Participant's Compensation for reimbursement for Dependent Care Expenses for the Period of Coverage, less any prior reimbursements). Payment will be made to the Participant in cash as reimbursement for Dependent Care Expenses incurred during the Period of Coverage for which the Participant's election is effective, provided that the other requirements of this Article VIII have been satisfied.
- (b) Maximum and Minimum Dollar Limits. The maximum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Dependent Care Expenses incurred in any Period of Coverage is \$5,000 or, if lower, the maximum amount that the Participant has reason to believe will be excludable from his or her income at the time the election is made as a result of the applicable statutory limit for the Participant. The applicable statutory limit for a Participant is the smallest of the following amounts:
 - the Participant's Earned Income for the calendar year;
 - the Earned Income of the Participant's Spouse for the calendar year (for this purpose, a Spouse who is not employed during a month in which the Participant incurs a Dependent Care Expense and is either (1) physically or mentally incapable of self-care, or (2) a Student will be deemed to have Earned Income in the amount specified in Code Section 21(d)(2)); or
 - either \$5,000 or \$2,500 for the calendar year, as applicable below:

(1) The amount is \$5,000 for the calendar year if one of the following applies: (a) the Participant is married and files a joint federal income tax return; (b) the Participant is married, files a separate federal income tax return, and meets the following conditions: (i) the Participant maintains as his or her home a household that constitutes (for more than half of the taxable year) the principal abode of a Qualifying Individual (i.e., the Dependent for whom the Participant is eligible to receive reimbursements under the DCAP); (ii) the Participant furnishes over half of the cost of maintaining such household during the taxable year; and (iii) during the last six months of the taxable year, the Participant's Spouse is not a member of such household (i.e., the Spouse maintained a separate residence); or (c) the Participant is single or is the head of the household for federal income tax purposes.

(2) The amounts is \$2,500 for the calendar year if the Participant is married and resides with the Spouse, but files a separate federal income tax return.

The minimum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Dependent Care Expenses incurred in any

Period of Coverage is \$0. (Note: The maximum annual benefit amount that a Participant may elect for the 2021 Plan year will not be affected by the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period from any amount that remains in his or her 2020 DCAP Account as of December 31, 2020 and the amount elected by the Participant for the 2021 Plan Year. In addition, the maximum annual benefit amount that a Participant may elect for the 2021 Plan year will not be affected by any balance available to such Participant during the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period from any amount that remains in his or her 2020 DCAP Account as of December 31, 2020 and the amount elected by the Participant for the 2021 Plan Year.)

- (c) Grace Period for 2020 Amounts; Special Rules for Claims Incurred During a Grace Period. Notwithstanding any contrary provision in this Plan and subject to the conditions of this Section 8.4(c), if an individual has amounts remaining in his or her DCAP Account as of December 31, 2020, such individual may be reimbursed for Dependent Care Expenses from such amount during the Grace Period from January 1, 2021 through December 31, 2021.
 - Amounts remaining in a DCAP Account as of December 31, 2020 may not be cashed out or converted to any other taxable or non-taxable benefit.
 - Dependent Care Expenses incurred during the Grace Period and approved for reimbursement in accordance will be reimbursed first from the 2020 DCAP Account amount and then from any amounts that are available to reimburse expenses that are incurred during the current Plan Year, except that if the DCAP Account is accessible by an electronic payment card (e.g., debit card, credit card, or similar arrangement), Dependent Care Expenses incurred during the Grace Period may need to be submitted manually in order to be reimbursed from 2020 DCAP Account amount if the card is unavailable for such reimbursement. An individual's 2020 DCAP Account amount will be debited for any reimbursement of Dependent Care Expenses incurred during the Grace Period that is made from such 2020 DCAP Account amount.
 - For purposes of the Grace Period for amounts relating to the 2020 Plan Year, claims for reimbursement incurred during the Grace Period from January 1, 2021 through December 31, 2021, must be submitted no later than April 30, 2022. Any Prior Plan Year Health FSA Amounts that remain after all reimbursements have been made for the Plan Year and its related Grace Period will not be carried over to reimburse the Participant for expenses incurred in any subsequent period. The Participant will forfeit all rights with respect to these amounts, which will be subject to the Plan's provisions regarding forfeitures in Section 8.6.
- (d) Changes; No Proration. For subsequent Plan Years, the maximum and minimum dollar limit may be changed by the Plan Administrator and will be communicated to Employees through the Election Form/Salary Reduction Agreement or other enrollment materials. If a Participant enters the DCAP Component mid-year or wishes to increase his or her election mid-year as permitted under Section 10.3, then there will be no proration rule i.e., the Participant may elect coverage up to the maximum dollar limit or may increase coverage up to the maximum dollar limit, as applicable.
- (e) *Effect on Maximum Benefits If Election Change Permitted.* Any change in an election under Article X affecting annual contributions to the DCAP Component also will change

the maximum reimbursement benefits for the balance of the Period of Coverage (commencing with the election change), as further limited by Sections 8.4(a) and (b). Such maximum reimbursement benefits for the balance of the Period of Coverage will be calculated by adding (1) the contributions, if any, made by the Participant as of the end of the portion of the Period of Coverage immediately preceding the change in election, to (2) the total contributions scheduled to be made by the Participant during the remainder of such Period of Coverage to the DCAP Account, reduced by (3) reimbursements during the Period of Coverage.

8.5 Establishment of DCAP Account

The Plan Administrator will establish and maintain a DCAP Account with respect to each Participant who has elected to participate in the DCAP Component, but it will not create a separate fund or otherwise segregate assets for this purpose. The Account so established will merely be a recordkeeping account with the purpose of keeping track of contributions and determining forfeitures under Section 8.6.

- (a) *Crediting of Accounts*. A Participant's DCAP Account will be credited periodically during each Period of Coverage with an amount equal to the Participant's Salary Reductions elected to be allocated to such Account.
- (b) *Debiting of Accounts*. A Participant's DCAP Account will be debited during each Period of Coverage for any reimbursement of Dependent Care Expenses incurred during the Period of Coverage.
- (c) Available Amount Is Based on Credited Amount. As described in Section 8.4, the amount available for reimbursement of Dependent Care Expenses may not exceed the year-to-date amount credited to the Participant's DCAP Account, less any prior reimbursements (i.e., it is based on the amount credited to the DCAP Account at a particular point in time). Thus, a Participant's DCAP Account may not have a negative balance during a Period of Coverage.

8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule

If any balance remains in the Participant's DCAP Account for a Period of Coverage after all reimbursements have been made for the Period of Coverage, then such balance will not be carried over to reimburse the Participant for Dependent Care Expenses incurred during a subsequent Plan Year. The Participant will forfeit all rights with respect to such balance. All forfeitures under this Plan will be used as follows: (1) first, to offset any losses experienced by the Employer during the Plan Year as a result of making reimbursements (i.e., providing DCAP Benefits) with respect to all Participants in excess of the contributions paid by such Participants through Salary Reductions; (2) second, to reduce the cost of administering the DCAP during the Plan Year or the subsequent Plan Year (all such administrative costs will be documented by the Plan Administrator); and (3) third, to provide increased benefits or compensation to Participants in subsequent years in any weighted or uniform fashion the Plan Administrator deems appropriate, consistent with applicable regulations. In addition, any DCAP Account benefit payments that are unclaimed (e.g., uncashed benefit checks) by the close of the Plan Year following the Period of Coverage in which the Dependent Care Expense was incurred will be forfeited and applied as described above. Notwithstanding, the foregoing, including but not limited to the overall maximum provided in Section 8.4(b), in response to the continuing COVID-19 pandemic and in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15, if a Participant has a balance in his or her DCAP Account as of December 31, 2020

and has a Qualifying Individual who attains the age of 13 either during the 2020 Plan Year, or during the 2021 Plan Year, such balance shall be carried over to the 2021 Plan Year. Such Participant may receive reimbursement for Dependent Care Expenses incurred during the 2021 Plan Year and submitted for reimbursement no later than March 31, 2022 from the balance carried over from the 2020 Plan Year for such Child that attained age 13 in the 2020 Plan Year or in the 2021 Plan Year only and not from any DCAP election for the 2021 Plan Year.

8.7 Reimbursement Claims Procedure for DCAP

- (a) Timing. Within 30 days after receipt by the Plan Administrator of a reimbursement claim from a Participant, the Employer will reimburse the Participant for the Participant's Dependent Care Expenses (if the Plan Administrator approves the claim), or the Plan Administrator will notify the Participant that his or her claim has been denied. This time period may be extended by an additional 15 days for matters beyond the control of the Plan Administrator, including in cases where a reimbursement claim is incomplete. The Plan Administrator will provide written notice of any extension, including the reasons for the extension, and will allow the Participant 45 days in which to complete the previously incomplete reimbursement claim.
- (b) Claims Substantiation. A Participant who has elected to receive DCAP Benefits for a Period of Coverage may apply for reimbursement by submitting a request for reimbursement in writing to the Plan Administrator in such form as the Plan Administrator may prescribe, by no later than the April 30 following the close of the Plan Year in which the Dependent Care Expense was incurred (except for a Participant who ceases to be eligible to participate, by no later than 90 days after the date that eligibility ceases, as described in Section 8.8), setting forth:
 - the person(s) on whose behalf Dependent Care Expenses have been incurred;
 - the nature and date of the expenses so incurred;
 - the amount of the requested reimbursement;
 - the name of the person, organization or entity to whom the expense was or is to be paid, and taxpayer identification number (Social Security number, if the recipient is a person);
 - a statement that such expenses have not otherwise been reimbursed and that the Participant will not seek reimbursement through any other source;
 - the Participant's certification that he or she has no reason to believe that the reimbursement requested, added to his or her other reimbursements to date for Dependent Care Expenses incurred during the same calendar year, will exceed the applicable statutory limit for the Participant as described in Section 8.4(b); and
 - other such details about the expenses that may be requested by the Plan Administrator in the reimbursement request form or otherwise (e.g., a more detailed certification from the Participant).

The application will be accompanied by bills, invoices, or other statements from an independent third party showing that the Dependent Care Expenses have been incurred and showing the amounts of such expenses, along with any additional documentation that the Plan Administrator may request. Except for the final reimbursement claim for a

Period of Coverage, no claim for reimbursement may be made unless and until the aggregate claim for reimbursement is at least \$25.

(c) *Claims Denied*. For reimbursement claims that are denied, see the appeals procedure in Article XI.

8.8 Reimbursements From DCAP After Termination of Participation

When a Participant ceases to be a Participant under Section 3.3, the Participant's Salary Reductions and election to participate will terminate. The Participant will not be able to receive reimbursements for Dependent Care Expenses incurred after the end of the day on which the Participant's employment terminates or the Participant otherwise ceases to be eligible, with one exception: such Participant (or the Participant's estate) may claim reimbursement for any Dependent Care Expenses incurred in the month that includes the date the Participant terminates employment or otherwise loses eligibility, provided that the Participant (or the Participant's estate) files a claim within 90 days after the date that the Participant's employment terminates or the Participant otherwise ceases to be eligible. Notwithstanding the foregoing, a Participant whose participation in the Plan terminates in 2020 may continue to receive reimbursements from any amount that remains in his or her DCAP Account as of the date of such termination through the end of the 2020 Plan Year and any related Grace Period.

ARTICLE IX. HIPAA PROVISIONS FOR HEALTH FSA

9.1 Provision of Protected Health Information to Employer

Members of the Employer's workforce have access to the individually identifiable health information of Plan participants for administrative functions of the Health FSA. When this health information is provided from the Health FSA to the Employer, it is Protected Health Information (PHI). The Health Insurance Portability and Accountability Act of 1996 (HIPAA) and its implementing regulations restrict the Employer's ability to use and disclose PHI. The following HIPAA definition of PHI applies for purposes of this Article IX:

Protected Health Information. Protected health information means information that is created or received by the Plan and relates to the past, present, or future physical or mental health or condition of a participant; the provision of health care to a participant; or the past, present, or future payment for the provision of health care to a participant; and that identifies the participant or for which there is a reasonable basis to believe the information can be used to identify the participant. Protected health information includes information of persons living or deceased.

The Employer will have access to PHI from the Health FSA only as permitted under this Article IX or as otherwise required or permitted by HIPAA. HIPAA and its implementing regulations were modified by the Health Information Technology for Economic and Clinical Health Act (HITECH Act), the statutory provisions of which are incorporated herein by reference.

9.2 Permitted Disclosure of Enrollment/Disenrollment Information

The Health FSA may disclose to the Employer information on whether the individual is participating in the Plan.

9.3 Permitted Uses and Disclosure of Summary Health Information

The Health FSA may disclose Summary Health Information to the Employer, provided that the Employer requests the Summary Health Information for the purpose of modifying, amending, or terminating the Health FSA.

"Summary Health Information" means information (a) that summarizes the claims history, claims expenses, or type of claims experienced by individuals for whom a plan sponsor had provided health benefits under a health plan; and (b) from which the information described at 42 CFR Section 164.514(b)(2)(i) has been deleted, except that the geographic information described in 42 CFR Section 164.514(b)(2)(i)(B) need only be aggregated to the level of a five-digit ZIP code.

9.4 Permitted and Required Uses and Disclosure of PHI for Plan Administration Purposes

Unless otherwise permitted by law, and subject to the conditions of disclosure described in Section 9.5 and obtaining written certification pursuant to Section 9.7, the Health FSA may disclose PHI to the Employer, provided that the Employer uses or discloses such PHI only for Plan administration purposes. "Plan administration purposes" means administration functions performed by the Employer on behalf of the Health FSA, such as quality assurance, claims processing, auditing, and monitoring. Plan administration functions do not include functions performed by the Employer in connection with any other benefit or benefit plan of the Employer, and they do not include any employment-related functions.

Notwithstanding the provisions of this Plan to the contrary, in no event will the Employer be permitted to use or disclose PHI in a manner that is inconsistent with 45 CFR Section 164.504(f).

9.5 Conditions of Disclosure for Plan Administration Purposes

The Employer agrees that with respect to any PHI (other than enrollment/disenrollment information and Summary Health Information, which are not subject to these restrictions) disclosed to it by the Health FSA, the Employer will:

- not use or further disclose the PHI other than as permitted or required by the Health FSA or as required by law;
- ensure that any agent, including a subcontractor, to whom it provides PHI received from the Health FSA agrees to the same restrictions and conditions that apply to the Employer with respect to PHI;
- not use or disclose the PHI for employment-related actions and decisions or in connection with any other benefit or employee benefit plan of the Employer;
- report to the Plan any use or disclosure of the information that is inconsistent with the uses or disclosures provided for of which it becomes aware;
- make available PHI to comply with HIPAA's right to access in accordance with 45 CFR Section 164.524;
- make available PHI for amendment and incorporate any amendments to PHI in accordance with 45 CFR Section 164.526;
- make available the information required to provide an accounting of disclosures in accordance with 45 CFR Section 164.528;

- make its internal practices, books, and records relating to the use and disclosure of PHI received from the Health FSA available to the Secretary of Health and Human Services for purposes of determining compliance by the Health FSA with HIPAA's privacy requirements;
- if feasible, return or destroy all PHI received from the Health FSA that the Employer still maintains in any form and retain no copies of such information when no longer needed for the purpose for which disclosure was made, except that, if such return or destruction is not feasible, limit further uses and disclosures to those purposes that make the return or destruction of the information infeasible; and
- ensure that the adequate separation between the Health FSA and the Employer (i.e., the "firewall"), required in 45 CFR Section 504(f)(2)(iii) is satisfied.

The Employer further agrees that if it creates, receives, maintains, or transmits any electronic PHI (other than enrollment/disenrollment information and Summary Health Information, which are not subject to these restrictions) on behalf of the Health FSA, it will implement administrative, physical, and technical safeguards that reasonably and appropriately protect the confidentiality, integrity, and availability of the electronic PHI, and it will ensure that any agents (including subcontractors) to whom it provides such electronic PHI agree to implement reasonable and appropriate security measures to protect the information. The Employer will report to the Health FSA any security incident of which it becomes aware.

9.6 Adequate Separation Between Plan and Employer

The Employer will allow the following persons access to PHI: Director of Finance, Accounting Manager, Senior Accountant, Accountant, the Plan Administrator, and payroll staff performing Health FSA functions and any other Employee who needs access to PHI in order to perform Plan administration functions that the Employer performs for the Health FSA (such as quality assurance, claims processing, auditing, monitoring, payroll, and appeals). No other persons will have access to PHI. These specified employees (or classes of employees) will only have access to and use PHI to the extent necessary to perform the plan administration functions that the Employer performs for the Health FSA. In the event that any of these specified employees does not comply with the provisions of this Section, that employee will be subject to disciplinary action by the Employer for non-compliance pursuant to the Employer's employee discipline and termination procedures.

The Employer will ensure that the provisions of this Section 9.6 are supported by reasonable and appropriate security measures to the extent that the designees have access to electronic PHI.

9.7 Certification of Plan Sponsor

The Health FSA will disclose PHI to the Employer only upon the receipt of a certification by the Employer that the Health FSA incorporates the provisions of 45 CFR Section 164.504(f)(2)(ii), and that the Employer agrees to the conditions of disclosure set forth in Section 9.5. Execution of the Plan by the Employer will serve as the required certification.

9.8 Privacy Official

The Employer will designate a Privacy Official, who will be responsible for the Plan's compliance with HIPAA. The Privacy Official may contract with or otherwise utilize the services of attorneys,

accountants, brokers, consultants, or other third party experts as the Privacy Official deems necessary or advisable. In addition and notwithstanding any provision of this Plan to the contrary, the Privacy Official will have the authority to and be responsible for:

- accepting and verifying the accuracy and completeness of any certification provided by the Employer under this Article;
- transmitting the certification to any third parties as may be necessary to permit them to disclose PHI to the Employer;
- establishing and implementing policies and procedures with respect to PHI that are designed to ensure compliance by the Plan with the requirements of HIPAA;
- establishing and overseeing proper training of personnel who will have access to PHI; and
- any other duty or responsibility that the Privacy Official, in his or her sole capacity, deems necessary or appropriate to comply with the provisions of HIPAA and the purposes of the Article IX.

9.9 Interpretation and Limited Applicability

This Article serves the sole purpose of complying with the requirements of HIPAA and will be interpreted and construed in a manner to effectuate this purpose. Neither this Article IX nor the duties, powers, responsibilities, and obligations listed herein will be taken into account in determining the amount or nature of the benefits provided to any person covered under the Health FSA Component, nor will they inure to the benefit of any third parties. To the extent that any of the provisions of this Article IX are no longer required by HIPAA or do not apply to the Plan because the Plan is otherwise excepted from HIPAA, they will be deemed deleted and will have no force or effect.

9.10 Service Performed for the Employer

Notwithstanding any other provisions of this Plan to the contrary, all services performed by a business associate for the Health FSA in accordance with the applicable service agreement will be deemed to be performed on behalf of the Health FSA and subject to the administrative simplification provisions of HIPAA contained in 45 C.F.R. Parts 160 through 164, except services that relate to eligibility and enrollment in the Health FSA. If a business associate of the Health FSA performs any services that relate to eligibility and enrollment in the Health FSA, these services will be deemed to be performed on behalf of the Employer in its capacity as Plan Sponsor and not on behalf of the Health FSA.

ARTICLE X. Irrevocability of Elections; Exceptions

10.1 Irrevocability of Elections

Except as described in this Article X, a Participant's election under the Plan is irrevocable for the duration of the Period of Coverage to which it relates. In other words, unless an exception applies, the Participant may not change any elections for the duration of the Period of Coverage regarding:

- (a) participation in this Plan;
- (b) Salary Reduction amounts; or
- (c) election of particular Benefit Plan Options.

10.2 Procedure for Making New Election If Exception to Irrevocability Applies

- (a) Timeframe for Making New Election. A Participant (or an Eligible Employee who, when first eligible under Section 4.2 or during the Open Enrollment Period under Section 4.3, declined to be a Participant) may make a new election within 30 days of the occurrence of an event described in Section 10.3 (or within 60 days of the occurrence of an event described in Section 10.3(e)(3) or (4)), as applicable, but only if the election under the new Election Form/Salary Reduction Agreement is made on account of and is consistent with the event. Notwithstanding the foregoing, a Change in Status (e.g., a divorce or a dependent's losing student status) that results in a beneficiary becoming ineligible for coverage under the Insurance Plans will automatically result in a corresponding election change, whether or not requested by the Participant within the normal 30-day period.
- (b) Effective Date of New Election. Elections made pursuant to this Section 10.2 will be effective for the balance of the Period of Coverage following the change of election unless a subsequent event allows for a further election change. Except as provided in Section 10.3(e) for HIPAA special enrollment rights in the event of birth, adoption, or placement for adoption, all election changes will be effective on a prospective basis only (i.e., election changes will become effective no earlier than the first day of the next calendar month following the date that the election change was filed, but, as determined by the Plan Administrator, election changes may become effective later to the extent that the coverage in the applicable Benefit Plan Option commences later).
- (c) *Effect of New Election Upon Amount of Benefits.* For the effect of a changed election upon the maximum and minimum benefits under the Health FSA and DCAP Components, see Sections 7.4 and 8.4 respectively.

10.3 Events Permitting Exception to Irrevocability Rule for All Benefits

A Participant may change an election as described below upon the occurrence of the stated events for the applicable component of this Plan:

- (a) *Open Enrollment Period (Applies to all Benefit Plan Options).* A Participant may change an election during the Open Enrollment Period in accordance with Section 4.3.
- (b) *Termination of Employment (Applies to all Benefit Plan Options).* A Participant's election will terminate under the Plan upon termination of employment in accordance with Section 3.3.
- (c) *Leaves of Absence (Applies to all Benefit Plan Options).* A Participant may change an election under the Plan upon FMLA leave in accordance with Section 3.5 and upon non-FMLA leave in accordance with Section 3.6.
- (d) Change in Status (Applies to Premium Payment <u>Benefit,sBenefits</u> and to Health FSA Benefits and DCAP Benefits as limited further below). A Participant may change his or her election under the Plan upon the occurrence of a Change in Status, but only if such election change is made on account of and corresponds with a Change in Status that

affects eligibility for coverage under a plan of the Employer or a plan of the Spouse's or Dependent's employer (referred to as the general consistency requirement). A Change in Status that affects eligibility for coverage under a plan of the Employer or a plan of the Spouse's or Dependent's employer includes a Change in Status that results in an increase or decrease in the number of an Employee's family members (i.e., a Spouse and/or Dependents) who may benefit from the coverage.

(e) HIPAA Special Enrollment Rights (Applies to Premium Payment Benefits under Medical Insurance Plans only, and not to any other Insurance Plan, Health FSA, or DCAP Benefits). If a Participant or his or her Spouse or Dependent is entitled to special enrollment rights under a group health plan (other than an excepted benefit), as required by HIPAA under Code Section 9801(f), then a Participant may revoke a prior election for group health plan coverage and make a new election (including, when required by HIPAA, an election to enroll in another group health plan), provided that the election change corresponds with such HIPAA special enrollment rights. As required by HIPAA, a special enrollment right will arise in the following circumstances:

(1) a Participant or his or her Spouse or Dependent declined to enroll in group health plan coverage because he or she had coverage, and eligibility for such coverage is subsequently lost because: (a) the coverage was provided under COBRA and the COBRA coverage was exhausted; or (b) the coverage was non-COBRA coverage and the coverage terminated due to loss of eligibility for coverage or the employer contributions for the coverage were terminated;

(2) a new Dependent is acquired as a result of marriage, birth, adoption, or placement for adoption;

(3) the Participant's or Dependent's coverage under a Medicaid plan or state children's health insurance program is terminated as a result of loss of eligibility for such coverage; or

(4) the Participant or Dependent becomes eligible for a state premium assistance subsidy from a Medicaid plan or through a state children's health insurance program with respect to coverage under the group health plan.

An election to add previously eligible Dependents as a result of the acquisition of a new Spouse or Dependent child will be considered to be consistent with the special enrollment right. An election change on account of a HIPAA special enrollment attributable to the birth, adoption, or placement for adoption of a new Dependent child may, subject to the provisions of the underlying group health plan, be effective retroactively (up to 30 days).

For purposes of Section 10.3(e)(1), a loss of eligibility includes (but is not limited to) loss of eligibility due to legal separation, divorce, cessation of dependent status, death of an employee, termination of employment, reduction of hours, or any loss of eligibility for coverage that is measured with reference to any of the foregoing; loss of coverage offered through an HMO that does not provide benefits to individuals who do not reside, live, or work in the service area because an individual no longer resides, lives, or works in the service area (whether or not within the choice of the individual), and in the case of HMO coverage in the group market, no other benefit Plan is available to the individual; a situation in which an individual incurs a claim that would meet or exceed a lifetime limit

on all benefits; and a situation in which a plan no longer offers any benefits to the class of similarly situated individuals that includes the individual.

- (f) Certain Judgments, Decrees and Orders (Applies to Premium Payment and Health FSA Benefits, but Not to DCAP Benefits). If a judgment, decree, or order (collectively, an "Order") resulting from a divorce, legal separation, annulment, or change in legal custody (including a National Medical Support Order) requires accident or health coverage (including an election for Health FSA Benefits) for a Participant's child (including a foster child who is a Dependent of the Participant), then a Participant may (1) change his or her election to provide coverage for the child (provided that the Order requires the Participant to provide coverage); or (2) change his or her election to revoke coverage for the child if the Order requires that another individual (including the Participant's Spouse or former Spouse) provide coverage under that individual's plan and such coverage is actually provided.
- Medicare and Medicaid (Applies to Premium Payment Benefits, to Health FSA Benefits (g) as Limited Below, but Not to DCAP Benefits). If a Participant or his or her Spouse or Dependent who is enrolled in a health or accident plan under this Plan becomes entitled to (i.e., becomes enrolled in) Medicare or Medicaid (other than coverage consisting solely of benefits under Section 1928 of the Social Security Act providing for pediatric vaccines), then the Participant may prospectively reduce or cancel the health or accident coverage of the person becoming entitled to Medicare or Medicaid and/or the Participant's Health FSA coverage may be canceled (but not reduced). Notwithstanding the foregoing, such cancellation will not become effective to the extent that it would reduce future contributions to the Health FSA to a point where the total contributions for the Plan Year are less than the amount already reimbursed for the Plan Year. Furthermore, if a Participant or his or her Spouse or Dependent who has been entitled to Medicare or Medicaid loses eligibility for such coverage, then the Participant may prospectively elect to commence or increase the accident or health coverage of the individual who loses Medicare or Medicaid eligibility and/or the Participant's Health FSA coverage may commence or increase.
- (h) Change in Cost (Applies to Premium Payment Benefits, to DCAP Benefits as Limited Below, but Not to Health FSA Benefits). For purposes of this Section 10.3(h), "similar coverage" means coverage for the same category of benefits for the same individuals (e.g., family to family or single to single). For example, two plans that provide major medical coverage are considered to be similar coverage. For purposes of this definition, (a) a health FSA is not similar coverage with respect to an accident or health plan that is not a health FSA; (b) an HMO and a PPO are considered to be similar coverage; and (c) coverage by another employer, such as a Spouse's or Dependent's employer, may be treated as similar coverage if it otherwise meets the requirements of similar coverage.

(1) Increase or Decrease for Insignificant Cost Changes. Participants are required to increase their elective contributions (by increasing Salary Reductions) to reflect insignificant increases in their required contribution for their Benefit Plan Option(s), and to decrease their elective contributions to reflect insignificant decreases in their required contribution. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will determine whether an increase or decrease is insignificant based upon all the surrounding facts and circumstances, including but not limited to the dollar amount or percentage of the cost change. The Plan Administrator, on a reasonable and consistent basis, will automatically effectuate this increase or decrease in affected employees' elective contributions on a prospective basis.

(2) *Significant Cost Increases.* If the Plan Administrator determines that the cost charged to an Employee of a Participant's Benefit Plan Option(s) significantly increases during a Period of Coverage, then the Participant may (a) make a corresponding prospective increase in his or her elective contributions (by increasing Salary Reductions); (b) revoke his or her election for that coverage, and in lieu thereof, receive on a prospective basis coverage under another Benefit Plan Option that provides similar coverage (such as an HMO, but not the Health FSA); or (c) drop coverage prospectively if there is no other Benefit Plan Option available that provides similar coverage. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether a cost increase is significant in accordance with prevailing IRS guidance.

(3) *Significant Cost Decreases.* If the Plan Administrator determines that the cost of any Benefit Plan Option significantly decreases during a Period of Coverage, then the Plan Administrator may permit the following election changes: (a) Participants enrolled in that Benefit Plan Option may make a corresponding prospective decrease in their elective contributions (by decreasing Salary Reductions); (b) Participants who are enrolled in another Benefit Plan Option (such as an HMO, but not the Health FSA) may change their election on a prospective basis to elect the Benefit Plan Option that has decreased in cost; or (c) Employees who are otherwise eligible under Section 3.1 may elect the Benefit Plan Option that has decreased in cost (such as the PPO) on a prospective basis, subject to the terms and limitations of the Benefit Plan Option. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether a cost decrease is significant in accordance with prevailing IRS guidance.

(4) Limitation on Change in Cost Provisions for DCAP Benefits. The above "Change in Cost" provisions (Sections 10.3(h)(1) through 10.3(h)(3)) apply to DCAP Benefits only if the cost change is imposed by a dependent care provider who is not a "relative" of the Employee. For this purpose, a relative is an individual who is related as described in Code Sections 152(d)(2)(A) through (G), incorporating the rules of Code Sections 152(f)(4).

(i) Change in Coverage (Applies to Premium Payment and DCAP Benefits, but Not to Health FSA Benefits).

The definition of "similar coverage" under Section 10.3(h) applies also to this Section 10.3(i).

(1) Significant Curtailment. If coverage is "significantly curtailed" (as defined below), Participants may elect coverage under another Benefit Plan Option that provides similar coverage. In addition, as set forth below, if the coverage curtailment results in a "Loss of Coverage" (as defined below), then Participants may drop coverage if no similar coverage is offered by the Employer. The Plan Administrator in its sole discretion, on a uniform and consistent basis, will decide, in accordance with prevailing IRS guidance, whether a curtailment is "significant," and whether a Loss of Coverage has occurred.

(a) *Significant Curtailment Without Loss of Coverage.* If the Plan Administrator determines that a Participant's coverage under a Benefit Plan Option under this Plan (or the Participant's Spouse's or Dependent's coverage under his or her employer's plan) is significantly curtailed without a Loss of Coverage (for example, when there is a significant increase in the deductible, the co-pay, or the out-of-pocket cost-sharing limit
under an accident or health plan) during a Period of Coverage, the Participant may revoke his or her election for the affected coverage, and in lieu thereof, prospectively elect coverage under another Benefit Plan Option that provides similar coverage (such as an HMO, but not the Health FSA). Coverage under a plan is deemed to be "significantly curtailed" only if there is an overall reduction in coverage provided under the plan so as to constitute reduced coverage generally.

(b) *Significant Curtailment With a Loss of Coverage*. If the Plan Administrator determines that a Participant's Benefit Plan Option coverage under this Plan (or the Participant's Spouse's or Dependent's coverage under his or her employer's plan) is significantly curtailed, and if such curtailment results in a Loss of Coverage during a Period of Coverage, then the Participant may revoke his or her election for the affected coverage and may either prospectively elect coverage under another Benefit Plan Option that provides similar coverage (such as an HMO, but not the Health FSA) or drop coverage if no other Benefit Plan Option providing similar coverage is offered by the Employer.

(c) *Definition of Loss of Coverage*. For purposes of this Section 10.3(i)(1), a "Loss of Coverage" means a complete loss of coverage (including the elimination of a Benefit Plan Option, an HMO ceasing to be available where the Participant or his or her Spouse or Dependent resides, or a Participant or his or her Spouse or Dependent losing all coverage under the Benefit Plan Option by reason of an overall lifetime or annual limitation). In addition, the Plan Administrator, in its sole discretion, on a uniform and consistent basis, may treat the following as a Loss of Coverage:

- a substantial decrease in the medical care providers available under the Benefit Plan Option (such as a major hospital ceasing to be a member of a preferred provider network or a substantial decrease in the number of physicians participating in a PPO or HMO);
- a reduction in benefits for a specific type of medical condition or treatment with respect to which the Participant or his or her Spouse or Dependent is currently in a course of treatment; or
- any other similar fundamental loss of coverage.

(d) *DCAP Coverage Changes*. A Participant may make a prospective election change that is on account of and corresponds with a change by the Participant in the dependent care service provider. For example: (i) if the Participant terminates one dependent care service provider and hires a new dependent care service provider, then the Participant may change coverage to reflect the cost of the new service provider; and (ii) if the Participant terminates a dependent care service provider because a relative becomes available to take care of the child at no charge, then the Participant may cancel coverage.

(2) Addition or Significant Improvement of a Benefit Plan Option. If during a Period of Coverage the Plan adds a new Benefit Plan Option or significantly improves an existing Benefit Plan Option, the Plan Administrator may permit the following election changes: (a) Participants who are enrolled in a Benefit Plan Option other than the newly added or significantly improved Benefit Plan Option may change their elections on a prospective basis to elect the newly added or significantly improved Benefit Plan Option; and (b) Employees who are otherwise eligible under Section 3.1 may elect the newly added or significantly improved Benefit Plan Option on a prospective basis, subject to the terms and limitations of the Benefit Plan Option. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether there has been an addition of, or a significant improvement in, a Benefit Plan Option in accordance with prevailing IRS guidance.

(3) Loss of Coverage Under Other Group Health Coverage. A Participant may prospectively change his or her election to add group health coverage for the Participant or his or her Spouse or Dependent, if such individual(s) loses coverage under any group health coverage sponsored by a governmental or educational institution, including (but not limited to) the following: a state children's health insurance program under Title XXI of the Social Security Act; a medical care program of an Indian Tribal government (as defined in Code Section 7701(a)(40), the Indian Health Service, or a tribal organization; a state health benefits risk pool; or a foreign government group health plan, subject to the terms and limitations of the applicable Benefit Plan Option(s).

(4) Change in Coverage Under An Employer Plan. A Participant may make a prospective election change that is on account of and corresponds with a change made under an employer plan (including a plan of the Employer or a plan of the Spouse's or Dependent's employer), so long as (a) the other cafeteria plan or qualified benefits plan permits its participants to make an election change that would be permitted under applicable IRS regulations; or (b) the Plan permits Participants to make an election for a Period of Coverage that is different from the plan year under the other cafeteria plan or qualified benefits plan. For example, if an election is made by the Participant's Spouse during his or her employer's open enrollment to drop coverage, the Participant may add coverage to replace the dropped coverage. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether a requested change is on account of and corresponds with a change made under the other employer plan, in accordance with prevailing IRS guidance.

Election changes may not be made to reduce Health FSA coverage during a Period of Coverage; however, election changes may be made to cancel Health FSA coverage completely due to the occurrence of any of the following events: death of a Spouse, divorce, legal separation, or annulment; death of a Dependent; change in employment status such that the Participant becomes ineligible for Health FSA coverage; or a Dependent's ceasing to satisfy eligibility requirements for Health FSA coverage. Notwithstanding the foregoing, such cancellation will not become effective to the extent that it would reduce future contributions to the Health FSA to a point where the total contributions for the Plan Year are less than the amount already reimbursed for the Plan Year. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will determine, based on prevailing IRS guidance, whether a requested change is on account of and corresponds with a Change in Status. Assuming that the general consistency requirement is satisfied, a requested election change must also satisfy the following specific consistency requirements in order for a Participant to be able to alter his or her election based on the specified Change in Status:

(4<u>5</u>) Loss of Spouse or Dependent Eligibility; Special COBRA Rules. For a Change in Status involving a Participant's divorce, annulment or legal separation from a Spouse, the death of a Spouse or a Dependent, or a Dependent's ceasing to satisfy the eligibility requirements for coverage, a Participant may only elect to cancel accident or health insurance coverage for (a) the Spouse involved in the divorce, annulment, or legal separation; (b) the deceased Spouse or Dependent; or (c) the Dependent that ceased to

satisfy the eligibility requirements. Canceling coverage for any other individual under these circumstances would fail to correspond with that Change in Status. Notwithstanding the foregoing, if the Participant or his or her Spouse or Dependent becomes eligible for COBRA (or similar health plan continuation coverage under state law) under the Employer's plan because of a reduction of hours or because the Participant's Dependent ceases to satisfy the eligibility requirements for coverage (and the Participant remains a Participant under this Plan), then the Participant may increase his or her election to pay for such coverage.

(26) Gain of Coverage Eligibility Under Another Employer's Plan. For a Change in Status in which a Participant or his or her Spouse or Dependent gains eligibility for coverage under a cafeteria plan or qualified benefit plan of the employer of the Participant's Spouse or Dependent as a result of a change in marital status or a change in employment status, a Participant may elect to cease or decrease coverage for that individual only if coverage for that individual becomes effective or is increased under the Spouse's or Dependent's employer's plan. The Plan Administrator may rely on a Participant's certification that the Participant has obtained or will obtain coverage under the Spouse's or Dependent's employer's plan, unless the Plan Administrator has reason to believe that the Participant's certification is incorrect.

(37) Special Consistency Rule for DCAP Benefits. With respect to the DCAP Benefits, a Participant may change or terminate his or her election upon a Change in Status if (a) such change or termination is made on account of and corresponds with a Change in Status that affects eligibility for coverage under an employer's plan; or (b) the election change is on account of and corresponds with a Change in Status that affects eligibility of Dependent Care Expenses for the tax exclusion under Code Section 129.

A Participant entitled to change an election as described in this Section 10.3 must do so in accordance with the procedures described in Section 10.2.

10.4 Election Modifications Required by Plan Administrator

The Plan Administrator may, at any time, require any Participant or class of Participants to amend the amount of their Salary Reductions for a Period of Coverage if the Plan Administrator determines that such action is necessary or advisable in order to (a) satisfy any of the Code's nondiscrimination requirements applicable to this Plan or other cafeteria plan; (b) prevent any Employee or class of Employees from having to recognize more income for federal income tax purposes from the receipt of benefits hereunder than would otherwise be recognized; (c) maintain the qualified status of benefits received under this Plan; or (d) satisfy Code nondiscrimination requirements or other limitations applicable to the Employer's qualified plans. In the event that contributions need to be reduced for a class of Participants, the Plan Administrator will reduce the Salary Reduction amounts for each affected Participant, beginning with the Participant in the class who had elected the highest Salary Reduction amount and continuing with the Participant in the class who had elected the next-highest Salary Reduction amount, and so forth, until the defect is corrected.

ARTICLE XI. Appeals Procedure

11.1 Procedure If Benefits Are Denied Under This Plan

If a claim for reimbursement under this Plan is wholly or partially denied, then claims will be administered in accordance with the claims procedure set forth in Appendix C of this Plan.

11.2 Claims Procedures for Insurance Benefits

Claims and reimbursement for benefits under any Insurance Plan will be administered in accordance with the claims procedures for the Insurance Plans, as set forth in their governing plan documents and/or summary plan descriptions.

ARTICLE XII. Recordkeeping and Administration

12.1 Plan Administrator

The administration of this Plan will be under the supervision of the Plan Administrator. It is the principal duty of the Plan Administrator to see that this Plan is carried out, in accordance with its terms, for the exclusive benefit of persons entitled to participate in this Plan without discrimination among them.

12.2 Powers of the Plan Administrator

The Plan Administrator will have such duties and powers as it considers necessary or appropriate to discharge its duties. It will have the exclusive right to interpret the Plan and to decide all matters thereunder, and all determinations of the Plan Administrator with respect to any matter hereunder will be conclusive and binding on all persons. Without limiting the generality of the foregoing, the Plan Administrator will have the following discretionary authority:

- to construe and interpret this Plan, including all possible ambiguities, inconsistencies, and omissions in the Plan and related documents, and to decide all questions of fact, questions relating to eligibility and participation, and questions of benefits under this Plan;
- (b) to prescribe procedures to be followed and the forms to be used by Employees and Participants to make elections pursuant to this Plan;
- (c) to prepare and distribute information explaining this Plan and the benefits under this Plan in such manner as the Plan Administrator determines to be appropriate;
- (d) to request and receive from all Employees and Participants such information as the Plan Administrator will from time to time determine to be necessary for the proper administration of this Plan;
- (e) to furnish each Employee and Participant with such reports with respect to the administration of this Plan as the Plan Administrator determines to be reasonable and appropriate, including appropriate statements setting forth the amounts by which a Participant's Compensation has been reduced in order to provide benefits under this Plan;
- (f) to receive, review, and keep on file such reports and information regarding the benefits covered by this Plan as the Plan Administrator determines from time to time to be necessary and proper;

- (g) to appoint and employ such individuals or entities to assist in the administration of this Plan as it determines to be necessary or advisable, including legal counsel and benefit consultants;
- (h) to sign documents for the purposes of administering this Plan, or to designate an individual or individuals to sign documents for the purposes of administering this Plan;
- (i) to secure independent medical or other advice and require such evidence as it deems necessary to decide any claim or appeal; and
- (j) to maintain the books of accounts, records, and other data in the manner necessary for proper administration of this Plan and to meet any applicable disclosure and reporting requirements.

12.3 Reliance on Participant, Tables, etc.

The Plan Administrator may rely upon the direction, information, or election of a Participant as being proper under the Plan and will not be responsible for any act or failure to act because of a direction or lack of direction by a Participant. The Plan Administrator will also be entitled, to the extent permitted by law, to rely conclusively on all tables, valuations, certificates, opinions, and reports that are furnished by accountants, attorneys, or other experts employed or engaged by the Plan Administrator.

12.4 Provision for Third-Party Plan Service Providers

The Plan Administrator, subject to approval of the Employer, may employ the services of such persons as it may deem necessary or desirable in connection with the operation of the Plan. Unless otherwise provided in the service agreement, obligations under this Plan will remain the obligation of the Plan Administrator or the Employer, as applicable.

12.5 Fiduciary Liability

To the extent permitted by law, the Plan Administrator will not incur any liability for any acts or for failure to act except for their own willful misconduct or willful breach of this Plan.

12.6 Compensation of Plan Administrator

Unless otherwise determined by the Employer and permitted by law, any Plan Administrator that is also an Employee of the Employer will serve without compensation for services rendered in such capacity, but all reasonable expenses incurred in the performance of their duties will be paid by the Employer.

12.7 Insurance Contracts

The Employer will have the right (a) to enter into a contract with one or more insurance companies for the purposes of providing any benefits under the Plan; and (b) to replace any of such insurance companies or contracts. Any dividends, retroactive rate adjustments, or other refunds of any type that may become payable under any such insurance contract will not be assets of the Plan but will be the property of and be retained by the Employer, to the extent that such amounts are less than aggregate Employer contributions toward such insurance.

12.8 Inability to Locate Payee

If the Plan Administrator is unable to make payment to any Participant or other person to whom a payment is due under the Plan because it cannot ascertain the identity or whereabouts of such Participant or other person after reasonable efforts have been made to identify or locate such person, then such payment and all subsequent payments otherwise due to such Participant or other person will be forfeited following a reasonable time after the date any such payment first became due.

12.9 Effect of Mistake

In the event of a mistake as to the eligibility or participation of an Employee, the allocations made to the account of any Participant, or the amount of benefits paid or to be paid to a Participant or other person, the Plan Administrator will, to the extent that it deems administratively possible and otherwise permissible under Code Section 125 or the regulations issued thereunder, cause to be allocated or cause to be withheld or accelerated, or otherwise make adjustment of, such amounts as it will in its judgment accord to such Participant or other person the credits to the account or distributions to which he or she is properly entitled under the Plan. Such action by the Plan Administrator may include withholding of any amounts due to the Plan or the Employer from Compensation paid by the Employer.

ARTICLE XIII. General Provisions

13.1 Expenses

All reasonable expenses incurred in administering the Plan are currently paid by forfeitures to the extent provided in Section 7.6 with respect to Health FSA Benefits and Section 8.6 with respect to DCAP Benefits, and then by the Employer.

13.2 No Contract of Employment

Nothing herein contained is intended to be or will be construed as constituting a contract or other arrangement between any Employee and the Employer to the effect that such Employee will be employed for any specific period of time.

13.3 Amendment and Termination

This Plan has been established with the intent of being maintained for an indefinite period of time. Nonetheless, the Employer may amend or terminate all or any part of this Plan at any time for any reason by resolution of the Employer's Board of Directors or by any person or persons authorized by the Board of Directors to take such action.

13.4 Governing Law

The provisions of the Plan will be construed, administered and enforced according to applicable federal law and, to the extent not preempted, the laws of the State of California.

13.5 Compliance With Code and Other Applicable Laws

It is intended that this Plan meet all applicable requirements of the Code and of all regulations issued thereunder. This Plan will be construed, operated, and administered accordingly, and in the event of any conflict between any part, clause, or provision of this Plan and the Code, the provisions of the Code will be deemed controlling, and any conflicting part, clause, or provision of this Plan will be deemed superseded to the extent of the conflict. In addition, the Plan will comply with the requirements of all other applicable laws.

13.6 No Guarantee of Tax Consequences

Neither the Plan Administrator nor the Employer makes any commitment or guarantee that any amounts paid to or for the benefit of a Participant under this Plan will be excludable from the Participant's gross income for federal, state, or local income tax purposes. It will be the obligation of each Participant to determine whether each payment under this Plan is excludable from the Participant's gross income for federal, state, and local income tax purposes and to notify the Plan Administrator if the Participant has any reason to believe that such payment is not so excludable.

13.7 Indemnification of Employer

If any Participant receives one or more payments or reimbursements under this Plan on a taxfree basis and if such payments do not qualify for such treatment under the Code, then such Participant will indemnify and reimburse the Employer for any liability that it may incur for failure to withhold federal income taxes, Social Security or Medicare taxes, or other taxes from such payments or reimbursements.

13.8 Non-Assignability of Rights

The right of any Participant to receive any reimbursement under this Plan will not be alienable by the Participant by assignment or any other method and will not be subject to claims by the Participant's creditors by any process whatsoever. Any attempt to cause such right to be so subjected will not be recognized, except to the extent required by law.

13.9 Headings

The headings of the various Articles and Sections are inserted for convenience of reference and are not to be regarded as part of this Plan or as indicating or controlling the meaning or construction of any provision.

13.10 Plan Provisions Controlling

In the event that the terms or provisions of any summary or description of this Plan are in any construction interpreted as being in conflict with the provisions of this Plan as set forth in this document, the provisions of this Plan will be controlling.

13.11 Severability

Should any part of this Plan subsequently be invalidated by a court of competent jurisdiction, the remainder of the Plan will be given effect to the maximum extent possible.

* * *

To record the adoption of the Plan, the Employer's authorized representative hereby executes this document on this _____ day of _____, <u>20122021</u>.

Alameda County Transportation Commission

Ву: _____

Title:

Date: _____

Appendix A

Benefit Plan Options

Benefit Plan Options will include the coverage available under the following plans maintained by the Alameda County Transportation Commission:

A. Insurance Plans

- 1. Medical Insurance under the Public Employees' Medical and Hospital Care Act (PEMHCA or "PERS Health")
- 2. Dental Insurance
- 3. Vision Insurance
- 4. Long-Term Disability Insurance
- 5. Short-Term Disability Insurance
- 6. Group-Term Life Insurance (on the life of an Employee only)

B. Health Care Flexible Spending Account

C. Dependent Care Flexible Spending Account

Appendix B

Exclusions: Medical Expenses that are Not Reimbursable from the Health FSA

The Alameda County Transportation Commission Cafeteria Plan document contains the general rules governing what expenses are reimbursable. This Appendix B, as referenced in the Plan document, specifies certain expenses that are excluded under this Plan with respect to reimbursement from the Health FSA—that is, expenses that *are not reimbursable*, even if they meet the definition of "medical care" under Code Section 213(d) and may otherwise be reimbursable under the regulations governing Health FSAs. <u>Such "medical care" also includes expenses incurred for menstrual care products</u>. A 'menstrual care product' means a tampon, pad, liner, cup, sponge, or similar product used by individuals with respect to menstruation or other genital-tract secretions.

Exclusions: The following expenses are not reimbursable from the Health FSA, even if they meet the definition of "medical care" under Code Section 213(d) and may otherwise be reimbursable under legal requirements applicable to health FSAs:

- Premiums for other health coverage, including but not limited to premiums for any other plan (whether or not sponsored by the Employer)
- Long-term care services
- Cosmetic surgery or other similar procedures, unless the surgery or procedure is necessary to ameliorate a deformity arising from, or directly related to, a congenital abnormality, a personal injury resulting from an accident or trauma, or a disfiguring disease. "Cosmetic surgery" means any procedure that is directed at improving the patient's appearance and does not meaningfully promote the proper function of the body or prevent or treat illness or disease.
- The salary expense of a nurse to care for a healthy newborn at home
- Funeral and burial expenses
- Household and domestic help (even if recommended by a qualified physician due to an Employee's or Dependent's inability to perform physical housework)
- Custodial care

Medicines or drugs (other than insulin) that have not been prescribed

- Costs for sending a problem child to a special school for benefits that the child may receive from the course of study and disciplinary methods
- Social activities, such as dance lessons (even if recommended by a physician for general health improvement)
- Bottled water
- Cosmetics, toiletries, toothpaste, etc.
- Uniforms or special clothing, such as maternity clothing
- Automobile insurance premiums
- Transportation expenses of any kind, including transportation expenses to receive medical care
- Marijuana and other controlled substances that are in violation of federal laws, even if prescribed by a physician
- Any item that does not constitute "medical care" as defined under Code Section 213(d)
- Any item that is not reimbursable due to the rules in Prop. Treas. Reg. Section 1.125-5(k)(4) or other applicable law or regulations

Appendix C

Claims Procedures

Capitalized terms in this Appendix C have the same meaning as the defined terms in the Alameda County Transportation Commission Cafeteria Plan.

Any Participant may file a claim with the Plan Administrator for a Plan benefit to which the claimant believes that he or she is entitled.

- 1. The Plan Administrator will receive all claims filed for benefits under the Plan. Upon receiving a claim, the Plan Administrator will review the claim and determine whether the claimant is entitled to receive any benefits pursuant to such claim. The Plan Administrator will notify the claimant in writing of any adverse decision with respect to his or her claim within 30 days after its submission. The notice of any adverse decision will be written in a manner calculated to be understood by the claimant and must include, as applicable: (i) the specific reason or reasons for the denial; (ii) specific references to the Plan provisions on which the denial is based; (iii) a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary; and (iv) an explanation of the Plan's claim review procedures.
- 2. If the circumstances require an extension of time for processing the initial claim, a written notice of the extension will be furnished to the claimant before the end of the initial 30-day period. This time period may be extended by an additional 15 days for matters beyond the control of the Plan Administrator, including in cases where a reimbursement claim is incomplete. The extension notice must indicate the circumstances requiring an extension of time.
- 3. If a claim for benefits is denied or if the Plan Administrator has given no response to such claim within the time period set out in the above paragraph (in which case the claim for benefits will be deemed to be denied), the claimant or his or her duly authorized representative, at the claimant's sole expense, may appeal the denial by submitting written notice of such appeal to the Plan Administrator within 90 days of the receipt of written notice of the denial or 60 days from the date such claim is deemed to be denied.
- 4. The claimant will be notified of the decision on the appeal within 90 days of receipt of the notice of appeal, unless circumstances require an extension of time for processing, in which case a decision will be rendered as soon as possible, but not later than 120 days after receipt of a notice of appeal. If such an extension of time is required, written notice of the extension will be furnished to the claimant before the end of the original 90-day period. The notice of decision on the appeal must be made in writing. If the decision on the appeal is not furnished within the time specified above, the appeal of the claim will be deemed denied.

This page intentionally left blank

ALAMEDA COUNTY TRANSPORTATION COMMISSION CAFETERIA PLAN

As Restated Effective January 1, 20202021

2962648.2



TABLE OF CONTENTS

ARTICL	E I. Introduction	1
	ESTABLISHMENT OF PLAN	4
1.1	ESTABLISHMENT OF PLAN	······
1.2	LEGAL STATUS	1
ARTICL	E II. Definitions	<u> </u>
2. 1	Definitions	2
	E III. Eligibility and Participation	6
3.1 —	ELIGIBILITY TO PARTICIPATE	 6
3.2	USE OF CONTRIBUTIONS	 6
3.3	TERMINATION OF PARTICIPATION	 6
3.4 —		 7
3.5	FMLA LEAVES OF ABSENCE	 7
3.6	Non-FMLA LEAVES OF ABSENCE	9
	E IV. Method and Timing of Elections	•
		 9
4.1	ELECTION TO PARTICIPATE	9
4.2	ELECTIONS WHEN FIRST ELIGIBLE	9
4.3—		9
4.4	FAILURE OF ELIGIBLE EMPLOYEE TO FILE AN ELECTION FORM/SALARY REDUCTION AGREEMENT	F 10
4.5	IRREVOCABILITY OF ELECTIONS.	10
		4.0
ARHCL	E V. Benefits Offered and Method of Funding	10
5.1	Benefits Offered	<u> </u>
5.2	Source of Benefit Funding	11
5.3	EMPLOYER CONTRIBUTIONS	<u> </u>
ARTICL	E VI. Premium Payment Component	11
6.1	Benefits	11
6.2	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE	11
6.3	BENEFITS PROVIDED UNDER THE INSURANCE PLANS	
6.4	Insurance Benefits; COBRA	
••••		
ARTICL	E VII. Health FSA Component	12
7.1	HEALTH FSA BENEFITS	12
7.2	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE OF HEALTH FSA BENEFITS	12
7.3	Eligible Medical Care Expenses for Health FSA	
7.4	MAXIMUM AND MINIMUM BENEFITS FOR HEALTH FSA	
7.5	ESTABLISHMENT OF HEALTH FSA ACCOUNT	15
7.6	Forfeiture of Health FSA Accounts; Use-or-Lose Rule	10
	REIMBURSEMENT CLAIMS PROCEDURE FOR HEALTH FSA	10
	REIMBURSEMENT CLAIMS FROCEDURE FOR HEALTH FSA	
	COORDINATION OF BENEFITS	
ARTICL	E VIII. DCAP Component	18
	DCAP BENEFITS	
8.1 —	DCAP BENEFITS PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE FOR DCAP BENEFITS	
8.3	ELIGIBLE DEPENDENT CARE EXPENSES	
8.4	MAXIMUM AND MINIMUM BENEFITS FOR DCAP	
8.5 —	ESTABLISHMENT OF DCAP ACCOUNT	
	Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule	
8.7 —		23

8.8 —	REIMBURSEMENTS FROM DCAP AFTER TERMINATION OF PARTICIPATION	24
ARTICL	E IX. HIPAA PROVISIONS FOR HEALTH FSA	24
9.1	PROVISION OF PROTECTED HEALTH INFORMATION TO EMPLOYER	24
9.2	PERMITTED DISCLOSURE OF ENROLLMENT/DISENROLLMENT INFORMATION	
9.3	PERMITTED USES AND DISCLOSURE OF SUMMARY HEALTH INFORMATION	
9.4	PERMITTED AND REQUIRED USES AND DISCLOSURE OF PHI FOR PLAN ADMINISTRATION PUT	
9.5	Conditions of Disclosure for Plan Administration Purposes	25
9.6	ADEQUATE SEPARATION BETWEEN PLAN AND EMPLOYER	-
9.7	CERTIFICATION OF PLAN SPONSOR	27
9.8	PRIVACY OFFICIAL	27
9.9	INTERPRETATION AND LIMITED APPLICABILITY	27
9.10		
ARTICL	E X. Irrevocability of Elections; Exceptions	28
10.1	IRREVOCABILITY OF ELECTIONS	28
10.1		
1012	Events Permitting Exception to Irrevocability Rule for All Benefits	-
10.3		
ARTICL	E XI. Appeals Procedure	35
11.1_	PROCEDURE IF BENEFITS ARE DENIED UNDER THIS PLAN	35
	CLAIMS PROCEDURES FOR INSURANCE BENEFITS	
ARTICL	E XII. Recordkeeping and Administration	36
12.1	PLAN ADMINISTRATOR	36
12.2	POWERS OF THE PLAN ADMINISTRATOR	
12.3		
12.4		37
12.5		37
12.6	COMPENSATION OF PLAN ADMINISTRATOR	37
12.7	Insurance Contracts	
12.8	INABILITY TO LOCATE PAYEE	37
12.9	EFFECT OF MISTAKE	38
ARTICL	E XIII. General Provisions	
40.4	EXPENSES	20
-		
13.3 13.4	AMENDMENT AND TERMINATION	
13.4 13.5	COMPLIANCE WITH CODE AND OTHER APPLICABLE LAWS	
13.0 13.6		
13.0 13.7	NO GUARANTEE OF TAX CONSEQUENCES	39 39
10./ 42.0		
13.0 42.0	HEADINGS	
13.3	PLAN PROVISIONS CONTROLLING	
10.10		
ARTICL	E I. Introduction	
<u>1.1</u>	ESTABLISHMENT OF PLAN	
<u>1.2</u>	LEGAL STATUS	<u></u> 1
ARTICL	E II. Definitions	2
2.1		
Z .		Z

ARTICLE	III. Eligibility and Participation	<u> </u>
3.1	ELIGIBILITY TO PARTICIPATE	
3.2	USE OF CONTRIBUTIONS	6
3.3	TERMINATION OF PARTICIPATION	6
3.4	PARTICIPATION FOLLOWING TERMINATION OF EMPLOYMENT OF LOSS OF ELIGIBILITY	7
3.5	FMLA LEAVES OF ABSENCE	
3.6	Non-FMLA LEAVES OF ABSENCE	9
ARTICLE	EIV. Method and Timing of Elections	<u></u> 9
4.1	ELECTION TO PARTICIPATE	
4.2	ELECTIONS WHEN FIRST ELIGIBLE	9
4.3	ELECTIONS DURING OPEN ENROLLMENT PERIOD	9
4.4	FAILURE OF ELIGIBLE EMPLOYEE TO FILE AN ELECTION FORM/SALARY REDUCTION AGREEMENT .	<u>.</u> 10
4.5	IRREVOCABILITY OF ELECTIONS	<u>.</u> 10
	V. Benefits Offered and Method of Funding	10
ANTICLE	-	
<u>5.1</u>	BENEFITS OFFERED	<u>.</u> 10
5.2	Source of Benefit Funding	
<u>5.3</u>	EMPLOYER CONTRIBUTIONS	<u>.</u> 11
	EVI. Premium Payment Component	11
<u>6.1</u>	BENEFITS	
<u>6.2</u>	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE	
<u>6.3</u>	BENEFITS PROVIDED UNDER THE INSURANCE PLANS	
<u>6.4</u>	INSURANCE BENEFITS; COBRA	<u>.</u> 12
ARTICLE	VII. Health FSA Component	. 12
7.1	HEALTH FSA BENEFITS	<u>-</u> 12
7.2	PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE OF HEALTH FSA BENEFITS	<u>-</u> 12
7.3	ELIGIBLE MEDICAL CARE EXPENSES FOR HEALTH FSA	<u>-12</u>
7.4	MAXIMUM AND MINIMUM BENEFITS FOR HEALTH FSA	
7.5	ESTABLISHMENT OF HEALTH FSA ACCOUNT	
7.6	FORFEITURE OF HEALTH FSA ACCOUNTS; USE-OR-LOSE RULE	<u>.</u> 16
7.7	REIMBURSEMENT CLAIMS PROCEDURE FOR HEALTH FSA	
7.8	REIMBURSEMENTS FROM HEALTH FSA AFTER TERMINATION OF PARTICIPATION; COBRA	
7.9	COORDINATION OF BENEFITS	
ARTICLE	E VIII. DCAP Component	.18
0.4		10
<u>8.1</u>	DCAP BENEFITS PARTICIPANT CONTRIBUTIONS FOR COST OF COVERAGE FOR DCAP BENEFITS	
<u>8.2</u>	ELIGIBLE DEPENDENT CARE EXPENSES.	
<u>8.3</u> 8.4	ELIGIBLE DEPENDENT CARE EXPENSES. MAXIMUM AND MINIMUM BENEFITS FOR DCAP	-10 10
<u>0.4</u> 8.5	ESTABLISHMENT OF DCAP ACCOUNT	
<u>8.6</u>	FORFEITURE OF DCAP ACCOUNTS; USE-IT-OR-LOSE-IT RULE.	<u>-</u> 22
<u>8.7</u>	REIMBURSEMENT CLAIMS PROCEDURE FOR DCAP	<u>-</u> 23
<u>8.8</u>	REIMBURSEMENT CLAIMS PROCEDURE FOR DCAP	<u>-</u> 23
ARTICLE	EIX. HIPAA PROVISIONS FOR HEALTH FSA	<u>.</u> 24
<u>9.1</u>	PROVISION OF PROTECTED HEALTH INFORMATION TO EMPLOYER	24
<u>9.1</u> 9.2	PROVISION OF PROTECTED HEALTH INFORMATION TO EMPLOYER PERMITTED DISCLOSURE OF ENROLLMENT/DISENROLLMENT INFORMATION	
<u>9.2</u> 9.3	PERMITTED USES AND DISCLOSURE OF SUMMARY HEALTH INFORMATION	
<u>9.3</u> 9.4	PERMITTED AND REQUIRED USES AND DISCLOSURE OF SOMMARY HEALTH INFORMATION.	
<u>J.4</u>	25	-0
9.5	Conditions of Disclosure for Plan Administration Purposes	25
<u>9.6</u>	ADEQUATE SEPARATION BETWEEN PLAN AND EMPLOYER	

<u>9.7</u>	CERTIFICATION OF PLAN SPONSOR	<u></u> 27
9.8	PRIVACY OFFICIAL	27
9.9	INTERPRETATION AND LIMITED APPLICABILITY	
9.10	SERVICE PERFORMED FOR THE EMPLOYER	
	X. Irrevocability of Elections; Exceptions	20
ARTICLE		
<u>10.1</u>	IRREVOCABILITY OF ELECTIONS	<u></u> 28
10.2	PROCEDURE FOR MAKING NEW ELECTION IF EXCEPTION TO IRREVOCABILITY APPLIES	
<u>10.3</u>	EVENTS PERMITTING EXCEPTION TO IRREVOCABILITY RULE FOR ALL BENEFITS	
<u>10.4</u>	ELECTION MODIFICATIONS REQUIRED BY PLAN ADMINISTRATOR	<u></u> 35
ARTICLE	XI. Appeals Procedure	
11.1		
11.2		
ARTICLE	XII. Recordkeeping and Administration	
12.1	PLAN ADMINISTRATOR	
12.2	POWERS OF THE PLAN ADMINISTRATOR	
12.3	RELIANCE ON PARTICIPANT, TABLES, ETC.	
12.4	PROVISION FOR THIRD-PARTY PLAN SERVICE PROVIDERS	
12.5	FIDUCIARY LIABILITY	
12.6	COMPENSATION OF PLAN ADMINISTRATOR	
12.7	INSURANCE CONTRACTS	
12.8	INABILITY TO LOCATE PAYEE	
12.9	EFFECT OF MISTAKE	
ARTICLE	XIII. General Provisions	
13.1	Expenses	
13.2	NO CONTRACT OF EMPLOYMENT	
13.3	AMENDMENT AND TERMINATION	
13.4	GOVERNING LAW	
13.5	COMPLIANCE WITH CODE AND OTHER APPLICABLE LAWS	
13.6	NO GUARANTEE OF TAX CONSEQUENCES	
13.7	INDEMNIFICATION OF EMPLOYER.	
13.8	NON-ASSIGNABILITY OF RIGHTS	
13.9	HEADINGS	
13.10	PLAN PROVISIONS CONTROLLING	
13.11	SEVERABILITY	

ALAMEDA COUNTY TRANSPORTATION COMMISSION CAFETERIA PLAN

As RestatedAdopted Effective January 1, 20202021

ARTICLE I. Introduction

1.1 Establishment of Plan

Alameda County Transportation Commission (the Employer) hereby restates Alameda County Transportation Commission Cafeteria Plan (the Plan) effective January 1, 20202021 (the Effective Date) to incorporate applicable COVID-19 related relief in accordance with the Consolidated Appropriations Act, 2021 and, IRS Notice 2021-15- and the American Rescue Plan of 2021. The Plan was established effective as of February 1, 2012 and was last restated effective January 1, 2020. Capitalized terms used in this Plan that are not otherwise defined have the meanings set forth in Article II.

This Plan is designed to permit an Eligible Employee to pay for his or her share of contributions for the Benefit Plan Options in Appendix A on a pre-tax Salary Reduction basis, and to contribute on a pre-tax Salary Reduction basis to an Employee's account for reimbursement of certain Medical Care Expenses (Health FSA Account) and/or to an account for reimbursement of certain Dependent Care Expenses (DCAP Account).

1.2 Legal Status

This Plan is intended to qualify as a cafeteria plan under Code Section 125 and will be interpreted and administered consistent with the requirements of Code Section 125 and the regulations issued thereunder.

The Health FSA Component is intended to qualify as a self-insured medical reimbursement plan under Code Section 105, and the Medical Care Expenses reimbursed thereunder are intended to be eligible for exclusion from participating Employees' gross income under Code Section 105(b). The DCAP Component is intended to qualify as a dependent care assistance program under Code Section 129, and the Dependent Care Expenses reimbursed thereunder are intended to be eligible for exclusion from participating Employees' gross income under Code Section 129(a).

Although reprinted within this document, the Health FSA Component and the DCAP Component are separate plans for purposes of administration and all reporting and nondiscrimination requirements imposed by Code Sections 105 and 129. The Health FSA Component is also a separate plan for purposes of applicable provisions of HIPAA and COBRA. In the event that the Health FSA Component is determined not to be a separate plan, the Plan will be designated as a hybrid entity for purposes of HIPAA, such that it will be a covered entity only with respect to the Health FSA Component.

ARTICLE II. Definitions

2.1 Definitions

Account(s) means the Health FSA Accounts and the DCAP Accounts described in Sections 7.5 and 8.5.

Benefit Plan Option means a qualified benefit under Code Section 125(f) that is available to a Participant under this Plan as set forth in Appendix A, as amended from time to time. The Employer may substitute, add, subtract, or revise at any time the menu of such Benefit Plan Options and/or the benefits, terms, and conditions of any such options or plans. Any such substitution, addition, subtraction, or revision will be communicated to Participants and will automatically be incorporated by reference under this Plan.

Board of Directors means the Board of Commissioners of the Alameda County Transportation Commission.

Change in Status means any of the events described below, as well as any other events included under subsequent changes to Code Section 125 or regulations issued thereunder, which the Plan Administrator, in its sole discretion and on a uniform and consistent basis, determines are permitted under IRS regulations and under this Plan:

- (a) *Legal Marital Status.* A change in a Participant's legal marital status, including marriage, death of a Spouse, divorce, legal separation, or annulment;
- (b) *Number of Dependents.* Events that change a Participant's number of Dependents, including birth, death, adoption, and placement for adoption;
- (c) Employment Status. Any of the following events that change the employment status of the Participant or his or her Spouse or Dependents: (1) a termination or commencement of employment; (2) a strike or lockout; (3) a commencement of or return from an unpaid leave of absence; (4) a change in worksite; and (5) the eligibility conditions of this Plan or other employee benefits plan of the Participant or his or her Spouse or Dependents depend on the employment status of that individual and there is a change in that individual's status with the consequence that the individual becomes (or ceases to be) eligible under this Plan or other employee benefits plan;
- (d) *Dependent Eligibility Requirements.* An event that causes a Dependent to satisfy or cease to satisfy the Dependent eligibility requirements for a particular benefit, such as attaining a specified age, student status, or any similar circumstance; and
- (e) *Change in Residence.* A change in the place of residence of the Participant or his or her Spouse or Dependents.

COBRA means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.

Code means the Internal Revenue Code of 1986 and the Treasury Regulations issued thereunder, as amended.

Compensation means the cash wages or salary paid to an Employee by the Employer.

DCAP means dependent care assistance program.

DCAP Account means the account described in Section 8.5.

DCAP Benefits has the meaning described in Section 8.1.

DCAP Component means the component of this Plan described in Article VIII.

Dependent means: (a) for purposes of accident or health coverage, (1) a dependent as defined in Code Section 152, determined without regard to subsections (b)(1), (b)(2), and (d)(1)(B) thereof, (2) any child (as defined in Code Section 152(f)(1) of the Participant who as of the end of the taxable year has not attained age 27, and (3) any child of the Participant to whom IRS Revenue Procedure 2008-48 applies (regarding certain children of divorced or separated parents who receive more than half of their support for the calendar year from one or both parents and are in the custody of one or both parents for more than half of the calendar year); and (b) for purposes of the DCAP Component, a Qualifying Individual. Notwithstanding the foregoing, the Health FSA Component will provide benefits in accordance with the applicable requirements of any National Medical Support Order, even if the child does not meet the definition of Dependent.

Dependent Care Expenses has the meaning described in Section 8.3.

Earned Income will have the meaning given such term in Code Section 129(e)(2).

Effective Date of this Plan means January 1, 20202021.

Election Form/Salary Reduction Agreement means the form provided by the Plan Administrator for the purpose of allowing an Eligible Employee to participate in this Plan by electing Benefit Plan Options(s) and authorizing Salary Reductions to pay for any of the Benefit Plan Options.

Eligible Employee means an Employee eligible to participate in this Plan, as provided in Section 3.1.

Employee means an individual that the Employer classifies as a common-law employee and who is on the Employer's W-2 payroll, but does not include the following: (a) any leased employee (including but not limited to those individuals defined as leased employees in Code Section 414(n) or an individual classified by the Employer as a contract worker, independent contractor, temporary employee, or casual employee for the period during which such individual is so classified, whether or not any such individual is on the Employer's W-2 payroll or is determined by the IRS or others to be a common-law employee of the Employer; (b) any individual who performs services for the period during which such individual is paid by such agency, whether or not such individual is determined by the IRS or others to be a common-law employee of the Employer; and (c) any employee covered under a collective bargaining agreement, unless that agreement provides for the employee's participation in the Plan. The term Employee does include former Employees for the limited purpose of allowing continued eligibility for benefits under the Plan for a limited duration following termination of employment

provided any required contributions are made and only to the extent specifically provided under this Plan.

Employer means the Alameda County Transportation Commission.

FMLA means the Family and Medical Leave Act of 1993, as amended.

Grace Period means the period that begins immediately following the close of a Plan Year and ends on the day that is two months plus 15 days following the close of that Plan Year. Notwithstanding the foregoing, for purposes of the Plan Year ending December 31, 20202021, the Grace Period is the period from January 1, 20212022 through December 31, 20212022. (Note: For purposes of the DCAP, as the Grace Period is available pursuant to the relief provided in the Consolidated Appropriations Act, 2021, the Grace Period is available for amounts in DCAP Accounts that remain as of December 31, 20202021 only. The Grace Period is available under the Health FSA irrespective of the relief.)

Health FSA means the health flexible spending arrangement.

Health FSA Account means the account described in Section 7.5.

Health FSA Benefits has the meaning described in Section 7.1.

Health FSA Component means the component of this Plan described in Article VII.

HIPAA means the Health Insurance Portability and Accountability Act of 1996, as amended.

Insurance Plan(s) means the plan(s) that the Employer maintains for its Employees (and for their Spouses, same-sex spouses, domestic partners, and Dependents that may be eligible under the terms of such plan), which provide benefits through a group insurance policy or policies (e.g., medical, dental and vision insurance). The Employer may substitute, add, subtract, or revise at any time the menu of such plans and/or the benefits, terms, and conditions of any such plans. Any such substitution, addition, subtraction, or revision will be communicated to Participants and will automatically be incorporated by reference under this Plan.

Medical Care Expenses has the meaning described in Section 7.3.

Nonelective Contribution(s) means any amount that the Employer, in its sole discretion, may contribute under the Plan to provide benefits for individual Participants and their Spouses, Dependents, domestic partners, and same-sex spouses, as applicable, under one or more of the Benefit Plan Options offered under the Plan.

Open Enrollment Period means the period during the Plan Year during which Eligible Employees may elect to participate in the Plan or make changes to their elections for the next Plan Year. The Employer will determine this period each Plan Year, which the Plan Administrator will make known in the Plan's open enrollment materials. **Participant** means a person who is an Eligible Employee and who is participating in this Plan in accordance with the provisions of Article III. Participants include those who elect one or more Benefit Plan Options under the Plan.

Period of Coverage means the Plan Year, with the following exceptions: (a) for Employees who first become eligible to participate, it will mean the portion of the Plan Year following the date on which participation commences, as described in Section 4.2; and (b) for Employees who terminate participation, it will mean the portion of the Plan Year prior to the date on which participation terminates, as described in Section 3.3.

Plan means the Alameda County Transportation Commission Cafeteria Plan as set forth herein, as amended from time to time.

Plan Administrator means the person(s), entity, or committee as may be appointed from time to time by the Board of Directors (or its authorized designee) to administer the Plan. If no such person, entity, or committee is appointed, the Plan Administrator is the Employer.

Plan Sponsor means the Employer.

Plan Year means the calendar year (i.e., the 12-month period commencing January 1 and ending on December 31), except in the case of a short plan year representing the initial Plan Year or where the Plan Year is being changed, in which case the Plan Year will be the entire short plan year.

Premium Payment Benefits means the Premium Payment Benefits described in Section 6.1.

Premium Payment Component means the component of this Plan described in Article VI.

Qualifying Dependent Care Services has the meaning described in Section 8.3.

Qualifying Individual means (a) a tax dependent of the Participant as defined in Code Section 152 who is under the age of 13 and who is the Participant's gualifying child as defined in Code Section 152(a)(1); (b) a tax dependent of the Participant as defined in Code Section 152, but determined without regard to subsections (b)(1), (b)(2), and (d)(1)(B) thereof, who is physically or mentally incapable of self-care and who has the same principal place of abode as the Participant for more than half of the year; or (c) a Participant's Spouse who is physically or mentally incapable of self-care, and who has the same principal place of abode as the Participant for more than half of the year. Notwithstanding the foregoing, in the case of divorced or separated parents, a Qualifying Individual who is a child will, as provided in Code Section 21(e)(5), be treated as a Qualifying Individual of the custodial parent (within the meaning of Code Section 152(e)) and will not be treated as a Qualifying Individual with respect to the noncustodial parent. Furthermore, notwithstanding the foregoing, in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15 and subject to Section 8.6,4 for purposes of the 20202021 Plan Year, the term "Qualifying Individual" shall also include a tax dependent of the Participant as defined in Code Section 152(a)(1) who attains the age of thirteen (13) at any time during the 20202021 calendar year, who has the same principal place of abode as the Participant for more than one-half of the Plan Year, and who has not provided over one-half of his or her own support for the Plan Year.

Salary Reduction means the amount by which the Participant's Compensation is reduced and applied by the Employer under this Plan to pay for one or more of the benefits, as permitted for the applicable component, before any applicable state and/or federal taxes have been deducted from the Participant's Compensation (i.e., on a pre-tax basis).

Spouse means an individual who is legally married to a Participant as determined under applicable state law and who is treated as a spouse under the Code. A domestic partner or same-sex spouse is not treated as a spouse under the Code. Notwithstanding the above, for purposes of the DCAP Component the term Spouse does not include (a) an individual legally separated from the Participant under a divorce or separate maintenance decree; or (b) an individual who, although married to the Participant, files a separate federal income tax return, maintains a principal residence separate from the Participant during the last six months of the taxable year, and does not furnish more than half of the cost of maintaining the principal place of abode of the Participant.

Student means an individual who, during each of five or more calendar months during the Plan Year, is a full-time student at any educational organization that normally maintains a regular faculty and curriculum and normally has an enrolled student body in attendance at the location where its educational activities are regularly carried on.

ARTICLE III. Eligibility and Participation

3.1 Eligibility to Participate

All full-time Employees are eligible to participate in the Plan. To become a Participant, an Eligible Employee must make a timely election to participate in accordance with Article IV. Eligibility for any Benefit Plan Option will be subject to the requirements specified in the governing plan documents of the applicable Benefit Plan Option. The provisions of this Article are not intended to override any eligibility requirement or waiting period specified in the applicable Benefit Plan Options and the terms of eligibility and participation for any Benefit Plan Option offered under the Plan are subject to the requirements specified in the Benefit Plan Option's governing documents.

3.2 Use of Contributions

As a Participant, an Employee will be permitted to (1) elect Benefit Plan Options for which he or she is eligible, (2) receive available Nonelective Contributions for which he or she is eligible in the manner set forth in the enrollment materials, (3) pay his or her share of the cost of his or her elected benefits with Salary Reduction contributions, and (4) if permitted under the terms of the Benefit Plan Options and uniform rules adopted by the Plan Administrator, pay his or her share of the costs of the elected benefits with after-tax dollars (e.g., if Salary Reduction contributions are not available or are insufficient to pay his or her share of the cost of the Benefit Plan Option). In addition, as a Participant, an Employee may be permitted to elect health coverage for an individual who is not the employee's Spouse or Dependent if permitted under the terms of the Benefit Plan Options and in accordance with uniform rules adopted by the Plan Administrator; provided, however, that the fair market value of such coverage will be included in the Employee's gross income to the extent required by applicable law, and the Employee will be treated as having purchased the coverage with after-tax dollars.

3.3 Termination of Participation

A Participant will cease to be a Participant in this Plan upon the earlier of:

- (a) the date the Participant makes a permitted election not to participate in the Plan;
- (b) the date that the Participant no longer satisfies the eligibility requirements of this Plan or all of the Benefit Plan Options. Notwithstanding the foregoing, for purposes of pre-tax COBRA coverage, certain Employees may continue eligibility for certain periods subject to the restrictions and terms otherwise described in this Plan; or
- (c) The date that the Plan is either terminated or amended to exclude the Participant or the class of employees to which the Participant belongs.

Termination of participation in this Plan will automatically revoke the Participant's elections. Benefits under any Insurance Plan will terminate as of the date(s) specified in the Insurance Plan. Reimbursements from the Health FSA and DCAP Accounts after termination of participation will be made pursuant to Section 7.8 for Health FSA Benefits and Section 8.8 for DCAP Benefits. If revocation occurs under this Section 3.3, no new election may be made by such Participant during the remainder of the Plan Year except as set forth in Section 3.4.

3.4 Participation Following Termination of Employment or Loss of Eligibility

If a Participant terminates his or her employment for any reason, including (but not limited to) disability, retirement, layoff, or voluntary resignation, or otherwise loses eligibility and then is rehired or becomes eligible once again within 30 days or less after the date of a termination of employment or loss of eligibility, then the Employee will be reinstated with the same elections that such individual had before termination or other loss of eligibility. If a former Participant is rehired more than 30 days following termination of employment or becomes eligible after 30 days following a loss of eligibility and is otherwise eligible to participate in the Plan, then the individual may make new elections as a new hire as described in Section 4.2. Notwithstanding the above, an election to participate in the Premium Payment Component will be reinstated only to the extent that coverage under the applicable Insurance Plan is reinstated.

3.5 FMLA Leaves of Absence

(a) *Health Insurance Benefits*. Notwithstanding any provision to the contrary in this Plan, if a Participant goes on a qualifying leave under the FMLA, then to the extent required by the FMLA, the Employer will continue to maintain the Participant's health insurance benefits and Health FSA Benefits on the same terms and conditions as if the Participant were still an active Employee. That is, if the Participant elects to continue his or her coverage while on leave, the Employer will continue to pay its share of the contributions for those benefits under this Plan.

An Employer may require participants to continue all health insurance benefits and Health FSA Benefits coverage for Participants while they are on paid leave (provided that Participants on non-FMLA paid leave are required to continue coverage). If so, the Participant's share of the contributions will be paid by the method normally used during any paid leave (e.g., on a pre-tax Salary Reduction basis).

In the event of unpaid FMLA leave (or paid FMLA leave where coverage is not required to be continued), a Participant may elect to continue his or her health insurance benefits and Health FSA Benefits during the leave. If the Participant elects to continue coverage

while on FMLA leave, then the Participant may pay his or her share of the contributions in one of the following ways:

- with after-tax dollars, by sending monthly payments to the Employer by the due date established by the Employer;
- with pre-tax dollars, by having such amounts withheld from the Participant's ongoing Compensation (if any), including unused sick days and vacation days, or pre-paying all or a portion of the contributions for the expected duration of the leave on a pre-tax Salary Reduction basis out of pre-leave Compensation. To pre-pay the contributions, the Participant must make a special election to that effect prior to the date that such Compensation would normally be made available (pre-tax dollars may not be used to fund coverage during the next Plan Year); or
- under another arrangement agreed upon between the Participant and the Plan Administrator (e.g., the Plan Administrator may fund coverage during the leave and withhold "catch-up" amounts from the Participant's Compensation on a pretax or after-tax basis) upon the Participant's return.

If the Employer requires all Participants to continue health insurance benefits and Health FSA Benefits during an unpaid FMLA leave, then the Participant may elect to discontinue payment of the Participant's required contributions until the Participant returns from leave. Upon returning from leave, the Participant will be required to repay the contributions not paid by the Participant during the leave. Payment will be withheld from the Participant's Compensation either on a pre-tax or after-tax basis, as agreed to by the Plan Administrator and the Participant.

If a Participant's health insurance benefits or Health FSA Benefits coverage ceases while on FMLA leave (e.g., for non-payment of required contributions), then the Participant is permitted to re-enter the Premium Payment Component or Health FSA Component as applicable, upon return from such leave on the same basis as when the Participant was participating in the Plan prior to the leave, or as otherwise required by the FMLA. In addition, the Plan may require Participants whose health insurance benefits or Health FSA Benefits coverage terminated during the leave to be reinstated in such coverage upon return from a period of unpaid leave, provided that Participants who return from a period of unpaid, non-FMLA leave are required to be reinstated in such coverage. Notwithstanding the preceding sentence, with regard to Health FSA Benefits, a Participant whose coverage ceased will be permitted to elect whether to be reinstated in the Health FSA Benefits at the same coverage level as was in effect before the FMLA leave (with increased contributions for the remaining period of coverage) or at a coverage level that is reduced pro rata for the period of FMLA leave during which the Participant did not pay contributions. If a Participant elects a coverage level that is reduced pro rata for the period of FMLA leave, then the amount withheld from a Participant's Compensation on a pay-period-by-pay-period basis for the purpose of paying for reinstated Health FSA Benefits will be equal to the amount withheld prior to the period of FMLA leave.

(b) _____ Non-Health Benefits. If a Participant goes on a qualifying leave under the FMLA, then entitlement to non-health benefits (such as DCAP Benefits) is to be determined by the Employer's policy for providing such benefits when the Participant is on non-FMLA leave, as described in Section 3.6. If such policy permits a Participant to discontinue contributions while on leave, then the Participant will, upon returning from leave, be required to repay the contributions not paid by the Participant during the leave. Payment will be withheld from the Participant's Compensation either on a pre-tax or after-tax basis, as may be agreed upon by the Plan Administrator and the Participant or as the Plan Administrator otherwise deems appropriate.

3.6 Non-FMLA Leaves of Absence

If a Participant goes on an unpaid leave of absence that does not affect eligibility, then the Participant will continue to participate and the contributions due for the Participant will be paid by pre-payment before going on leave, by after-tax contributions while on leave, or with catchup contributions after the leave ends, as may be determined by the Plan Administrator. If a Participant goes on an unpaid leave that affects eligibility, then the applicable election change rules in Section 10.3 will apply.

ARTICLE IV. Method and Timing of Elections

4.1 Election to Participate

To become a Participant, an Eligible Employee must submit a completed and signed Election Form/Salary Reduction Agreement to the Plan Administrator in the time and in the manner required by the Plan Administrator.

4.2 Elections When First Eligible

- (a) *Currently Eligible Employees.* An Employee who is eligible to participate in this Plan as of the Effective Date must complete, sign, and file an Election Form/Salary Reduction Agreement with the Plan Administrator during the election period (as specified by the Plan Administrator) immediately preceding the Effective Date of the Plan to become a Participant on the Effective Date. The elections made by the Eligible Employee on this initial Election Form/Salary Reduction Agreement will be effective for the Plan Year beginning on the Effective Date.
- (b) New Employees or Newly Eligible Employees. An Employee who first becomes eligible to participate in the Plan mid-year (and after the Effective Date) may elect to commence participation in the Plan after the eligibility requirements of Section 3.1 have been satisfied by completing, signing, and filing an Election Form/Salary Reduction Agreement with the Plan Administrator in the time and in the manner required by the Plan Administrator. Participation in the Plan will commence on the first day of the month following the Plan Administrator's receipt of a properly completed and signed Election Form/Salary Reduction Agreement. An Employee who does not elect benefits when first eligible may not enroll until the next Open Enrollment Period, unless an event occurs that would justify a mid-year election change, as described under Section 10.3. Eligibility for Premium Payment Benefits will be subject to the additional requirements, if any, specified in the applicable Insurance Plans.

4.3 Elections During Open Enrollment Period

During each Open Enrollment Period with respect to a Plan Year, the Plan Administrator will provide an Election Form/Salary Reduction Agreement to each Employee who is eligible to participate in this Plan. The Election Form/Salary Reduction Agreement will enable the Employee to elect to participate in the various components of this Plan for the next Plan Year and to authorize the necessary Salary Reductions to pay for the benefits elected. The Election Form/Salary Reduction Agreement must be returned to the Plan Administrator on or before the last day of the Open Enrollment Period, and it will become effective on the first day of the next Plan Year. If an Eligible Employee fails to return the Election Form/Salary Reduction Agreement during the Open Enrollment Period, then the Employee may not elect any benefits under this Plan until the next Open Enrollment Period, unless an event occurs that would justify a mid-year election change, as described under Section 10.3.

4.4 Failure of Eligible Employee to File an Election Form/Salary Reduction Agreement

If an Eligible Employee fails to file an Election Form/Salary Reduction Agreement within the time period described in Sections 4.2 and 4.3, then the Employee may not elect any benefits under the Plan (a) until the next Open Enrollment Period; or (b) until an event occurs that would justify a mid-year election change, as described under Section 10.3. Notwithstanding any contrary provision in the Plan, if an Employee who fails to file an Election Form/Salary Reduction Agreement is eligible for benefits under an Insurance Plan and has made an effective election for such benefits outside the Plan, then the Employee's share of the contributions for such benefits will automatically be paid with pre-tax dollars and will be deemed a "default election" under the Plan. Such default elections cannot be changed until such time as the Employee files, during a subsequent Open Enrollment Period (or after an event occurs that would justify a mid-year election change as described under Section 10.3), a timely Election Form/Salary Reduction Form/Salary Reduction Agreement to elect Premium Payment Benefits. No default elections are permitted for Health FSA or DCAP Benefits.

4.5 Irrevocability of Elections

Unless an exception applies (as described in Article X), a Participant's election under the Plan is irrevocable for the duration of the Period of Coverage to which it relates.

ARTICLE V. Benefits Offered and Method of Funding

5.1 Benefits Offered

When first eligible or during the Open Enrollment Period as described under Article IV, Participants will be given the opportunity to elect one or more of the following benefits:

- (a) Premium Payment Benefits, as described in Article VI;
- (b) Health FSA Benefits, as described in Article VII.
- (c) DCAP Benefits, as described in Article VIII.

In no event will benefits under the Plan be provided in the form of deferred compensation. Notwithstanding the foregoing, amounts remaining in a Participant's Health FSA Account at the end of a Plan Year can be used to reimburse the Participant for Medical Care Expenses that are incurred during the Grace Period immediately following the close of that Plan Year as provided in Article VII. No Grace Period is available for DCAP Benefits.

5.2 Source of Benefit Funding

The cost of coverage under the component Benefit Plan Options will be funded by a Participant's Salary Reductions, Nonelective Contributions provided by the Employer, or a combination of the foregoing. The required contributions for each of the Benefit Plan Options offered under the Plan will be made known to employees in annual enrollment materials. Salary Reduction Contributions that are allocated to any Benefit Plan Option will equal the contributions required from the Participant less any available Nonelective Contributions allocated to that option. A Participant may elect to receive Nonelective Contributions in the form of cash to the extent described in the applicable annual enrollment materials. The maximum amount of employee contributions, plus any Nonelective Contributions made available by the Employer, will not exceed the aggregate cost of the Benefit Plan Options elected.

5.3 Employer Contributions

The Employer may, in its sole discretion, make Nonelective Contributions on behalf of a Participant toward the cost of one or more Benefit Plan Options. The amount of Nonelective Contributions that may be applied towards the cost of each of the Benefit Plan Option(s) for any Participant will be subject to the sole discretion of the Employer and may be adjusted upward or downward at any time in the Employer's sole discretion. The amount will be calculated for each Plan Year in a uniform and nondiscriminatory manner and may be based upon the Participant's dependent status, commencement or termination date of the Participant's employment during the Plan Year, and such other factors as the Employer may prescribe.

No provision of this Plan will be construed to require the Employer or Plan Administrator to maintain any fund or segregate any amount for the benefit of any Participant, and no Participant or other person will have any claim against, right to, or security or other interest in, any fund, account or asset of the Employer from which any payment under the Plan may be made. The Plan does not create a trust in favor of a Participant or any person claiming on a Participant's behalf.

ARTICLE VI. Premium Payment Component

6.1 Benefits

An Eligible Employee can elect to participate in the Premium Payment Component by electing (a) to receive benefits under the Insurance Plans described in Appendix A; and (b) to pay for his or her share of the contributions for those benefits on a pre-tax Salary Reduction basis. Unless an exception applies (as described in Article X), such election is irrevocable for the duration of the Period of Coverage to which it relates. Notwithstanding any other provision in this Plan, insurance benefits under the Insurance Plans are subject to the terms and conditions of the Insurance Plans, and no changes can be made with respect to such plans (such as mid-year changes in election) if such changes are not permitted under the applicable Insurance Plan.

6.2 Participant Contributions for Cost of Coverage

The annual contribution for a Participant's portion of the Premium Payment Benefits is equal to the amount as set by the Employer in the annual enrollment materials.

6.3 Benefits Provided Under the Insurance Plans

Insurance benefits will be provided by the Insurance Plans in accordance with their governing documents, and not this Plan. The types and amounts of insurance benefits, the requirements for participating in the Insurance Plans, and the other terms and conditions of coverage and benefits of such plans are set forth in their governing documents. All claims to receive benefits under the Insurance Plans will be subject to and governed by the terms and conditions of the Insurance Plans and the rules, regulations, policies, and procedures adopted in accordance with those plans, as may be amended from time to time.

6.4 Insurance Benefits; COBRA

Notwithstanding any provision to the contrary in this Plan, to the extent required by COBRA, a Participant and his or her Spouse and Dependents, as applicable, whose health coverage terminates under an Insurance Plan because of a COBRA qualifying event (and who is a qualified beneficiary as defined under COBRA), will be given the opportunity to continue on a self-pay basis the same health coverage that he or she had under the applicable Insurance Plan the day before the qualifying event for the periods prescribed by COBRA. Such continuation coverage will be subject to all conditions and limitations under COBRA.

Contributions for COBRA coverage under an Insurance Plan may be paid on a pre-tax basis for current Employees receiving taxable compensation (as may be permitted by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year) where COBRA coverage arises either (a) because the Employee ceases to be eligible because of a reduction in hours; or (b) because the Employee's Dependent ceases to satisfy the eligibility requirements for coverage. For all other individuals (e.g., Employees who cease to be eligible because of retirement, termination of employment, or layoff), contributions for COBRA coverage for Insurance Plan benefits will be paid on an after-tax basis (unless as may be otherwise permitted by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year).

ARTICLE VII. Health FSA Component

7.1 Health FSA Benefits

An Eligible Employee can elect to participate in the Health FSA Component by electing (a) to receive benefits in the form of reimbursements for Medical Care Expenses from the Health FSA (Health FSA Benefits); and (b) to pay his or her contribution for such Health FSA Benefits on a pre-tax Salary Reduction basis. Unless an exception applies (as described in Article X), any such election is irrevocable for the duration of the Period of Coverage to which it relates.

7.2 Participant Contributions for Cost of Coverage of Health FSA Benefits

The annual contribution for a Participant's portion of the Health FSA Benefits is equal to the annual benefit amount elected by the Participant, subject to the dollar limits set forth in the annual enrollment materials.

7.3 Eligible Medical Care Expenses for Health FSA

Under the Health FSA Component, a Participant may receive reimbursement for Medical Care Expenses incurred during the Period of Coverage for which an election is in force. In addition,

certain individuals may receive reimbursement for Medical Care Expenses incurred during the Grace Period immediately following the close of a Plan Year from amounts remaining in their Health FSA Accounts for that Plan Year in accordance with Section 7.4(e).

- (a) *Incurred*. A Medical Care Expense is incurred at the time the medical care or service giving rise to the expense is furnished and not when the Participant is formally billed for, is charged for, or pays for the medical care.
- (b) Medical Care Expenses. "Medical Care Expenses" means expenses incurred by a Participant or his or her Spouse or Dependents for medical care, as defined in Code Section 213(d), but only to the extent that the expense has not been reimbursed through insurance or otherwise. If only a portion of a Medical Care Expense has been reimbursed elsewhere, then the Health FSA can reimburse the remaining portion of such Medical Care Expense if it otherwise meets the requirements of this Article VII. Notwithstanding the foregoing, the term Medical Care Expenses does not include:
 - premium payments for other health coverage, including but not limited to health insurance premiums for any other plan (whether or not sponsored by the Employer);
 - cosmetic surgery or other similar procedures, unless the surgery or procedure is necessary to ameliorate a deformity arising from, or directly related to, a congenital abnormality, a personal injury resulting from an accident or trauma, or a disfiguring disease (for this purpose, "cosmetic surgery" means any procedure that is directed at improving the patient's appearance and does not meaningfully promote the proper function of the body or prevent or treat illness or disease); or
 - any other expense excluded under Appendix B or otherwise under the terms of this Plan.

The Plan Administrator may promulgate procedures regarding the eligibility of various expenses for reimbursement as Medical Care Expenses and may limit reimbursement of expenses described in such procedures.

7.4 Maximum and Minimum Benefits for Health FSA

(a) Maximum Reimbursement Available; Uniform Coverage. The maximum dollar amount elected by the Participant for reimbursement of Medical Care Expenses incurred during a Period of Coverage (reduced by prior reimbursements during the Period of Coverage) will be available at all times during the Period of Coverage, regardless of the actual amounts credited to the Participant's Health FSA Account pursuant to Section 7.5. Notwithstanding the foregoing, no reimbursements will be available for Medical Care Expenses incurred after coverage under this Plan has terminated, unless the Participant has elected COBRA as provided in Section 7.8 or is entitled to submit expenses incurred during a Grace Period as provided in Section 7.4(e). Payment will be made to the Participant in cash as reimbursement for Medical Care Expenses incurred during the Period of Coverage for which the Participant's election is effective (or during a Grace Period, if applicable under Section 7.4(e)), provided that the other requirements of this Article VII have been satisfied. Notwithstanding the foregoing, in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15, a Participant whose participation in the Plan terminates in 20202021 may continue to receive reimbursements from any contributions that remain in his or her Health FSA as of the date of such termination through the end of the 20202021 Plan Year, including any Grace Period.

- (b) Maximum and Minimum Dollar Limits. The maximum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Medical Care Expenses incurred in any Period of Coverage will be set forth in the enrollment materials, but shall be no more than the contribution limit announced by the IRS for the applicable year. The minimum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Medical Care Expenses incurred in any Period of Coverage is \$0. Reimbursements due for Medical Care Expenses incurred by the Participant's Spouse or Dependents will be charged against the Participant's Health FSA Account. Unless otherwise permitted under applicable law, the maximum annual benefit may not exceed the maximum limit provided under such law.
- (c) Changes; No Proration. For each Plan Year, the maximum and minimum dollar limit may be changed by the Plan Administrator and will be communicated to Employees through the Election Form/Salary Reduction Agreement or other enrollment materials. If a Participant enters the Health FSA Component mid-year or wishes to increase his or her election mid-year as permitted under Section 10.3, then there will be no proration rule i.e., the Participant may elect coverage up to the maximum dollar limit or may increase coverage to the maximum dollar limit, as applicable.
- (d) Effect on Maximum Benefits If Election Change Permitted. Any change in an election under Article X (other than under Section 10.3(c) for FMLA leave) that increases contributions to the Health FSA Component also will change the maximum reimbursement benefits for the balance of the Period of Coverage commencing with the election change. Such maximum reimbursement benefits for the balance of the Period of Coverage will be calculated by adding (1) the contributions (if any) made by the Participant as of the end of the portion of the Period of Coverage immediately preceding the change in election, to (2) the total contributions scheduled to be made by the Participant during the remainder of such Period of Coverage to the Health FSA Account, reduced by (3) all reimbursements made during the entire Period of Coverage. Any change in an election under Section 10.3(c) for FMLA leave will change the maximum reimbursement benefits in accordance with the regulations governing the effect of the FMLA on the operation of cafeteria plans.
- (e) Grace Periods; Special Rules for Claims Incurred During a Grace Period. Notwithstanding any contrary provision in this Plan and subject to the conditions of this Section 7.4(e), an individual may be reimbursed for Medical Care Expenses incurred during a Grace Period from amounts remaining in his or her Health FSA Account at the end of the Plan Year to which that Grace Period relates ("Prior Plan Year Health FSA Amounts") if he or she is either: (1) a Participant with Health FSA coverage that is in effect on the last day of that Plan Year; or (2) a qualified beneficiary (as defined under COBRA) who has COBRA coverage under the Health FSA Component on the last day of that Plan Year.
 - Prior Plan Year Health FSA Amounts may not be cashed out or converted to any other taxable or non-taxable benefit. For example, Prior Plan Year Health FSA Amounts may not be used to reimburse Dependent Care Expenses.
 - Medical Care Expenses incurred during a Grace Period and approved for reimbursement in accordance with Section 7.7 will be reimbursed first from any available Prior Plan Year Health FSA Amounts and then from any amounts that are available to reimburse expenses that are incurred during the current Plan Year, except that if the Health FSA is accessible by an electronic payment card

(e.g., debit card, credit card, or similar arrangement), Medical Care Expenses incurred during the Grace Period may need to be submitted manually in order to be reimbursed from Prior Plan Year Health FSA Amounts if the card is unavailable for such reimbursement. An individual's Prior Plan Year Health FSA Amounts will be debited for any reimbursement of Medical Care Expenses incurred during the Grace Period that is made from such Prior Plan Year Health FSA Amounts.

Claims for reimbursement of Medical Care Expenses incurred during a Grace Period must be submitted no later than the April 30 following the close of the Plan Year to which the Grace Period relates in order to be reimbursed from Prior Plan Year Health FSA Amounts. For purposes of the Grace Period: (i) for amounts relating to the 2020 Plan Year, claims for reimbursement incurred during the Grace Period from January 1, 2021 through December 31, 2021, must be submitted no later than April 30, 2022-; and (ii) for purposes of amounts relating to the 2021 Plan Year, claims for reimbursement incurred during the Grace Period from January 1, 2022 through December 31, 2022, must be submitted no later than April 30, 2023. Any Prior Plan Year Health FSA Amounts that remain after all reimbursements have been made for the Plan Year and its related Grace Period will not be carried over to reimburse the Participant for expenses incurred in any subsequent period. The Participant will forfeit all rights with respect to these amounts, which will be subject to the Plan's provisions regarding forfeitures in Section 7.6(b).

7.5 Establishment of Health FSA Account

The Plan Administrator will establish and maintain a Health FSA Account with respect to each Participant for each Plan Year or other Period of Coverage for which the Participant elects to participate in the Health FSA Component, but it will not create a separate fund or otherwise segregate assets for this purpose. The Account so established will merely be a recordkeeping account with the purpose of keeping track of contributions and determining forfeitures under Section 7.6.

- (a) *Crediting of Accounts*. A Participant's Health FSA Account for a Plan Year or other Period of Coverage will be credited periodically during such period with an amount equal to the Participant's Salary Reductions elected to be allocated to such Account.
- (b) Debiting of Accounts. A Participant's Health FSA Account for a Plan Year or other Period of Coverage will be debited for any reimbursement of Medical Care Expenses incurred during such period (or for reimbursement of Medical Care Expenses incurred during any Grace Period to which he or she is entitled as provided in Section 7.4(e)).
- (c) Available Amount Not Based on Credited Amount. As described in Section 7.4, the amount available for reimbursement of Medical Care Expenses is the Participant's annual benefit amount, reduced by prior reimbursements for Medical Care Expenses incurred during the Plan Year or other Period of Coverage (or during the Grace Period, if applicable); it is not based on the amount credited to the Health FSA Account at a particular point in time. Thus, a Participant's Health FSA Account may have a negative balance during a Plan Year or other Period of Coverage, but the aggregate amount of reimbursement will in no event exceed the maximum dollar amount elected by the Participant under this Plan.

7.6 Forfeiture of Health FSA Accounts; Use-or-Lose Rule

- (a) Use-or-Lose Rule. Except as otherwise provided in Section 7.4(e) (regarding certain individuals who may be reimbursed from Prior Plan Year Health FSA Amounts for expenses incurred during a Grace Period), if any balance remains in the Participant's Health FSA Account for a Period of Coverage after all reimbursements have been made for the Period of Coverage, then such balance will not be carried over to reimburse the Participant for Medical Care Expenses incurred during a subsequent Plan Year. The Participant will forfeit all rights with respect to such balance.
- (b) Use of Forfeitures. All forfeitures under this Plan will be used as follows: (1) first, to offset any losses experienced by the Employer during the Plan Year as a result of making reimbursements (i.e., providing Health FSA Benefits) with respect to all Participants in excess of the contributions paid by such Participants through Salary Reductions; (2) second, to reduce the cost of administering the Health FSA Component during the Plan Year or the subsequent Plan Year (all such administrative costs will be documented by the Plan Administrator); and (3) third, to provide increased benefits or compensation to Participants in subsequent years in any weighted or uniform fashion that the Plan Administrator deems appropriate, consistent with applicable regulations. In addition, any Health FSA Account benefit payments that are unclaimed (e.g., uncashed benefit checks) by the close of the Plan Year following the Period of Coverage in which the Medical Care Expense was incurred will be forfeited and applied as described above.

7.7 Reimbursement Claims Procedure for Health FSA

- (a) *Timing.* Within 30 days after receipt by the Plan Administrator of a reimbursement claim from a Participant, the Employer will reimburse the Participant for the Participant's Medical Care Expenses (if the Plan Administrator approves the claim), or the Plan Administrator will notify the Participant that his or her claim has been denied. This time period may be extended by an additional 15 days for matters beyond the control of the Plan Administrator, including in cases where a reimbursement claim is incomplete. The Plan Administrator will provide written notice of any extension, including the reasons for the extension, and will allow the Participant 45 days in which to complete the previously incomplete reimbursement claim.
- (b) Claims Substantiation. A Participant who has elected to receive Health FSA Benefits for a Period of Coverage may apply for reimbursement by submitting a request in writing to the Plan Administrator in such form as the Plan Administrator may prescribe, by no later than the April 30 following the close of the Plan Year in which the Medical Care Expense was incurred (except that for a Participant who ceases to be eligible to participate, this must be done no later than 90 days after the date that eligibility ceases, as described in Section 7.8) setting forth:
 - the person(s) on whose behalf Medical Care Expenses have been incurred;
 - the nature and date of the expenses so incurred;
 - the amount of the requested reimbursement;
 - a statement that such expenses have not otherwise been reimbursed and that the Participant will not seek reimbursement through any other source; and
 - other such details about the expenses that may be requested by the Plan Administrator in the reimbursement request form or otherwise (e.g., a statement from a medical practitioner that the expense is to treat a specific medical

condition, documentation that a medicine or drug was prescribed, or a more detailed certification from the Participant).

The application must be accompanied by bills, invoices, or other statements from an independent third party showing that the Medical Care Expenses have been incurred and showing the amounts of such expenses, along with any additional documentation that the Plan Administrator may request. Except for the final reimbursement claim for a Participant's Health FSA Account for a Plan Year or other Period of Coverage, no claim for reimbursement may be made unless and until the aggregate claim for reimbursement is at least \$25. If the Health FSA is accessible by an electronic payment card (e.g., debit card, credit card, or similar arrangement), the Participant will be required to comply with substantiation procedures established by the Plan Administrator in accordance with Rev. Rul. 2003-43, IRS Notice 2006-69, or other IRS guidance.

- (c) *Claims Denied*. For reimbursement claims that are denied, see the appeals procedure in Article XI.
- (d) Claims Ordering; No Reprocessing. All claims for reimbursement under the Health FSA Component will be paid in the order in which they are approved. Once paid, a claim will not be reprocessed or otherwise recharacterized solely for the purpose of paying it (or treating it as paid) from amounts attributable to a different Plan Year or Period of Coverage.

7.8 Reimbursements From Health FSA After Termination of Participation; COBRA

When a Participant ceases to be a Participant under Section 3.3, the Participant's Salary Reductions and election to participate will terminate. Except as otherwise provided in Section 7.4(e) (regarding certain individuals who may be reimbursed from Prior Plan Year Health FSA Amounts for expenses incurred during a Grace Period), the Participant will not be able to receive reimbursements for Medical Care Expenses incurred after the end of the day on which the Participant's employment terminates or the Participant otherwise ceases to be eligible. However, such Participant (or the Participant's estate) may claim reimbursement for any Medical Care Expenses incurred during the Period of Coverage prior to the date that the Participant ceases to be eligible (or during any Grace Period to which he or she is entitled as provided in Section 7.4(e)), provided that the Participant (or the Participant's estate) files a claim within 90 days after the date that the Participant ceases to be a Participant.

Notwithstanding any provision to the contrary in this Plan, to the extent required by COBRA, a Participant and his or her Spouse and Dependents, as applicable, whose coverage terminates under the Health FSA Component because of a COBRA qualifying event (and who is a qualified beneficiary as defined under COBRA) will be given the opportunity to continue on a self-pay basis the same coverage that he or she had under the Health FSA Component the day before the qualifying event for the periods prescribed by COBRA. Specifically, such individuals will be eligible for COBRA continuation coverage only if, under Section 7.5, they have a positive Health FSA Account balance at the time of a COBRA qualifying event (taking into account all claims submitted before the date of the qualifying event). Such individuals will be notified if they are eligible for COBRA continuation coverage. If COBRA is elected, it will be available only for the remainder of the Plan Year in which the qualifying event occurs; such COBRA coverage for the Health FSA Component will cease at the end of the Plan Year and cannot be continued for the next Plan Year. Such continuation coverage will be subject to all conditions and limitations under COBRA. Notwithstanding the foregoing, a qualified beneficiary (as defined under COBRA) who has COBRA coverage under the Health FSA Component on the last day of a Plan

Year may be entitled to reimbursement of Medical Care Expenses incurred during the Grace Period following that Plan Year in accordance with the provisions of Section 7.4(e).

Contributions for coverage for Health FSA Benefits may be paid on a pre-tax basis for current Employees receiving taxable compensation (as may be permitted by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year) where COBRA coverage arises either (a) because the Employee ceases to be eligible because of a reduction of hours or (b) because the Employee's Dependent ceases to satisfy the eligibility requirements for coverage. For all other individuals (e.g., Employees who cease to be eligible because of retirement, termination of employment, or layoff), contributions for COBRA coverage for Health FSA Benefits must be paid on an after-tax basis (unless permitted otherwise by the Plan Administrator on a uniform and consistent basis, but may not be prepaid from contributions in one Plan Year to provide coverage that extends into a subsequent Plan Year).

Notwithstanding the foregoing, a Participant whose participation in the Plan terminates in 20202021 may continue to receive reimbursements from any contributions that remain in his or her Health FSA as of the date of such termination through the end of the 20202021 Plan Year and, including any related Grace Period.

7.9 Coordination of Benefits

Health FSA Benefits are intended to pay benefits solely for Medical Care Expenses for which Participants have not been previously reimbursed and will not seek reimbursement elsewhere. Accordingly, the Health FSA will not be considered to be a group health plan for coordination of benefits purposes, and Health FSA Benefits will not be taken into account when determining benefits payable under any other plan.

ARTICLE VIII. DCAP Component

8.1 DCAP Benefits

An Eligible Employee can elect to participate in the DCAP Component by electing (a) to receive benefits in the form of reimbursements for Dependent Care Expenses from the DCAP Component (DCAP Benefits), and (b) to pay his or her contribution for such DCAP Benefits on a pre-tax Salary Reduction basis. Unless an exception applies (as described in Article X), such election is irrevocable for the duration of the Period of Coverage to which it relates.

8.2 Participant Contributions for Cost of Coverage for DCAP Benefits

The annual Contribution for a Participant's portion of the DCAP Benefits is equal to the annual benefit amount elected by the Participant, subject to the dollar limits set forth in Section 8.4(b). (For example, if the maximum \$5,000 annual benefit amount is elected, then the annual contribution amount is also \$5,000.)

8.3 Eligible Dependent Care Expenses

Under the DCAP Component, a Participant may receive reimbursement for Dependent Care Expenses incurred during the Period of Coverage for which an election is in force.

- (a) *Incurred.* A Dependent Care Expense is incurred at the time the Qualifying Dependent Care Services giving rise to the expense is furnished, not when the Participant is formally billed for, is charged for, or pays for the Qualifying Dependent Care Services (e.g., services rendered for the month of June are not fully incurred until June 30 and cannot be reimbursed in full until then).
- (b) Dependent Care Expenses. "Dependent Care Expenses" are expenses that are considered to be employment-related expenses under Code Section 21(b)(2) (relating to expenses for the care of a Qualifying Individual necessary for gainful employment of the Employee and Spouse, if any, and expenses for incidental household services), if paid for by the Eligible Employee to obtain Qualifying Dependent Care Services; provided, however, that this term will not include any expenses for which the Participant or other person incurring the expense is reimbursed for the expense through insurance or any other plan. If only a portion of a Dependent Care Expense has been reimbursed elsewhere (e.g., because the Spouse's DCAP imposes maximum benefit limitations), the DCAP can reimburse the remaining portion of such Expense if it otherwise meets the requirements of this Article VIII.
- (c) *Qualifying Dependent Care Services.* "Qualifying Dependent Care Services" means services that: (1) relate to the care of a Qualifying Individual that enable the Participant and his or her Spouse to remain gainfully employed after the date of participation in the DCAP Component and during the Period of Coverage; and (2) are performed—
 - in the Participant's home; or
 - outside the Participant's home for (1) the care of a Participant's qualifying child who is under age 13; or (2) the care of any other Qualifying Individual who regularly spends at least eight hours per day in the Participant's household. In addition, if the expenses are incurred for services provided by a dependent care center (i.e., a facility (including a day camp) that provides care for more than six individuals (other than individuals residing at the facility) on a regular basis and receives a fee, payment, or grant for such services), then the center must comply with all applicable state and local laws and regulations.
- (d) *Exclusion*. Dependent Care Expenses do not include amounts paid to:
 - an individual with respect to whom a personal exemption is allowable under Code Section 151(c) to a Participant or his or her Spouse;
 - a Participant's Spouse;
 - a Participant's child (as defined in Code Section 152(f)(1)) who is under 19 years of age at the end of the year in which the expenses were incurred; or
 - a parent of a Participant's under age 13 qualifying child as defined in Code Section 152(a)(1) (e.g., a former spouse who is the child's noncustodial parent).

8.4 Maximum and Minimum Benefits for DCAP

(a) Maximum Reimbursement Available. The maximum dollar amount elected by the Participant for reimbursement of Dependent Care Expenses incurred during a Period of Coverage (reduced by prior reimbursements during the Period of Coverage) and any balance covered by a Grace Period will only be available during the Period of Coverage to the extent of the actual amounts credited to the Participant's DCAP Account pursuant to Section 8.5. (No reimbursement will be made to the extent that such reimbursement would exceed the balance in the Participant's Account (that is, the year-to-date amount
that has been withheld from the Participant's Compensation for reimbursement for Dependent Care Expenses for the Period of Coverage, less any prior reimbursements). Payment will be made to the Participant in cash as reimbursement for Dependent Care Expenses incurred during the Period of Coverage for which the Participant's election is effective, provided that the other requirements of this Article VIII have been satisfied.

- (b) Maximum and Minimum Dollar Limits. The maximum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Dependent Care Expenses incurred in any Period of Coverage is \$5,000 or, if lower, the maximum amount that the Participant has reason to believe will be excludable from his or her income at the time the election is made as a result of the applicable statutory limit for the Participant. The applicable statutory limit for a Participant is the smallest of the following amounts:
 - the Participant's Earned Income for the calendar year;
 - the Earned Income of the Participant's Spouse for the calendar year (for this purpose, a Spouse who is not employed during a month in which the Participant incurs a Dependent Care Expense and is either (1) physically or mentally incapable of self-care, or (2) a Student will be deemed to have Earned Income in the amount specified in Code Section 21(d)(2)); or
 - either \$5,000 or \$2,500 for the calendar year, as applicable below:

(1) The amount is \$5,000 for the calendar year if one of the following applies: (a) the Participant is married and files a joint federal income tax return; (b) the Participant is married, files a separate federal income tax return, and meets the following conditions: (i) the Participant maintains as his or her home a household that constitutes (for more than half of the taxable year) the principal abode of a Qualifying Individual (i.e., the Dependent for whom the Participant is eligible to receive reimbursements under the DCAP); (ii) the Participant furnishes over half of the cost of maintaining such household during the taxable year; and (iii) during the last six months of the taxable year, the Participant's Spouse is not a member of such household (i.e., the Spouse maintained a separate residence); or (c) the Participant is single or is the head of the household for federal income tax purposes.

(2) The amounts is \$2,500 for the calendar year if the Participant is married and resides with the Spouse, but files a separate federal income tax return.

The minimum annual benefit amount that a Participant may elect to receive under this Plan in the form of reimbursements for Dependent Care Expenses incurred in any Period of Coverage is \$0. (Note: The maximum annual benefit amount that a Participant may elect for the 2021 Plan year will not be affected by the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period from any amount that remains in his or her 2020 DCAP Account as of December 31, 2020 and the amount elected by the Participant for the 2021 Plan Year. In addition, the maximum annual benefit amount that a Participant may elect for the 20212022 Plan year will not be affected by any balance available to such Participant during the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period provided for in Section 8.4(c). A Participant may receive reimbursement during the Grace Period from any amount that remains in his or her 20202021 DCAP Account as of December 31, 20202021 and the amount elected by the Participant for the 20212022 Plan Year.)

- (c) Grace Period for 2020 Amounts; Special Rules for Claims Incurred During a Grace Period. Notwithstanding any contrary provision in this Plan and subject to the conditions of this Section 8.4(c), if an individual has amounts remaining in his or her DCAP Account as of December 31, 2020, such individual may be reimbursed for Dependent Care Expenses from such amount during the Grace Period from January 1, 2021 through December 31, 2021.
 - Amounts remaining in a DCAP Account as of December 31, 2020 may not be cashed out or converted to any other taxable or non-taxable benefit.
 - Dependent Care Expenses incurred during the Grace Period and approved for reimbursement in accordance will be reimbursed first from the 2020 DCAP Account amount and then from any amounts that are available to reimburse expenses that are incurred during the current Plan Year, except that if the DCAP Account is accessible by an electronic payment card (e.g., debit card, credit card, or similar arrangement), Dependent Care Expenses incurred during the Grace Period may need to be submitted manually in order to be reimbursed from 2020 DCAP Account amount if the card is unavailable for such reimbursement. An individual's 2020 DCAP Account amount will be debited for any reimbursement of Dependent Care Expenses incurred during the Grace Period that is made from such 2020 DCAP Account amount.
 - For purposes of the Grace Period for amounts relating to the 2020 Plan Year, claims for reimbursement incurred during the Grace Period from January 1, 2021 through December 31, 2021, must be submitted no later than April 30, 2022. Any Prior Plan Year Health FSA Amounts that remain after all reimbursements have been made for the Plan Year and its related Grace Period will not be carried over to reimburse the Participant for expenses incurred in any subsequent period. The Participant will forfeit all rights with respect to these amounts, which will be subject to the Plan's provisions regarding forfeitures in Section 8.6.
- (d) Grace Period for 2021 Amounts; Special Rules for Claims Incurred During a Grace Period. Notwithstanding any contrary provision in this Plan and subject to the conditions of this Section 8.4(d), if an individual has amounts remaining in his or her DCAP Account as of December 31, 2021, such individual may be reimbursed for Dependent Care Expenses from such amount during the Grace Period from January 1, 2022 through December 31, 2022.
 - Amounts remaining in a DCAP Account as of December 31, 2021 may not be cashed out or converted to any other taxable or non-taxable benefit.
 - Dependent Care Expenses incurred during the Grace Period and approved for reimbursement in accordance will be reimbursed first from the 2021 DCAP Account amount and then from any amounts that are available to reimburse expenses that are incurred during the current Plan Year, except that if the DCAP Account is accessible by an electronic payment card (e.g., debit card, credit card, or similar arrangement), Dependent Care Expenses incurred during the Grace Period may need to be submitted manually in order to be reimbursed from 2021 DCAP Account amount if the card is unavailable for such reimbursement. An individual's 2021 DCAP Account amount will be debited for any reimbursement of Dependent Care Expenses incurred during the Grace Period that is made from such 2021 DCAP Account amount.

- For purposes of the Grace Period for amounts relating to the 2021 Plan Year, claims for reimbursement incurred during the Grace Period from January 1, 2022 through December 31, 2022, must be submitted no later than April 30, 2023. Any Prior Plan Year Health FSA Amounts that remain after all reimbursements have been made for the Plan Year and its related Grace Period will not be carried over to reimburse the Participant for expenses incurred in any subsequent period. The Participant will forfeit all rights with respect to these amounts, which will be subject to the Plan's provisions regarding forfeitures in Section 8.6.
- (e) *Changes; No Proration.* For subsequent Plan Years, the maximum and minimum dollar limit may be changed by the Plan Administrator and will be communicated to Employees through the Election Form/Salary Reduction Agreement or other enrollment materials. If a Participant enters the DCAP Component mid-year or wishes to increase his or her election mid-year as permitted under Section 10.3, then there will be no proration rule i.e., the Participant may elect coverage up to the maximum dollar limit or may increase coverage up to the maximum dollar limit, as applicable.
- (f) Effect on Maximum Benefits If Election Change Permitted. Any change in an election under Article X affecting annual contributions to the DCAP Component also will change the maximum reimbursement benefits for the balance of the Period of Coverage (commencing with the election change), as further limited by Sections 8.4(a) and (b). Such maximum reimbursement benefits for the balance of the Period of Coverage will be calculated by adding (1) the contributions, if any, made by the Participant as of the end of the portion of the Period of Coverage immediately preceding the change in election, to (2) the total contributions scheduled to be made by the Participant during the remainder of such Period of Coverage to the DCAP Account, reduced by (3) reimbursements during the Period of Coverage.

8.5 Establishment of DCAP Account

The Plan Administrator will establish and maintain a DCAP Account with respect to each Participant who has elected to participate in the DCAP Component, but it will not create a separate fund or otherwise segregate assets for this purpose. The Account so established will merely be a recordkeeping account with the purpose of keeping track of contributions and determining forfeitures under Section 8.6.

- (a) *Crediting of Accounts*. A Participant's DCAP Account will be credited periodically during each Period of Coverage with an amount equal to the Participant's Salary Reductions elected to be allocated to such Account.
- (b) *Debiting of Accounts*. A Participant's DCAP Account will be debited during each Period of Coverage for any reimbursement of Dependent Care Expenses incurred during the Period of Coverage.
- (c) Available Amount Is Based on Credited Amount. As described in Section 8.4, the amount available for reimbursement of Dependent Care Expenses may not exceed the year-to-date amount credited to the Participant's DCAP Account, less any prior reimbursements (i.e., it is based on the amount credited to the DCAP Account at a particular point in time). Thus, a Participant's DCAP Account may not have a negative balance during a Period of Coverage.

8.6 Forfeiture of DCAP Accounts; Use-It-or-Lose-It Rule

If any balance remains in the Participant's DCAP Account for a Period of Coverage after all reimbursements have been made for the Period of Coverage, then such balance will not be carried over to reimburse the Participant for Dependent Care Expenses incurred during a subsequent Plan Year. The Participant will forfeit all rights with respect to such balance. All forfeitures under this Plan will be used as follows: (1) first, to offset any losses experienced by the Employer during the Plan Year as a result of making reimbursements (i.e., providing DCAP Benefits) with respect to all Participants in excess of the contributions paid by such Participants through Salary Reductions; (2) second, to reduce the cost of administering the DCAP during the Plan Year or the subsequent Plan Year (all such administrative costs will be documented by the Plan Administrator); and (3) third, to provide increased benefits or compensation to Participants in subsequent years in any weighted or uniform fashion the Plan Administrator deems appropriate, consistent with applicable regulations. In addition, any DCAP Account benefit payments that are unclaimed (e.g., uncashed benefit checks) by the close of the Plan Year following the Period of Coverage in which the Dependent Care Expense was incurred will be forfeited and applied as described above. Notwithstanding, the foregoing, including but not limited to the overall maximum provided in Section 8.4(b), in response to the continuing COVID-19 pandemic and in accordance with the Consolidated Appropriations Act, 2021 and IRS Notice 2021-15, if a Participant has a balance in his or her DCAP Account as of December 31, 2020 and has a Qualifying Individual who attains the age of 13 either during the 2020 Plan Year, or during the 2021 Plan Year, such balance shall be carried over to the 2021 Plan Year. Such Participant may receive reimbursement for Dependent Care Expenses incurred during the 2021 Plan Year and submitted for reimbursement no later than March 31, 2022 from the balance carried over from the 2020 Plan Year for such Child that attained age 13 in the 2020 Plan Year or in the 2021 Plan Year only and not from any DCAP election for the 2021 Plan Year.

8.7 Reimbursement Claims Procedure for DCAP

- (a) Timing. Within 30 days after receipt by the Plan Administrator of a reimbursement claim from a Participant, the Employer will reimburse the Participant for the Participant's Dependent Care Expenses (if the Plan Administrator approves the claim), or the Plan Administrator will notify the Participant that his or her claim has been denied. This time period may be extended by an additional 15 days for matters beyond the control of the Plan Administrator, including in cases where a reimbursement claim is incomplete. The Plan Administrator will provide written notice of any extension, including the reasons for the extension, and will allow the Participant 45 days in which to complete the previously incomplete reimbursement claim.
- (b) Claims Substantiation. A Participant who has elected to receive DCAP Benefits for a Period of Coverage may apply for reimbursement by submitting a request for reimbursement in writing to the Plan Administrator in such form as the Plan Administrator may prescribe, by no later than the April 30 following the close of the Plan Year in which the Dependent Care Expense was incurred (except for a Participant who ceases to be eligible to participate, by no later than 90 days after the date that eligibility ceases, as described in Section 8.8), setting forth:
 - the person(s) on whose behalf Dependent Care Expenses have been incurred;
 - the nature and date of the expenses so incurred;
 - the amount of the requested reimbursement;

- the name of the person, organization or entity to whom the expense was or is to be paid, and taxpayer identification number (Social Security number, if the recipient is a person);
- a statement that such expenses have not otherwise been reimbursed and that the Participant will not seek reimbursement through any other source;
- the Participant's certification that he or she has no reason to believe that the reimbursement requested, added to his or her other reimbursements to date for Dependent Care Expenses incurred during the same calendar year, will exceed the applicable statutory limit for the Participant as described in Section 8.4(b); and
- other such details about the expenses that may be requested by the Plan Administrator in the reimbursement request form or otherwise (e.g., a more detailed certification from the Participant).

The application will be accompanied by bills, invoices, or other statements from an independent third party showing that the Dependent Care Expenses have been incurred and showing the amounts of such expenses, along with any additional documentation that the Plan Administrator may request. Except for the final reimbursement claim for a Period of Coverage, no claim for reimbursement may be made unless and until the aggregate claim for reimbursement is at least \$25.

(c) *Claims Denied*. For reimbursement claims that are denied, see the appeals procedure in Article XI.

8.8 Reimbursements From DCAP After Termination of Participation

When a Participant ceases to be a Participant under Section 3.3, the Participant's Salary Reductions and election to participate will terminate. The Participant will not be able to receive reimbursements for Dependent Care Expenses incurred after the end of the day on which the Participant's employment terminates or the Participant otherwise ceases to be eligible, with one exception: such Participant (or the Participant's estate) may claim reimbursement for any Dependent Care Expenses incurred in the month that includes the date the Participant terminates employment or otherwise loses eligibility, provided that the Participant (or the Participant's estate) files a claim within 90 days after the date that the Participant's employment terminates or the Participant otherwise ceases to be eligible. Notwithstanding the foregoing, a Participant whose participation in the Plan terminates in 20202021 may continue to receive reimbursements from any amount that remains in his or her DCAP Account as of the date of such termination through the end of the 20202021 Plan Year and any related Grace Period.

ARTICLE IX. HIPAA PROVISIONS FOR HEALTH FSA

9.1 Provision of Protected Health Information to Employer

Members of the Employer's workforce have access to the individually identifiable health information of Plan participants for administrative functions of the Health FSA. When this health information is provided from the Health FSA to the Employer, it is Protected Health Information (PHI). The Health Insurance Portability and Accountability Act of 1996 (HIPAA) and its implementing regulations restrict the Employer's ability to use and disclose PHI. The following HIPAA definition of PHI applies for purposes of this Article IX:

Protected Health Information. Protected health information means information that is created or received by the Plan and relates to the past, present, or future physical or mental health or condition of a participant; the provision of health care to a participant; or the past, present, or future payment for the provision of health care to a participant; and that identifies the participant or for which there is a reasonable basis to believe the information can be used to identify the participant. Protected health information includes information of persons living or deceased.

The Employer will have access to PHI from the Health FSA only as permitted under this Article IX or as otherwise required or permitted by HIPAA. HIPAA and its implementing regulations were modified by the Health Information Technology for Economic and Clinical Health Act (HITECH Act), the statutory provisions of which are incorporated herein by reference.

9.2 Permitted Disclosure of Enrollment/Disenrollment Information

The Health FSA may disclose to the Employer information on whether the individual is participating in the Plan.

9.3 Permitted Uses and Disclosure of Summary Health Information

The Health FSA may disclose Summary Health Information to the Employer, provided that the Employer requests the Summary Health Information for the purpose of modifying, amending, or terminating the Health FSA.

"Summary Health Information" means information (a) that summarizes the claims history, claims expenses, or type of claims experienced by individuals for whom a plan sponsor had provided health benefits under a health plan; and (b) from which the information described at 42 CFR Section 164.514(b)(2)(i) has been deleted, except that the geographic information described in 42 CFR Section 164.514(b)(2)(i)(B) need only be aggregated to the level of a five-digit ZIP code.

9.4 Permitted and Required Uses and Disclosure of PHI for Plan Administration Purposes

Unless otherwise permitted by law, and subject to the conditions of disclosure described in Section 9.5 and obtaining written certification pursuant to Section 9.7, the Health FSA may disclose PHI to the Employer, provided that the Employer uses or discloses such PHI only for Plan administration purposes. "Plan administration purposes" means administration functions performed by the Employer on behalf of the Health FSA, such as quality assurance, claims processing, auditing, and monitoring. Plan administration functions do not include functions performed by the Employer in connection with any other benefit or benefit plan of the Employer, and they do not include any employment-related functions.

Notwithstanding the provisions of this Plan to the contrary, in no event will the Employer be permitted to use or disclose PHI in a manner that is inconsistent with 45 CFR Section 164.504(f).

9.5 Conditions of Disclosure for Plan Administration Purposes

The Employer agrees that with respect to any PHI (other than enrollment/disenrollment information and Summary Health Information, which are not subject to these restrictions) disclosed to it by the Health FSA, the Employer will:

- not use or further disclose the PHI other than as permitted or required by the Health FSA or as required by law;
- ensure that any agent, including a subcontractor, to whom it provides PHI received from the Health FSA agrees to the same restrictions and conditions that apply to the Employer with respect to PHI;
- not use or disclose the PHI for employment-related actions and decisions or in connection with any other benefit or employee benefit plan of the Employer;
- report to the Plan any use or disclosure of the information that is inconsistent with the uses or disclosures provided for of which it becomes aware;
- make available PHI to comply with HIPAA's right to access in accordance with 45 CFR Section 164.524;
- make available PHI for amendment and incorporate any amendments to PHI in accordance with 45 CFR Section 164.526;
- make available the information required to provide an accounting of disclosures in accordance with 45 CFR Section 164.528;
- make its internal practices, books, and records relating to the use and disclosure of PHI received from the Health FSA available to the Secretary of Health and Human Services for purposes of determining compliance by the Health FSA with HIPAA's privacy requirements;
- if feasible, return or destroy all PHI received from the Health FSA that the Employer still maintains in any form and retain no copies of such information when no longer needed for the purpose for which disclosure was made, except that, if such return or destruction is not feasible, limit further uses and disclosures to those purposes that make the return or destruction of the information infeasible; and
- ensure that the adequate separation between the Health FSA and the Employer (i.e., the "firewall"), required in 45 CFR Section 504(f)(2)(iii) is satisfied.

The Employer further agrees that if it creates, receives, maintains, or transmits any electronic PHI (other than enrollment/disenrollment information and Summary Health Information, which are not subject to these restrictions) on behalf of the Health FSA, it will implement administrative, physical, and technical safeguards that reasonably and appropriately protect the confidentiality, integrity, and availability of the electronic PHI, and it will ensure that any agents (including subcontractors) to whom it provides such electronic PHI agree to implement reasonable and appropriate security measures to protect the information. The Employer will report to the Health FSA any security incident of which it becomes aware.

9.6 Adequate Separation Between Plan and Employer

The Employer will allow the following persons access to PHI: Director of Finance, Accounting Manager, Senior Accountant, Accountant, the Plan Administrator, and payroll staff performing Health FSA functions and any other Employee who needs access to PHI in order to perform Plan administration functions that the Employer performs for the Health FSA (such as quality assurance, claims processing, auditing, monitoring, payroll, and appeals). No other persons will have access to PHI. These specified employees (or classes of employees) will only have access to and use PHI to the extent necessary to perform the plan administration functions that

the Employer performs for the Health FSA. In the event that any of these specified employees does not comply with the provisions of this Section, that employee will be subject to disciplinary action by the Employer for non-compliance pursuant to the Employer's employee discipline and termination procedures.

The Employer will ensure that the provisions of this Section 9.6 are supported by reasonable and appropriate security measures to the extent that the designees have access to electronic PHI.

9.7 Certification of Plan Sponsor

The Health FSA will disclose PHI to the Employer only upon the receipt of a certification by the Employer that the Health FSA incorporates the provisions of 45 CFR Section 164.504(f)(2)(ii), and that the Employer agrees to the conditions of disclosure set forth in Section 9.5. Execution of the Plan by the Employer will serve as the required certification.

9.8 Privacy Official

The Employer will designate a Privacy Official, who will be responsible for the Plan's compliance with HIPAA. The Privacy Official may contract with or otherwise utilize the services of attorneys, accountants, brokers, consultants, or other third party experts as the Privacy Official deems necessary or advisable. In addition and notwithstanding any provision of this Plan to the contrary, the Privacy Official will have the authority to and be responsible for:

- accepting and verifying the accuracy and completeness of any certification provided by the Employer under this Article;
- transmitting the certification to any third parties as may be necessary to permit them to disclose PHI to the Employer;
- establishing and implementing policies and procedures with respect to PHI that are designed to ensure compliance by the Plan with the requirements of HIPAA;
- establishing and overseeing proper training of personnel who will have access to PHI; and
- any other duty or responsibility that the Privacy Official, in his or her sole capacity, deems necessary or appropriate to comply with the provisions of HIPAA and the purposes of the Article IX.

9.9 Interpretation and Limited Applicability

This Article serves the sole purpose of complying with the requirements of HIPAA and will be interpreted and construed in a manner to effectuate this purpose. Neither this Article IX nor the duties, powers, responsibilities, and obligations listed herein will be taken into account in determining the amount or nature of the benefits provided to any person covered under the Health FSA Component, nor will they inure to the benefit of any third parties. To the extent that any of the provisions of this Article IX are no longer required by HIPAA or do not apply to the Plan because the Plan is otherwise excepted from HIPAA, they will be deemed deleted and will have no force or effect.

9.10 Service Performed for the Employer

Notwithstanding any other provisions of this Plan to the contrary, all services performed by a business associate for the Health FSA in accordance with the applicable service agreement will be deemed to be performed on behalf of the Health FSA and subject to the administrative simplification provisions of HIPAA contained in 45 C.F.R. Parts 160 through 164, except services that relate to eligibility and enrollment in the Health FSA. If a business associate of the Health FSA performs any services that relate to eligibility and enrollment in the Health FSA, these services will be deemed to be performed on behalf of the Employer in its capacity as Plan Sponsor and not on behalf of the Health FSA.

ARTICLE X. Irrevocability of Elections; Exceptions

10.1 Irrevocability of Elections

Except as described in this Article X, a Participant's election under the Plan is irrevocable for the duration of the Period of Coverage to which it relates. In other words, unless an exception applies, the Participant may not change any elections for the duration of the Period of Coverage regarding:

- (a) participation in this Plan;
- (b) Salary Reduction amounts; or
- (c) election of particular Benefit Plan Options.

10.2 Procedure for Making New Election If Exception to Irrevocability Applies

- (a) Timeframe for Making New Election. A Participant (or an Eligible Employee who, when first eligible under Section 4.2 or during the Open Enrollment Period under Section 4.3, declined to be a Participant) may make a new election within 30 days of the occurrence of an event described in Section 10.3 (or within 60 days of the occurrence of an event described in Section 10.3(e)(3) or (4)), as applicable, but only if the election under the new Election Form/Salary Reduction Agreement is made on account of and is consistent with the event. Notwithstanding the foregoing, a Change in Status (e.g., a divorce or a dependent's losing student status) that results in a beneficiary becoming ineligible for coverage under the Insurance Plans will automatically result in a corresponding election change, whether or not requested by the Participant within the normal 30-day period.
- (b) Effective Date of New Election. Elections made pursuant to this Section 10.2 will be effective for the balance of the Period of Coverage following the change of election unless a subsequent event allows for a further election change. Except as provided in Section 10.3(e) for HIPAA special enrollment rights in the event of birth, adoption, or placement for adoption, all election changes will be effective on a prospective basis only (i.e., election changes will become effective no earlier than the first day of the next calendar month following the date that the election change was filed, but, as determined by the Plan Administrator, election changes may become effective later to the extent that the coverage in the applicable Benefit Plan Option commences later).
- (c) *Effect of New Election Upon Amount of Benefits.* For the effect of a changed election upon the maximum and minimum benefits under the Health FSA and DCAP Components, see Sections 7.4 and 8.4 respectively.

10.3 Events Permitting Exception to Irrevocability Rule for All Benefits

A Participant may change an election as described below upon the occurrence of the stated events for the applicable component of this Plan:

- (a) *Open Enrollment Period (Applies to all Benefit Plan Options).* A Participant may change an election during the Open Enrollment Period in accordance with Section 4.3.
- (b) *Termination of Employment (Applies to all Benefit Plan Options).* A Participant's election will terminate under the Plan upon termination of employment in accordance with Section 3.3.
- (c) *Leaves of Absence (Applies to all Benefit Plan Options).* A Participant may change an election under the Plan upon FMLA leave in accordance with Section 3.5 and upon non-FMLA leave in accordance with Section 3.6.
- (d) Change in Status (Applies to Premium Payment Benefits and to Health FSA Benefits and DCAP Benefits as limited further below). A Participant may change his or her election under the Plan upon the occurrence of a Change in Status, but only if such election change is made on account of and corresponds with a Change in Status that affects eligibility for coverage under a plan of the Employer or a plan of the Spouse's or Dependent's employer (referred to as the general consistency requirement). A Change in Status that affects eligibility for coverage under a plan of the Employer or a plan of the Spouse's or Dependent's employer includes a Change in Status that results in an increase or decrease in the number of an Employee's family members (i.e., a Spouse and/or Dependents) who may benefit from the coverage.
- (e) HIPAA Special Enrollment Rights (Applies to Premium Payment Benefits under Medical Insurance Plans only, and not to any other Insurance Plan, Health FSA, or DCAP Benefits). If a Participant or his or her Spouse or Dependent is entitled to special enrollment rights under a group health plan (other than an excepted benefit), as required by HIPAA under Code Section 9801(f), then a Participant may revoke a prior election for group health plan coverage and make a new election (including, when required by HIPAA, an election to enroll in another group health plan), provided that the election change corresponds with such HIPAA special enrollment rights. As required by HIPAA, a special enrollment right will arise in the following circumstances:

(1) a Participant or his or her Spouse or Dependent declined to enroll in group health plan coverage because he or she had coverage, and eligibility for such coverage is subsequently lost because: (a) the coverage was provided under COBRA and the COBRA coverage was exhausted; or (b) the coverage was non-COBRA coverage and the coverage terminated due to loss of eligibility for coverage or the employer contributions for the coverage were terminated;

(2) a new Dependent is acquired as a result of marriage, birth, adoption, or placement for adoption;

(3) the Participant's or Dependent's coverage under a Medicaid plan or state children's health insurance program is terminated as a result of loss of eligibility for such coverage; or

(4) the Participant or Dependent becomes eligible for a state premium assistance subsidy from a Medicaid plan or through a state children's health insurance program with respect to coverage under the group health plan.

An election to add previously eligible Dependents as a result of the acquisition of a new Spouse or Dependent child will be considered to be consistent with the special enrollment right. An election change on account of a HIPAA special enrollment attributable to the birth, adoption, or placement for adoption of a new Dependent child may, subject to the provisions of the underlying group health plan, be effective retroactively (up to 30 days).

For purposes of Section 10.3(e)(1), a loss of eligibility includes (but is not limited to) loss of eligibility due to legal separation, divorce, cessation of dependent status, death of an employee, termination of employment, reduction of hours, or any loss of eligibility for coverage that is measured with reference to any of the foregoing; loss of coverage offered through an HMO that does not provide benefits to individuals who do not reside, live, or work in the service area because an individual no longer resides, lives, or works in the service area (whether or not within the choice of the individual), and in the case of HMO coverage in the group market, no other benefit Plan is available to the individual; a situation in which an individual incurs a claim that would meet or exceed a lifetime limit on all benefits; and a situation in which a plan no longer offers any benefits to the class of similarly situated individuals that includes the individual.

- (f) Certain Judgments, Decrees and Orders (Applies to Premium Payment and Health FSA Benefits, but Not to DCAP Benefits). If a judgment, decree, or order (collectively, an "Order") resulting from a divorce, legal separation, annulment, or change in legal custody (including a National Medical Support Order) requires accident or health coverage (including an election for Health FSA Benefits) for a Participant's child (including a foster child who is a Dependent of the Participant), then a Participant may (1) change his or her election to provide coverage for the child (provided that the Order requires the Participant to provide coverage); or (2) change his or her election to revoke coverage for the child if the Order requires that another individual (including the Participant's Spouse or former Spouse) provide coverage under that individual's plan and such coverage is actually provided.
- Medicare and Medicaid (Applies to Premium Payment Benefits, to Health FSA Benefits (g) as Limited Below, but Not to DCAP Benefits). If a Participant or his or her Spouse or Dependent who is enrolled in a health or accident plan under this Plan becomes entitled to (i.e., becomes enrolled in) Medicare or Medicaid (other than coverage consisting solely of benefits under Section 1928 of the Social Security Act providing for pediatric vaccines), then the Participant may prospectively reduce or cancel the health or accident coverage of the person becoming entitled to Medicare or Medicaid and/or the Participant's Health FSA coverage may be canceled (but not reduced). Notwithstanding the foregoing, such cancellation will not become effective to the extent that it would reduce future contributions to the Health FSA to a point where the total contributions for the Plan Year are less than the amount already reimbursed for the Plan Year. Furthermore, if a Participant or his or her Spouse or Dependent who has been entitled to Medicare or Medicaid loses eligibility for such coverage, then the Participant may prospectively elect to commence or increase the accident or health coverage of the individual who loses Medicare or Medicaid eligibility and/or the Participant's Health FSA coverage may commence or increase.

(h) Change in Cost (Applies to Premium Payment Benefits, to DCAP Benefits as Limited Below, but Not to Health FSA Benefits). For purposes of this Section 10.3(h), "similar coverage" means coverage for the same category of benefits for the same individuals (e.g., family to family or single to single). For example, two plans that provide major medical coverage are considered to be similar coverage. For purposes of this definition, (a) a health FSA is not similar coverage with respect to an accident or health plan that is not a health FSA; (b) an HMO and a PPO are considered to be similar coverage; and (c) coverage by another employer, such as a Spouse's or Dependent's employer, may be treated as similar coverage if it otherwise meets the requirements of similar coverage.

(1) Increase or Decrease for Insignificant Cost Changes. Participants are required to increase their elective contributions (by increasing Salary Reductions) to reflect insignificant increases in their required contribution for their Benefit Plan Option(s), and to decrease their elective contributions to reflect insignificant decreases in their required contribution. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will determine whether an increase or decrease is insignificant based upon all the surrounding facts and circumstances, including but not limited to the dollar amount or percentage of the cost change. The Plan Administrator, on a reasonable and consistent basis, will automatically effectuate this increase or decrease in affected employees' elective contributions on a prospective basis.

(2) Significant Cost Increases. If the Plan Administrator determines that the cost charged to an Employee of a Participant's Benefit Plan Option(s) significantly increases during a Period of Coverage, then the Participant may (a) make a corresponding prospective increase in his or her elective contributions (by increasing Salary Reductions); (b) revoke his or her election for that coverage, and in lieu thereof, receive on a prospective basis coverage under another Benefit Plan Option that provides similar coverage (such as an HMO, but not the Health FSA); or (c) drop coverage prospectively if there is no other Benefit Plan Option available that provides similar coverage. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether a cost increase is significant in accordance with prevailing IRS guidance.

(3) *Significant Cost Decreases.* If the Plan Administrator determines that the cost of any Benefit Plan Option significantly decreases during a Period of Coverage, then the Plan Administrator may permit the following election changes: (a) Participants enrolled in that Benefit Plan Option may make a corresponding prospective decrease in their elective contributions (by decreasing Salary Reductions); (b) Participants who are enrolled in another Benefit Plan Option (such as an HMO, but not the Health FSA) may change their election on a prospective basis to elect the Benefit Plan Option that has decreased in cost; or (c) Employees who are otherwise eligible under Section 3.1 may elect the Benefit Plan Option that has decreased in cost (such as the PPO) on a prospective basis, subject to the terms and limitations of the Benefit Plan Option. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether a cost decrease is significant in accordance with prevailing IRS guidance.

(4) Limitation on Change in Cost Provisions for DCAP Benefits. The above "Change in Cost" provisions (Sections 10.3(h)(1) through 10.3(h)(3)) apply to DCAP Benefits only if the cost change is imposed by a dependent care provider who is not a "relative" of the Employee. For this purpose, a relative is an individual who is related as described in Code Sections 152(d)(2)(A) through (G), incorporating the rules of Code Sections 152(f)(4).

(i) Change in Coverage (Applies to Premium Payment and DCAP Benefits, but Not to Health FSA Benefits).

The definition of "similar coverage" under Section 10.3(h) applies also to this Section 10.3(i).

(1) Significant Curtailment. If coverage is "significantly curtailed" (as defined below), Participants may elect coverage under another Benefit Plan Option that provides similar coverage. In addition, as set forth below, if the coverage curtailment results in a "Loss of Coverage" (as defined below), then Participants may drop coverage if no similar coverage is offered by the Employer. The Plan Administrator in its sole discretion, on a uniform and consistent basis, will decide, in accordance with prevailing IRS guidance, whether a curtailment is "significant," and whether a Loss of Coverage has occurred.

(a) Significant Curtailment Without Loss of Coverage. If the Plan Administrator determines that a Participant's coverage under a Benefit Plan Option under this Plan (or the Participant's Spouse's or Dependent's coverage under his or her employer's plan) is significantly curtailed without a Loss of Coverage (for example, when there is a significant increase in the deductible, the co-pay, or the out-of-pocket cost-sharing limit under an accident or health plan) during a Period of Coverage, the Participant may revoke his or her election for the affected coverage, and in lieu thereof, prospectively elect coverage under another Benefit Plan Option that provides similar coverage (such as an HMO, but not the Health FSA). Coverage under a plan is deemed to be "significantly curtailed" only if there is an overall reduction in coverage provided under the plan so as to constitute reduced coverage generally.

(b) *Significant Curtailment With a Loss of Coverage*. If the Plan Administrator determines that a Participant's Benefit Plan Option coverage under this Plan (or the Participant's Spouse's or Dependent's coverage under his or her employer's plan) is significantly curtailed, and if such curtailment results in a Loss of Coverage during a Period of Coverage, then the Participant may revoke his or her election for the affected coverage and may either prospectively elect coverage under another Benefit Plan Option that provides similar coverage (such as an HMO, but not the Health FSA) or drop coverage if no other Benefit Plan Option providing similar coverage is offered by the Employer.

(c) *Definition of Loss of Coverage*. For purposes of this Section 10.3(i)(1), a "Loss of Coverage" means a complete loss of coverage (including the elimination of a Benefit Plan Option, an HMO ceasing to be available where the Participant or his or her Spouse or Dependent resides, or a Participant or his or her Spouse or Dependent losing all coverage under the Benefit Plan Option by reason of an overall lifetime or annual limitation). In addition, the Plan Administrator, in its sole discretion, on a uniform and consistent basis, may treat the following as a Loss of Coverage:

• a substantial decrease in the medical care providers available under the Benefit Plan Option (such as a major hospital ceasing to be a member of a preferred provider network or a substantial decrease in the number of physicians participating in a PPO or HMO);

- a reduction in benefits for a specific type of medical condition or treatment with respect to which the Participant or his or her Spouse or Dependent is currently in a course of treatment; or
- any other similar fundamental loss of coverage.

(d) *DCAP Coverage Changes.* A Participant may make a prospective election change that is on account of and corresponds with a change by the Participant in the dependent care service provider. For example: (i) if the Participant terminates one dependent care service provider and hires a new dependent care service provider, then the Participant may change coverage to reflect the cost of the new service provider; and (ii) if the Participant terminates a dependent care service provider because a relative becomes available to take care of the child at no charge, then the Participant may cancel coverage.

(2) Addition or Significant Improvement of a Benefit Plan Option. If during a Period of Coverage the Plan adds a new Benefit Plan Option or significantly improves an existing Benefit Plan Option, the Plan Administrator may permit the following election changes: (a) Participants who are enrolled in a Benefit Plan Option other than the newly added or significantly improved Benefit Plan Option may change their elections on a prospective basis to elect the newly added or significantly improved Benefit Plan Option; and (b) Employees who are otherwise eligible under Section 3.1 may elect the newly added or significantly improved Benefit Plan Option, in its sole discretion and on a uniform and consistent basis, will decide whether there has been an addition of, or a significant improvement in, a Benefit Plan Option in accordance with prevailing IRS guidance.

(3) Loss of Coverage Under Other Group Health Coverage. A Participant may prospectively change his or her election to add group health coverage for the Participant or his or her Spouse or Dependent, if such individual(s) loses coverage under any group health coverage sponsored by a governmental or educational institution, including (but not limited to) the following: a state children's health insurance program under Title XXI of the Social Security Act; a medical care program of an Indian Tribal government (as defined in Code Section 7701(a)(40), the Indian Health Service, or a tribal organization; a state health benefits risk pool; or a foreign government group health plan, subject to the terms and limitations of the applicable Benefit Plan Option(s).

(4) Change in Coverage Under An Employer Plan. A Participant may make a prospective election change that is on account of and corresponds with a change made under an employer plan (including a plan of the Employer or a plan of the Spouse's or Dependent's employer), so long as (a) the other cafeteria plan or qualified benefits plan permits its participants to make an election change that would be permitted under applicable IRS regulations; or (b) the Plan permits Participants to make an election for a Period of Coverage that is different from the plan year under the other cafeteria plan or qualified benefits plan. For example, if an election is made by the Participant's Spouse during his or her employer's open enrollment to drop coverage, the Participant may add coverage to replace the dropped coverage. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will decide whether a requested change is on account of and corresponds with a change made under the other employer plan, in accordance with prevailing IRS guidance.

Election changes may not be made to reduce Health FSA coverage during a Period of Coverage; however, election changes may be made to cancel Health FSA coverage completely due to the occurrence of any of the following events: death of a Spouse, divorce, legal separation, or annulment; death of a Dependent; change in employment status such that the Participant becomes ineligible for Health FSA coverage; or a Dependent's ceasing to satisfy eligibility requirements for Health FSA coverage. Notwithstanding the foregoing, such cancellation will not become effective to the extent that it would reduce future contributions to the Health FSA to a point where the total contributions for the Plan Year are less than the amount already reimbursed for the Plan Year. The Plan Administrator, in its sole discretion and on a uniform and consistent basis, will determine, based on prevailing IRS guidance, whether a requested change is on account of and corresponds with a Change in Status. Assuming that the general consistency requirement is satisfied, a requested election change must also satisfy the following specific consistency requirements in order for a Participant to be able to alter his or her election based on the specified Change in Status:

(5) Loss of Spouse or Dependent Eligibility; Special COBRA Rules. For a Change in Status involving a Participant's divorce, annulment or legal separation from a Spouse, the death of a Spouse or a Dependent, or a Dependent's ceasing to satisfy the eligibility requirements for coverage, a Participant may only elect to cancel accident or health insurance coverage for (a) the Spouse involved in the divorce, annulment, or legal separation; (b) the deceased Spouse or Dependent; or (c) the Dependent that ceased to satisfy the eligibility requirements. Canceling coverage for any other individual under these circumstances would fail to correspond with that Change in Status. Notwithstanding the foregoing, if the Participant or his or her Spouse or Dependent becomes eligible for COBRA (or similar health plan continuation coverage under state law) under the Employer's plan because of a reduction of hours or because the Participant's Dependent ceases to satisfy the eligibility requirements for coverage (and the Participant remains a Participant under this Plan), then the Participant may increase his or her election to pay for such coverage.

(6) Gain of Coverage Eligibility Under Another Employer's Plan. For a Change in Status in which a Participant or his or her Spouse or Dependent gains eligibility for coverage under a cafeteria plan or qualified benefit plan of the employer of the Participant's Spouse or Dependent as a result of a change in marital status or a change in employment status, a Participant may elect to cease or decrease coverage for that individual only if coverage for that individual becomes effective or is increased under the Spouse's or Dependent's employer's plan. The Plan Administrator may rely on a Participant's certification that the Participant has obtained or will obtain coverage under the Spouse's or Dependent's employer's plan, unless the Plan Administrator has reason to believe that the Participant's certification is incorrect.

(7) Special Consistency Rule for DCAP Benefits. With respect to the DCAP Benefits, a Participant may change or terminate his or her election upon a Change in Status if (a) such change or termination is made on account of and corresponds with a Change in Status that affects eligibility for coverage under an employer's plan; or (b) the election change is on account of and corresponds with a Change in Status that affects eligibility of Dependent Care Expenses for the tax exclusion under Code Section 129.

(j) Mid-Year Election Changes Permitted by Consolidated Appropriations Act, 2021

Notwithstanding anything in the Plan to the contrary, subject to and in accordance with the Consolidated Appropriations Act, 2021 and any guidance implementing thereto, including but not limited to IRS Notice 2021-15, the following mid-year election plan changes are permissible:

- During the period from June 1, 2021 through August 31, 2021, Participants may, on a prospective basis, request to revoke, increase or decrease their 2021 Plan Year Salary Reduction amount or make a new election under the applicable Health FSA. A Participant may change his or her election pursuant to this provision not more than three (3) times during the period from June 1, 2021 through August 31, 2021;
 - Any election to decrease a Health FSA election will be limited to amount already reimbursed to or contributed by the Participant, whichever is greater;
- During the period from June 1, 2021 through August 31, 2021, Participants may, on a prospective basis, request to revoke, increase or decrease their 2021 Plan Year Salary Reduction amount under the DCAP Account;
 - For purposes of this Section 10.3(i), the \$5,000 and \$2,500 maximums provided in Section 8.4(b) are replaced with \$10,500 and \$5,250, respectively;
 - Any election to decrease a DCAP Account election will be limited to the amount already reimbursed to or contributed by the Participant, whichever is greater;
- The above referenced changes shall be effective as of the first payroll period after the required submission date that follows the Plan Administrator's receipt of a proper election change request, and if a Participant requests to reduce his or her Health FSA election to the amount of reimbursements received or the amount already contributed, Salary Reductions for the remainder of the Plan Year will be adjusted accordingly. Notwithstanding the foregoing, while Salary Reduction amounts are applied prospectively under any revised election, amounts contributed to the Health FSA and/or DCAP Account after the revised election may be used to reimburse any Medical Care Expense or Dependent Care Expenses, as applicable, incurred on or after January 1, 2021 or through the end of the 2021 Plan Year, subject to any Grace Period.

A Participant entitled to change an election as described in this Section 10.3 must do so in accordance with the procedures described in Section 10.2.

10.4 Election Modifications Required by Plan Administrator

The Plan Administrator may, at any time, require any Participant or class of Participants to amend the amount of their Salary Reductions for a Period of Coverage if the Plan Administrator determines that such action is necessary or advisable in order to (a) satisfy any of the Code's nondiscrimination requirements applicable to this Plan or other cafeteria plan; (b) prevent any Employee or class of Employees from having to recognize more income for federal income tax purposes from the receipt of benefits hereunder than would otherwise be recognized; (c) maintain the qualified status of benefits received under this Plan; or (d) satisfy Code nondiscrimination requirements or other limitations applicable to the Employer's qualified plans. In the event that contributions need to be reduced for a class of Participants, the Plan Administrator will reduce the Salary Reduction amounts for each affected Participant, beginning with the Participant in the class who had elected the highest Salary Reduction amount and continuing with the Participant in the class who had elected the next-highest Salary Reduction amount, and so forth, until the defect is corrected.

ARTICLE XI. Appeals Procedure

11.1 Procedure If Benefits Are Denied Under This Plan

If a claim for reimbursement under this Plan is wholly or partially denied, then claims will be administered in accordance with the claims procedure set forth in Appendix C of this Plan.

11.2 Claims Procedures for Insurance Benefits

Claims and reimbursement for benefits under any Insurance Plan will be administered in accordance with the claims procedures for the Insurance Plans, as set forth in their governing plan documents and/or summary plan descriptions.

ARTICLE XII. Recordkeeping and Administration

12.1 Plan Administrator

The administration of this Plan will be under the supervision of the Plan Administrator. It is the principal duty of the Plan Administrator to see that this Plan is carried out, in accordance with its terms, for the exclusive benefit of persons entitled to participate in this Plan without discrimination among them.

12.2 Powers of the Plan Administrator

The Plan Administrator will have such duties and powers as it considers necessary or appropriate to discharge its duties. It will have the exclusive right to interpret the Plan and to decide all matters thereunder, and all determinations of the Plan Administrator with respect to any matter hereunder will be conclusive and binding on all persons. Without limiting the generality of the foregoing, the Plan Administrator will have the following discretionary authority:

- to construe and interpret this Plan, including all possible ambiguities, inconsistencies, and omissions in the Plan and related documents, and to decide all questions of fact, questions relating to eligibility and participation, and questions of benefits under this Plan;
- (b) to prescribe procedures to be followed and the forms to be used by Employees and Participants to make elections pursuant to this Plan;
- (c) to prepare and distribute information explaining this Plan and the benefits under this Plan in such manner as the Plan Administrator determines to be appropriate;
- (d) to request and receive from all Employees and Participants such information as the Plan Administrator will from time to time determine to be necessary for the proper administration of this Plan;
- (e) to furnish each Employee and Participant with such reports with respect to the administration of this Plan as the Plan Administrator determines to be reasonable and appropriate, including appropriate statements setting forth the amounts by which a Participant's Compensation has been reduced in order to provide benefits under this Plan;
- (f) to receive, review, and keep on file such reports and information regarding the benefits covered by this Plan as the Plan Administrator determines from time to time to be necessary and proper;

- (g) to appoint and employ such individuals or entities to assist in the administration of this Plan as it determines to be necessary or advisable, including legal counsel and benefit consultants;
- (h) to sign documents for the purposes of administering this Plan, or to designate an individual or individuals to sign documents for the purposes of administering this Plan;
- (i) to secure independent medical or other advice and require such evidence as it deems necessary to decide any claim or appeal; and
- (j) to maintain the books of accounts, records, and other data in the manner necessary for proper administration of this Plan and to meet any applicable disclosure and reporting requirements.

12.3 Reliance on Participant, Tables, etc.

The Plan Administrator may rely upon the direction, information, or election of a Participant as being proper under the Plan and will not be responsible for any act or failure to act because of a direction or lack of direction by a Participant. The Plan Administrator will also be entitled, to the extent permitted by law, to rely conclusively on all tables, valuations, certificates, opinions, and reports that are furnished by accountants, attorneys, or other experts employed or engaged by the Plan Administrator.

12.4 Provision for Third-Party Plan Service Providers

The Plan Administrator, subject to approval of the Employer, may employ the services of such persons as it may deem necessary or desirable in connection with the operation of the Plan. Unless otherwise provided in the service agreement, obligations under this Plan will remain the obligation of the Plan Administrator or the Employer, as applicable.

12.5 Fiduciary Liability

To the extent permitted by law, the Plan Administrator will not incur any liability for any acts or for failure to act except for their own willful misconduct or willful breach of this Plan.

12.6 Compensation of Plan Administrator

Unless otherwise determined by the Employer and permitted by law, any Plan Administrator that is also an Employee of the Employer will serve without compensation for services rendered in such capacity, but all reasonable expenses incurred in the performance of their duties will be paid by the Employer.

12.7 Insurance Contracts

The Employer will have the right (a) to enter into a contract with one or more insurance companies for the purposes of providing any benefits under the Plan; and (b) to replace any of such insurance companies or contracts. Any dividends, retroactive rate adjustments, or other refunds of any type that may become payable under any such insurance contract will not be assets of the Plan but will be the property of and be retained by the Employer, to the extent that such amounts are less than aggregate Employer contributions toward such insurance.

12.8 Inability to Locate Payee

If the Plan Administrator is unable to make payment to any Participant or other person to whom a payment is due under the Plan because it cannot ascertain the identity or whereabouts of

such Participant or other person after reasonable efforts have been made to identify or locate such person, then such payment and all subsequent payments otherwise due to such Participant or other person will be forfeited following a reasonable time after the date any such payment first became due.

12.9 Effect of Mistake

In the event of a mistake as to the eligibility or participation of an Employee, the allocations made to the account of any Participant, or the amount of benefits paid or to be paid to a Participant or other person, the Plan Administrator will, to the extent that it deems administratively possible and otherwise permissible under Code Section 125 or the regulations issued thereunder, cause to be allocated or cause to be withheld or accelerated, or otherwise make adjustment of, such amounts as it will in its judgment accord to such Participant or other person the credits to the account or distributions to which he or she is properly entitled under the Plan. Such action by the Plan Administrator may include withholding of any amounts due to the Plan or the Employer from Compensation paid by the Employer.

ARTICLE XIII. General Provisions

13.1 Expenses

All reasonable expenses incurred in administering the Plan are currently paid by forfeitures to the extent provided in Section 7.6 with respect to Health FSA Benefits and Section 8.6 with respect to DCAP Benefits, and then by the Employer.

13.2 No Contract of Employment

Nothing herein contained is intended to be or will be construed as constituting a contract or other arrangement between any Employee and the Employer to the effect that such Employee will be employed for any specific period of time.

13.3 Amendment and Termination

This Plan has been established with the intent of being maintained for an indefinite period of time. Nonetheless, the Employer may amend or terminate all or any part of this Plan at any time for any reason by resolution of the Employer's Board of Directors or by any person or persons authorized by the Board of Directors to take such action.

13.4 Governing Law

The provisions of the Plan will be construed, administered and enforced according to applicable federal law and, to the extent not preempted, the laws of the State of California.

13.5 Compliance With Code and Other Applicable Laws

It is intended that this Plan meet all applicable requirements of the Code and of all regulations issued thereunder. This Plan will be construed, operated, and administered accordingly, and in the event of any conflict between any part, clause, or provision of this Plan and the Code, the

provisions of the Code will be deemed controlling, and any conflicting part, clause, or provision of this Plan will be deemed superseded to the extent of the conflict. In addition, the Plan will comply with the requirements of all other applicable laws.

13.6 No Guarantee of Tax Consequences

Neither the Plan Administrator nor the Employer makes any commitment or guarantee that any amounts paid to or for the benefit of a Participant under this Plan will be excludable from the Participant's gross income for federal, state, or local income tax purposes. It will be the obligation of each Participant to determine whether each payment under this Plan is excludable from the Participant's gross income for federal, state, and local income tax purposes and to notify the Plan Administrator if the Participant has any reason to believe that such payment is not so excludable.

13.7 Indemnification of Employer

If any Participant receives one or more payments or reimbursements under this Plan on a taxfree basis and if such payments do not qualify for such treatment under the Code, then such Participant will indemnify and reimburse the Employer for any liability that it may incur for failure to withhold federal income taxes, Social Security or Medicare taxes, or other taxes from such payments or reimbursements.

13.8 Non-Assignability of Rights

The right of any Participant to receive any reimbursement under this Plan will not be alienable by the Participant by assignment or any other method and will not be subject to claims by the Participant's creditors by any process whatsoever. Any attempt to cause such right to be so subjected will not be recognized, except to the extent required by law.

13.9 Headings

The headings of the various Articles and Sections are inserted for convenience of reference and are not to be regarded as part of this Plan or as indicating or controlling the meaning or construction of any provision.

13.10 Plan Provisions Controlling

In the event that the terms or provisions of any summary or description of this Plan are in any construction interpreted as being in conflict with the provisions of this Plan as set forth in this document, the provisions of this Plan will be controlling.

13.11 Severability

Should any part of this Plan subsequently be invalidated by a court of competent jurisdiction, the remainder of the Plan will be given effect to the maximum extent possible.

* * *

To record the adoption of the Plan, the Employer's authorized representative hereby executes this document on this ______ day of ______, 2021.

Alameda County Transportation Commission

Ву: _____

Title:

Date: _____

Appendix A

Benefit Plan Options

Benefit Plan Options will include the coverage available under the following plans maintained by the Alameda County Transportation Commission:

A. Insurance Plans

- 1. Medical Insurance under the Public Employees' Medical and Hospital Care Act (PEMHCA or "PERS Health")
- 2. Dental Insurance
- 3. Vision Insurance
- 4. Long-Term Disability Insurance
- 5. Short-Term Disability Insurance
- 6. Group-Term Life Insurance (on the life of an Employee only)

B. Health Care Flexible Spending Account

C. Dependent Care Flexible Spending Account

Appendix B

Exclusions: Medical Expenses that are Not Reimbursable from the Health FSA

The Alameda County Transportation Commission Cafeteria Plan document contains the general rules governing what expenses are reimbursable. This Appendix B, as referenced in the Plan document, specifies certain expenses that are excluded under this Plan with respect to reimbursement from the Health FSA—that is, expenses that *are not reimbursable*, even if they meet the definition of "medical care" under Code Section 213(d) and may otherwise be reimbursable under the regulations governing Health FSAs. Such "medical care" also includes expenses incurred for menstrual care products. A 'menstrual care product' means a tampon, pad, liner, cup, sponge, or similar product used by individuals with respect to menstruation or other genital-tract secretions.

Exclusions: The following expenses are not reimbursable from the Health FSA, even if they meet the definition of "medical care" under Code Section 213(d) and may otherwise be reimbursable under legal requirements applicable to health FSAs:

- Premiums for other health coverage, including but not limited to premiums for any other plan (whether or not sponsored by the Employer)
- Long-term care services
- Cosmetic surgery or other similar procedures, unless the surgery or procedure is necessary to ameliorate a deformity arising from, or directly related to, a congenital abnormality, a personal injury resulting from an accident or trauma, or a disfiguring disease. "Cosmetic surgery" means any procedure that is directed at improving the patient's appearance and does not meaningfully promote the proper function of the body or prevent or treat illness or disease.
- The salary expense of a nurse to care for a healthy newborn at home
- Funeral and burial expenses
- Household and domestic help (even if recommended by a qualified physician due to an Employee's or Dependent's inability to perform physical housework)
- Custodial care

- Costs for sending a problem child to a special school for benefits that the child may receive from the course of study and disciplinary methods
- Social activities, such as dance lessons (even if recommended by a physician for general health improvement)
- Bottled water
- Cosmetics, toiletries, toothpaste, etc.
- Uniforms or special clothing, such as maternity clothing
- Automobile insurance premiums
- Transportation expenses of any kind, including transportation expenses to receive medical care
- Marijuana and other controlled substances that are in violation of federal laws, even if prescribed by a physician
- Any item that does not constitute "medical care" as defined under Code Section 213(d)
- Any item that is not reimbursable due to the rules in Prop. Treas. Reg. Section 1.125-5(k)(4) or other applicable law or regulations

Appendix C

Claims Procedures

Capitalized terms in this Appendix C have the same meaning as the defined terms in the Alameda County Transportation Commission Cafeteria Plan.

Any Participant may file a claim with the Plan Administrator for a Plan benefit to which the claimant believes that he or she is entitled.

- 1. The Plan Administrator will receive all claims filed for benefits under the Plan. Upon receiving a claim, the Plan Administrator will review the claim and determine whether the claimant is entitled to receive any benefits pursuant to such claim. The Plan Administrator will notify the claimant in writing of any adverse decision with respect to his or her claim within 30 days after its submission. The notice of any adverse decision will be written in a manner calculated to be understood by the claimant and must include, as applicable: (i) the specific reason or reasons for the denial; (ii) specific references to the Plan provisions on which the denial is based; (iii) a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary; and (iv) an explanation of the Plan's claim review procedures.
- 2. If the circumstances require an extension of time for processing the initial claim, a written notice of the extension will be furnished to the claimant before the end of the initial 30-day period. This time period may be extended by an additional 15 days for matters beyond the control of the Plan Administrator, including in cases where a reimbursement claim is incomplete. The extension notice must indicate the circumstances requiring an extension of time.
- 3. If a claim for benefits is denied or if the Plan Administrator has given no response to such claim within the time period set out in the above paragraph (in which case the claim for benefits will be deemed to be denied), the claimant or his or her duly authorized representative, at the claimant's sole expense, may appeal the denial by submitting written notice of such appeal to the Plan Administrator within 90 days of the receipt of written notice of the denial or 60 days from the date such claim is deemed to be denied.
- 4. The claimant will be notified of the decision on the appeal within 90 days of receipt of the notice of appeal, unless circumstances require an extension of time for processing, in which case a decision will be rendered as soon as possible, but not later than 120 days after receipt of a notice of appeal. If such an extension of time is required, written notice of the extension will be furnished to the claimant before the end of the original 90-day period. The notice of decision on the appeal must be made in writing. If the decision on the appeal is not furnished within the time specified above, the appeal of the claim will be deemed denied.



1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

www.AlamedaCTC.org

Commission Chair Mayor Pauline Russo Cutter City of San Leandro

Commission Vice Chair Councilmember John Bauters City of Emeryville

AC Transit Board President Elsa Ortiz

Alameda County

Supervisor David Haubert, District 1 Supervisor Richard Valle, District 2 Supervisor Wilma Chan, District 3 Supervisor Nate Miley, District 4 Supervisor Keith Carson, District 5

BART

Vice President Rebecca Saltzman

City of Alameda Mayor Marilyn Ezzy Ashcraft

City of Albany Councilmember Rochelle Nason

City of Berkeley Councilmember Lori Droste

City of Dublin Mayor Melissa Hernandez

City of Fremont Mayor Lily Mei

City of Hayward Mayor Barbara Halliday

City of Livermore Mayor Bob Woerner

City of Newark Councilmember Luis Freitas

City of Oakland Councilmember At-Large Rebecca Kaplan Councilmember Sheng Thao

City of Piedmont Councilmember Jen Cavenaugh

City of Pleasanton Mayor Karla Brown

City of Union City Mayor Carol Dutra-Vernaci

Executive Director Tess Lengyel

ALAMEDA COUNTY TRANSPORTATION COMMISSION

RESOLUTION 21-009

Restating the Cafeteria Plan of the Alameda County Transportation Commission

Whereas, Alameda County Transportation Commission ("Alameda CTC") adopted the Alameda County Transportation Commission Cafeteria Plan ("the Plan"), effective February 1, 2012;

Whereas, the Plan was designed to permit an eligible employee to pay for his or her share of contributions for options allowed in the Plan on a pre-tax, salary reduction basis, and to contribute on a pre-tax, salary reduction basis to an employee's account for reimbursement of certain medical care expenses and/or to an account for reimbursement of certain dependent care expenses;

Whereas, all full-time employees are eligible to participate in the Plan;

Whereas, Alameda CTC's governing body ("Commission") has determined it to be in the best interest of the Alameda CTC to restate the Plan for 2020 to include provisions allowed in the Consolidated Appropriations Act, 2021 IRS Notice 2021-15; and

Whereas, the Commission has further determined it to be in the best interest of the Alameda CTC to restate the Plan for 2021 to include provisions allowed in the Consolidated Appropriations Act, 2021 IRS Notice 2021-15, and the American Rescue Plan of 2021.

Now, Therefore, Be It Resolved by the Commission of the Alameda CTC as follows:

A. Amendment of Cafeteria Plan

Section 1. Effective January 1, 2020, the Commission hereby amends the Alameda County Transportation Commission Cafeteria Plan, substantially in the form attached as Exhibit A, to permit provisions allowed for in the Consolidated Appropriations Act and 2021 IRS Notice 2021-15; and Section 2. Effective January 1, 2021, the Commission hereby amends the Alameda County Transportation Commission Cafeteria Plan, substantially in the form attached as Exhibit B, to permit provisions allowed for in the Consolidated Appropriations Act, 2021 IRS Notice 2021-15, and the American Rescue Plan of 2021.

DULY PASSED AND ADOPTED by the Alameda County Transportation Commission at the regular meeting of the Commission held on Thursday, May 27, 2021 in Oakland, California, by the following votes:

AYES:

NOES:

ABSTAIN:

ABSENT:

SIGNED:

ATTEST:

Pauline Russo Cutter Chair, Alameda CTC Vanessa Lee Clerk of the Commission



Memorandum

PH: (510) 208-7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Gary Huisingh, Deputy Executive Director of Project Delivery
SUBJECT:	Approve Cost Reimbursement Agreement with the Tri-Valley San Joaquin Valley Regional Rail Authority

Recommendation

It is recommended that the Commission approve and authorize the Executive Director or designee to execute a Cost Reimbursement Agreement with the Tri-Valley San Joaquin Valley Regional Rail Authority (TVSJVRRA).

Summary

TVSJVRRA is a new transit agency established by state law AB 758, which was effective January 1, 2018, and proposes to design and construct a section of the Valley Link rail route within the median of Interstate I-580 that would include a transfer station facility between Valley Link and BART at the Dublin/Pleasanton BART Station.

Alameda CTC has informed TVSJVRRA that the installation of the Valley Link route within the I-580 corridor would require modifications to the existing express lanes and related tolling equipment within the I-580 corridor between the I-580/I-680 Interchange and Greenville Road in Livermore, which Alameda CTC operates. TVSJRRVA recognizes that the design of the project within the I-580 corridor needs to meet Caltrans and Alameda CTC design standards and requirements for the roadway improvements and tolling equipment, and to address any impacts the Valley Link system may have on the existing roadway, I-580 express lanes, and tolling facilities.

TVSJVRRA has been engaged with Alameda CTC in the review of the proposed project and has agreed to provide \$992,000 to Alameda CTC for its participation in the development of the proposed project.

Staff recommends the Commission authorize the Executive Director or designee to execute a Cost Reimbursement Agreement with TVSJVRRA. The anticipated Alameda CTC consultant costs to be reimbursed through this agreement include preliminary engineering and design review and related support, as well as consultant support for

procurement/contracting of toll system integrator to ensure the next phase of the project (final design and construction) is not delayed.

Fiscal Impact: There is no fiscal impact associated with the requested action.



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

www.AlamedaCTC.org

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Vivek Bhat, Director of Programming and Project Controls Jacki Taylor, Senior Program Analyst
SUBJECT:	Metropolitan Transportation Commission's Safe and Seamless Mobility Quick-Strike Program Update

Recommendation

This item is to provide the Commission with an update on the Metropolitan Transportation Commission's Safe and Seamless Mobility Quick-Strike Program. This item is for information only.

Background

On February 5, 2021, the Metropolitan Transportation Commission (MTC) released a call for project nominations for the Safe and Seamless Mobility Quick-Strike Program (Program), a one-time, competitive grant program within its One Bay Area Grant Cycle 2 (OBAG 2) program framework.

To encourage community-based project investments and address local needs throughout the region, MTC had requested County Transportation Agencies (CTAs) to submit project nominations for their county area by March 30, 2021. County targets provided were to be used as a guide based on the OBAG 2 county program distribution, and Alameda County's funding target was 19.9% (approximately \$9.8 million).

In response, Alameda CTC staff submitted 12 project nominations requesting a total of \$15.1 million (Attachment A). MTC received a total of 71 project nominations from the nine Bay Area CTA's for a total amount requested of more than \$108 million. MTC staff has released an initial staff recommendation that includes nine (9) projects from Alameda County for a total of approximately \$9.6 million (Attachment B). A final recommendation is scheduled for consideration and approval by MTC in June 2021.

In order for projects to be included in the final recommendation for MTC's approval in June, all supporting documentation (complete application material) for the recommended projects is due to MTC (via Alameda CTC) by May 21, 2021. The required

Page 169

items are: a resolution of local support and, as applicable, a resolution for Surplus Lands Act compliance; completion of the Local Compliance Checklist; Complete Streets Checklist(s) and summarized responses to any comments received from the Bicycle and Pedestrian Advisory Committee; and a complete Transportation Improvement Program (TIP) application.

Potential Program Augmentation

At its May meeting, MTC will consider the distribution of an additional \$34 million available through the federal Coronavirus Response and Relief Supplemental Appropriations Act (CRRSAA) highway infrastructure funds. MTC staff's recommendation includes augmenting the Program with the CRRSAA funds as follows:

- \$17.2 million directed to fund additional projects nominated by CTAs primarily, this
 would be projects remaining from the original pool of nominations, but for counties
 that limited their nominations to within their county target, a limited number of new
 projects may be submitted; and
- \$17.2 million directed to safe and seamless mobility projects of regional significance – projects that were co-nominated by CTAs and MTC as part of the call for project nominations process will be shifted over to this category, creating capacity to fund additional CTA nominated projects.

MTC staff plans to present the final recommended program of projects to the MTC Commission for consideration and approval at its June 2021 meeting.

Next Steps

Sponsors of nominated projects are to continue finalizing the required project application material due to MTC (via Alameda CTC) by May 21, 2021. This includes completing TIP applications in MTC's online Fund Management System (FMS) by May 14th so they may be reviewed and submitted by Alameda CTC staff by the May 21st deadline. Following approval by MTC in June and federal TIP approval in summer 2021, project sponsors will have until September 30, 2022 to obligate the awarded federal funding.

Fiscal Impact: There is no fiscal impact. This is an information item only.

Attachments:

- A. Alameda CTC's Project Nominations
- B. MTC Safe and Seamless Mobility Quick-strike Program Initial Staff Recommendations

6.10A

Index #	Jurisdiction	Project / Program	Requested Amount
1	Alameda CTC	Countywide Safe Routes to School Operations	\$1,500,000
2	AC Transit	Quick Builds	\$954,000
3	AC Transit	Tempo Quick Build Transit Lane Delineation	\$300,000
4	Alameda County	Anita Avenue Streetscape Improvements	\$2,000,000
5	Bike East Bay	Various youth and adult bicycle promotion and education programs	\$160,000
6	City of Dublin	Safe Routes to School Improvements in Dublin	\$2,000,000
7	City of Fremont	Fremont Boulevard/Walnut Avenue Protected Intersection Project	\$1,271,000
8	City of Fremont	Fremont Boulevard/Grimmer Boulevard Improvement Project	\$1,415,000
9	LAVTA	Passenger Facilities Enhancements	\$2,000,000
10	МТС	Bay Bridge Forward, I-580 WB HOV Lane Extension	\$1,500,000
11	City of Oakland	14th Street Complete Streets Project	\$1,000,000
12	City of Oakland	East Bay Greenway Segment II	\$1,000,000
		Total	\$15,100,000

Projects Nominated for MTC's Safe and Seamless Mobility Quick Strike Program

This page intentionally left blank

6.10B

Bay Area Partnership Board

April 23, 2021

Agenda Item 4a

Safe and Seamless Mobility Quick-Strike Program

Subject:Discussion of staff recommendations for the Safe and Seamless Mobility Quick-
Strike program within the One Bay Area Grant (OBAG 2) program framework.

Background: On February 5, 2021, MTC staff released a call for project nominations for the Safe and Seamless Mobility Quick-Strike program. The materials distributed through the call for project nominations is provided as **Attachment A**.

This memo provides an overview of the project evaluation and prioritization process, and the staff recommended program of projects. Staff seeks feedback from the Board on this recommendation. Final recommendations will be presented for consideration and approval at the Programming and Allocations Committee meeting, June 9, 2021.

Project Prioritization Process

To encourage community-based project investments and address local needs throughout the region, County Transportation Agencies (CTA's) were invited to nominate projects for their county area. Funding targets were provided to guide the CTA's in developing nominations; however, final funding levels for each county are not guaranteed to correspond exactly to the initial targets. The final program of projects is based on the evaluation of individual projects and reflect regional priorities.

On March 30, 2021, MTC received 71 project nominations from CTA's for a total grant request of more than \$108 million. MTC staff evaluated projects against the six evaluation criteria established in the program guidelines:

- Alignment with the Connected Mobility Framework Values and Goals.
- Relationship to priority planning areas including, but not limited to, Priority Development Areas (PDAs) and Communities of Concern (CoCs).
- Inclusion of community engagement.
- Addressing a gap in transit connectivity.
- Demonstration of partnership among multiple partners.
- Ability to deliver the project quickly while meeting all federal funding requirements.

Each project was evaluated and scored individually by an MTC staff evaluation team. Following the initial evaluation, the committee took into consideration any tiering or prioritization communicated by a CTA. The full list of projects nominated by CTAs, including project scoring, is provided as **Attachment B**.

Recommendation

With the \$49.4 million currently available within the Quick-Strike framework, staff recommends funding 40 projects across the region (including 2 projects with partial funding recommendations). The list of projects recommended for funding is included as **Attachment C**.

	Initial Nomination		Staff	
	Targets		Recommendation	
	%	\$	%	\$
Alameda	19.9%	\$9.8	19.4%	\$9.6
Contra Costa	14.6%	\$7.2	15.2%	\$7.5
Marin	2.8%	\$1.4	2.4%	\$1.2
Napa	2.1%	\$1.0	2.4%	\$1.2
San Francisco	12.5%	\$6.2	13.8%	\$6.8
San Mateo	8.4%	\$4.1	8.2%	\$4.1
Santa Clara	27.0%	\$13.3	26.1%	\$12.9
Solano	5.5%	\$2.7	6.1%	\$3.0
Sonoma	7.2%	\$3.6	6.3%	\$3.1
Total	100%	\$49.4	100%	\$49.4

Notes: Amounts shown in millions. Totals may not add due to rounding. Total does not include \$5 million set aside for BRTRTF early implementation.

Potential Program Augmentation

At their May meeting, the Commission will consider the distribution of \$34 million in federal Coronavirus Response and Relief Supplemental Appropriations Act (CRRSAA) highway infrastructure funds.¹ Staff recommends augmenting the Quick-Strike program with these funds as follows:

- Half of the funds (\$17.2 million) directed to fund additional projects nominated by CTAs; and
- The remaining half of augmented funds (\$17.2 million) directed to safe and seamless mobility projects of regional significance – including projects that were co-nominated by CTAs and MTC as part of the call for project nominations process. Capacity created by shifting CTA and MTC co-nominated projects to this category will be made available for additional CTA nominated projects.

Pending Commission approval of the program augmentation described above, staff will revise its recommended program of projects.

Staff anticipates presenting the recommended program of projects to the MTC Commission for consideration and approval at its June 2021 meeting.

Issues: None.

Recommendation: Information.

Attachments:Attachment A: Call for Project NominationsAttachment B: Recommended Program of ProjectsAttachment C: List of All Project Nominations Received

Therese W. McMillan

¹ An additional \$31 million in CRRSAA highway infrastructure program funds will be distributed by the CTC through the regional portion of the STIP.

Bay Area Partnership Board April 23, 2021 Page 1 of 13

M

Scott Haggerty, Chair Alameda County

Alfredo Pedroza, Vice Chair Napa County and Cities

> Margaret Abe-Koga Cities of Santa Clara County

Eddie Abn San Francisco Bay Conservation and Development Commission

> **David Canepa** San Mateo County

Cindy Chavez Santa Clara County

Damon Connolly Marin County and Cities

Carol Dutra-Vernaci Cities of Alameda County

Dina El-Tawansy California State Transportation Agency

Dorene M. Giacopini U.S. Department of Transportation

> Federal D. Glover Contra Costa County

Nick Josefowitz San Francisco Mayor's Appointee

> Sam Liccardo San Jose Mayor's Appointee

Jake Mackenzie Sonoma County and Cities

Gina Papan Cities of San Mateo County

David Rabbitt Association of Bay Area Governments

Hillary Ronen City and County of San Francisco

> *Libby Schaaf* Oakland Mayor's Appointee

James P. Spering Solano County and Cities

Amy R. Worth Cities of Contra Costa County

Vacant U.S. Department of Housing and Urban Development

Therese W. McMillan Executive Director

Alix Bockelman Deputy Executive Director, Policy

Andrew B. Fremier Deputy Executive Director, Operations

> Brad Paul Deputy Executive Director, Local Government Services

METROPOLITAN TRANSPORTATION COMMISSION Attachment A Agenda Item 4a Bay Area Metro Center 375 Beale Street, Suite 800 San Francisco, CA 94105 415.778.6700 www.mtc.ca.gov

February 5, 2021

TO: CTA Executive Directors

RE: Safe and Seamless Mobility Quick-Strike Program - Call for Nominations

Dear CTA Executive Directors:

On January 27, 2021, the Metropolitan Transportation Commission (MTC) approved MTC Resolution No. 4202, Revised, which included the policy framework for the Safe and Seamless Mobility Quick-Strike program. The detailed program guidelines, located in Appendix A-11 to the One Bay Area Grant (OBAG 2) program resolution, are provided as an attachment (Attachment 1). OBAG 2 policies, procedures, and requirements apply to the Safe & Seamless program unless specified otherwise in Appendix A-11.

The purpose of this letter is to release the call for project nomination letters for the Safe & Seamless grant program and provide additional guidance on the solicitation process. Project nomination letters for projects submitted as part of county targets are to be submitted by the Bay Area County Transportation Agencies (CTAs). Project sponsors and interested stakeholders are encouraged to work with the applicable CTA (or multiple CTAs for multi-county projects) for submittal of project nominations. This call does not include the projects to be identified by the Blue Ribbon Transit Recovery Task Force, which will follow a different process.

Project Nominations

CTAs are invited to submit project nomination letters to MTC for projects located within their counties, as well as multi-county and regional projects for the respective county target. Letters should describe the CTA's process to identify and prioritize projects for this competitive grant opportunity. CTAs should also list, in narrative form or in a table, the projects being nominated, along with brief project descriptions and the amount of funds requested for each project.

Attached to the project nomination letters, CTAs must also provide completed project information forms for each project (**Attachment 2**).

In addition to these materials required to be submitted directly by the CTAs, project sponsors must submit a Complete Streets checklist for each nominated project into MTC's Complete Streets Database: <u>https://completestreets.mtc.ca.gov/</u>.

Bay Area Partnership Board April 23, 2021 Page 2 of 13 Safe & Seamless Call for Nomination Letters Page 2

Nomination letters, project information forms, and Complete Streets checklists must be submitted no later than Tuesday, March 30, 2021. Nomination letters and project information forms should be sent to Mallory Atkinson at <u>matkinson@bayareametro.gov</u>. Complete Streets checklists should be uploaded directly into the online database, linked above.

Project Evaluation & Final Project Applications

In April, MTC staff will evaluate project nominations using the established program criteria, funding eligibility, and focus areas. Staff will consider each CTA's nominations independently as well as in relation to other county submissions and regional priorities to develop its initial funding proposal. Staff will share its initial funding proposal with the Bay Area Partnership Board for discussion and feedback.

CTAs and project sponsors will be notified of MTC staff's funding recommendation by May 3, 2021. Project sponsors recommended for funding must submit the final application materials to MTC by May 21, 2021.

Final application materials include:

- Project submission in MTC's Financial Management System (FMS) <u>https://fms.mtc.ca.gov/fms/home.ds</u>, which will include detailed information on project scope, funding, and performance metrics.
- Written response to any remaining project-specific questions from MTC's evaluation team.
- Project map with sufficient detail to clearly identify the location and extent of the project.
- A signed Local Agency Compliance Checklist (Attachment 3). CTAs and local agencies should review this checklist carefully. Although these requirements were included in the OBAG 2 County Program, additional actions will be required for sponsors to satisfy the requirements for the Safe & Seamless grant program. These requirements include a review of the project's Complete Streets checklist by the appropriate Bicycle and Pedestrian Advisory Council, submission of the Housing Element annual progress report for 2020, and adoption of a Resolution of Local Support for the project. In addition, sponsors that have not yet adopted a resolution affirming compliance with California's Surplus Lands Act must now do so. This final requirement will primarily affect charter cities, which were not required to adopt such a resolution at the time of the OBAG 2 County Program adoption.

Please note that project sponsors have only *two weeks* to submit the final required materials to MTC. To meet this aggressive timeline, project sponsors are encouraged to submit their project into FMS in advance of being notified of MTC staff's funding recommendation. Additionally, sponsors should seek early Council or Board approvals of the resolutions required in the Local Agency Compliance Checklist.

The responses to project-specific questions, project maps, and the completed Local Agency Compliance Checklist must be submitted no later than Friday, May 21, 2021. Project data should be uploaded directly into FMS, linked above. Responses to project questions, project maps, and checklists should be sent to Mallory Atkinson at <u>matkinson@bayareametro.gov</u>.
Bay Area Partnership Board April 23, 2021 Page 3 of 13 Safe & Seamless Call for Nomination Letters Page 3

Program Approval

Staff anticipates presenting its recommended program of projects to the MTC Commission for consideration and approval at its June 2021 meeting.

Sincerely,

Theresa Romell

Theresa Romell Funding Policy and Programs

Attachments

Attachment 1: Safe & Seamless Mobility Quick-Strike Program – MTC Resolution No. 4202, Revised, Appendix A-11 Attachment 2: Project Information Form Attachment 3: Local Agency Compliance Checklist

TR:MA J:\PROJECT\Funding\T5-FAST\STP-CMAQ\FHIP - STP-Bump\Safe & Seamless Quick-Strike

Appendix A-11: Safe and Seamless Mobility Quick-Strike Program

The Safe and Seamless Mobility Quick-Strike program is a one-time, competitive grant program within the One Bay Area Grant program (OBAG 2) framework. Federal funding is available to support local and regional projects that can be implemented quickly to benefit communities responding and adapting to the COVID-19 environment.

Available funding includes a mix of Surface Transportation Block Grant Program (STP), Congestion Mitigation and Air Quality Improvement (CMAQ) and Federal Highway Infrastructure Program (FHIP) funds, with FHIP funds exchanged with STP/CMAQ funds to the extent possible to meet federal other funding deadlines and requirements. CMAQ funds will be used for eligible projects that demonstrate air quality benefits and implement Plan Bay Area's climate initiative goals and priorities.

Project Eligibility & Focus Areas

The program emphasizes bicycle/pedestrian safety and mobility, connections to transit, and projects that advance equitable mobility. Eligible project types include:

- Quick-build bike, pedestrian, and transit improvements; including bike share enhancements.
- Local safe and seamless mobility projects, including projects that advance equitable mobility; invest in bicycle/pedestrian safety; improve connections to transit; or implement seamless strategies within a corridor.
- In addition to capital projects, programs that support safe and seamless mobility or advance equitable mobility are also eligible (ex. safe routes to school/transit programs); a limited amount of funding, (up to \$200,000 per county) may also be directed towards countywide implementation of safe and seamless mobility planning and programming efforts).
- Other near-term implementation of strategies emerging from the Blue-Ribbon Transit Recovery Task Force and Partnership Board's Connected Mobility Subcommittee.

Fund commitments for specific focus areas include:

- One-quarter of the total program is targeted for bicycle/pedestrian safety (including local road safety).
- \$5 million is set aside to support early implementation efforts anticipated from the Blue-Ribbon Transit Recovery Task Force.

Evaluation Criteria

MTC staff will evaluate nominated projects against the following program criteria. Nominated projects should:

- Align with Connected Mobility Framework Values and Goals (see inset below)
- Be the direct result or outcome of a community engagement process
- Be within or directly connected to a Priority Development Area (PDA) or Transportation Priority Area (TPA) and/or serve a Community of Concern (CoC), Community Air Risk Evaluation (CARE) program area, or similar local designation. PDAs and TPAs may be existing or recently designated as part of the Plan Bay Area 2050 growth framework.
- Addresses transit connectivity gaps, especially in areas significantly impacted from the pandemic

- Demonstrate partnership among jurisdictions, transit agencies, and counties.
- Demonstrate ability to quickly deliver, and meet federal funding requirements, as funds must be obligated by September 30, 2022.

To ensure consistency with the implementation of county and regional plans and priorities, as well as encourage discussion and coordination in developing investment proposals, projects conominated by MTC and a CTA will be given extra consideration if meeting regional goals and priorities.

Below are the regional connected mobility values and goals guiding these investments:

Values	Goals
Think Regionally Act Locally	Be coordinated, interconnected, and contiguous
Provide Great Travel Choices	Provide choices that are better than driving alone, are viable and intuitive for all trips
Put the Traveler First	Ensure a dignified traveler experience, focusing on customer care and needs
Be Equitable & Inclusive	Address disparities and be transparent for all people and all trips
Be Sustainable	Strive for a healthy planet, people, and full-cost accounting

CONNECTED MOBILITY VALUES AND GOALS

Project Nominations

To address local needs throughout the region, and encourage community-based project investments, each County Transportation Agency (CTA) will act on MTC's behalf and submit project nominations for their county area. County targets have been provided as a guide, for each county (see table at right). However, final project selection by MTC will not necessarily adhere to these targets. Target amounts are based on the OBAG 2 county program distribution.

In addition to county submissions, MTC may consider projects that would be implemented regionwide or in more than one county. Where applicable, MTC staff will work with CTAs to coordinate on co-nominations for regional projects.

As the final program of projects must reflect regional or multicounty priorities, in addition to local priorities within each county, the final programming per county will not correspond exactly to nomination targets.

County Nomination Targets

(\$ millions, rounded)

	%
Alameda	19.9%
Contra Costa	14.6%
Marin	2.8%
Napa	2.1%
San Francisco	12.5%
San Mateo	8.4%
Santa Clara	27.0%
Solano	5.5%
Sonoma	7.2%
	100.0%

Note: Final project selection and fund programming will not correspond exactly to nomination targets.

To ensure each county is provided sufficient funding to have a meaningful community impact, each county's nomination target will be a minimum of \$1 million.

Project Selection Process

The prioritization process is designed to quickly distribute funds to competitive and impactful investments throughout the region.

- Letters of Interest: County Transportation Agencies (CTAs) submit Letters of Interest to nominate projects within their counties. In addition to basic project information (project description, sponsor, total cost, funding request), submittals should also describe how the project meets the program eligibility requirements and evaluation criteria, and how well the proposed project sponsor meets state and federal funding requirements.
- **Evaluation:** MTC staff evaluate CTA nominations as well as regional program considerations to develop a recommended program of projects. Program recommendations presented to Bay Area Partnership Board for review and discussion.
- **Project Applications:** MTC and CTA staff work with project sponsors to submit project applications with a detailed scope, delivery schedule, and funding plan.
- **Program Approval:** MTC Commission consideration and approval of projects and fund programming.

Programming Policies and Requirements

Unless otherwise noted within these guidelines, OBAG 2 General Programming Policies (see MTC Resolution No. 4202, Attachment A, pages 6-11), and Regional Project Funding Delivery Policy (MTC Resolution No. 3606) apply.

- **Project sponsors:** Eligible sponsors are those approved by Caltrans to receive FHWA federal-aid funds (including cities, counties, transit agencies, CTAs, and MTC). Sponsors must also have a demonstrated ability to meet timely use of funds deadlines and requirements (see Project Delivery and Monitoring, below).
- **Minimum Grant Size:** Project nominations should be consistent with OBAG 2 minimum grant size requirements per county (\$500,000 grant minimum for counties with population over 1 million, and \$250,000 minimum for all other counties). Final funding awards may deviate from grant minimums per county, should one or more grant awards span multiple counties or regionwide.

Additionally, deviations from the OBAG 2 minimum grant size requirements for project nominations may be considered on a project-by-project basis. However, grant awards must be at least \$100,000.

- Local Match: Toll credits may be requested in lieu of non-federal cash match.
- **Supplanting of Funds Prohibited**: Supplanting of existing funds on fully-funded projects is prohibited, as the program is intended to infuse transportation investment into communities responding and adapting to the COVID-19 environment. If funds are



requested to address a funding shortfall on a project due to reduced local revenues, CTAs must demonstrate why the project should be a priority for regional funding, if it was not the highest priority for available local funding. In their nomination, CTAs should describe how the county and local jurisdictions determined which projects are prioritized for reduced local revenues.

- Project Phases: The Environmental (ENV), Plans, Specifications and Estimates (PS&E), Preliminary Engineering (PE) and Right Of Way (ROW) phases are eligible for capital projects as long as the construction (CON) phase of the project is delivered and funds obligated by September 30, 2022.
- **Project Delivery and Monitoring:** Project sponsors must have a record of consistently meeting state and federal timely use of funds deadlines and requirements, or demonstrate/identify revised/new internal processes to ensure they will meet funding deadlines and requirements moving forward at the time of project nomination. In addition to the provisions of the Regional Project Funding Delivery Policy (MTC Resolution No. 3606), the following specific funding deadlines/requirements apply:
 - Funds must be obligated (authorized in a federal E-76, or transferred to FTA) no later than September 30, 2022.
 - Funds must be encumbered or awarded in a contract within 6 months of federal obligation.
 - Funds must be invoiced against within 3 months of encumbrance/award and invoiced against and receive a federal reimbursement quarterly thereafter.
 - If there could be complications with invoicing against the construction phase within 9 months of federal obligation, then the sponsor should consider including Construction Engineering (CE) in the federal obligation so that eligible costs may be invoiced in order to meet the invoicing deadline.
 - Project sponsor must meet all other timely use of funds deadlines and requirements, for all other state and federal transportation funds received by the agency, during the duration of project implementation (such as, but not limited to, project award, federal invoicing, and project reporting).
 - To help ensure compliance with state and federal invoicing requirements, as part of the application submittal, the Finance/Accounting Manager/Director for the agency receiving the funds must provide written documentation on the agency's internal process and procedures for complying with FHWA federal-aid timely use of funds requirements, especially with regards to meeting federal invoicing requirements.
 - CTAs nominating successful projects must monitor the project sponsors within their respective county in meeting the timely use of funds deadline requirements in MTC Resolution No. 3606 and report quarterly to MTC on the agency's status in meeting regional, state, and federal timely use of funds deadlines and requirements.

• Additional Requirements Apply:

 Project sponsor must comply with MTC's Complete Street Policy and submit a Complete Streets Checklist for the project.

- Project sponsor must adopt a Resolution of Local Support prior to adding the project into the Transportation Improvement Program (TIP).
- Project sponsor must satisfy the OBAG 2 housing policy requirements have a certified Housing Element, submit the Annual Progress Report for the Housing Element, and have adopted a resolution affirming compliance with the California Surplus Lands Act.
- CTAs must make each project's Complete Streets Checklist available for review by the appropriate Bicycle and Pedestrian Advisory Committee (BPAC) prior to MTC Commission approval of projects and fund programming. Documentation this has occurred must be included with the project application.

	Basic Project Informatio	n	
Project Name:	Project name		
Project Sponsor:	Project sponsor		
Sponsor Contact	Contact name		
Information:	Contact phone		
	Contact email		
Project Location:	Project location		
Brief Project Description: Please limit to 100 word maximum	Project description		
	Program Focus Areas & Evaluatio	on Criteria	
Program Focus Areas:	Identify the type of project to be completed. Select all that apply:	Improved connections to transit	
	 Quick-build bicycle and/or pedestrian improvement 	Programming to support safe and	
	Quick-build transit improvement	seamless mobility	
	Bike share enhancement	CTA planning or programming to	
	 Bicycle and/or pedestrian safety 	support safe and seamless mobility	
	improvement	Other project type consistent with	
	Local safe & seamless mobility	the Blue-Ribbon Transit Recovery	
	improvement	Task Force or the Partnership	
	Safe & seamless mobility improvement in a corridor	Board's Connected Mobility Framework	
Priority Planning Areas:	Identify the location of the project to be con prioritized geographies. Select all that apply		
	Priority Development Area (PDA)		
	□ Transit Priority Area (TPA)		
	Community of Concern (COC)		
		ammunit.	
	Community Air Risk Evaluation (CARE) co		
	Other project area – for a project that is not located in the above areas, please describe how this project advances safe and seamless mobility for populations that are low-income or that have been historically-disadvantaged		
	Describe how project located outside of a PDA, TPA, COC, or CARE advances equitable mobility		
Connected Mobility	Describe how the project aligns with the values & goals of the Partnership Board's Connected Mobility Framework:		
Framework: Please limit to 200 word maximum	Project alignment with Connected Mobility F	ramework	
Community	Describe the community outreach that has I	been completed related to this project,	
Engagement & Planning Processes:	and also reference any local or regional plan		

Please limit to 200 word maximum	community-based transportation plan, station area or specific plans, bicycle or pedestrian plans, etc.):	
	Community engagement & planning processes	
	Describe the outreach that has been completed with transit operators to ensure the project does not conflict with existing or planned transit service:	
	Coordination and outreach with transit operators	
Transit Connectivity: Please limit to 100	Describe how this project addresses a gap in transit connectivity, particularly in areas significantly impacted from pandemic: <i>Transit connectivity</i>	
word maximum		
Project Partnerships:	Describe any partnerships in place for this project (jurisdictions, CTAs, transit agencies, community groups, etc.):	
Indicate if project is anticipated to be co-	Project partnerships	
nominated by MTC	\Box Project is co-nominated by MTC.	
Project Readiness: Please limit to 100 word maximum	Describe the readiness of the project, including right-of-way impacts, the type of environmental document/clearance required, and consistency with Plan Bay Area 2040.	
	Project readiness, right-of-way, environment	
	If the project touches Caltrans right-of-way, include the status and timeline of the necessary Caltrans approvals and documents, the status and timeline of Caltrans requirements, and approvals such as planning documents (PSR or equivalent) environmental approval, encroachment permit. Also, include a statement of Caltrans' level of support for the project.	
	Caltrans approvals status and timeline; level of support	
Deliverability: Please limit to 200	Describe the project's timeline and status, as well as the sponsor's ability to meet the September 30, 2022 obligation deadline.	
word maximum, or include as	Project timeline, status, and obligation deadline	
attachment	Identify any known risks to the project schedule, and how the CTA and project sponsor will mitigate and respond to those risks.	
	Project risks and mitigation strategies	
	Describe the sponsor's ability to meet state and federal requirements after fund obligation. Include confirmation of ability to award within 6 months of obligation and a discussion of the agency's delivery history for federal funds; note any documented internal processes in place to ensure full adherence to invoicing and timely use of funds policies, and describe CTA's role in delivery and monitoring.	
	Project sponsor ability to meet delivery and monitoring requirements	

Fund exchange:	 Project involves a local fund exchange. If yes, please describe. Clarify which project will receive federal funds directly, which project will receive non-federal funds, and the timing of both projects. Fund exchange description
Grant minimum: Project does not meet the minimum grant size requirement. If yes, describe an exception to this requirement should be considered. <i>Exception request to minimum grant size</i> 	
Supplanting of existing funds:	□ Grant funds would supplant existing funds previously programmed to the project. If yes, describe why an exception to this requirement should be considered. If funds are requested to address a funding shortfall on a project due to reduced local revenues, describe how the county and/or local jurisdiction(s) determined which projects should be prioritized for the use of the remaining local revenues. Response should demonstrate why the project should be prioritized for regional funding if it was not the highest priority for local funding. <i>Exception request to fund supplanting requirement</i>
Toll credit request:	Toll credits are requested; no local match is provided. Notes on toll credit request, optional

Project Cost & Funding

Project Cost & Status:

		Fund Source by Phase			Project Status by Phase	
Phase	Total Cost	Safe & Seamless (Grant Request)	Other Funds		% Complete	
Planning/ Conceptual	\$	\$	\$	Fund source; notes	% complete	
Environmental Studies (PA&ED)	\$	\$	\$	Fund source; notes	% complete	
Design Engineering (PS&E)	\$	\$	\$ Fund source; notes % complete		% complete	
Right-of-way	\$	\$	\$	Fund source; notes	% complete	
Construction	\$	\$	\$ Fund source; notes		% complete	
Total	\$	\$	\$			

Project Investment by Mode:

Mode	Share of project investment	
Auto	%	
Transit	%	
Bicycle/Pedestrian	%	
Other	%	
Total	100%	

Local Compliance Checklist				
Jurisdiction:	Local jurisdiction			
MTC's Complete	□ Jurisdiction complies with MTC's Complete Street Policy, either by:			
Streets Policy:	 Adopting a Complete Streets resolution incorporating MTC's nine required complete streets elements; or 			
	2. Adopting a significant revision to the General Plan Circulation Element after January 1, 2010 that complies with the California Complete Streets Act of 2008.			
	□ Complete Streets checklist for project was reviewed by the appropriate Bicycle and Pedestrian Advisory Committee (BPAC) prior to May 21, 2021. Provide the date of BPAC review, describe any comments that were received, and the jurisdiction's response to feedback.			
	Date of BPAC review & discussion of BPAC comments			
Resolution of Local Support:	□ Jurisdiction has adopted a Resolution of Local Support by May 21, 2021.			
Housing Element:	□ Jurisdiction's Housing Element has been certified by California Department of Housing and Community Development (HCD). Note: all Bay Area jurisdictions satisfied this requirement prior to the One Bay Area Grant (OBAG 2) County Program adoption.			
□ Jurisdiction's Housing Element annual progress report for calendar ye been submitted to HCD.				
Surplus Lands Act:	□ Jurisdiction has met MTC's Surplus Land Requirements prior to May 21, 2021, through the adoption of a resolution demonstrating compliance with the State's Surplus Land Act.			
	Note for Charter Cities: At the time of the adoption of the OBAG 2 County Program, this requirement applied only to general law cities and counties. However, as a final court decision has now been rendered confirming that the Act does apply to charter cities, funding eligibility through the Safe & Seamless Mobility Quick-Strike program is contingent upon the adoption, by <i>all</i> cities and counties, of a resolution affirming the jurisdiction's compliance with the Surplus Lands Act.			

This checklist was approved for submission by:

Signature

Date Date

Signature

Name (print)

City Manager/Administrator or designee

Safe & Seamless Mobility Quick-Strike Grant Program

List of Project Nominations (Sorted by County, Sponsor, Project)

County	Sponsor	Project Title	Funds Requested
Alameda	AC Transit	Quick Builds Transit Lanes	\$954,000
Alameda	AC Transit	Tempo Quick Build Transit Lane Delineation	\$300,000
Alameda	ACTC	Alameda County Safe Routes to Schools	\$1,500,000
Alameda	Alameda County	Anita Avenue Safe and Accessible Route to School and Transit	\$2,000,000
Alameda	Dublin	Dublin Safe Routes to School - Safety and Access Improvements	\$2,000,000
Alameda	Fremont	Fremont Boulevard/Grimmer Boulevard Protected Intersection	\$1,415,000
Alameda	Fremont	Fremont Boulevard/Walnut Avenue Protected Intersection Project	\$1,271,000
Alameda	LAVTA	Passenger Facilities Enhancements	\$2,000,000
Alameda	MTC / ACTC	I-580 Westbound High Occupancy Vehicle Lane Extension	\$1,500,000
Alameda	Oakland	14th Street Complete Streets Project	\$1,000,000
Alameda	Oakland	East Bay Greenway Segment II	\$1,000,000
Alameda	TBD	Various Youth and Adult Bicycle Promotion & Education Programs	\$160,000
Contra Costa	BART	Bicycle, Pedestrian, and ADA Imps. at Pittsburg/Bay Point BART	\$1,510,000
Contra Costa	BATA	RSR Forward: I-580 WB Open Road Tolling & HOV Lane Extension	\$2,000,000
Contra Costa	Concord	East Downtown Concord PDA Access & Safe Routes to Transit	\$2,164,000
Contra Costa	Danville	Diablo Road Trail	\$2,000,000
Contra Costa	Lafayette / BART	Lafayette Town Center Pathway and BART Bike Station Project	\$1,825,000
Contra Costa	Pinole	Pedestrian and Bicycle Safety Imps. at Appian Way and Marlesta Rd	\$350,000
Contra Costa	Pleasant Hill	Contra Costa Blvd Complete Streets (Harriet Drive to Viking Drive)	\$4,792,000
Contra Costa	Richmond	13th Street Complete Streets Project	\$2,821,000
Contra Costa	Richmond	San Pablo Ave Complete Streets Phase 2	\$6,000,000
Marin	Corte Madera	Casa Buena Drive Complete Streets Regional Improvements	\$600,000
Marin	Larkspur	Doherty Drive Multi-Use Pathway at Redwood High School - Phase 2	\$500,000
Marin	Marin Transit	Marin County Bus Stop Improvements	\$1,200,000
Marin	Mill Valley	East Blithedale Pathway to Transit	\$1,000,000
Marin	SMART	SMART Pathway - San Rafael (McInnis Pkwy to Smith Ranch Rd)	\$2,158,026
Marin	ТАМ	Marin County Street Smarts Bicycle and Pedestrian Safety Program	\$250,000
Napa	MTC	Napa Forward SR 29 Safety and Operational Improvements	\$11,000,000
Napa	NVTA	Napa Valley Safe Routes to School	\$100,000
San Francisco	BART	Embarcadero Station Platform Elevator Capacity & Redundancy	\$3,144,302
San Francisco	SFCTA	Downtown San Francisco Congestion Pricing Study	\$200,000
San Francisco	SFMTA	Folsom Streetscape Project	\$5,000,000
San Francisco	SFMTA	Safe Routes to School Non-Infrastructure Program	\$2,100,000
San Mateo	Brisbane	Bayshore Blvd/Van Waters and Rodgers Rd Bus Stop Improvements	\$360,000
San Mateo	Burlingame	Carmelita Ave and Paloma Ave Traffic Calming and Bicycle Safe Routes	\$850,000

Bay Area Partnership Board April 23, 2021 Page 2 of 2

County	Sponsor	Project Title	Funds Requested
San Mateo	Burlingame	City-Wide Pedestrian Safe Routes and Mobility Improvements	\$200,000
San Mateo	C/CAG	Planning and Programming of safe and seamless mobility projects	\$200,000
San Mateo	Daly City	Southgate Avenue and School Street Safety Improvements Project	\$350,000
San Mateo	Millbrae	Millbrae - Citywide Virtual Mobility Detection	\$223,591
San Mateo	Millbrae	Park Blvd, Santa Teresa Way, San Anselmo Ave Traffic Calming Quick Build	\$347,250
San Mateo	Redwood City	Roosevelt Avenue Quick-build Traffic Calming Project	\$755,000
San Mateo	San Bruno	San Bruno Citywide Bicycle Route Network	\$615,000
San Mateo	San Bruno	Transit Corridor Pedestrian Connection Phase 4	\$385,000
San Mateo	San Mateo	Delaware Street Quick-Build Bike Improvements*	\$610,007
San Mateo	San Mateo County	Coleman-Ringwood Pedestrian and Bicycle SRTS Improvements	\$808,562
San Mateo	San Mateo County	San Mateo County Broadmoor SRTS Pedestrian Safety & Mobility Imps	\$1,418,412
San Mateo	South San Francisco	East of 101 Transit Expansion Project	\$480,000
San Mateo	South San Francisco	El Camino Real Grand Boulevard Initiative Phase III	\$2,120,000
Santa Clara	Los Altos	St. Joseph Avenue Mobility Connector	\$950,000
Santa Clara	Los Gatos	Los Gatos Creek Trail to Highway 9 Trailhead Connector	\$5,097,400
Santa Clara	Mountain View	Stierlin Road Bicycle and Pedestrian Improvements	\$4,007,000
Santa Clara	San Jose	Bascom Avenue Protected Bike Lanes & Complete Street	\$690,000
Santa Clara	San Jose	En Movimiento Quick Build Network for East San Jose	\$1,325,000
Santa Clara	San Jose	Julian Street & McKee Road Vision Zero Complete Street	\$705,000
Santa Clara	San Jose	San Jose - Downtown Bikeways	\$4,025,000
Santa Clara	Saratoga	Blue Hills Elementary Pedestrian Crossing at UPRR	\$1,800,000
Santa Clara	Sunnyvale	Sunnyvale Bicycle, Pedestrian and SRTS Safety Improvements	\$1,900,000
Santa Clara	Sunnyvale	Sunnyvale Saratoga Class IIB Buffered Bicycle Lanes	\$2,200,000
Santa Clara	VTA	Electronic Locker Upgrade and Replacement	\$1,940,060
Santa Clara	VTA	Evaluating on-demand shuttle strategies for improved transit access	\$200,000
Solano	BAIFA	FasTrak START Means-Based Managed Lanes Pilot Evaluation	\$900,000
Solano	Fairfield	Fairfield/Vacaville Hannigan Station Capacity Improvements	\$1,000,000
Solano	STA	STA Mobility Planning	\$200,000
Solano	Vallejo	Bay Trail/Vine Trail Gap Closure Segment	\$1,800,000
Sonoma	Cotati	Downtown and Civic Center Connectivity and Safety Project	\$1,250,000
Sonoma	Healdsburg	Healdsburg Bike Share	\$250,000
Sonoma	Rohnert Park	2022 Pedestrian and Bicycle Safety Improvements in Rohnert Park	\$522,000
Sonoma	Santa Rosa	Santa Rosa Transit Mall Roadbed Rehabilitation	\$868,230
Sonoma	SCTA	Countywide Active Transportation Plan	\$200,000
Sonoma	Sebastopol	SR 116 and Bodega Ave Pedestrian Access and Mobility Enhancements	\$476,310
Sonoma	SMART	SMART Pathway - Payran to Lakeville in Petaluma	\$806,000
	-	Total Funding Request:	¢109 640 150

Total Funding Request: \$108,649,150

Applications Received: 71

Bay Area Partnership Board April 23, 2021 Page 1 of 2

Safe & Seamless Mobility Quick-Strike Grant Program

Initial Staff Recommendation (Sorted by County, Sponsor, Project)

Recommendations are pending further review. Final recommended program of projects and specific grant amounts subject to change.

Color Key:

Initial Staff Recommended Program of Projects

			Total available = \$49,400,000	
County	Sponsor	Project Title	Funds Requested	Cumulative Funding Recommendation
Alameda Cou	nty		Initial Fund Target:	\$9.8M
Alameda	AC Transit	Quick Builds Transit Lanes	\$954,000	\$954,000
Alameda	AC Transit	Tempo Quick Build Transit Lane Delineation	\$300,000	\$1,254,000
Alameda	ACTC	Alameda County Safe Routes to Schools	\$1,500,000	\$2,754,000
Alameda	Alameda County	Anita Avenue Safe and Accessible Route to School and Transit	\$2,000,000	
Alameda	Dublin	Dublin Safe Routes to School - Safety and Access Improvements	\$2,000,000	
Alameda	Fremont	Fremont Boulevard/Grimmer Boulevard Protected Intersection	\$1,415,000	\$4,169,000
Alameda	Fremont	Fremont Boulevard/Walnut Avenue Protected Intersection Project	\$1,271,000	\$5,440,000
Alameda	LAVTA	Passenger Facilities Enhancements	\$2,000,000	\$7,440,000
Alameda	MTC/ACTC	I-580 Westbound High Occupancy Vehicle Lane Extension	\$1,500,000	
Alameda	Oakland	14th Street Complete Streets Project	\$1,000,000	\$8,440,000
Alameda	Oakland	East Bay Greenway Segment II	\$1,000,000	\$9,440,000
Alameda	TBD	Various Youth and Adult Bicycle Promotion & Education Programs*	\$160,000	\$9,600,000
Contra Costa	County		Initial Fund Target:	\$7.2M
Contra Costa	BART	Bicycle, Pedestrian, and ADA Imps. at Pittsburg/Bay Point BART	\$1,510,000	\$1,510,000
Contra Costa	BATA	RSR Forward: I-580 WB Open Road Tolling & HOV Lane Extension**	\$2,000,000	\$3,510,000
Contra Costa	Concord	East Downtown Concord PDA Access & Safe Routes to Transit	\$2,164,000	\$5,674,000
Contra Costa	Danville	Diablo Road Trail	\$2,000,000	
Contra Costa	Lafayette / BART	Lafayette Town Center Pathway and BART Bike Station Project	\$1,825,000	\$7,499,000
Contra Costa	Pinole	Pedestrian and Bicycle Safety Imps. at Appian Way and Marlesta Rd	\$350,000	
Contra Costa	Pleasant Hill	Contra Costa Blvd Complete Streets (Harriet Drive to Viking Drive)	\$4,792,000	
Contra Costa	Richmond	13th Street Complete Streets Project	\$2,821,000	
Contra Costa	Richmond	San Pablo Ave Complete Streets Phase 2	\$6,000,000	
Marin County			Initial Fund Target:	\$1.4M
Marin	Corte Madera	Casa Buena Drive Complete Streets Regional Improvements	\$600,000	
Marin	Larkspur	Doherty Drive Multi-Use Pathway at Redwood High School - Phase 2	\$500,000	
Marin	Marin Transit	Marin County Bus Stop Improvements	\$1,200,000	\$1,200,000
Marin	Mill Valley	East Blithedale Pathway to Transit	\$1,000,000	
Marin	SMART	SMART Pathway - San Rafael (McInnis Pkwy to Smith Ranch Rd)	\$2,158,026	
Marin	ТАМ	Marin County Street Smarts Bicycle and Pedestrian Safety Program	\$250,000	
Napa County			Initial Fund Target:	\$1.0M
Napa	MTC	Napa Forward SR 29 Safety and Operational Imps. (\$1.1M partial funding)***	\$11,000,000	\$1,100,000
Napa	NVTA	Napa Valley Safe Routes to School	\$100,000	\$1,200,000
San Francisco	-		Initial Fund Target:	\$6.2M
San Francisco		Embarcadero Station Platform Elevator Capacity & Redundancy	\$3,144,302	
San Francisco	-	Downtown San Francisco Congestion Pricing Study	\$200,000	\$200,000
San Francisco		Folsom Streetscape Project	\$5,000,000	\$5,200,000
San Francisco		Safe Routes to School Non-Infrastructure Program (\$1.8M partial funding)***	\$2,100,000	\$7,002,710
San Mateo Co			Initial Fund Target:	\$4.1M
San Mateo	Brisbane	Bayshore Blvd/Van Waters and Rodgers Rd Bus Stop Improvements	\$360,000	
San Mateo	Burlingame	Carmelita Ave and Paloma Ave Traffic Calming and Bicycle Safe Routes	\$850,000	
San Mateo	Burlingame	City-Wide Pedestrian Safe Routes and Mobility Improvements	\$200,000	
San Mateo	C/CAG	Planning and Programming of safe and seamless mobility projects	\$200,000	\$200,000

Bay Area Partnership Board April 23, 2021 Page 2 of 2

County	Sponsor	Project Title	Funds Requested	Cumulative Funding Recommendation
San Mateo	Daly City	Southgate Avenue and School Street Safety Improvements Project	\$350,000	
San Mateo	Millbrae	Millbrae - Citywide Virtual Mobility Detection	\$223,591	
San Mateo	Millbrae	Park Blvd, Santa Teresa Way, San Anselmo Ave Traffic Calming Quick Build	\$347,250	
San Mateo	Redwood City	Roosevelt Avenue Quick-build Traffic Calming Project	\$755,000	\$955,000
San Mateo	San Bruno	San Bruno Citywide Bicycle Route Network	\$615,000	
San Mateo	San Bruno	Transit Corridor Pedestrian Connection Phase 4	\$385,000	\$1,340,000
San Mateo	San Mateo	Delaware Street Quick-Build Bike Improvements****	\$610,007	
San Mateo	San Mateo County	Coleman-Ringwood Pedestrian and Bicycle SRTS Improvements	\$808,562	
San Mateo	San Mateo County	San Mateo County Broadmoor SRTS Pedestrian Safety & Mobility Imps	\$1,418,412	
San Mateo	South San Francisco	East of 101 Transit Expansion Project	\$480,000	\$1,820,000
San Mateo	South San Francisco	El Camino Real Grand Boulevard Initiative Phase III	\$2,120,000	\$3,460,000
Santa Clara C	ounty	·	nitial Fund Target:	\$13.3M
Santa Clara	Los Altos	St. Joseph Avenue Mobility Connector	\$950,000	
Santa Clara	Los Gatos	Los Gatos Creek Trail to Highway 9 Trailhead Connector	\$5,097,400	
Santa Clara	Mountain View	Stierlin Road Bicycle and Pedestrian Improvements	\$4,007,000	\$4,007,000
Santa Clara	San Jose	Bascom Avenue Protected Bike Lanes & Complete Street	\$690,000	\$4,697,000
Santa Clara	San Jose	En Movimiento Quick Build Network for East San Jose	\$1,325,000	\$6,022,000
Santa Clara	San Jose	Julian Street & McKee Road Vision Zero Complete Street	\$705,000	\$6,727,000
Santa Clara	San Jose	San Jose - Downtown Bikeways	\$4,025,000	\$10,752,000
Santa Clara	Saratoga	Blue Hills Elementary Pedestrian Crossing at UPRR	\$1,800,000	
Santa Clara	Sunnyvale	Sunnyvale Bicycle, Pedestrian and SRTS Safety Improvements	\$1,900,000	
Santa Clara	Sunnyvale	Sunnyvale Saratoga Class IIB Buffered Bicycle Lanes	\$2,200,000	
Santa Clara	VTA	Electronic Locker Upgrade and Replacement	\$1,940,060	\$12,692,060
Santa Clara	VTA	Evaluating on-demand shuttle strategies for improved transit access	\$200,000	\$12,892,060
Solano Count	у	·	nitial Fund Target:	\$2.7M
Solano	BAIFA	FasTrak START Means-Based Managed Lanes Pilot Evaluation	\$900,000	
Solano	Fairfield	Fairfield/Vacaville Hannigan Station Capacity Improvements	\$1,000,000	\$1,000,000
Solano	STA	STA Mobility Planning	\$200,000	\$1,200,000
Solano	Vallejo	Bay Trail/Vine Trail Gap Closure Segment	\$1,800,000	\$3,000,000
Sonoma Cour	nty		nitial Fund Target:	\$3.6M
Sonoma	Cotati	Downtown and Civic Center Connectivity and Safety Project (\$0.9M partial funding)***	\$1,250,000	\$900,000
Sonoma	Healdsburg	Healdsburg Bike Share	\$250,000	\$1,150,000
Sonoma	Rohnert Park	2022 Pedestrian and Bicycle Safety Improvements in Rohnert Park	\$522,000	\$1,672,000
Sonoma	Santa Rosa	Santa Rosa Transit Mall Roadbed Rehabilitation	\$868,230	\$2,540,230
Sonoma	SCTA	Countywide Active Transportation Plan	\$200,000	\$2,740,230
Sonoma	Sebastopol	SR 116 and Bodega Ave Pedestrian Access and Mobility Enhancements	\$476,310	
Sonoma	SMART	SMART Pathway - Payran to Lakeville in Petaluma	\$806,000	\$3,546,230

Total Funding Request: \$108,649,150 Cumulative Funding Recommendation: \$49,400,000

*Recommendation contingent upon identification of an eligible project sponsor.

**Recommendation pending further review.

***Project recommended to receive partial funding.

****Project recommended to receive Regional ATP funds, pending Commission approval.

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:May 20, 2021TO:Alameda County Transportation CommissionFROM:Vivek Bhat, Director of Programming and Project Controls
Jacki Taylor, Senior Program AnalystSUBJECT:Approve Alameda County 2021 Mid-Cycle Augmentation of the 2020
State Transportation Improvement Program

Recommendation

It is recommended that the Commission approve the following actions related to the midcycle augmentation of the Alameda County 2020 State Transportation Improvement Program (2021 Mid-Cycle STIP):

- Approve Resolution 21-010 (Attachment A) regarding the approval of a 2021Mid-Cycle STIP program of projects, which identifies a total of \$6.308 million of new 2021Mid-Cycle STIP funding;
- 2. Approve Resolution 21-011 (Attachment B), the required resolution of local support for new STIP projects; and
- 3. Authorize the Executive Director or designee to enter into any necessary agreements.

Summary

The State Transportation Improvement Program (STIP) is a multi-year capital improvement program of transportation projects on and off the State Highway System, funded with revenues from the State Highway Account and other funding sources administered by the California Transportation Commission (CTC), including Senate Bill 1 (SB 1). The 2020 STIP was approved by the CTC in Spring 2020 and covers Fiscal Years (FYs) 2020-21 through 2024-25. Through the approval of the federal Coronavirus Response and Relief Supplemental Appropriations Act of 2021 (CRRSAA), \$912 million of COVID-19 relief funds were apportioned to the State. From this amount, the CTC authorized \$243 million to be distributed through the STIP as a mid-cycle augmentation to the 2020 STIP (2021 Mid-cycle STIP Program), as follows: seventy-five (75%) percent through the Regional Improvement Program (\$182 million) and twenty-five percent (25%) through the Interregional Improvement Program (\$61 million).

Alameda County's share of the State's 2021 Mid-cycle STIP Fund Estimate is \$6.308 million of new programming capacity for projects. The Alameda CTC is to adopt and forward a program of STIP projects to the Metropolitan Transportation Commission (MTC) for inclusion in MTC's 2021 Mid-Cycle Regional STIP program (2021Mid-Cycle RTIP). MTC approves the region's RTIP and submits it to the CTC for inclusion in the STIP.

Staff is recommending Commission approval of the Alameda County 2021 Mid-cycle STIP Program (Attachment A). Due to the short programming timeline, draft 2021 Mid-Cycle STIP project lists were due to MTC by April 19, 2021. A final draft project list and supporting documentation is due to MTC by May 1, 2021, with concurrent approval of the proposed project lists by Alameda CTC, MTC and CTC all scheduled during the month of May. The MTC and CTC approvals will be contingent upon Alameda CTC's action.

Background

The STIP is a multi-year capital improvement program of transportation projects on and off the State Highway System that is administered by the CTC and funded with revenues from the State Highway Account and other State and federal funding sources, including SB 1. The STIP is composed of two sub-elements with 75% of the STIP funds reserved for the Regional Transportation Improvement Program (RTIP) and 25% for the Interregional Transportation Improvement Program (ITIP).

Senate Bill 45 (SB 45) was signed into law in 1996 and had significant impacts on the regional transportation planning and programming process. The statute delegated major funding decisions to the local level and allows the Congestion Management Agencies/County Transportation Agencies (CMAs/CTAs) to have a more active role in selecting and programming transportation projects. SB 45 changed the transportation funding structure and modified the transportation programming cycle, program components, and expenditure priorities.

For each STIP cycle, Alameda CTC adopts and forwards a program of STIP projects to MTC. As the Regional Transportation Planning Agency (RTPA) for the nine-county Bay Area, MTC is responsible for developing the regional priorities for the RTIP. MTC approves the region's RTIP and submits it to the CTC for inclusion in the STIP. Caltrans is responsible for developing the ITIP.

The 2020 STIP was approved by the CTC in Spring 2020 and covers Fiscal Years (FYs) 2020-21 through 2024-25. On March 24, 2021, the CTC adopted the 2021 Mid-Cycle STIP schedule and Share Distribution and released draft Mid-Cycle STIP Guidelines.

2021 Mid-Cycle STIP Program

Through the approval of the 2021 federal CRRSAA, \$912 million of COVID-19 relief funds were apportioned to the State. From this amount, the CTC authorized \$243 million to be distributed through the STIP as a mid-cycle augmentation to the 2020 STIP (2021 Mid-Cycle STIP), as follows: seventy-five (75%) percent through the Regional Improvement Program (\$182 million) and twenty-five percent (25%) through the Interregional Improvement

Program (\$61 million). Alameda County's share of the State's 2021 Mid-Cycle STIP Fund Estimate is \$6.308 million of new programming capacity for projects. MTC staff released draft 2021 Mid-Cycle RTIP Guidelines and Fund Estimate on April 8, 2021. Unlike regular STIP cycles, the mid-cycle funding can be programmed in any Fiscal Year within the existing 2020 STIP period, starting with FY 2021-22.

I-680 Southbound Express Lanes from SR-84 to Alcosta Boulevard Project

The I-680 Express Lanes Gap Closure (SR-84 to Alcosta Boulevard) project (Project) is currently in the Design phase and scheduled to go into construction by fall 2021. The delivery of the Project is being coordinated with a Caltrans pavement rehabilitation SHOPP project within the same limits, resulting in delivery efficiencies and cost savings. In February 2021, the Commission approved an amendment to the 2020 STIP to program up to \$40 million of future STIP funding to the Project for CTC's consideration in June 2021.

The availability of CRRSAA funding through the STIP provides the opportunity to program \$5.793 million of STIP funds to the project in FY 2021-22. Additional STIP funding will be programmed in June if additional unused STIP capacity is identified by either MTC or CTC. The project has a mix of Competitive and Formula-based Senate Bill 1 Local Partnership Program (SB-1 LPP) funding totaling \$37 million that requires a full funding plan in June in order for the funds to remain programmed to the project. Additionally, MTC has committed \$90 million of regional funds. In July 2021, staff will update the Commission regarding the availability, amount and timing of the STIP funds and other related Project actions.

New projects proposed for 2021 Mid-Cycle Program

The 2021 Mid-Cycle Fund Estimate for Alameda County includes \$315,000 of for ongoing STIP Planning, Programming and Monitoring (PPM) activities. Staff is also recommending the remaining \$200,000 to be programmed to four Alameda CTC-implemented projects currently in the Environmental phase, as follows:

		Amount x \$1,000			
Index	Project	Total Cost	ldentified Funding	TBD	STIP amount recommended
1	Oakland/Alameda Access (I-880 Broadway-Jackson)*	129,900	79,991	49,909	50
2	I-80 Ashby Interchange Improvements*	117,174	52,000	65,174	50
3	I-880 Interchanges (Whipple Road/Industrial Parkway Southwest & Ind. Parkway West) Improvements*	220,000	104,000	116,000	50

4	I-880 Interchanges Winton Avenue and A Street Improvements	114,308	5,308	109,000	50
	Total	\$581,382	\$241,299	\$340,083	\$200

* Named project in 2014 MBB Transportation Expenditure Plan

These four projects included in the 2014 Transportation Expenditure Plan have a collective need of approximately \$340 million. Alameda CTC's strategic project delivery vision includes leveraging local sales tax investments with external grant opportunities. In order to allow the projects to compete for future federal funding opportunities, staff is working towards obtaining federal National Environmental Policy Act (NEPA) clearance for the above four projects. Caltrans has indicated that in order to perform its review and approval of the NEPA documents, the projects first need to have federal funds programmed. The 2021 Mid-Cycle STIP provides an opportunity to include federal funds on these projects.

Staff recommends the Commission approve the following actions related to the 2021 Mid-Cycle STIP:

- 1. Approve Resolution 21-010 regarding the approval of a 2021Mid-Cycle STIP program of projects, which identifies a total of \$6.308 million of new 2021Mid-Cycle STIP funding;
- 2. Approve Resolution 21-011, the required resolution of local support for new STIP projects; and
- 3. Authorize the Executive Director or designee to enter into any necessary agreements.

Next Steps

Due to the condensed programming schedule for the 2021 Mid-Cycle STIP Program, MTC and CTC approvals of the Alameda County 2021 Mid-Cycle STIP project list are also scheduled to occur in May 2021, contingent upon the Commission's action.

Fiscal Impact: There is no fiscal impact associated with the requested item.

Attachments:

- A. Resolution 21-010, Alameda County 2021 Mid-Cycle STIP Program
- B. Resolution 21-011, Resolution of Local Support for new STIP projects



Commission Chair Mayor Pauline Russo Cutter City of San Leandro

Commission Vice Chair Councilmember John Bauters City of Emeryville

AC Transit Board President Elsa Ortiz

Alameda County

Supervisor David Haubert, District 1 Supervisor Richard Valle, District 2 Supervisor Wilma Chan, District 3 Supervisor Nate Miley, District 4 Supervisor Keith Carson, District 5

BART Vice President Rebecca Saltzman

City of Alameda Mayor Marilyn Ezzy Ashcraft

City of Albany Councilmember Rochelle Nason

City of Berkeley Councilmember Lori Droste

City of Dublin Melissa Hernandez, Mayor

City of Fremont Mayor Lily Mei

City of Hayward Mayor Barbara Halliday

City of Livermore Mayor Bob Woerner

City of Newark Councilmember Luis Freitas

City of Oakland Councilmember At-Large Rebecca Kaplan Councilmember Sheng Thao

City of Piedmont Councilmember Jen Cavenaugh

City of Pleasanton Mayor Karla Brown

City of Union City Mayor Carol Dutra-Vernaci

Executive Director Tess Lengyel

6.11A

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

www.AlamedaCTC.org

ALAMEDA COUNTY TRANSPORTATION COMMISSION RESOLUTION 21-010

Approval of the Alameda County 2021 Mid-Cycle Augmentation of the 2020 State Transportation Improvement (STIP) Program

WHEREAS, SB 45 (Chapter 622, Statutes 1997) substantially revised the process for estimating the amount of state and federal funds available for transportation projects in the state and for appropriating and allocating the available funds to these projects; and

WHEREAS, as part of this process, the Alameda County Transportation Commission (Alameda CTC) is responsible for programming projects eligible for Regional Improvement Program (RIP) funds, pursuant to Government Code Section 14527 (a), for inclusion in the Regional Transportation Improvement Program, and submission to the Metropolitan Transportation Commission (MTC) for inclusion in the MTC Regional Transportation Improvement Program (RTIP) and then to the California Transportation Commission (CTC), for inclusion in the State Transportation Improvement Program (STIP); and

WHEREAS, projects recommended for inclusion in the 2020 STIP must be consistent with the Commission-approved 2020 STIP Principles and satisfy all STIP programming, allocation and delivery requirements; and

WHEREAS, the CTC authorized federal Coronavirus Response and Relief Supplemental Appropriations Act of 2021 (CRRSAA) funds apportioned to the State to be distributed through the STIP as a midcycle augmentation to the 2020 STIP (2021 Mid-cycle STIP Program);

WHEREAS, the funding identified in the 2021 Mid-Cycle STIP Program Fund Estimate for Alameda County of \$6.308 million includes \$5.993 million of new STIP funding for projects and \$315,000 for Planning, Programming and Monitoring (PPM).

NOW, **THERFORE BE IT RESOLVED**, that the Alameda CTC approves the 2021 Mid-Cycle STIP Program detailed in Exhibit A.

DULY PASSED AND ADOPTED by the Alameda CTC Commission at the regular Commission meeting held on Thursday, May 27, 2021 in Oakland, California, by the following vote:

AYES: N

NOES: ABSTAIN:

ABSENT:

SIGNED:

Attest:

Pauline Russo Cutter Chair, Alameda CTC Vanessa Lee, Clerk of the Commission

EXHIBIT A

Alameda County 2021 Mid-cycle STIP Program

Index #	Project	Recommended 2021 Mid-Cycle STIP Funding		
1	I-680 Southbound Express Lanes from SR-84 to Alcosta Boulevard Project 1	\$5,793,000		
2	Alameda CTC STIP Administration (Planning, Programming and Monitoring)	\$315,000		
3	Oakland/Alameda Access (I-880 Broadway- Jackson)	\$50,000		
4	I-80 Ashby Interchange Improvements	\$50,000		
5	I-880 Interchanges (Whipple Road/Industrial Parkway Southwest & Industrial Parkway West) Improvements	\$50,000		
6	I-880 Interchanges (Winton Avenue and A Street)	\$50,000		
	Total	\$6,308,000		
Table Notes:1. In February 2021, the Alameda CTC approved the I-680 Southbound Express Lanes from SR-84 to Alcosta Boulevard Project to be amended into the 2020 STIP.				

This page intentionally left blank



1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

Commission Chair Mayor Pauline Russo Cutter City of San Leandro

Commission Vice Chair Councilmember John Bauters City of Emeryville

AC Transit Board President Elsa Ortiz

Alameda County

Supervisor David Haubert, District 1 Supervisor Richard Valle, District 2 Supervisor Wilma Chan, District 3 Supervisor Nate Miley, District 4 Supervisor Keith Carson, District 5

BART Vice President Rebecca Saltzman

City of Alameda Mayor Marilyn Ezzy Ashcraft

City of Albany Councilmember Rochelle Nason

City of Berkeley Councilmember Lori Droste

City of Dublin Melissa Hernandez, Mayor

City of Fremont Mayor Lily Mei

City of Hayward Mayor Barbara Halliday

City of Livermore Mayor Bob Woerner

City of Newark Councilmember Luis Freitas

City of Oakland Councilmember At-Large Rebecca Kaplan Councilmember Sheng Thao

City of Piedmont Councilmember Jen Cavenaugh

City of Pleasanton Mayor Karla Brown

City of Union City Mayor Carol Dutra-Vernaci

Executive Director Tess Lengyel

ALAMEDA COUNTY TRANSPORTATION COMMISSION Resolution No. 21-011

Resolution of Local Support for Oakland/Alameda Access (I-880 Broadway-Jackson); I-80 Ashby Interchange Improvements; I-880 Interchanges (Whipple Road/Industrial Parkway Southwest & Industrial Parkway West) Improvements; I-880 Interchanges at Winton Avenue and A Street Projects

Authorizing the filing of an application for funding assigned to MTC and committing any necessary matching funds and stating assurance to complete the projects

WHEREAS, Alameda County Transportation Commission (Alameda CTC) (herein referred to as APPLICANT) is submitting an application to the Metropolitan Transportation Commission (MTC) for up to \$200,000 in funding assigned to MTC for programming discretion, which includes federal funding administered by the Federal Highway Administration (FHWA) and federal or state funding administered by the California Transportation Commission (CTC) such as Surface Transportation Block Grant Program (STP) funding, Congestion Mitigation and Air Quality Improvement Program (CMAQ) funding, Transportation Alternatives (TA) set-aside/Active Transportation Program (ATP) funding, and Regional Transportation Improvement Program (RTIP) funding (herein collectively referred to as REGIONAL DISCRETIONARY FUNDING) for the Oakland/Alameda Access (I-880 Broadway-Jackson); I-80 Ashby Interchange Improvements; I-880 Interchanges (Whipple Road/Industrial Parkway Southwest & Industrial Parkway West) Improvements; I-880 Interchanges at Winton Avenue and A Street Projects (herein referred to as PROJECTS) for the 2021 Mid-Cycle RTIP (herein referred to as PROGRAM); and

WHEREAS, the United States Congress from time to time enacts and amends legislation to provide funding for various transportation needs and programs, (collectively, the FEDERAL TRANSPORTATION ACT) including, but not limited to the Surface Transportation Block Grant Program (STP) (23 U.S.C. § 133), the Congestion Mitigation and Air Quality Improvement Program (CMAQ) (23 U.S.C. § 149) and the Transportation Alternatives (TA) set-aside (23 U.S.C. § 133); and

WHEREAS, state statutes, including California Streets and Highways Code §182.6, §182.7, and §2381(a)(1), and California Government Code §14527, provide various funding programs for the programming discretion of the Metropolitan Planning Organization (MPO) and the Regional Transportation Planning Agency (RTPA); and

WHEREAS, pursuant to the FEDERAL TRANSPORTATION ACT, and any regulations promulgated thereunder, eligible project sponsors wishing to receive federal or state funds for a regionally-significant project shall submit an application first with the appropriate MPO, or RTPA, as applicable, for review and inclusion in the federal Transportation Improvement Program (TIP); and

WHEREAS, MTC is the MPO and RTPA for the nine counties of the San Francisco Bay region; and

WHEREAS, MTC has adopted a Regional Project Funding Delivery Policy (MTC Resolution No. 3606, revised) that sets out procedures governing the application and use of REGIONAL DISCRETIONARY FUNDING; and

WHEREAS, APPLICANT is an eligible sponsor for REGIONAL DISCRETIONARY FUNDING; and

WHEREAS, as part of the application for REGIONAL DISCRETIONARY FUNDING, MTC requires a resolution adopted by the responsible implementing agency stating the following:

- the commitment of any required matching funds; and
- that the sponsor understands that the REGIONAL DISCRETIONARY FUNDING is fixed at the programmed amount, and therefore any cost increase cannot be expected to be funded with additional REGIONAL DISCRETIONARY FUNDING; and
- that the PROJECTS will comply with the procedures, delivery milestones and funding deadlines specified in the Regional Project Funding Delivery Policy (MTC Resolution No. 3606, revised); and
- the assurance of the sponsor to complete the PROJECTS as described in the application, subject to environmental clearance, and if approved, as included in MTC's federal Transportation Improvement Program (TIP); and
- that the PROJECTS will have adequate staffing resources to deliver and complete the PROJECTS within the schedule submitted with the project application; and
- that the PROJECTS will comply with all project-specific requirements as set forth in the PROGRAM; and
- that APPLICANT has assigned, and will maintain a single point of contact for all FHWAand CTC-funded transportation projects to coordinate within the agency and with the respective Congestion Management Agency (CMA), MTC, Caltrans, FHWA, and CTC on all communications, inquires or issues that may arise during the federal programming and delivery process for all FHWA- and CTC-funded transportation and transit projects implemented by APPLICANT; and
- in the case of a transit project, the PROJECTS will comply with MTC Resolution No. 3866, revised, which sets forth the requirements of MTC's Transit Coordination Implementation Plan to more efficiently deliver transit projects in the region; and

- in the case of a highway project, the PROJECTS will comply with MTC Resolution No.
 4104, which sets forth MTC's Traffic Operations System (TOS) Policy to install and activate TOS elements on new major freeway projects; and
- in the case of an RTIP project, state law requires PROJECTS be included in a local congestion management plan, or be consistent with the capital improvement program adopted pursuant to MTC's funding agreement with the countywide transportation agency; and

WHEREAS, that APPLICANT is authorized to submit an application for REGIONAL DISCRETIONARY FUNDING for the PROJECTS; and

WHEREAS, there is no legal impediment to APPLICANT making applications for the funds; and

WHEREAS, there is no pending or threatened litigation that might in any way adversely affect the proposed PROJECTS, or the ability of APPLICANT to deliver such PROJECTS; and

WHEREAS, APPLICANT authorizes its Executive Director, General Manager, or designee to execute and file an application with MTC for REGIONAL DISCRETIONARY FUNDING for the PROJECTS as referenced in this resolution; and

WHEREAS, MTC requires that a copy of this resolution be transmitted to the MTC in conjunction with the filing of the application.

NOW, **THEREFORE**, **BE IT RESOLVED** that the APPLICANT is authorized to execute and file an application for funding for the PROJECTS for REGIONAL DISCRETIONARY FUNDING under the FEDERAL TRANSPORTATION ACT or continued funding; and be it further

RESOLVED that APPLICANT will provide any required matching funds; and be it further

RESOLVED that APPLICANT understands that the REGIONAL DISCRETIONARY FUNDING for the project is fixed at the MTC approved programmed amount, and that any cost increases must be funded by the APPLICANT from other funds, and that APPLICANT does not expect any cost increases to be funded with additional REGIONAL DISCRETIONARY FUNDING; and be it further

RESOLVED that APPLICANT understands the funding deadlines associated with these funds and will comply with the provisions and requirements of the Regional Project Funding Delivery Policy (MTC Resolution No. 3606, revised) and APPLICANT has, and will retain the expertise, knowledge and resources necessary to deliver federally-funded transportation and transit projects, and has assigned, and will maintain a single point of contact for all FHWA- and CTCfunded transportation projects to coordinate within the agency and with the respective Congestion Management Agency (CMA), MTC, Caltrans, FHWA, and CTC on all communications, inquires or issues that may arise during the federal programming and delivery process for all FHWA- and CTC-funded transportation and transit projects implemented by APPLICANT; and be it further

RESOLVED that PROJECTS will be implemented as described in the complete application

and in this resolution, subject to environmental clearance, and, if approved, for the amount approved by MTC and programmed in the federal TIP; and be it further

RESOLVED that APPLICANT has reviewed the PROJECTS and has adequate staffing resources to deliver and complete the PROJECTS within the schedule submitted with the project application; and be it further

RESOLVED that PROJECTS will comply with the requirements as set forth in MTC programming guidelines and project selection procedures for the PROGRAM; and be it further

RESOLVED that, in the case of a transit project, APPLICANT agrees to comply with the requirements of MTC's Transit Coordination Implementation Plan as set forth in MTC Resolution No. 3866, revised; and be it further

RESOLVED that, in the case of a highway project, APPLICANT agrees to comply with the requirements of MTC's Traffic Operations System (TOS) Policy as set forth in MTC Resolution No. 4104; and be it further

RESOLVED that, in the case of an RTIP project, PROJECTS is included in a local congestion management plan, or is consistent with the capital improvement program adopted pursuant to MTC's funding agreement with the countywide transportation agency; and be it further

RESOLVED that APPLICANT is an eligible sponsor of REGIONAL DISCRETIONARY FUNDING funded projects; and be it further

RESOLVED that APPLICANT is authorized to submit an application for REGIONAL DISCRETIONARY FUNDING for the PROJECTS; and be it further

RESOLVED that there is no legal impediment to APPLICANT making applications for the funds; and be it further

RESOLVED that there is no pending or threatened litigation that might in any way adversely affect the proposed PROJECTS, or the ability of APPLICANT to deliver such PROJECT; and be it further

RESOLVED that APPLICANT authorizes its Executive Director or designee to execute and file an application with MTC for REGIONAL DISCRETIONARY FUNDING for the PROJECTS as referenced in this resolution; and be it further

RESOLVED that a copy of this resolution will be transmitted to the MTC in conjunction with the filing of the application; and be it further

RESOLVED that the MTC is requested to support the application for the PROJECTS described in the resolution, and if approved, to include the PROJECT Sin MTC's federal TIP upon submittal by the project sponsor for TIP programming.

DULY PASSED AND ADOPTED by the Alameda CTC Commission at the regular Commission meeting held on Thursday, May 27, 2021 in Oakland, California, by the following vote:

AYES:	NOES:	ABSTAIN:	ABSENT:		
SIGNED:		ATTEST:			
Pauline Rus	so Cutter	Vanessa	Lee		
Chair, Alameda CTC		Clerk of t	Clerk of the Commission		

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:May 20, 2021TO:Alameda County Transportation CommissionFROM:Vivek Bhat, Director of Programming and Project Controls
Jacki Taylor, Senior Program AnalystSUBJECT:Approve actions associated with allocation of Regional Measure 2 funds for
Livermore Amador Valley Transit Authority's Rapid Bus Stop Improvement
project

Recommendation

It is recommended that the Commission approve the following actions associated with the Livermore Amador Valley Transit Authority's (LAVTA) Rapid Bus Stop Improvement project:

- Approve Resolution 21-012 (Attachment A) that approves the Regional Measure 2 (RM2) Initial Project Report (IPR) and designates LAVTA as the Implementing Agency for the design and construction phases of the LAVTA Rapid Bus Stop Improvement project (Project), under RM2 Capital Project #32 (I-580 Tri Valley Rapid Transit Corridor Improvements; Sponsor Alameda CTC).
- 2. Approve LAVTA's RM2 allocation request of \$2.3 million (Attachment B) to the Metropolitan Transportation Commission (MTC) for the Plans Specifications and Estimates (PS&E) and Construction phases of the Project.

Summary

Alameda CTC is the sponsor of the I-580 Tri Valley Rapid Transit Corridor Improvements (RM2 Capital Project #32). In December 2020, MTC notified LAVTA that a balance of approximately \$5 million remained on RM2 Capital Project #32 and is available for allocation to eligible transit related projects in the corridor and requested proposal(s) from LAVTA that could utilize the funds. LAVTA proposes to request \$2.3 million of these funds towards the PS&E and Construction phases of the LAVTA Rapid Bus Stop Improvement project (Project).

As the project sponsor of RM2 Capital Project #32, Alameda CTC is required to submit allocation request paperwork to MTC. In order to reduce administrative burden for both agencies and create efficiencies, Alameda CTC requested MTC if there was a pathway for LAVTA to access the RM2 funds directly. MTC notified staff this was possible if Alameda



CTC was supportive of LAVTA's allocation request and provided a Commission Resolution (Attachment A) that approves the RM2 IPR and designates LAVTA as the Implementing Agency for the design and construction phases of the Project.

Background

In 2004, Senate Bill 916 established the Regional Traffic Relief Plan, including a list of projects eligible to receive funding authorized by Regional Measure 2, which increased tolls on the seven state-owned toll bridges in the Bay Area by \$1 to fund various traffic relief programs and projects in eligible bridge corridors. SB 916 identified the Alameda CTC as the project sponsor of \$65 million in anticipated revenues to be allocated for RM2 Project #32, I-580 (Tri-Valley) Rapid Transit Corridor Improvements in Alameda County. To date \$60 million has been allocated in the corridor on construction of the I-580 High Occupancy Toll (HOT) lanes and other HOV improvements, improvements to the I-580/I-680 interchange, and to the construction phase of the Dublin/Pleasanton BART Parking Garage.

In December 2020, MTC notified LAVTA that a balance of approximately \$5 million remained on RM2 Capital Project #32 that is available for allocation to eligible transit related projects in the corridor that could utilize the funds. In response to MTC's notification, LAVTA's proposal includes \$2.3 million towards the PS&E and Construction phases of the Project.

The initial project scope calls for design, construction and installation of Rapid signage (both real-time bus arrival digital signs and other signage in display cases), shelters, benches, trash receptacles, and other passenger amenities for Rapid bus stops that are yet to receive them following realignment of the Rapid routes in 2016. The proposed amenities will bring consistency to the corridors served and continue to grow ridership. The project includes work on 47 bus stops, including equipment and installation costs for 47 Rapid signs, 9 Rapid-style shelters (including solar real time signage and lighting), 36 Rapid-style benches and trash receptacles and 18 Rapid-style bike loops for shelter locations and other high-use areas.

Of the 47 stops, 30 are on the Route 30R and 17 are on the 10R focused in the following locations:

- 26 in Livermore (notably the East Avenue and Portola Avenue corridors)
- 13 in Pleasanton (Stanley Blvd. and First St. corridors)
- 6 in Dublin (Dublin Blvd. corridor)
- 2 in Alameda County (Stanley Blvd. at Shadow Cliffs)

MTC staff has evaluated LAVTA's initial proposal and confirmed its eligibility for RM2 Capital Project #32 funds. Per MTC's RM2 Policies and Procedures (MTC Resolution No. 3636) project sponsors must submit a governing-board certification of compliance with RM2 provisions in order to receive allocations. Because the RM2 legislation identifies Alameda CTC as the project sponsor of RM2 Capital Project #32, Alameda CTC must also submit a resolution of local support for the project. In coordinating with MTC and LAVTA on this request, it was determined that an efficient way to deliver this project would be if LAVTA had direct access to the funds from MTC, rather than being passed through via Alameda CTC. MTC notified staff this was possible if Alameda CTC was supportive of LAVTA's allocation request and provided a Commission Resolution that designates LAVTA as the Implementing Agency for the design and construction phases of the Project.

LAVTA Board will be considering this RM2 allocation request approval at their May 3, 2021 meeting (Attachment B). RM2 Policies and Procedures require each allocation to fund a minimum useable segment and/or deliverable. LAVTA's initial allocation will request \$230,000 for the Project's design phase only. Pending acceptance of 100% PS&E for the Project, MTC will consider LAVTA's \$2.07 million construction phase allocation request at a later date.

Staff has reviewed LAVTA's request and is recommending Commission approve the following actions associated with the Project:

- 1. Approve Resolution 21-012 that approves the RM2 IPR and designates LAVTA as the Implementing Agency for the design and construction phases of Project, under RM2 Capital Project #32.
- 2. Approve and support LAVTA's RM2 allocation request of \$2.3 million to the MTC for the PS&E and Construction phases of the Project.

Next Steps

Upon approvals by the Alameda CTC and LAVTA Board, MTC will consider this allocation request at their next Commission meeting. Alameda CTC will continue to work with BART and LAVTA to identify eligible transit capital needs on the remaining balance of RM2 Capital Project #32 funds.

Fiscal Impact: There is no fiscal impact associated with the requested item.

Attachments

- A. Alameda CTC Resolution 21-012
- B. LAVTA staff report

This page intentionally left blank



1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

www.AlamedaCTC.org

Commission Chair Mayor Pauline Russo Cutter City of San Leandro

Commission Vice Chair Councilmember John Bauters City of Emeryville

AC Transit Board President Elsa Ortiz

Alameda County

Supervisor David Haubert, District 1 Supervisor Richard Valle, District 2 Supervisor Wilma Chan, District 3 Supervisor Nate Miley, District 4 Supervisor Keith Carson, District 5

BART

Vice President Rebecca Saltzman

City of Alameda Mayor Marilyn Ezzy Ashcraft

City of Albany Councilmember Rochelle Nason

City of Berkeley Councilmember Lori Droste

City of Dublin Melissa Hernandez, Mayor

City of Fremont Mayor Lily Mei

City of Hayward Mayor Barbara Halliday

City of Livermore Mayor Bob Woerner

City of Newark Councilmember Luis Freitas

City of Oakland Councilmember At-Large Rebecca Kaplan Councilmember Sheng Thao

City of Piedmont Councilmember Jen Cavenaugh

City of Pleasanton Mayor Karla Brown

City of Union City Mayor Carol Dutra-Vernaci

Executive Director Tess Lengyel

ALAMEDA COUNTY TRANSPORTATION COMMISSION RESOLUTION 21-012

Approval of the Regional Measure 2 Initial Project Report for the Livermore Amador Valley Transit Authority (LAVTA) Rapid Bus Stop Improvement Project and for LAVTA to serve as the RM2 Implementing Agency

RM2 Sponsor: Alameda County Transportation Commission

RM2 Implementing Livermore Amador Valley Transit Authority Agency: (LAVTA)

RM2 Project Title: LAVTA Rapid Bus Stop Improvements

WHEREAS, SB 916 (Chapter 715, Statutes 2004), commonly referred as Regional Measure 2, identified projects eligible to receive funding under the Regional Traffic Relief Plan; and

WHEREAS, the Metropolitan Transportation Commission (MTC) is responsible for funding projects eligible for Regional Measure 2 funds, pursuant to Streets and Highways Code Section 30914(c) and (d); and

WHEREAS, MTC has established a process whereby eligible transportation project sponsors may submit allocation requests for Regional Measure 2 funding; and

WHEREAS, allocations to MTC must be submitted consistent with procedures and conditions as outlined in Regional Measure 2 Policy and Procedures; and

WHEREAS, the Alameda County Transportation Commission (Alameda CTC) and Livermore Amador Valley Transit Authority (LAVTA) are each an eligible sponsor of transportation project(s) in Regional Measure 2, Regional Traffic Relief Plan funds; and

WHEREAS, the LAVTA Rapid Bus Stop Improvements Project is eligible for consideration in the Regional Traffic Relief Plan of Regional Measure 2, as identified in California Streets and Highways Code Section 30914(c) or (d); and

kland, CA 94607 •

Alameda CTC Resolution No. 21-012 LAVTA RM2 Initial Project Report Page 2 of 4

WHEREAS, LAVTA has prepared the Regional Measure 2 allocation request, attached hereto in the Initial Project Report and incorporated herein as though set forth at length, which lists the project, purpose, schedule, budget, expenditure and cash flow plan for which LAVTA is requesting that MTC allocate Regional Measure 2 funds; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA and its agents shall comply with the provisions of the Metropolitan Transportation Commission's Regional Measure 2 Policy Guidance (MTC Resolution No. 3636); and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA certifies that the project is consistent with the Regional Transportation Plan (RTP); and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that the year of funding for any design, right-of-way and/or construction phases has taken into consideration the time necessary to obtain environmental clearance and permitting approval for the project; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that the Regional Measure 2 phase or segment is fully funded, and results in an operable and useable segment; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA approves the cash flow plan, attached to this resolution; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA has reviewed the project needs and has adequate staffing resources to deliver and complete the project within the schedule set forth in the updated Initial Project Report, attached to this resolution; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA is an eligible sponsor of projects in the Regional Measure 2 Regional Traffic Relief Plan, Capital Program, in accordance with California Streets and Highways Code 30914(c); and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA is authorized to submit an application for Regional Measure 2 funds for (project name) in accordance with California Streets and Highways Code 30914(c); and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA certifies that the projects and purposes for which RM2 funds are being requested is in compliance with the requirements of the California Environmental Quality Act (Public Resources Code Section 21000 <u>et seq</u>.), and with the State Environmental Impact Report Guidelines (I4 California Code of Regulations Section 15000 <u>et seq</u>.) and if relevant the National Environmental Policy Act (NEPA), 42 USC Section 4-1 et. seq. and the applicable regulations thereunder; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that there is no legal impediment to LAVTA making allocation requests for Regional Measure 2 funds; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that there is no pending or threatened litigation which might in any way adversely affect the proposed project, or the ability of LAVTA to deliver such project; and be it further
Alameda CTC Resolution No. 21-012 LAVTA RM2 Initial Project Report Page 3 of 4

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA agrees to comply with the requirements of MTC's Transit Coordination Implementation Plan as set forth in MTC Resolution 3866; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA shall, if any revenues or profits from any non-governmental use of property (or project) that those revenues or profits shall be used exclusively for the public transportation services for which the project was initially approved, either for capital improvements or maintenance and operational costs, otherwise the Metropolitan Transportation Commission is entitled to a proportionate share equal to MTC's percentage participation in the projects(s); and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that assets purchased with RM2 funds including facilities and equipment shall be used for the public transportation uses intended, and should said facilities and equipment cease to be operated or maintained for their intended public transportation purposes for its useful life, that the Metropolitan Transportation Commission (MTC) shall be entitled to a present day value refund or credit (at MTC's option) based on MTC's share of the Fair Market Value of the said facilities and equipment at the time the public transportation uses ceased, which shall be paid back to MTC in the same proportion that Regional Measure 2 funds were originally used; and

WHEREAS, LAVTA has passed LAVTA Resolution No. 14-2021, which states that LAVTA shall post on both ends of the construction site(s) at least two signs visible to the public stating that the Project is funded with Regional Measure 2 Toll Revenues; now, therefore, be it

RESOLVED, that Alameda CTC approves the Initial Project Report, attached to this resolution as Exhibit A; and be it further

RESOLVED, that Alameda CTC designates LAVTA as the implementing agency for the design and construction phases of the LAVTA Rapid Bus Stop Improvement project, under RM2 Project 32, I-580 (Tri Valley) Rapid Transit Corridor Improvements; and be it further

RESOLVED, that Alameda CTC indemnifies and holds harmless MTC, its Commissioners, representatives, agents, and employees from and against all claims, injury, suits, demands, liability, losses, damages, and expenses, whether direct or indirect (including any and all costs and expenses in connection therewith), incurred by reason of any act or failure to act of Alameda CTC, its officers, employees or agents, or subcontractors or any of them in connection with its performance of services under this allocation of RM2 funds; and be it further

RESOLVED, that Alameda CTC authorizes LAVTA to execute and submit allocation requests for the design and construction phases with MTC for Regional Measure 2 funds in the amount of \$2.3 million, for the project, purposes and amounts included in the project application attached to this resolution as Exhibit A; and be it further

RESOLVED, that LAVTA is hereby authorized to make non-substantive changes or minor amendments to the IPR as LAVTA deems appropriate; and be it further

RESOLVED, that a copy of this resolution shall be transmitted to MTC in conjunction with the filing of the LAVTA application referenced herein.

Alameda CTC Resolution No. 21-012 LAVTA RM2 Initial Project Report Page 4 of 4

DULY PASSED AND ADOPTED by the Alameda CTC Commission at the regular Commission meeting held on Thursday, May 27, 2021 in Oakland, California, by the following vote:

AYES: NOES: ABSTAIN: ABSENT:

SIGNED:

Attest:

Pauline Russo Cutter Chair, Alameda CTC

Vanessa Lee, Clerk of the Commission

RESOLUTION NO. 14-2021

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE LIVERMORE AMADOR VALLEY TRANSIT AUTHORITY AS IMPLEMENTING AGENCY FOR REGIONAL MEASURE 2 FUNDING FOR THE RAPID BUS STOP IMPROVEMENT PROJECT

WHEREAS, SB 916 (Chapter 715, Statutes 2004), commonly referred as Regional Measure 2, identified projects eligible to receive funding under the Regional Traffic Relief Plan; and

WHEREAS, the Metropolitan Transportation Commission (MTC) is responsible for funding projects eligible for Regional Measure 2 funds, pursuant to Streets and Highways Code Section 30914(c) and (d); and

WHEREAS, MTC has established a process whereby eligible transportation project sponsors may submit allocation requests for Regional Measure 2 funding; and

WHEREAS, allocations to MTC must be submitted consistent with procedures and conditions as outlined in Regional Measure 2 Policy and Procedures; and

WHEREAS, Streets and Highways Code Section 30914(c) and (d) identifies the Alameda County Transportation Commission as Project Sponsor for RM2 Project 32, I-580 (Tri-Valley) Rapid Transit Corridor Improvements in Alameda County; and

WHEREAS, the Alameda County Transportation Commission plans to designate the Livermore Amador Valley Transit Authority (LAVTA) as implementing agency for the design and construction of the Rapid Bus Stop Improvement Project, an eligible project under RM2 Project 32, I-580 (Tri-Valley) Rapid Transit Corridor Improvements; and

WHEREAS, LAVTA is an eligible implementing agency for transportation project(s) in Regional Measure 2, Regional Traffic Relief Plan funds; and

WHEREAS, the Rapid Bus Stop Improvement Project is eligible for consideration in the Regional Traffic Relief Plan of Regional Measure 2, as identified in California Streets and Highways Code Section 30914(c) or (d); and

WHEREAS, the Regional Measure 2 allocation request, attached hereto in the Initial Project Report and incorporated herein as though set forth at length, lists the project, purpose, schedule, budget, expenditure and cash flow plan for which LAVTA is requesting that MTC allocate Regional Measure 2 funds; now, therefore, be it

RESOLVED, that LAVTA, and its agents shall comply with the provisions of the Metropolitan Transportation Commission's Regional Measure 2 Policy Guidance (MTC Resolution No. 3636); and be it further

RESOLVED, that LAVTA certifies that the project is consistent with the Regional Transportation Plan (RTP); and be it further

RESOLVED, that the year of funding for any design, right-of-way and/or construction phases has taken into consideration the time necessary to obtain environmental clearance and permitting approval for the project; and be it further

RESOLVED, that the Regional Measure 2 phase or segment is fully funded, and results in an operable and useable segment; and be it further

RESOLVED, that LAVTA approves the updated Initial Project Report, attached to this resolution; and be it further

RESOLVED, that LAVTA approves the cash flow plan, attached to this resolution; and be it further

RESOLVED, that LAVTA has reviewed the project needs and has adequate staffing resources to deliver and complete the project within the schedule set forth in the updated Initial Project Report, attached to this resolution; and, be it further

RESOLVED, that LAVTA is an eligible sponsor of projects in the Regional Measure 2 Regional Traffic Relief Plan, Capital Program, in accordance with California Streets and Highways Code 30914(c); and be it further

RESOLVED, that LAVTA is authorized to submit an application for Regional Measure 2 funds for the Rapid Bus Stop Improvement Project in accordance with California Streets and Highways Code 30914(c); and be it further

RESOLVED, that LAVTA certifies that the projects and purposes for which RM2 funds are being requested is in compliance with the requirements of the California Environmental Quality Act (Public Resources Code Section 21000 <u>et seq</u>.), and with the State Environmental Impact Report Guidelines (14 California Code of Regulations Section 15000 <u>et seq</u>.) and if relevant the National Environmental Policy Act (NEPA), 42 USC Section 4-1 et. seq. and the applicable regulations thereunder; and be it further

RESOLVED, that there is no legal impediment to LAVTA making allocation requests for Regional Measure 2 funds; and be it further

RESOLVED, that there is no pending or threatened litigation which might in any way adversely affect the proposed project, or the ability of LAVTA to deliver such project; and be it further

RESOLVED, that LAVTA agrees to comply with the requirements of MTC's Transit Coordination Implementation Plan as set forth in MTC Resolution 3866; and be it further **RESOLVED**, that LAVTA indemnifies and holds harmless MTC, its Commissioners, representatives, agents, and employees from and against all claims, injury, suits, demands, liability, losses, damages, and expenses, whether direct or indirect (including any and all costs and expenses in connection therewith), incurred by reason of any act or failure to act of LAVTA, its officers, employees or agents, or subcontractors or any of them in connection with its performance of services under this allocation of RM2 funds. In addition to any other remedy authorized by law, so much of the funding due under this allocation of RM2 funds as shall reasonably be considered necessary by MTC may be retained until disposition has been made of any claim for damages, and be it further

RESOLVED, that LAVTA shall, if any revenues or profits from any non-governmental use of property (or project) that those revenues or profits shall be used exclusively for the public transportation services for which the project was initially approved, either for capital improvements or maintenance and operational costs, otherwise the Metropolitan Transportation Commission is entitled to a proportionate share equal to MTC's percentage participation in the projects(s); and be it further

RESOLVED, that assets purchased with RM2 funds including facilities and equipment shall be used for the public transportation uses intended, and should said facilities and equipment cease to be operated or maintained for their intended public transportation purposes for its useful life, that the Metropolitan Transportation Commission (MTC) shall be entitled to a present day value refund or credit (at MTC's option) based on MTC's share of the Fair Market Value of the said facilities and equipment at the time the public transportation uses ceased, which shall be paid back to MTC in the same proportion that Regional Measure 2 funds were originally used; and be it further

RESOLVED, that LAVTA shall post on both ends of the construction site(s) at least two signs visible to the public stating that the Project is funded with Regional Measure 2 Toll Revenues; and be it further

RESOLVED, that LAVTA authorizes its Executive Director or his/her designee to execute and submit an allocation request for the design phase with MTC for Regional Measure 2 funds in the amount of two hundred thirty thousand dollars (\$230,000), for the project, purposes and amounts included in the project application attached to this resolution; and be it further

RESOLVED, that the Executive Director or his/her designee is hereby delegated the authority to make non-substantive changes or minor amendments to the IPR as he/she deems appropriate; and be it further

RESOLVED, that a copy of this resolution shall be transmitted to MTC in conjunction with the filing of the LAVTA application referenced herein; and be it further

RESOLVED, that allocation of Regional Measure 2 funding for the Rapid Bus Stop Improvement Project is contingent upon action by the Alameda County Transportation Commission designating LAVTA as implementing agency for the Project and the Metropolitan Transportation Commission's approval of this designation with the allocation request. **PASSED AND ADOPTED BY** the governing board of the Livermore Amador Valley Transit Authority on this 3rd day of May 2021.

Bob Woerner, Chair

Attest:

Michael Tree, Executive Director

Regional Measure 2 Initial Project Report (IPR)

Project Title:	LAVTA Rapid Bus Stop Improvement Project
RM2 Project No.	

Allocation History:

	MTC Approval Date	Amount	Phase
#1:			
#2			
#3			
	Total: \$		

Current Allocation Request:

IPR Date	Amount Being	Phase Requested
	Requested	
February 24,	\$230,000	PS&E
2021		
February 24,	\$2,070,000	Construction
2021		

I. OVERALL PROJECT INFORMATION

A. Project Sponsor / Co-sponsor(s) / Implementing Agency

Livermore Amador Valley Transit Authority (LAVTA)

B. Project Purpose

In 2008 LAVTA introduced its first Rapid route (10R) with 15-minute all day service in an effort to attract choice riders commuting within the Tri-Valley and throughout the Bay Area. The introduction of the Rapid service coincided with an \$8 million investment by the FTA in Rapid bus stops, that included upgraded shelters and amenities, such as lighting, signage, real time information, displays, etc.

LAVTA initiated a comprehensive operational analysis in 2014 of the Wheels system and focused on attracting choice riders, adding an additional Rapid route (30R) that doubled the amount of 15-minute all day service in the Wheels system. The 30R began service in 2016 and today the Rapid routes account for 55% of all riders in the Wheels system, with the majority of Rapid passengers boarding and alighting at the two BART stations in the LAVTA service area. The increase in ridership on the Rapid routes since implementation has a positive impact at reducing congestion on I-580 as Tri-Valley residents forgo their mostly single occupancy commute vehicles for the Rapid, which connects them with BART for travel throughout the Bay Area, including transbay trips to San Francisco and the Peninsula.

The purpose of the LAVTA Rapid Bus Stop Improvement Project is to construct and install Rapid signage (both real-time bus arrival digital signs and other signage in display cases), shelters, benches, trash receptacles, and other passenger amenities for Rapid bus stops that are yet to receive them. Such amenities will bring consistency to the corridors served and continue to grow ridership.

C. Project Description (please provide details) Project Graphics to be sent electronically with This Application

This project includes Rapid bus stops on the 10R and 30R that have yet to receive Rapid-branded signage, shelters, benches, trash receptacles and other amenities. It includes work on 47 bus stops as detailed in the attached project scoping plan, including:

- 47 Rapid signs and installation costs
- 9 Rapid style shelters (including solar real time signage and lighting) and installation costs
- 36 Rapid style benches and trash receptacles and installation costs
- 18 Rapid style bike loops for shelter locations and other high-use areas, and installation costs

The total cost for the Rapid Bus Stop Improvement Project is \$2,300,000, which includes \$230,000 in project design and engineering work, and \$2,070,000 in construction costs (including equipment acquisition and installation and any necessary concrete improvements), escalated to 2022 dollars, the year it is assumed construction will take place. The project can be completed within 15 months of authorization.

D. Impediments to Project Completion

None identified.

E. Operability

The maintenance costs associated with the Rapid bus stops are included in the annual LAVTA Operating and Maintenance budget. Maintenance includes regular removal of trash and cleaning of shelters, as well as immediate repair of Rapid shelters when damaged, as well as less-frequent ongoing maintenance such as painting.

II. PROJECT PHASE DESCRIPTION and STATUS

F. Environmental -

Does NEPA Apply: 🗌 Yes 🔀 No

N/A - these improvements should qualify for Categorical Exclusion.

G. Design –

LAVTA currently has an on-call engineering contract in place with Kimley-Horn and Associates. It is anticipated that upon allocation of RM2 funding, LAVTA would execute a Task Order with Kimley-Horn to prepare 100% PS&E documents ready to advertise for construction and equipment acquisition, based on the quantities and locations specified in the attached project scoping plan. Equipment design of the Rapid bus stop signage, shelters and amenities has been selected by the member agencies in collaboration with LAVTA, and would be selected to match branding of existing facilities in each jurisdiction.

H. Right-of-Way Activities / Acquisition -

All Rapid bus stop improvements are located in the public right-of-way. Permits with applicable agencies will provide easements for the bus stop improvements as well as LAVTA's ongoing maintenance responsibilities.

I. Construction / Vehicle Acquisition -

At multiple Rapid bus stops in the project area, minor concrete improvements will be required to support installation of the bus stop improvements. Accordingly, 5,000 sq ft of concrete flatwork has been assumed and budgeted for the project, among other ancillary construction costs associated with typical contracts for similar improvements, including mobilization, demo/salvage, etc. Additionally, bus stop signage, shelters, benches and other amenities will need to be procured from available manufacturers before installation. The anticipated lead time for delivery of customized equipment is 4-6 months. Additional details are provided in the Work Plan in Section P below.

III. PROJECT BUDGET

J. Project Budget (Escalated to year of expenditure)

Phase	Total Amount - Escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$230
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,070
Total Project Budget (in thousands)	\$2,300

K. Project Budget (De-escalated to current year)

Phase	Total Amount - De-escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$225
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,023
Total Project Budget (in thousands)	\$2,248

L. Project Budget – Deliverable Segment (Escalated to year of expenditure)

Phase	Total Amount - Escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$230
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,070
Total Project Budget (in thousands)	\$2,300

M. Project Budget – Deliverable Segment(De-escalated to current year)

Phase	Total Amount - De-escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$225
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,023
Total Project Budget (in thousands)	\$2,248

IV. OVERALL PROJECT SCHEDULE

	Planned (Update as needed)		
Phase-Milestone	Start Date	Completion Date	
Environmental Document	N/A	N/A	
Environmental Studies, Preliminary Eng. (ENV / PE / PA&ED)	N/A	N/A	
Final Design - Plans, Specs. & Estimates (PS&E)	June 2021	November 2021	
Right-of-Way Activities /Acquisition (R/W)	N/A	N/A	
Construction (Begin – Open for Use) / Acquisition / Operating Service (CON)	March 2022	October 2022	

V. ALLOCATION REQUEST INFORMATION

N. Detailed Description of Allocation Request

Describe the scope of the allocation request. Provide background and other details as necessary.

Amount being requested (in escalated dollars)	\$2,300,000
Project Phase being requested	PS&E / CON
Are there other fund sources involved in this phase?	🗌 Yes 🖾 No
Date of anticipated Implementing Agency Board approval the RM2 IPR Resolution for the allocation being requested	April 5, 2021
Month/year being requested for MTC Commission approval of allocation	April 2021

O. Status of Previous Allocations (if any)

N/A

P. Workplan

Workplan in Alternate Format Enclosed

TASK NO	Description	Deliverables	Completion Date
1.	Award Design Engineering Contract	Board resolution authorizing design- engineering contract/scope of work	June 2021
2.	Design completion/ready to advertise/procure equipment	100% Plans, Specifications & Estimates	November 2021
3.	Award third-party equipment procurement/construction contracts	Board resolution(s) authorizing contracts for equipment-purchase and construction contracts	March 2022
4.	Delivery of all required equipment	Equipment delivered per specifications, vendor invoices paid	April 2022
5.	Completion of all construction and equipment installations	Construction contract acceptance/closeout	October 2022

Q. Impediments to Allocation Implementation

Circumstances surrounding COVID-19 could potentially delay in manufacture of equipment, depending on circumstances of individual vendors.

VI. RM-2 FUNDING INFORMATION

R. RM-2 Funding Expenditures for funds being allocated

The companion Microsoft Excel Project Funding Spreadsheet to this IPR is included

S. Next Anticipated RM2 Allocation Request.

LAVTA does not anticipate future RM2 expenditure requests for Rapid bus stop improvements at this time.

VII. GOVERNING BOARD ACTION

Check the box that applies:

- Governing Board Resolution attached
- Governing Board Resolution to be provided on or before: April 5, 2021

VIII. CONTACT / PREPARATION INFORMATION

Contact for Applicant's Agency

Name: Michael Tree Phone: 925-455-7555 Title: Executive Director E-mail: mtree@lavta.org Address: 1362 Rutan Court, Suite 100, Livermore, CA 94551

Information on Person Preparing IPR

Name: Jennifer Yeamans Phone: 925-455-7561 Title: Senior Grants & Management Specialist E-mail: jyeamans@lavta.org Address: 1362 Rutan Court, Suite 100, Livermore, CA 94551

Applicant Agency's Accounting Contact

Name: Tamara Edwards Phone: 925-455-7566 Title: Director of Finance E-mail: tedwards@lavta.org Address: 1362 Rutan Court, Suite 100, Livermore, CA 94551

Revised IPR 120905.doc

LAVTA Rapid Bus Stops Improvement Project

Initial Equipment/Materials/Labor Cost Estimates

Equipment		nit Cost	Unit	Qty	Total
12' Passenger shelter-Livermore Style	\$	36,900	EA	5	\$ 184,500
12' Passenger shelter-D/P style	\$	8,700	EA	4	\$ 34,800
Rapid benches - Livermore Style	\$	3,300	EA	18	\$ 59,400
Rapid benches - Dublin/Pleasanton Style	\$	650	EA	18	\$ 11,700
Trash & Recycling Receptacle Pair - Livermore Style	\$	2,800	EA	16	\$ 44,800
Trash & Recycling Receptacle Pair - D/P Style	\$	2,900	EA	18	\$ 52,200
RT signs	\$	7,000	EA	9	\$ 63,000
Bicycle Rack	\$	360	EA	18	\$ 6,480
Rapid Flag Sign	\$	12,500	EA	47	\$ 587,500
Schedule holders	\$	550	EA	9	\$ 4,950
Equipment Subtotal					\$ 1,049,000

Construction	Uı	nit Cost	Unit	Qty	Total
Mobilization	\$	40,000	LS	1	\$ 40,000
Demo/Disposal	\$	17,000	LS	1	\$ 17,000
Minor concrete (SW)-SF	\$	27	SF	4,000	\$ 108,000
Equipment Installation					
Passenger Shelter w/solar system	\$	11,600	EA	9	\$ 104,400
RT info signs	\$	11,200	EA	9	\$ 100,800
Bike racks	\$	750	EA	18	\$ 13,500
Bench	\$	750	EA	36	\$ 27,000
Waste Receptacles	\$	770	EA	34	\$ 26,180
Rapid flag sign	\$	5,000	EA	47	\$ 235,000
Schedule holders	\$	500	EA	9	\$ 4,500
Construction Subtotal					\$ 676,000
CM Services (@10% of subtotals)	\$	172,500	LS	1	\$ 172,500
Contingency (@ 10% of subtotals)	\$	172,500	LS	1	\$ 172,500
GRAND TOTAL (rounded to nearest 1000)					\$ 2,070,000

Livermore Amador Valley Transit Authority

STAFF REPORT

SUBJECT:	Resolution in Support of Allocation Request for Regional Measure 2 Funding for the Rapid Bus Stop Improvement Project
FROM:	Jennifer Yeamans, Senior Grants & Management Specialist
DATE:	April 27, 2021

Action Requested

Staff requests the Finance and Administration Committee recommend the Board of Directors approve Resolution 14-2021 in support of an allocation request to the Metropolitan Transportation Commission (MTC) for \$230,000 for the design phase of the Rapid Bus Stop Improvement Project. This resolution is required to request an allocation of this funding from MTC.

Background

In 2004, Senate Bill 916 established the Regional Traffic Relief Plan, including a list of projects eligible to receive funding authorized by Regional Measure 2, which increased tolls on the seven state-owned toll bridges in the Bay Area by \$1 to fund various traffic relief programs and projects in eligible bridge corridors. SB 916 identified the Alameda County Transportation Commission (CTC) as the project sponsor of \$65 million in anticipated revenues to be allocated for RM2 Project 32, *I-580 (Tri-Valley) Rapid Transit Corridor Improvements in Alameda County*. To date funds have been allocated in the corridor on construction of the I-580 High Occupancy Toll (HOT) lanes and other HOV improvements, improvements to the I-580/I-680 interchange, and construction of the Dublin/Pleasanton BART Parking Garage.

In late 2020, MTC notified LAVTA staff that a balance of approximately \$5 million remained on the Project available for allocation to eligible transit-related projects in the corridor and requested proposal(s) from LAVTA that could utilize the funds. In December 2020, Alameda CTC approved the update to the Countywide Transportation Program, which included several LAVTA priority projects, including \$3 million for systemwide passenger facilities rehabilitation and enhancements, and \$2 million for capital costs related to Phase 2 deployment of the Shared Autonomous Vehicle (SAV) project. LAVTA initiated formal requests to MTC for RM2 capital funding for both projects. To date, MTC staff has only approved the allocation request submitted for Rapid Bus Stop Improvements, as discussions continue regarding the SAV proposal.

Discussion

Per MTC Regional Measure 2 Policies and Procedures (MTC Resolution 3636), project sponsors must submit a governing-board certification of compliance with RM2 provisions (Attachment 1) in order to receive allocations. Because the RM2 legislation identifies Alameda CTC as the project sponsor, Alameda CTC must also submit a resolution of local support for the project. At its May 27 meeting, Alameda CTC intends to approve its resolution and in doing so designate LAVTA as the project's Implementing Agency, delegating responsibility to LAVTA for compliance with all RM2 Policies and Procedures. MTC's approval of the allocation request is anticipated on May 26 and will be contingent up actions by both the LAVTA Board on May 5 and Alameda CTC on May 27.

RM2 Policies and Procedures require each allocation to fund a minimum useable segment and/or deliverable. Thus MTC's initial allocation will fund \$230,000 budgeted for the project's design phase only. Pending acceptance of 100% plans, specifications, and estimates for the project, MTC will consider allocating an additional \$2.07 million for the construction phase as described in the Initial Project Report (IPR), shown in Attachment 2.

The initial project scope defined in the IPR calls for design, construction and installation of Rapid signage (both real-time bus arrival digital signs and other signage in display cases), shelters, benches, trash receptacles, and other passenger amenities for Rapid bus stops that are yet to receive them following realignment of the Rapid routes in 2016. Such amenities will bring consistency to the corridors served and continue to grow ridership. The project includes work on 47 bus stops as detailed in the attached project scoping plan (Attachment 3), including equipment and installation costs for:

- 47 Rapid signs
- 9 Rapid-style shelters (including solar real time signage and lighting)
- 36 Rapid-style benches and trash receptacles
- 18 Rapid-style bike loops for shelter locations and other high-use areas

Of the 47 stops, 30 are on the Route 30R and 17 are on the 10R focused in the following locations:

- 26 in Livermore (notably the East Avenue and Portola Avenue corridors)
- 13 in Pleasanton (Stanley Blvd. and First St. corridors)
- 6 in Dublin (Dublin Blvd. corridor)
- 2 in Alameda County (Stanley Blvd. at Shadow Cliffs)

Budget

The total project budget is funded 100% by RM2 funds as follows by phase.

	RM2
PS&E (current allocation)	\$230,000
Construction (<i>future allocation</i>)	\$2,070,000
Total	\$2,300,000

Next Steps

Following MTC and Alameda CTC approval of the RM2 allocation in late May, LAVTA will initiate a Task Order Request with its on-call design-engineering firm, Kimley-Horn, and

bring a detailed scope of work for design-engineering services to the Projects & Services Committee for review and approval. LAVTA will continue to work with MTC staff to allocate the remaining available balance to eligible project(s), including SAV Phase 2 Deployment activities.

Because the proposed corridor-level Rapid Bus Stops Improvements Project funded by RM2 complements MTC's proposed programming of \$2 million in available federal Safe and Seamless Quick-Strike funds (Agenda item 9) for the node-based Passenger Facilities Enhancements Project, staff will consult with MTC on the feasibility of combining the two projects to the maximum extent possible in order to achieve cost-efficiencies in design-engineering and project-management effort, while still satisfying the requirements of all individual fund sources involved.

Recommendation

Staff recommends Finance & Administration Committee refer Resolution 14-2021 to the Board of Directors for approval in support of an allocation request to the Metropolitan Transportation Commission for \$230,000 for the design phase of the Rapid Bus Stop Improvement Project.

Attachments:

- 1. Resolution 14-2021
- 2. Initial Project Report: LAVTA Rapid Bus Stop Improvement Project
- 3. Rapid Bus Stop Improvement Project Scoping Plan

RESOLUTION NO. 14-2021

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE LIVERMORE AMADOR VALLEY TRANSIT AUTHORITY AS IMPLEMENTING AGENCY FOR REGIONAL MEASURE 2 FUNDING FOR THE RAPID BUS STOP IMPROVEMENT PROJECT

WHEREAS, SB 916 (Chapter 715, Statutes 2004), commonly referred as Regional Measure 2, identified projects eligible to receive funding under the Regional Traffic Relief Plan; and

WHEREAS, the Metropolitan Transportation Commission (MTC) is responsible for funding projects eligible for Regional Measure 2 funds, pursuant to Streets and Highways Code Section 30914(c) and (d); and

WHEREAS, MTC has established a process whereby eligible transportation project sponsors may submit allocation requests for Regional Measure 2 funding; and

WHEREAS, allocations to MTC must be submitted consistent with procedures and conditions as outlined in Regional Measure 2 Policy and Procedures; and

WHEREAS, Streets and Highways Code Section 30914(c) and (d) identifies the Alameda County Transportation Commission as Project Sponsor for RM2 Project 32, I-580 (Tri-Valley) Rapid Transit Corridor Improvements in Alameda County; and

WHEREAS, the Alameda County Transportation Commission plans to designate the Livermore Amador Valley Transit Authority (LAVTA) as implementing agency for the design and construction of the Rapid Bus Stop Improvement Project, an eligible project under RM2 Project 32, I-580 (Tri-Valley) Rapid Transit Corridor Improvements; and

WHEREAS, LAVTA is an eligible implementing agency for transportation project(s) in Regional Measure 2, Regional Traffic Relief Plan funds; and

WHEREAS, the Rapid Bus Stop Improvement Project is eligible for consideration in the Regional Traffic Relief Plan of Regional Measure 2, as identified in California Streets and Highways Code Section 30914(c) or (d); and

WHEREAS, the Regional Measure 2 allocation request, attached hereto in the Initial Project Report and incorporated herein as though set forth at length, lists the project, purpose, schedule, budget, expenditure and cash flow plan for which LAVTA is requesting that MTC allocate Regional Measure 2 funds; now, therefore, be it

RESOLVED, that LAVTA, and its agents shall comply with the provisions of the Metropolitan Transportation Commission's Regional Measure 2 Policy Guidance (MTC Resolution No. 3636); and be it further

RESOLVED, that LAVTA certifies that the project is consistent with the Regional Transportation Plan (RTP); and be it further

RESOLVED, that the year of funding for any design, right-of-way and/or construction phases has taken into consideration the time necessary to obtain environmental clearance and permitting approval for the project; and be it further

RESOLVED, that the Regional Measure 2 phase or segment is fully funded, and results in an operable and useable segment; and be it further

RESOLVED, that LAVTA approves the updated Initial Project Report, attached to this resolution; and be it further

RESOLVED, that LAVTA approves the cash flow plan, attached to this resolution; and be it further

RESOLVED, that LAVTA has reviewed the project needs and has adequate staffing resources to deliver and complete the project within the schedule set forth in the updated Initial Project Report, attached to this resolution; and, be it further

RESOLVED, that LAVTA is an eligible sponsor of projects in the Regional Measure 2 Regional Traffic Relief Plan, Capital Program, in accordance with California Streets and Highways Code 30914(c); and be it further

RESOLVED, that LAVTA is authorized to submit an application for Regional Measure 2 funds for the Rapid Bus Stop Improvement Project in accordance with California Streets and Highways Code 30914(c); and be it further

RESOLVED, that LAVTA certifies that the projects and purposes for which RM2 funds are being requested is in compliance with the requirements of the California Environmental Quality Act (Public Resources Code Section 21000 <u>et seq</u>.), and with the State Environmental Impact Report Guidelines (14 California Code of Regulations Section 15000 <u>et seq</u>.) and if relevant the National Environmental Policy Act (NEPA), 42 USC Section 4-1 et. seq. and the applicable regulations thereunder; and be it further

RESOLVED, that there is no legal impediment to LAVTA making allocation requests for Regional Measure 2 funds; and be it further

RESOLVED, that there is no pending or threatened litigation which might in any way adversely affect the proposed project, or the ability of LAVTA to deliver such project; and be it further

RESOLVED, that LAVTA agrees to comply with the requirements of MTC's Transit Coordination Implementation Plan as set forth in MTC Resolution 3866; and be it further **RESOLVED**, that LAVTA indemnifies and holds harmless MTC, its Commissioners, representatives, agents, and employees from and against all claims, injury, suits, demands, liability, losses, damages, and expenses, whether direct or indirect (including any and all costs and expenses in connection therewith), incurred by reason of any act or failure to act of LAVTA, its officers, employees or agents, or subcontractors or any of them in connection with its performance of services under this allocation of RM2 funds. In addition to any other remedy authorized by law, so much of the funding due under this allocation of RM2 funds as shall reasonably be considered necessary by MTC may be retained until disposition has been made of any claim for damages, and be it further

RESOLVED, that LAVTA shall, if any revenues or profits from any non-governmental use of property (or project) that those revenues or profits shall be used exclusively for the public transportation services for which the project was initially approved, either for capital improvements or maintenance and operational costs, otherwise the Metropolitan Transportation Commission is entitled to a proportionate share equal to MTC's percentage participation in the projects(s); and be it further

RESOLVED, that assets purchased with RM2 funds including facilities and equipment shall be used for the public transportation uses intended, and should said facilities and equipment cease to be operated or maintained for their intended public transportation purposes for its useful life, that the Metropolitan Transportation Commission (MTC) shall be entitled to a present day value refund or credit (at MTC's option) based on MTC's share of the Fair Market Value of the said facilities and equipment at the time the public transportation uses ceased, which shall be paid back to MTC in the same proportion that Regional Measure 2 funds were originally used; and be it further

RESOLVED, that LAVTA shall post on both ends of the construction site(s) at least two signs visible to the public stating that the Project is funded with Regional Measure 2 Toll Revenues; and be it further

RESOLVED, that LAVTA authorizes its Executive Director or his/her designee to execute and submit an allocation request for the design phase with MTC for Regional Measure 2 funds in the amount of two hundred thirty thousand dollars (\$230,000), for the project, purposes and amounts included in the project application attached to this resolution; and be it further

RESOLVED, that the Executive Director or his/her designee is hereby delegated the authority to make non-substantive changes or minor amendments to the IPR as he/she deems appropriate; and be it further

RESOLVED, that a copy of this resolution shall be transmitted to MTC in conjunction with the filing of the LAVTA application referenced herein; and be it further

RESOLVED, that allocation of Regional Measure 2 funding for the Rapid Bus Stop Improvement Project is contingent upon action by the Alameda County Transportation Commission designating LAVTA as implementing agency for the Project and the Metropolitan Transportation Commission's approval of this designation with the allocation request. **PASSED AND ADOPTED BY** the governing board of the Livermore Amador Valley Transit Authority on this 3rd day of May 2021.

Bob Woerner, Chair

Attest:

Michael Tree, Executive Director

Regional Measure 2 Initial Project Report (IPR)

Project Title:	LAVTA Rapid Bus Stop Improvement Project
RM2 Project No.	

Allocation History:

	MTC Approval Date	Amount	Phase
#1:			
#2			
#3			
	Total: \$		

Current Allocation Request:

IPR Date	Amount Being	Phase Requested
	Requested	
February 24,	\$230,000	PS&E
2021		
February 24,	\$2,070,000	Construction
2021		

I. OVERALL PROJECT INFORMATION

A. Project Sponsor / Co-sponsor(s) / Implementing Agency

Livermore Amador Valley Transit Authority (LAVTA)

B. Project Purpose

In 2008 LAVTA introduced its first Rapid route (10R) with 15-minute all day service in an effort to attract choice riders commuting within the Tri-Valley and throughout the Bay Area. The introduction of the Rapid service coincided with an \$8 million investment by the FTA in Rapid bus stops, that included upgraded shelters and amenities, such as lighting, signage, real time information, displays, etc.

LAVTA initiated a comprehensive operational analysis in 2014 of the Wheels system and focused on attracting choice riders, adding an additional Rapid route (30R) that doubled the amount of 15-minute all day service in the Wheels system. The 30R began service in 2016 and today the Rapid routes account for 55% of all riders in the Wheels system, with the majority of Rapid passengers boarding and alighting at the two BART stations in the LAVTA service area. The increase in ridership on the Rapid routes since implementation has a positive impact at reducing congestion on I-580 as Tri-Valley residents forgo their mostly single occupancy commute vehicles for the Rapid, which connects them with BART for travel throughout the Bay Area, including transbay trips to San Francisco and the Peninsula.

The purpose of the LAVTA Rapid Bus Stop Improvement Project is to construct and install Rapid signage (both real-time bus arrival digital signs and other signage in display cases), shelters, benches, trash receptacles, and other passenger amenities for Rapid bus stops that are yet to receive them. Such amenities will bring consistency to the corridors served and continue to grow ridership.

C. Project Description (please provide details) Project Graphics to be sent electronically with This Application

This project includes Rapid bus stops on the 10R and 30R that have yet to receive Rapid-branded signage, shelters, benches, trash receptacles and other amenities. It includes work on 47 bus stops as detailed in the attached project scoping plan, including:

- 47 Rapid signs and installation costs
- 9 Rapid style shelters (including solar real time signage and lighting) and installation costs
- 36 Rapid style benches and trash receptacles and installation costs
- 18 Rapid style bike loops for shelter locations and other high-use areas, and installation costs

The total cost for the Rapid Bus Stop Improvement Project is \$2,300,000, which includes \$230,000 in project design and engineering work, and \$2,070,000 in construction costs (including equipment acquisition and installation and any necessary concrete improvements), escalated to 2022 dollars, the year it is assumed construction will take place. The project can be completed within 15 months of authorization.

D. Impediments to Project Completion

None identified.

E. Operability

The maintenance costs associated with the Rapid bus stops are included in the annual LAVTA Operating and Maintenance budget. Maintenance includes regular removal of trash and cleaning of shelters, as well as immediate repair of Rapid shelters when damaged, as well as less-frequent ongoing maintenance such as painting.

II. PROJECT PHASE DESCRIPTION and STATUS

F. Environmental -

Does NEPA Apply: 🗌 Yes 🔀 No

N/A – these improvements should qualify for Categorical Exclusion.

G. Design –

LAVTA currently has an on-call engineering contract in place with Kimley-Horn and Associates. It is anticipated that upon allocation of RM2 funding, LAVTA would execute a Task Order with Kimley-Horn to prepare 100% PS&E documents ready to advertise for construction and equipment acquisition, based on the quantities and locations specified in the attached project scoping plan. Equipment design of the Rapid bus stop signage, shelters and amenities has been selected by the member agencies in collaboration with LAVTA, and would be selected to match branding of existing facilities in each jurisdiction.

H. Right-of-Way Activities / Acquisition -

All Rapid bus stop improvements are located in the public right-of-way. Permits with applicable agencies will provide easements for the bus stop improvements as well as LAVTA's ongoing maintenance responsibilities.

I. Construction / Vehicle Acquisition -

At multiple Rapid bus stops in the project area, minor concrete improvements will be required to support installation of the bus stop improvements. Accordingly, 5,000 sq ft of concrete flatwork has been assumed and budgeted for the project, among other ancillary construction costs associated with typical contracts for similar improvements, including mobilization, demo/salvage, etc. Additionally, bus stop signage, shelters, benches and other amenities will need to be procured from available manufacturers before installation. The anticipated lead time for delivery of customized equipment is 4-6 months. Additional details are provided in the Work Plan in Section P below.

III. PROJECT BUDGET

J. Project Budget (Escalated to year of expenditure)

Phase	Total Amount - Escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$230
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,070
Total Project Budget (in thousands)	\$2,300

K. Project Budget (De-escalated to current year)

Phase	Total Amount - De-escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$225
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,023
Total Project Budget (in thousands)	\$2,248

L. Project Budget – Deliverable Segment (Escalated to year of expenditure)

Phase	Total Amount - Escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$230
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,070
Total Project Budget (in thousands)	\$2,300

M. Project Budget – Deliverable Segment(De-escalated to current year)

Phase	Total Amount - De-escalated - (Thousands)
Environmental Studies & Preliminary Eng (ENV / PE / PA&ED)	N/A
Design - Plans, Specifications and Estimates (PS&E)	\$225
Right-of-Way Activities /Acquisition (R/W)	N/A
Construction / Rolling Stock Acquisition (CON)	\$2,023
Total Project Budget (in thousands)	\$2,248

IV. OVERALL PROJECT SCHEDULE

	Planned (Update as needed)	
Phase-Milestone	Start Date	Completion Date
Environmental Document	N/A	N/A
Environmental Studies, Preliminary Eng. (ENV / PE / PA&ED)	N/A	N/A
Final Design - Plans, Specs. & Estimates (PS&E)	June 2021	November 2021
Right-of-Way Activities /Acquisition (R/W)	N/A	N/A
Construction (Begin – Open for Use) / Acquisition / Operating Service (CON)	March 2022	October 2022

V. ALLOCATION REQUEST INFORMATION

N. Detailed Description of Allocation Request

Describe the scope of the allocation request. Provide background and other details as necessary.

Amount being requested (in escalated dollars)	\$2,300,000
Project Phase being requested	PS&E / CON
Are there other fund sources involved in this phase?	🗌 Yes 🖾 No
Date of anticipated Implementing Agency Board approval the RM2 IPR Resolution for the allocation being requested	April 5, 2021
Month/year being requested for MTC Commission approval of allocation	April 2021

O. Status of Previous Allocations (if any)

N/A

P. Workplan

Workplan in Alternate Format Enclosed

|--|

TASK NO	Description	Deliverables	Completion Date
1.	Award Design Engineering Contract	Board resolution authorizing design- engineering contract/scope of work	June 2021
2.	Design completion/ready to advertise/procure equipment	100% Plans, Specifications & Estimates	November 2021
3.	Award third-party equipment procurement/construction contracts	Board resolution(s) authorizing contracts for equipment-purchase and construction contracts	March 2022
4.	Delivery of all required equipment	Equipment delivered per specifications, vendor invoices paid	April 2022
5.	Completion of all construction and equipment installations	Construction contract acceptance/closeout	October 2022

Q. Impediments to Allocation Implementation

Circumstances surrounding COVID-19 could potentially delay in manufacture of equipment, depending on circumstances of individual vendors.

VI. RM-2 FUNDING INFORMATION

R. RM-2 Funding Expenditures for funds being allocated

The companion Microsoft Excel Project Funding Spreadsheet to this IPR is included

S. Next Anticipated RM2 Allocation Request.

LAVTA does not anticipate future RM2 expenditure requests for Rapid bus stop improvements at this time.

VII. GOVERNING BOARD ACTION

Check the box that applies:

- Governing Board Resolution attached
- Governing Board Resolution to be provided on or before: April 5, 2021

VIII. CONTACT / PREPARATION INFORMATION

Contact for Applicant's Agency

Name: Michael Tree Phone: 925-455-7555 Title: Executive Director E-mail: mtree@lavta.org Address: 1362 Rutan Court, Suite 100, Livermore, CA 94551

Information on Person Preparing IPR

Name: Jennifer Yeamans Phone: 925-455-7561 Title: Senior Grants & Management Specialist E-mail: jyeamans@lavta.org Address: 1362 Rutan Court, Suite 100, Livermore, CA 94551

Applicant Agency's Accounting Contact

Name: Tamara Edwards Phone: 925-455-7566 Title: Director of Finance E-mail: tedwards@lavta.org Address: 1362 Rutan Court, Suite 100, Livermore, CA 94551

Revised IPR 120905.doc

LAVTA Rapid Bus Stops Improvement Project

Initial Equipment/Materials/Labor Cost Estimates

Equipment	Ur	nit Cost	Unit	Qty	Total
12' Passenger shelter-Livermore Style	\$	36,900	EA	5	\$ 184,500
12' Passenger shelter-D/P style	\$	8,700	EA	4	\$ 34,800
Rapid benches - Livermore Style	\$	3,300	EA	18	\$ 59,400
Rapid benches - Dublin/Pleasanton Style	\$	650	EA	18	\$ 11,700
Trash & Recycling Receptacle Pair - Livermore Style	\$	2,800	EA	16	\$ 44,800
Trash & Recycling Receptacle Pair - D/P Style	\$	2,900	EA	18	\$ 52,200
RT signs	\$	7,000	EA	9	\$ 63,000
Bicycle Rack	\$	360	EA	18	\$ 6,480
Rapid Flag Sign	\$	12,500	EA	47	\$ 587,500
Schedule holders	\$	550	EA	9	\$ 4,950
Equipment Subtotal					\$ 1,049,000

Construction	Uı	nit Cost	Unit	Qty	Total
Mobilization	\$	40,000	LS	1	\$ 40,000
Demo/Disposal	\$	17,000	LS	1	\$ 17,000
Minor concrete (SW)-SF	\$	27	SF	4,000	\$ 108,000
Equipment Installation					
Passenger Shelter w/solar system	\$	11,600	EA	9	\$ 104,400
RT info signs	\$	11,200	EA	9	\$ 100,800
Bike racks	\$	750	EA	18	\$ 13,500
Bench	\$	750	EA	36	\$ 27,000
Waste Receptacles	\$	770	EA	34	\$ 26,180
Rapid flag sign	\$	5,000	EA	47	\$ 235,000
Schedule holders	\$	500	EA	9	\$ 4,500
Construction Subtotal					\$ 676,000
CM Services (@10% of subtotals)	\$	172,500	LS	1	\$ 172,500
Contingency (@ 10% of subtotals)	\$	172,500	LS	1	\$ 172,500
GRAND TOTAL (rounded to nearest 1000)					\$ 2,070,000

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:May 20, 2021TO:Alameda County Transportation CommissionFROM:Gary Huisingh, Deputy Executive Director of Project Delivery
Joy Sharma, Director of Project Delivery
Vivek Bhat, Director of Programming and Project ControlsSUBJECT:Approve actions necessary to initiate and complete the preparation of
Plans, Specifications, and Estimate (PS&E) and Right of Way Phases for
Ready to List package for the Oakland Alameda Access Project

Recommendation

It is recommended that the Commission approve the following actions related to the Oakland Alameda Access Project (Project):

- Allocate \$10M of 2014 Measure BB funds from Transportation Expenditure Plan Project 37 (TEP-37), the Oakland Alameda Access project, to the Plans, Specifications and Estimates (PS&E) phase of the Project;
- 2. Allocate \$5.966M of 2014 Measure BB funds from TEP-37, the Oakland Alameda Access project, to the Right of Way phase of the Project
- 3. Authorize staff to issue a Request for Proposals (RFP) for professional services for the preparation of the PS&E and completion of Right of Way phases for a Ready to List package, and authorize the Executive Director to negotiate with the top ranked firms; and
- 4. Authorize the Executive Director to execute all necessary agreements for the preparation of PS&E and completion of Right of Way phases of the Project.

Summary

Alameda CTC is the project sponsor for the Oakland Alameda Access Project (Project). The Project, previously known as the I-880 Broadway-Jackson Interchange Project, has been in the planning stages for nearly 30 years due to the lack of consensus among key stakeholders. The Project is a named capital project in the 2000 Measure B and the 2014 Measure BB Transportation Expenditure Plan (TEP) and has a combined earmark of \$83,101,000 in Measure funds. To date, the Commission has approved a total allocation of \$13,901,000 of Measure funds for the Project. The Project is located along I-880 between Oak Street and Washington Street in Oakland, including the Webster Tube and Posey Tube, up to Atlantic Avenue in Alameda. The Project proposes to construct a new horseshoe ramp, add approximately 3.0 miles of new bicycle/pedestrian facilities, remove and modify existing freeway ramps, modify the Posey tube exit and implement various safety and complete streets improvements. The Project is currently in the Project Approval & Environmental Document (PA&ED) phase and the draft environmental document (Environmental Impact Report/ Environmental Assessment) was released on September 29, 2020. The 60-day public comment period ended on November 30, 2020 and environmental clearance for the National Environmental Policy Act (NEPA) and the California Environmental Quality Act (CEQA) is anticipated by September 2021. For additional project details, refer to Attachment A - Project Fact Sheet.

Staff recommends that the Commission approve the above actions in order to advance the design of the Project. Upon approval of this item, staff intends to issue an RFP for professional services for PS&E and Right of Way phases in May 2021, and expects to return to the Commission late 2021 with an award recommendation. The estimated duration to complete the PS&E and Right of Way phases is 30 months.

This project has been nominated for \$3 million of federal funding through a congressional earmark. The Measure BB funding recommendation is contingent upon the outcome of the federal funding request. If the project is awarded federal funding through an earmark, \$3 million of the recommended \$10 million of Measure BB funding will be rescinded, while the balance will remain programmed to meet the local matching fund requirements of the awarded federal funds.

Background

The Oakland Alameda Access Project, previously known as the I-880 Broadway Jackson Project, has been in the planning stages for nearly 30 years. The Project was initially introduced as part of the 2000 Measure B TEP as the I-880 Jackson/Broadway Interchange Project. Due to the lack of consensus among the various stakeholders, agencies and Caltrans on an acceptable solution, previous iterations of this project have not advanced beyond the Scoping phase. The most recent Project Study Report developed for this project was approved by Caltrans in March 2011. The recommended alternative did not move forward as it did not have the support of the local community, particularly key stakeholders in Chinatown.

In November 2014, the Project was revived with the passage of Measure BB. The 2014 TEP included \$75 million for the I-880 Broadway/Jackson multimodal transportation and circulation improvements. Project is currently in the Project Approval & Environmental Document (PA&ED) phase and the draft environmental document (Environmental Impact Report/ Environmental Assessment) was released on September 29, 2020. The 60-day public comment period ended on November 30, 2020 and environmental clearance for the National Environmental Policy Act (NEPA) and the California Environmental Quality Act (CEQA) is anticipated by September 2021.

The Project improvements include:

- Removal and modification of existing freeway ramps;
- Construction of a new horseshoe ramp from Posey Tube that would connect to the existing I-880;
- Modification of the Posey Tube exit in the City of Oakland; and
- Construction of approximately 3.0 miles of new bicycle/pedestrian facility;
- Implementation of various safety and "complete streets" improvements to facilitate mobility across I-880 between downtown Oakland and Jack London neighborhoods.
- Refine design elements that address pedestrian/bicycle safety and connectivity (e.g. lighting improvements, sidewalks, bulb-outs and signal pre-emption).

The work under this RFP will include all services needed to prepare PS&E, right of way engineering and utility coordination and preparation of utility agreements, any necessary environmental revalidation, environmental permitting, and design support during contract advertisement and award. The estimated duration to complete the PS&E and Right of Way phases is 30 months.

Staff recommends that the Commission approve the following actions in order to advance design of the Project:

- 1. Allocate \$10M of 2014 Measure BB funds from TEP-37, the Oakland Alameda Access project, to the PS&E phase of the Project;
- 2. Allocate \$5.966M of 2014 Measure BB funds from TEP-37, the Oakland Alameda Access project, to the Right of Way phase of the Project;
- 3. Authorize staff to issue a Request for Proposals (RFP) for professional services for the preparation of the PS&E and completion of Right of Way phases for Ready to List package, and authorize the Executive Director to negotiate with the top ranked firms; and
- 4. Authorize the Executive Director to execute all necessary agreements for the preparation of PS&E and completion of Right of Way phases of the Project.

The Project is a named project in the 2014 MBB TEP (TEP-37) with a total MBB commitment of \$75 million. Since 2014, the Commission has approved allocations through prior Comprehensive Investment Plan (CIP) actions as listed below in Table A.

Table A - Summary of TEP-37 Project Funding Commitments					
Description	Date Authorized	Amount	Commitment Balance		
TEP-37 Project Commitment	November 2014	\$75,000,000	\$75,000,000		
Preliminary Engineering/ Environmental Phase Allocation	October 2018	\$5,000,000	\$70,000,000		
Preliminary Engineering/ Environmental Phase Allocation	November 2020	\$800,000	\$69,200,000		

Scoping Phase Allocation (West End Bike Ped Bridge- Sponsor City of Alameda)	January 2021	\$1,555,000	\$67,645,000
Design Phase Allocation (This request)	May 2021	\$10,000,000	\$57,645,000
Right-of-Way Phase Allocation (This request)	May 2021	\$5,966,000	\$51,679,000
	\$51,679,000		

Upon approval of this item, staff intends to issue an RFP for professional services for PS&E and Right of Way phases in May 2021 in a manner that will allow use of the federal earmark funds if granted, and expects to return to the Commission in late 2021 with an award recommendation.

Fiscal Impact: The action will authorize the allocation of \$15.966M of federal and local MBB funds for subsequent expenditure. This amount will be committed to the project funding plan, and sufficient budget will be included in the Alameda CTC FY 2021-22 Capital Program Budget.

Attachment:

A. Oakland Alameda Access Project Fact Sheet



Oakland Alameda Access Project

APRIL 2021

6.13A

PROJECT OVERVIEW

The Alameda County Transportation Commission (Alameda CTC) is currently working to identify potential freeway access and arterial roadway improvements as part of the Oakland Alameda Access Project, formerly the Broadway-Jackson Interchange Improvements Project. Today, motorists traveling between the I-880 and I-980 freeways and the Webster and Posey Tubes, which connect the cities of Oakland and Alameda, must travel along congested city streets causing heavy bottlenecks, long delays and potential vehiclepedestrian-bicycle conflicts. A proposed alternative that best meets the project's purpose and need has been selected. This alternative will be documented in the Final Environmental Document and will be carried into the final design phase.

PROJECT NEED

- Access between the freeway and the roadway networks between I-880 and the Tubes is limited and indirect and access to/from the cities of Oakland and Alameda is circuitous
- Oakland Chinatown has a high volume of pedestrian activity and experiences substantial vehicle-pedestrian conflicts
- The I-880 viaduct limits bicycle and pedestrian connectivity between downtown Oakland and the Jack London District



PROJECT BENEFITS

- Improves multimodal safety and reduces conflicts between regional and local traffic
- Enhances bicycle and pedestrian accessibility and connectivity within the project study area
- Improves mobility and accessibility between
 I-880, SR-260, City of Oakland downtown
 neighborhoods and the City of Alameda
- Reduces freeway-bound regional traffic and congestion on local roadways and in area neighborhoods





Aerial view of Oakland-Alameda Access Project.

STATUS

Implementing Agency: Alameda CTC

- Current Phase: Preliminary Engineering and Environmental Document (EIR/Complex EA)
- Environmental Document: Environmental Impact Report/ Environmental Assessment
- Project Study Report-Project Development Support (PSR-PDS) approved in spring 2011
- Public scoping meeting held on September 28, 2017
- Reached consensus on one alternative in late 2019
- Draft Environmental Document/Draft Project Report (DED/DPR) completed on September 29, 2020
- Virtual public hearing held on October 20, 2020
- Public Comment Period, which began September 29, 2020, ended on November 30, 2020
- Final Project Approval and Environmental Document (PA&ED) in fall 2021

PARTNERS AND STAKEHOLDERS

Federal Highway Administration, California Department of Transportation, the cities of Oakland and Alameda, regional organizations, local advocacy groups, businesses and residential organizations in Alameda, Chinatown and Jack London District

www.alamedactc.org/oakland-alamedaproject

COST ESTIMATE BY PHASE (\$ x 1,000)

Scoping	\$2,172
Preliminary Engineering/ Environmental	\$11,762
Final Design (PS&E)	\$10,000
Right-of-Way	\$5,966
Construction	\$100,000
Total Expenditures	\$129,900

FUNDING SOURCES (\$ X 1,000)

Measure BB	\$73,445
Measure B	\$8,101
Federal	\$0
State	\$0
Regional	\$0
TBD	\$48,354
Total Revenue	\$129,900

SCHEDULE BY PHASE

	Begin	End
Scoping	Late 2014	Fall 2017
Preliminary Engineering/ Environmental	Fall 2017	Fall 2021
Final Design	Early 2022	Spring 2024
Right-of-Way	Early 2022	Spring 2024
Construction	Summer 2024	Late 2027

Note: Information on this fact sheet is subject to periodic updates. Schedule assumes just-in-time funding.






PH: (510) 208-7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Chris G. Marks, Associate Transportation Planner
SUBJECT:	Congestion Management Program (CMP): Summary of the Alameda CTC's Review and Comments on Environmental Documents and General Plan Amendments

Recommendation

This item updates the Commission with a summary of Alameda CTC's review and comments on Environmental Documents and General Plan Amendments. This item is for information only.

Summary

This item fulfills one of the requirements under the Land Use Analysis Program (LUAP) element of the Congestion Management Program. As part of the LUAP, Alameda CTC reviews Notices of Preparations (NOPs), General Plan Amendments (GPAs), and Environmental Impact Reports (EIRs) prepared by local jurisdictions and comments on the potential impact of proposed land development on the regional transportation system.

Since the last update on April 11, 2021, Alameda CTC did not review any environmental documents.

Fiscal Impact: There is no fiscal impact. This is an information item only.

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607 . 5

• 510.208.7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Commission
FROM:	Cathleen Sullivan, Director of Planning and Policy Kristen Villanueva, Principal Transportation Planner
SUBJECT:	Approve Release of a Request for Proposals (RFP) for the I-580 Sustainable Corridor Strategy

Recommendation

It is recommended that the Commission approve the release of a Request for Proposals (RFP) for Professional Services for the I-580 Sustainable Corridor Strategy and authorize the Executive Director or designee to negotiate with the top ranked firm.

Summary

Interstate 580 (I-580) is one of Northern California's key transportation routes, carrying over 200,000 vehicles per day in its most heavily-used segments and providing an interregional freight connection between the Central Valley and industrial areas along I-880 and the Port of Oakland. In September of 2018, staff presented an informational item on a work program for the I-580 and I-680 corridors that described several planning and project development activities for each segment of I-580. The work plan included completion of an I-580 Design Alternatives Assessment with MTC on the northern segment of the corridor which was completed in 2019 and recommended the I-580 HOV lane extension and other bridge access improvements which are currently moving forward as part of the Bay Bridge Forward program. The work plan also recommended advancing planning for two gaps along the corridor along the Dublin Grade and Altamont Pass as subsequent steps. Since 2018, the region developed its next regional plan, Plan Bay Area 2050, the Alameda CTC Commission provided additional policy guidance to staff through development and adoption of the 2020 Countywide Transportation Plan (CTP), and the COVID-19 pandemic impacted travel patterns and behavior. This warrants a fresh look at the corridor via a more comprehensive strategy that focuses on reducing vehicle-miles travelled (VMT), improving safety and air quality, and advancing equity.

Staff seeks to release an RFP that solicits professional services for an I-580 Sustainable Corridor Strategy that will take a holistic look at serving existing and growing travel demand in the corridor, from the Bay Bridge to San Joaquin Valley. This Strategy will take broad policy guidance from the county, regional and state levels around pricing, equity, environment and health and "ground truth" these policies along one of Alameda County's core, heavilytravelled interstates.

The Strategy will focus on VMT-reducing strategies that can increase commute choices beyond the single-occupant vehicle as well as strategies that will improve safety, air quality, and equity. In acknowledgement of the key role that parallel arterials play in this travel corridor and the interactions between those streets and the freeway, the Strategy will also include consideration of key parallel arterials including exploration of multimodal safety and bus priority treatments. Finally, given the current uncertainty around commute travel, the I-580 Sustainable Corridor Strategy will also explore post-pandemic scenarios and commute possibilities and seek to understand where express bus, Park-n-Ride, vanpool, carpool and other travel demand management (TDM) strategies can be effective along the corridor.

It is recommended that the Commission approve release of an RFP for Professional Services for the I-580 Sustainable Corridor Strategy, which is a planning phase effort, and authorize the Executive Director, or designee, to negotiate with the top-ranked firm.

Background

The I-580 corridor in Alameda County is a significant corridor serving both inter-regional and inter-county commute trips, as well as trips within Alameda County. This corridor is part of the National Primary Highway Freight Network, connecting the Port of Oakland with agricultural producers and warehousing the Central Valley, and also serving as a key delivery corridor for Alameda County consumers and businesses. It is also a core Alameda County corridor, carrying significant intra-county flows, serving the Tri-Valley, central county area and north county. Alameda CTC has made significant investments and constructed improvements along I-580 over the past two decades including construction and operation of express lanes in the Tri-Valley.

In September 2018, staff presented a summary of planning and project development efforts <u>along I-580 and I-680</u> as part of a work program for the I-580 and I-680 corridors. For I-580, this included several efforts completed or underway at the time. Key efforts, including their current status, are:

- A managed lanes feasibility assessment from the Bay Bridge to I-238 called a "Design Alternatives Assessment" (completed by MTC in partnership with Alameda CTC and presented to the Commission in <u>September 2019</u>),
- A Project Study Report for the I-580/I-680 Interchange (completed in 2009),
- The I-580 Express Lanes implementation in the Tri-Valley and its After Study (presented to the Commission in <u>September 2018</u> and finalized for the <u>state legislature</u>),
- Project development of Valley Link (currently underway, led by the Tri-Valley San Joaquin Valley Regional Rail Authority), and
- San Joaquin County I-205 Managed Lane Project (currently underway, led by <u>San</u> <u>Joaquin Council of Governments and Caltrans District 10</u>)

The work program recommended advancing planning in subsequent years for two additional segments – the Dublin Grade (from I-238 to I-680) and Altamont Pass (from Greenville Road in Livermore to I-205 in San Joaquin County) – to ultimately create a connected network of managed lanes and supportive transit and TDM services.

The I-580 Design Alternatives Assessment (DAA) from the Bay Bridge to I-238 recommended extending the carpool lane from the Bay Bridge Toll Plaza to east of the I-980/SR-24 interchange in the near-term and converting a general-purpose lane to an express lane farther east to I-238 in the mid-term. This assessment noted that arterial transit improvements including express bus and park-and-ride lots should also be developed for the near-term. MTC has since incorporated the carpool lane extension near the Toll Plaza as part of the Bay Bridge Forward effort and the project is advancing through project development.

Since presentation of the 2018 work program and completion of the I-580 DAA in North County, there has been a significant evolution in policy at the state, regional and local levels, including a renewed focus on reducing VMT and greenhouse gas emissions and more robust consideration of equity and safety. These themes were strongly reflected in the 2020 CTP which sets policy priorities for Alameda County (Attachment A includes relevant CTP strategies related to this effort). This change in approach warrants a fresh look at the I-580 corridor. Additionally, the previous plans have not articulated details on how to create attractive, affordable, seamless multimodal connections that will truly spur the mode shift required to convert a general purpose lane and reduce vehicle miles traveled. There are large implementation gaps related to express bus service, TDM and integration with parkand-ride lots, and impacts and integration with parallel arterials.

The I-580 Sustainable Corridor Strategy is an opportunity to understand what is required to sustainably and equitably reduce VMT and improve safety and air quality. The strategy will ground policy recommendations in the tangible constraints, operations, and demand characteristics of the I-580 corridor in Alameda County, consider interactions with and issues on key parallel arterials, and begin to consider lasting pandemic-related effects on commuting in corridor planning.

I-580 Sustainable Corridor Strategy RFP

Following from the policy direction of the 2020 CTP and Plan Bay Area 2050, staff proposes to focus on mobility strategies that start from the joint premises of equitably reducing VMT and improving safety and air quality. Strategies will be developed and evaluated at the corridor level, from the Bay Bridge to San Joaquin County. Based on performance analysis, discussions with partner agencies, stakeholders and the public, and guidance from the Commission, those showing the most promise and having the most support will be further advanced with implementation plans/actions.

Based on a review of recent plans and discussions with jurisdictions and transit agencies along the corridor, the following strategies are likely to be considered, as well as additional viable ideas that arise from community and partner engagement:

- Express bus services deemed viable based on travel demand analysis, such as within Oakland, and/or between Central and East County to Oakland
- TDM programs, park-and-ride opportunties, and employer partnerships with the largest employment clusters along the corridor such as medical facilities in uptown Oakland and the Hacienda Business Park
- Corridor-based clean fueling opportunities, especially for heavy and medium-duty trucks
- Multimodal access to/from rail stations and integration with rail services (BART, Valley Link, and ACE)
- Multimodal arterial connections, impacts and integration, especially along MacArthur Blvd. and Castro Valley Blvd.
- Managed lanes and infrastructure for transit, vanpool, and carpool along I-580, as well as the role and equity of tolling these lanes

The scope of work will include a market assessment and deep look at demand generators along the corridor, strategy development and evaluation, and an implementation plan for the highest-performing strategies. A stakeholder and public engagement plan will be developed to ensure communities along the corridor are engaged in the planning work. Engagement will be conducted throughout and at key milestones, emphasizing outreach with the low-income and communities of color that live adjacent to I-580 in Oakland, San Leandro, and central Alameda County.

Staff anticipates releasing the RFP after Commission approval in summer 2021. Once the procurement is complete and a contract is executed, the I-580 Sustainable Corridor Strategy is anticipated to be an 18 to 24-month effort and the Strategy itself is anticipated to have a 10-15 year time horizon. The contract will be funded with local funds. As such, the Alameda CTC Local Business Contract Equity Program requirements applicable to such contracts will apply.

Fiscal Impact: Funding for the I-580 Sustainable Corridor Strategy contract will be included in Alameda CTC's annual proposed budget for Commission approval. The funding will be included in Alameda CTC's annual budget starting in FY 2021/2022.

Attachment:

A. 2020 CTP Strategies that will be advanced through I-580 Sustainable Corridor Strategy

Attachment A

2020 CTP Strategies Advanced through I-580 Sustainable Corridor Strategy								
Advance Equity								
Safe Systems Approach								
1. Modernize Interchanges for safer multimodal travel								
Complete Corridors								
2. Improve bus service frequency, reliability, quality and travel time								
3. Coordinate with Caltrans for faster project advancement and innovation								
Transit Accessibility and TDM								
4. Use incentives to reduce drive-alone trips and VMT								
5. Improve fare integration and explore affordable fare options								
6. Expand first/last-mile options and improve access to major transit hubs								
 Explore innovative, agile solutions to supplement transit, e.g. in low density settings or to serve older adults; consider potential impacts of innovative strategies 								
Partnerships to Address Regional and Megaregional Issues								
8. Create a continuous managed lane network								
9. Provide express bus service and bus prioritization on freeways and approaches								
New Mobility and an Automated, Low-Emission and Shared Future								
10. Integrate regional and local TDM programs and utilize data, incentives and digital platforms to shift traveler behavior and reduce VMT								
 Establish a coordinated approach to promoting electrified and low-emission mobility for al modes, including goods movement (e.g. hydrogen fuel cell technologies, infrastructure for near-zero/zero-emission truck technology) 								
12. Use new mobility as a tool to promote equitable outcomes for Alameda County communities								

This page intentionally left blank



TO:

Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE: May 20, 2021 Alameda County Transportation Commission

FROM: Denise Turner, Associate Program Analyst/Program Manager

SUBJECT: Alameda County Safe Routes to Schools Program Update and Approve Contract Amendments and Funding Allocation for the Safe Routes to Schools Program

Recommendation

This item will provide the Commission with an update and approval is recommended of the associated actions required for the implementation of the Alameda County Safe Routes to Schools (SR2S) Program:

- 1. Allocate \$1.7 million of 2000 Measure B Discretionary Bicycle and Pedestrian funds required to complete the funding plan and implement year five of the program;
- 2. Approve Amendment No. 4 to Agreement No. A17-0075 with Alta Planning + Design, Inc. for an additional \$1,254,020 for a total not-to-exceed amount of \$5,408,778 for Direct Student Safety Training services and a one-year time extension through June 30, 2022:
- 3. Approve Amendment No. 5 to Agreement No. A17-0076 with Alta Planning + Design, Inc. for an additional \$523,747 for a total not-to-exceed amount of \$2,298,966 for School Site Assessments, Data Collection and Analysis and Program Evaluation services and a one-year time extension through June 30, 2022;
- 4. Approve Amendment No. 5 to Agreement No. A17-0077 with Toole Design Group, LLC, for an additional \$1,349,674 for a total not-to-exceed amount of \$5,878,425 for Education and Outreach services and a one-year time extension through June 30, 2022; and
- 5. Authorize the Executive Director or designee to execute amendments to the above listed professional services agreements.

Summary

The Alameda County Safe Routes to Schools (SR2S) Program is administered and managed by the Alameda County Transportation Commission (Alameda CTC). SR2S promotes safe, active and shared transportation choices as fun and easy options for parents and students to travel to and from school. The program offers direct support and various programs to public elementary, middle, and high schools in Alameda County, and it fosters partnerships and collaboration with school communities across the county to promote activities (walking and rolling) and shared (carpooling and transit) transportation options while educating students on safety.

This program has been nominated for \$1.5 million of federal funding through the Metropolitan Transportation Commission's (MTC) Safe and Seamless Quick-Strike Program. The 2000 Measure B Bike and Pedestrian funding recommendation is contingent upon the outcome of the federal funding request. If the program is awarded federal funding through MTC in June, \$1.5 million of the recommended \$1.7 million of Measure B funding will be rescinded, while the \$200,000 balance will remain programmed to meet the local matching fund requirements of the awarded federal funds.

Background

The Alameda County SR2S Program was established in 2006 through a local grant-funded pilot program. The program has grown significantly over time (see Figure 1). Initially, resources focused on developing programs while encouraging walking and rolling to school through three main events (International Walk and Roll to School Day, the Golden Sneaker Contest, and Bike to School Day) held throughout the school year. As the program grew, additional innovative program elements were introduced.



FIGURE 1. ALAMEDA COUNTY SR2S PROGRAM GROWTH AND MAJOR MILESTONES

In 2016, staff assessed the long-term viability and structure of the program. The findings from this assessment showed that rather than focusing on events, the program needed to be re-balanced among the Six E's framework of Safe Routes to School (Education, Encouragement, Engineering, Evaluation, Equity, and Engagement) in order to ensure

Page 258

program success and sustainability. As a result, the Commission adopted a new policy and program framework in early 2017, which led to the Commission's adoption of new program implementation goals (see Figure 2).

FIGURE 2. SR2S PROGRAM IMPLEMENTATION GOALS AND DESIRED PROGRAM OUTCOMES

SR2S Program Goals > Goal 1: Provide a comprehensive and equitable program. > Goal 2: Every student has access to safety training. > Goal 3: Establish and maintain strong, effective partnerships. > Goal 4: Support improvements to the built environment near schools. > Goal 5: Encourage the adoption of SR2S policies and curriculum. > Goal 6: Evaluate the SR2S program. > Goal 7: Engage parents as the transportation mode "decision maker." Increase Mode Shift Increase Safety

In recent years, the SR2S Program has increased participation in events and activities to encourage shifts in student travel mode and safety practices. In addition, the program added educational programming for rail safety, transit travel training, and a focus on schools in low-income, under-resourced communities through the Access Safe Routes program. Between 2017-18 and 2019-20, the percentage of eligible schools enrolled in the program increased considerably: from 68% of elementary schools to 77%, from 49% of middle schools to 68%, and from 34% of high schools to 43%.

Since the program's inception, 223 <u>School Safety Assessments</u> (SSAs) have been completed across Alameda County, which identify infrastructure and programmatic recommendations to improve safety for pedestrian and bicyclists traveling to school (see Figure 3). Countywide, SSAs have been completed at 61% of schools that are eligible to participate in the Safe Routes Program. This number also includes SSAs conducted by the City of Fremont and Alameda County Public Works. This year, the program offered virtual and no contact SSAs through an interactive mapping tool. FIGURE 3. SSAS COMPLETED BY PLANNING AREA



2020-2021 School Year Program Delivery Achievements

To date, there are 262 schools enrolled in the SR2S Program. This past year, local public schools implemented physically distanced learning due to the COVID-19 pandemic. In response, the SR2S program worked with schools and peer SR2S programs to identify the needs of the community and customized unique programming to continue serving students. Remote programing resulted in record breaking turn-outs for Task Force Meetings and participation in Back to School Meetings. The program and engineering teams have hosted meetings with city staff and school districts to offer arrival and departure technical assistance for schools as they develop return to in-person learning implementation plans.

The SR2S program successfully produced a <u>portfolio of online services</u> that incorporated Safe Routes material into distanced learning curriculum and virtual school assemblies. Many schools and school districts incorporated the new online curriculum into their physical education and health lesson plans as schools shifted to online learning. The expanded programming includes customized resources to engage children in Safe Routes programming through a user-friendly suite of dynamic and creative training resources offered in multiple languages, including Wellness Walks, Pen Pals, and Transportation Storytime.

The SR2S program also re-imagined the large scale county-wide outreach and encouragement events.

- This year's International Walk & Roll Week included participation of 126 schools, and celebrated 96 student winners.
- The newly designed Creation for Transportation Arts Projects targeted high school participation this winter with participant incentives for three exemplary projects.

- The Golden Sneaker Contest included 64 participating schools with a winning classroom at each school and the Platinum top winning school announced during the March 25th Alameda CTC Commission meeting.
- This month Bike to School Day will be adapted to Bike to the Moon to allow students to participate regardless of remote or in-person school status.

Adding to our traffic circulation offerings this year, engineering for the implementation of <u>school slow streets</u> is now available for the physically distanced, return to in-person learning. Technical resources for planners and engineers were developed including collision density data, compiled into visualized heat maps. We have also finalized information documents such as <u>school district snapshots</u> and <u>school snapshots</u>.

Additionally, in January 2021, Alameda CTC launched the Safe Routes to Schools <u>Mini-Grant</u> <u>Program</u>. The Mini Grant opportunity aims to support jurisdictions with quick build improvements that enhance the conditions for cyclists and pedestrians traveling to and from school in Alameda County. Safety improvements proposed under this mini grant program must be those identified through an SSA conducted by the Alameda CTC SR2S Program, a Site Assessment conducted by a local SR2S Program, SSA improvements at one school location or propose a package of improvements at multiple school locations.

The Mini Grant Program goals are to:

- Create, support, and enhance safe access and mobility to schools
- Implement safety countermeasures identified in a School Site Assessment (SSA) conducted by Alameda CTC or by a local SR2S School program
- Encourage students to bike and walk to school
- Support improvements that are implementation ready to provide immediate benefits to bicycle/pedestrian travel routes to schools

The Mini Grant Program contains \$1.7 million in local funds, distributed on a formulaic distribution share based on student population, with special consideration toward small cities who will receive a fixed allocation amount. All Project Sponsors are expected to complete proposed improvements by June 30, 2023.

Professional Services Contract Amendments

The SR2S Program is administered via three contracts, with close program management by Alameda CTC. The proposed contract amendments detailed in the recommendation extend all three contracts to add one additional year of program delivery to each contract, and the funding necessary to continue providing the SR2S program at levels consistent with those of the previous contract years, with the addition of the ATP Access Safe Routes Program expansion. These contracts were adopted after a competitive bid process and the extensions are within the five-year eligible contract extension timeframe.

Levine Act Statement: Alta Planning + Design, Inc. and its subconsultants, and Toole Design Group, LLC, and its subconsultants did not report any conflicts in accordance with the Levine Act.

Fiscal Impact

The action will authorize \$3,127,441 of federal, state and local funding for subsequent encumbrance and expenditure, which has been included in the proposed budget for FY2021-22. Upon approval, contract funding will be included in the agency's 2022 Comprehensive Investment Plan.

Attachments:

A. Alameda County Safe Routes To School Annual Report: 2019-20 Year-End Report





April 5, 2021

Alameda County Transportation Commission: Paratransit Advisory and Planning Commission

Attn: Commission Secretary 1111 Broadway Suite 800 Oakland, CA 94607

RE: Emeryville City Council Reorganization and Representative Appointment

To Commission Secretary:

This letter serves to inform you that on March 16, 2021, the City Council of the City of Emeryville appointed the following City representative to serve on the Alameda County Transportation Commission: Paratransit Advisory and Planning Commission, per City of Emeryville Resolution No. 21-16, for the term 03/16/2021 - 12/31/2022.

Primary Representative	John Suter		
Alternate Representative	Not Applicable		

If any of the designated City representatives are no longer able to attend meetings due to a conflict of schedules, a newly appointed representative will be designated at the City's earliest convenience.

If you have any questions, please contact me at 510-596-4383 / lleong@emeryville.org.

Sincerely,

Lorrayne Leong, Deputy City Clerk

Encl: City of Emeryville Resolution No. 21-16

6.17

RESOLUTION NO. 21-16

Resolution Of The City Council Of The City Of Emeryville Appointing John Suter As Resident To Serve On The Alameda County Transportation Commission's Paratransit Advisory And Planning Commission For A Term Commencing Immediately And Ending December 31, 2022; Amending The Publicly Noticed FPPC Form 806 As Required By FPPC Regulation 18705.5; And Amending The 2021 City Roster Of County, Regional And Statewide Bodies Accordingly

WHEREAS, the City of Emeryville has established various local advisory bodies designed to aid the City Council in the conduct of the public's business; and

WHEREAS, the City of Emeryville has representation on various county, regional and statewide joint powers authorities and advisory bodies as a means of furthering the City's interests; and

WHEREAS, the Alameda County Transportation Commission's Paratransit Advisory and Planning Commission ("Commission") has a current vacancy for one Resident with a term scheduled to end December 31, 2022; and

WHEREAS, on August 21, 2012, the City Council adopted Resolution No. 12-146 thereby directing the City Clerk to complete and publicly notice the Fair Political Practices Commission ("FPPC") Form No. 806 of City representatives appointed to compensated positions on county, regional and statewide bodies, as required by FPPC Regulation 18705.5, and accordingly all Council Members may participate in a decision to appoint a representative to a compensated position; and

WHEREAS, the vacancy on the Commission was announced and noticed pursuant to the terms of Government Code Section 54970, et. seq. (the "Maddy Act"); and

WHEREAS, the City Council received applications from qualifying community members interested in serving on the Commission; and

WHEREAS, the City Council hereby finds that John Suter possesses the necessary qualifications to serve on the Commission; now, therefore, be it

RESOLVED, by the City Council of the City of Emeryville that John Suter is hereby appointed as Resident to serve on the Alameda County Transportation Commission's Paratransit Advisory and Planning Commission for a term commencing immediately and ending December 31, 2022; and, be it, further

RESOLVED, by the City Council of the City of Emeryville directing the City Clerk to amend the publicly noticed FPPC Form No. 806 as required by FPPC Regulation 18705.5 disclosing the appointment of City representatives to compensated positions on county, regional and statewide bodies; and, be it, further Resolution No. 21-16 ACTC PAPCO Appointment City Council Meeting | March 16, 2021 Page 2 of 2

RESOLVED, by the City Council of the City of Emeryville that the 2021 City Roster of County, Regional and Statewide Bodies is hereby amended with the aforementioned appointment.

ADOPTED, by the City Council of the City of Emeryville at a special meeting held Tuesday, March 16, 2021, by the following vote:

AYES:	5	Mayor Martinez, Vice Mayor Donahue and Council Members Bauters, Medina, and Patz
NOES:	0	
ABSTAIN:	0	
ABSENT:	0	
		Docusigned by: manne Marting
		BBAF287D25AB463
		MAYOR
ATTEST:		APPROVED AS TO FORM:
— DocuSigned by:		Illi len le .
Sheri Hartz		mychael Kunder

CITY CLERK

CITY ATTORNEY

ATTACHMENTS

• Exhibit A – ACTC Paratransit Advisory and Planning Commission Roster

EXHIBIT A

City of Emeryville | 2021 City Roster of County, Regional and Statewide Bodies Amended: March 16, 2021

ALAMEDA COUNTY TRANSPORTATION COMMISSION: PARATRANSIT ADVISORY AND PLANNING COMMISSION

Created	To serve as Paratransit Coordinating Council for the Alameda County Transportation Improvement Authority, and incorporated into the Alameda County Transportation Commission upon its establishment in 2010 as the successor to the Transportation Improvement Authority and the Alameda County Congestion Management Authority.								
Official Action(s)	Joint Powers Agreement of the fourteen cities in Alameda County, the County Board of Supervisors, Alameda County Transit, and Bay Area Rapid Transit; ratified by the Alameda County Congestion Management Agency and Alameda County Transportation Improvement Authority Board of Directors.								
Definition/Purpose	PAPCO makes recommendations to improve the planning and coordination of transportation services for seniors and people with disabilities in Alameda County. PAPCO members advise Alameda CTC on the development and implementation of paratransit programs, including a grant program.								
Committee Requirements and Composition	 All 23 members must be Alameda County residents who use transportation that supports seniors and people with disabilities. Members are appointed for a two-year term as follows: One per county supervisor (five total) One per city (14 total) One per transit agency (AC Transit, BART, LAVTA, Union City Transit) 								
Term(s)	2 Years								
Regular Meeting	Monthly 4th Monday 1:00 PM 111 Broadway, Suite 800, Oakland								
Contact	Alameda County Transportation Commission 1111 Broadway, Suite 800. Oakland, CA 94607 510-208-7400 www.alamedactc.org/about-us/committees								
Emeryville	Member: John Suter		lon 1 2020 Dec 21 2022						
Representatives	Alternate: Not Applicable		Jan 1, 2020 – Dec 31, 2022						
Stipend	tipend \$50 per meeting (including up to two Board meetings per month; eligible sub-committee meetings; and an annual outreach event)								

Alameda CTC Community Advisory Committee Appointment Detail for Supervisor David Haubert, Alameda County, District 1

Check the box and date and sign this form to approve reappointment of the Bicycle and Pedestrian Advisory Planning Committee (BPAC) member.

Bicycle and Pedestrian Advisory Planning Committee

Reappointment (action required) David Fishbaugh



Term Began: March 2019 Term Expires: March 2021

14-2021

Date

Supervisor David Haubert, Alameda County, District 1

To fill a vacancy, submit a <u>committee application</u> and corresponding resume to the Alameda County Transportation Commission (Alameda CTC) for each new member. Return the form(s) by email, mail, or fax to:

Alameda CTC Attn: Angie Ayers 1111 Broadway, Suite 800 Oakland, CA 94607 Email: <u>aayers@alamedactc.org</u> Fax: (510) 893-6489 This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

DATE:	May 20, 2021
TO:	Planning, Policy and Legislation Committee
FROM:	Carolyn Clevenger, Deputy Executive Director of Planning and Policy
SUBJECT:	Plan Bay Area 2050 Update

Recommendation

This item will provide the Commission with an update from the Metropolitan Transportation Commission on Plan Bay Area 2050. This item is for information only.

Summary

The Metropolitan Transportation Commission (MTC) is the metropolitan transportation planning organization for the nine-county Bay Area. As such, MTC is required to develop a long-range transportation plan every four years. As part of MTC's comprehensive outreach for the current long-range plan, Plan Bay Area 2050, MTC staff is presenting to the governing boards of each of the nine county transportation agencies.

Background

Plan Bay Area 2050 is the next-generation long-range plan for the nine-county San Francisco Bay Area, focusing on four key issues – transportation, housing, the economy, and the environment. By identifying 35 strategies designed to make the region more equitable for all residents and more resilient to unexpected challenges, Plan Bay Area 2050 outlines a path forward over the coming decades. This includes critical infrastructure to improve our transportation system and to protect communities from rising sea levels — as well as the types of public policies necessary to realize a future growth pattern for housing and jobs. Plan Bay Area 2050 also meets and exceeds federal and state planning requirements, allowing it to serve as the Regional Transportation Plan and Sustainable Communities Strategy.

In January 2021, the Metropolitan Transportation Commission (MTC) and the Association of Bay Area Governments (ABAG) adopted the Plan Bay Area 2050 Final Blueprint as the preferred alternative to evaluate through the environmental process. In the coming days, MTC and ABAG will release the Draft Plan Document, Draft EIR, and Draft Implementation Plan for public comment, with virtual public workshops and hearings slated for June and July. The Final Plan Bay Area 2050 is anticipated to be brought forward for adoption by MTC and ABAG in fall 2021, with updates every four years going forward.

Throughout development of Plan Bay Area 2050, MTC and ABAG have engaged Alameda CTC and other stakeholders via multiple advisory committees, meetings and public outreach. As this is a key milestone in the long-range planning process that commenced in 2018, MTC and ABAG are conducting briefings on Plan Bay Area 2050 with elected officials on the county and city levels, consistent with the requirements of Senate Bill 375.

Fiscal Impact: There is no fiscal impact.

Attachment

A. MTC/ABAG Plan Bay Area 2050 Final Blueprint/Preferred Alternative Compendium



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:May 20, 2021TO:Alameda County Transportation CommissionFROM:Patricia Reavey, Deputy Executive Director of Finance
and Administration
Jeannie Chen, Director of FinanceSUBJECT:Approve Measure B and Measure BB Sales Tax Budget Update for
FY2020-21

Recommendation

It is recommended that the Commission approve:

- An increase to the Alameda CTC Measure B sales tax revenue budget for FY2020-21 from the currently adopted amount of \$145.0 million to \$155.0 million for an increase of \$10.0 million and an increase in the corresponding direct local distribution expenditures based on the formulas established in the 2000 transportation expenditure plan, and
- An increase to the Alameda CTC Measure BB sales tax revenue budget for FY2020-21 from the currently adopted amount of \$145.0 million to \$155.0 million for an increase of \$10.0 million and an increase in the corresponding direct local distribution expenditures based on the formulas established in the 2014 transportation expenditure plan.

Summary

The proposed Measure B and Measure BB revenue increases to the budget are 6.9 percent higher than the currently adopted budget. Based on receipts to date, sales tax revenues are projected to out-perform the conservative projection in the budget proposed during the height of the pandemic by at least 6.9 percent. Overall receipts for the first half of the fiscal year were higher than budget by about 10.4 percent and generally came in slightly higher in the first half of the fiscal year mostly due to holiday spending. This revised projection brings the agency much closer to the projections for last fiscal year before the pandemic hit of \$320 million, which still is not as high as the agency's historical peak collection level of \$334 million which occurred in FY2018-19. These revised sales tax projections will be included as a budget adjustment to the FY2020-21 budget, increasing projected revenues overall by \$20.0 million and the corresponding direct local distribution (DLD) budgeted expenditures based on the formulas established in the 2000 and 2014 transportation expenditure plans.



Background

An adjustment to sales tax revenues in the budget is not always necessary once it has been established, however, when actual revenues are coming in higher than budget, it is imperative that we ensure that the Agency has adequate legal expenditure authority to allow for the transfer of all DLD funds to the member agencies as sales tax revenues are collected. For this reason, staff recommends the Commission adopt this sales tax revenue budget update for FY2020-21.

Fiscal Impact: The fiscal impact of approving the proposed FY2020-21 sales tax revenue and expenditure budget update would be to provide additional resources of \$20.0 million and authorize the corresponding DLD expenditures to member agencies based on the formulas established in the 2000 and 2014 transportation expenditure plans.



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:	May 20, 2021
TO:	Alameda County Transportation Committee
FROM:	Patricia Reavey, Deputy Executive Director of Finance and Admin. Jeannie Chen, Director of Finance
SUBJECT:	Approve the Alameda CTC FY2021-22 Proposed Budget

Recommendation

It is recommended that the Commission approve the Alameda CTC Proposed Budget for FY2021-22.

Summary

The Alameda CTC FY2021-22 Proposed Consolidated Budget (Proposed Budget) demonstrates a sustainable, balanced budget utilizing projected revenues and fund balance in some funds to fund total expenditures. The Proposed Budget has been prepared based on the modified accrual basis of accounting, which is consistent with the basis of accounting utilized to prepare the agency's audited financial statements. It has been segregated by fund type and includes an adjustment column in order to eliminate interagency revenues and expenditures on a consolidated basis. The fund types included are General Funds, Express Lanes Fund, Special Revenue Funds, Exchange Fund, Debt Service Fund, and Capital Projects Funds.

The Proposed Budget is summarized in Attachment A of this staff report. The FY2021-22 budget contains projected revenues totaling \$360.2 million and anticipated expenditures of \$306.9 million. Salaries and benefits expenditures are nominal as compared to total budgeted expenditures. These revenue and expenditure totals constitute a net increase in fund balance of \$53.3 million and a projected consolidated ending fund balance of \$419.6 million.

The proposed budget assumes a short-term, inter-fund loan of up to \$125.0 million from the 1986 Measure B Capital Fund to the Measure BB Capital program, which would delay the need for external financing for the Measure BB Capital program to FY2022-23 based on most recent cash flow projections.

Approval of the Proposed Capital Program budget is requested for the total amount found in the "Proposed FY2021-22 Capital Budget w/ Estimated Rollover" column on the attached FY2021-22 Proposed Capital Programs Budget sheet (Attachment B). This column includes both the additional capital budget amount requested for FY2021-22 as well as an estimated rollover balance from the adopted FY2020-21 budget. The capital program amount carried forward to the Alameda CTC FY2021-22 Proposed Consolidated Budget sheet (Attachment A) does not include the roll forward budget authority because the expenditure amount is still included in the approved budget for FY2020-21 and, therefore, is already netted out of the projected roll forward fund balance from the FY2020-21 adopted budget. During the midyear budget update process, the roll forward fund balance will be updated to actual based on audited financial statements. Consequently, the capital program budget amount on the FY2021-22 Proposed Consolidated Mid-Year Budget Update spreadsheet, which will come to the Commission for approval later in FY2021-22, will be for the full capital budget including both the actual roll forward balance from FY2020-21 and any additional requested capital budget for FY2021-22. This methodology is necessary to ensure accurate and reliable fund balance information in the Alameda CTC budget.

The Proposed Budget includes revenues and expenditures necessary to provide vital programs and planning projects for Alameda County and to deliver significant capital projects that expand access and improve mobility in Alameda County consistent with the Comprehensive Investment Plan (CIP).

In January 2014, the Commission adopted a General Fund Balance Reserve Policy to conform to best practices in mitigating risk for the agency. The policy was developed in accordance with best practice recommendations by the Government Finance Officers' Association. Alameda CTC has included the General Fund balance reserve amount in this budget, which is calculated based on 2 months' worth of expenditures in the General Fund and 1 months' worth of expenditures in all other funds and amounts to \$28.6 million. The Express Lanes Fund includes a maintenance reserve carried over from prior years of \$5.0 million, and the operational risk reserve in this fund remains at the goal level of \$20.0 million, which was established in the approved I-580 Express Lane 20 Year Expenditure Plan. The reserves in this budget are in line with best practices and are necessary to ensure financial stability and the ability to fund maintenance when needed on the I-580 Express Lanes and to cover unanticipated expenses, such as those related to commitments in the operations and maintenance agreement with Caltrans, and loss of revenue due to unexpected events such as the COVID-19 pandemic, catastrophic failure of the toll lane systems, or a natural disaster, which are not covered by insurance. This budget also allows for the I-580 Express Lanes Fund to begin to pay back the loan from 2000 Measure B used to construct the lane. The total amount of all reserves in the Proposed Budget is \$48.6 million which is 9.3 percent of total expenditures including the roll forward capital budget.

Background

Development of the Proposed Budget for FY2021-22 focused on the mission and core functions of Alameda CTC that will enable Alameda CTC to plan, fund and deliver

transportation programs and projects that expand access and improve mobility in Alameda County. It includes funding for:

- Critical planning activities to implement accessible, affordable, and equitable multimodal projects,
- Programming activities for various internal and external funds under purview of the Alameda CTC to advance projects, programs and policies that support sustainable transportation, reducing emissions and increasing mobility options, and
- Delivery of multimodal projects to support investments in active transportation, economic vitality, jobs, safety and access in Alameda County.

The Proposed Budget includes funding for all approved agency positions filled or planned to be filled in FY2021-22. Salaries and benefits in the Proposed Budget account for 1.69 percent of budgeted expenditures including roll forward capital budget authority.

The 2000 Measure B and 2014 Measure BB Salary and Benefits Limitation ratio and the Administrative Cost Limitation ratio were calculated based on the revenues and expenditures in the Proposed Budget and were found to be compliant with requirements in the Transportation Expenditure Plans and the Public Utility Code.

Fiscal Impact: The fiscal impact of the FY2021-22 Proposed Consolidated Budget will be to provide resources of \$360.2 million and authorize expenditures of \$306.9 million, with an overall increase in fund balance of \$53.3 million for a projected ending fund balance of \$419.6 million.

Attachments:

- A. Alameda CTC FY2021-22 Proposed Consolidated Budget
- B. Alameda CTC FY2021-22 Proposed Capital Programs Budget

This page intentionally left blank

Alameda County Transportation Commission FY2021-22 Proposed Consolidated Budget

		General Funds	Ex	press Lanes Fund		Special Revenue Funds	E	xchange Fund	D	ebt Service Fund		Capital Project Fund	A	Inter-Agency Adjustments/ Eliminations		Total
Projected Beginning Fund Balance:	\$	89,631,179	\$	29,326,279	\$	138,141,301	\$	5,868,846	\$	9,822,864	\$	93,492,042	\$	-	\$	366,282,511
Revenues:																
Sales Tax Revenues	Ś	13,155,000	Ś	-	\$	197,519,940	ŝ	-	\$	-	\$	103,325,060	Ś	-	\$	314,000,000
Investment Income	Ŷ	210,000	Ŷ	115,000	Ŷ	390,000	-	45,000	Ŷ	2,500	Ŷ	670,000	Ŷ		Ŷ	1,432,500
Member Agency Fees		1,550,368		-		-		-				-				1,550,368
VRF Funds		-		-		12,000,000		-		-		-				12,000,000
TFCA Funds		-		-		2,395,951		-		-		-		-		2,395,951
Toll Revenues		-		7,500,000		_,		-		-		-				7,500,000
Toll Violation and Penalty Revenue		-		1,500,000		-		-		-		-				1,500,000
Other Revenues		-		_,,		13,100		-		26,473,200		-		(26,486,300)		_,,
Regional/State/Federal Grants		3,106,229		-		3,024,172		-				11,200,000		-		17,330,400
Local and Other Grants		-		-		-		42,109		-		2,413,000		-		2,455,109
Total Revenues		18,021,597		9,115,000		215,343,163		87,109		26,475,700		117,608,060		(26,486,300)		360,164,329
Evpandituras																
Expenditures: Administration																
Salaries and Benefits		2,791,195				-						8,565				2,799,760
General Office Expenses		2,791,195		-		400		-		-		9,300		(400)		2,322,300
				-		400		-		-		9,500		(400)		
Travel Expense Debt Service		32,000		-		-		-		- 26,473,200		-		-		32,000 26,473,200
Professional Services		- 2,472,271		-		-		-		20,475,200		26,473,200 225,000		(26,473,200)		2,697,271
Commission and Community Support		2,472,271		-		- 12,700		-		-		- 225,000		- (12,700)		2,097,271
		500,000		-		-		-		-		-		(12,700)		500,000
Contingency Freeway Operations		500,000		-		-		-		-		-				500,000
Salaries and Benefits				326,590								_				326,590
Operating Expenditures		_		6,233,920		-				_		_				6,233,920
Special Project Expenditures		-		5,375,000		-		-		-		-		-		5,375,000
Planning				3,373,000												3,373,000
Salaries and Benefits		1,656,091				_		_		_		-				1,656,091
Programs		1,050,051														1,050,051
Salaries and Benefits		134,425				2,452,370		42,109		_		-		(97,202)		2,531,703
Programs Management and Support		400,000				2,038,132		42,105						(37,202)		2,438,132
Safe Routes to School Programs		400,000				3,127,441						-		-		3,127,441
VRF Programming		_		-		9,303,473								_		9,303,473
Measure B/BB Direct Local Distribution		_		_		161,658,967				_		_				161,658,967
Grant Awards		-		-		12,267,536		-		-		-		_		12,267,536
TFCA Programming		_		_		2,702,668				_		_		_		2,702,668
Capital Projects						2,702,000										2,702,000
Salaries and Benefits		_		12,396		_		_		_		1,602,196		(87,243)		1,527,349
Capital Project Expenditures				(10,496,107)		6,677,424		-		-		66,486,259		(87,243)		62,667,575
Indirect Cost Recovery/Allocation				(10,450,107)		0,077,424						00,400,200				02,007,575
Indirect Cost Recovery from Capital, Spec Rev & Exch Funds		(184,445)		-		-		-		-		-		184,445		-
Total Expenditures		10,337,237		1,451,799		200,241,110		42,109		26,473,200		94,804,520		(26,486,300)		306,863,676
Net Change in Fund Balance		7,684,360		7,663,201		15,102,052		45,000		2,500		22,803,540		-		53,300,653
Projected Ending Fund Balance	\$	97,315,539	\$	36,989,480	\$	153,243,353	ŝ	5,913,846	\$	9,825,364	\$	116,295,582	\$	-	\$	419,583,164
Freeway Maintenance Contributions		_		5,000,000		_		_		_		-		_		5,000,000
Fund Balance/Operational Reserves		- 28,640,601		20,000,000		-		-		-		-		-		48,640,601
Loan Repayment I-580 EL to MB				20,000,000 11,989,480				-								48,640,601 11,989,480
Projected Net Fund Balance	ć	68,674,938	ć		¢	153,243,353	\$	5,913,846	ć	9 825 364	ć	116,295,582	¢		\$	353,953,083
	Ļ	55,077,558	Ŷ		Ŷ	133,273,333 ,	r	5,515,640	Ļ	5,025,504	Ļ	110,233,302	Ļ		Ļ	555,555,005

This page intentionally left blank

7.3B

Alameda CTC Fiscal Year 2021-22 Proposed Capital Programs Budget

	(A)	(B)	(/	A) - (B) = (C)		(D)		C) + (D) = (E)			Funding					
Capital Programs	Adopted FY 2020-21 apital Budget	Estimated FY 2020-21 xpenditures	l	Estimated FY 2020-21 Rollover to FY 2021-22	I	Proposed FY 2021-22 pital Budget Request	C	Proposed FY 2021-22 Capital Budget w/ Estimated Rollover		Total Local	Total Regional	Total State	Total Federal			
1986 Measure B Capital Program	\$ 5,460,506	\$ 560,000	\$	4,900,506	\$	1,400,000	\$	6,300,506	\$	6,300,506 \$	- \$	- \$	-			
2000 Measure B Capital Program	62,223,130	17,025,209		45,197,921		(16,206)		45,181,715		45,181,715	-	-	-			
2014 Measure BB Capital Program	275,188,360	146,087,894		129,100,466		64,292,000		193,392,466		150,223,620	9,000,000	31,770,413	2,398,434			
2014 Measure BB SRF Discretionary Capital Program	18,734,694	8,431,781		10,302,913		6,677,424		16,980,336		16,980,336	-	-	-			
Non-Sales Tax Capital Program	7,644,494	2,301,883		5,342,611		2,412,661		7,755,272		4,284,266	1,877,814	1,593,192	-			
Non-Sales Tax Exchange Fund Capital Program	8,897,319	1,800,000		7,097,319		-		7,097,319		7,097,319						
Non-Sales Tax SRF Capital Program	985,002	-		985,002		-		985,002		985,002	-	-	-			
Express Lanes Capital Program	 14,858,858	 -		14,858,858		(10,483,711)		4,375,147		4,375,147	-	-	-			
	\$ 393,992,365	\$ 176,206,767	\$	217,785,597	\$	64,282,168	\$	282,067,765	\$	235,427,912 \$	10,877,814 \$	33,363,605 \$	2,398,434			

This page intentionally left blank



Memorandum

1111 Broadway, Suite 800, Oakland, CA 94607

510.208.7400

DATE:May 20, 2021TO:Alameda County Transportation CommissionFROM:Carolyn Clevenger, Deputy Executive Director of Planning and Policy
Maisha Everhart, Director of Government Affairs and CommunicationsSUBJECT:Federal, state, regional, and local legislative activities update

Recommendation

This item is to provide the Commission with an update on federal, state, regional, and local legislative activities. Staff recommends the Commission approve positions on the bills, as detailed in Table 1.

Background

The Commission approved the 2021 Legislative Program in January 2021. The purpose of the legislative program is to establish funding, regulatory, and administrative principles to guide Alameda CTC's legislative advocacy.

Each month, staff brings updates to the Commission on legislative issues related to the adopted legislative program, including recommended positions on bills as well as legislative and policy updates. Attachment A is the Alameda CTC 2021 adopted Legislative Program.

Federal Update

In April, lawmakers started official discussions with the Administration focused on the path forward for a comprehensive infrastructure package. President Joseph Biden opened the negotiations in an Oval Office meeting with the bipartisan leaders of the Senate committees of jurisdiction that will have a role in the process. A group of Republican Senators released a counteroffer the \$2.25 trillion American Jobs Plan proposing \$568 billion over five years for core infrastructure, to be funded by user fees and repurposing unspent COVID-19 relief funds. Negotiations are anticipated to be ongoing and staff will provide updates as available.

On April 15, 2021, Transportation Secretary Pete Buttigieg detailed the Administration's infrastructure priorities in a budget hearing before the House Appropriations Subcommittee on Transportation, Housing and Urban Development, and Related Agencies. The Secretary advocated for programs proposed in President Biden's American Jobs Plan as well as increased funding for passenger rail, the Capital Investment Grant (CIG) Program, the Rebuilding American Infrastructure with Sustainability and Equity (RAISE) Program (formerly known as the BUILD or TIGER Discretionary Grant Programs), and a new Thriving Communities Program intended to address transportation barriers to work, business, and education.

During the last week of April, the White House held a virtual climate summit. The President established a new target for the country to reduce greenhouse gas emissions by roughly 50 percent below 2005 levels by 2030. Specifically related to transportation, the U.S. Department of Transportation highlighted a number of climate mitigation and sustainability initiatives it supports, and announced a renewed focus on zero emissions and electrification of the transportation system.

State Update

Staff continues to review proposed legislation to identify bills relevant to our 2021 Legislative Program. Below are two bills and recommended positions for the Commission's consideration. In addition, staff are monitoring the progress of the bills the Commission has taken a position on and will provide updates as appropriate at the May Committee and Commission meetings.

Bill Number	Bill Information	Analysis	Recommended Position
SB 671	Establishes the Clean	The 2021 Legislative	Support
(Gonzalez)	Freight Corridor	Program calls for the	
	Efficiency Assessment, to	support of efforts to	
	be developed by the	address climate change	
	California Transportation	as well as support for	
	Commission (CTC) in	emerging technologies	
	coordination with other	such as alternative fuels	
	state agencies. The bill	and technology to	
	would require the CTC to	reduce emissions. This	
	identify freight corridors	bill would require the	
	throughout the state	CTC to identify freight	
	that would be priority	corridors throughout the	
	candidates for the	state that would be	
	deployment of zero-	priority candidates for	
	emission medium- and	the deployment of zero	
	heavy-duty vehicles. This	emission medium- and	
	bill would provide that	heavy-duty vehicles.	
	projects that employ	Alameda County is	
	advanced and	home to major goods	
	innovative technology to	movement corridors,	
	improve the flow of	including I-880 and I-	

Table 1. Recommended Bill Positions

	freight and also environmental and community mitigation or reduce environmental impacts of freight movement are eligible for Trade Corridors funding.	580, which carry the highest number of trucks in the region. The Commission recently provided matching funding and support for a pilot program for zero emissions hydrogen fuel cell drayage trucks, and expressed continued support for advancing zero emission projects.	
AB 476 (Mullin)	Department of Transportation: state highways: transit bus pilot program. Caltrans to establish up to 8 projects to authorize a transit operator to use buses on the shoulders of state highways in partnership with a regional transportation agency. The bill would require the applicable regional transportation agency to be responsible for all costs attributable to the project. Two years after commencing a project, the bill would require an operator or operators, in conjunction with the applicable regional transportation agency, to submit a report. The California Transit Association is the sponsor of the bill.	The 2021 Legislative Program calls for the support of efforts to increase transit priority throughout the transportation system, such as on freeway corridors and bridges serving the county including express bus on shoulder opportunities. This legislation would create a pilot program of up to 8 projects to authorize a transit operator to use buses on shoulders on state highways. Staff recommends seeking to remove the language in the bill stating that the regional transportation agency that submits the application with the transit operator shall be responsible for all costs attributable to the project. In developing full funding plans for projects, agencies often	Support and seek amendments.

rely on multiple f sources. This curr language appea overly restrictive could make it m challenging to d transit priority pro Caltrans is requir develop guidelir the program sho move forward, so issue of cost-sha be covered in th process.	rent ars and lore deliver ojects. red to nes for buld it o the ring can
---	---

On May 14, 2021, Governor Newsom issued the 2021-22 May Revision to the Governor's budget. The state now has a projected \$75.7 billion surplus. The May Revise proposes over \$11 billion of state investment in transportation infrastructure. The <u>transportation segment to the May Revise</u> details proposed investments, including a focus on rail and transit, bicycle and pedestrian access, safety, zero-emissions technologies, and maintaining the state's commitment to "fix-it-first" projects on the state's highways and bridges.

Fiscal Impact: There is no fiscal impact.

Attachment:

A. Alameda CTC 2021 Legislative Program



2021 Alameda County Transportation Commission Legislative Program

The legislative program herein supports Alameda CTC's transportation vision below adopted for the 2020Countywide Transportation Plan:

"Alameda County residents, businesses and visitors will be served by a premier transportation system that supports a vibrant and livable Alameda County through a connected and integrated multimodal transportation system promoting sustainability, access, transit operations, public health and economic opportunities." Our vision recognizes the need to maintain and operate our existing transportation infrastructure and services while developing new investments that are targeted, effective, financially sound and supported by appropriate land uses. Mobility in Alameda County will be guided by transparent decision-making and measurable performance indicators. Our transportation system will be:

- Accessible, Affordable and Equitable Improve and expand connected multimodal choices that are available for people of all abilities, affordable to all income levels.
- Safe, Healthy and Sustainable Create safe facilities to walk, bike and access public transportation to promote healthy outcomes and support strategies that reduce adverse impacts of pollutants and greenhouse gas emissions by reducing reliance on single-occupant vehicles.
- High Quality and Modern Infrastructure Upgrade infrastructure such that the system is of a high quality, is well-maintained, resilient and maximizes the benefits of new technologies for the public.
- Economic Vitality Support the growth of Alameda County's economy and vibrancy of local communities through an integrated, reliable, efficient, cost-effective and high-capacity • transportation system."

Issue	Priority	Strategy Concepts
Transportation Funding	Increase transportation funding	 Seek COVID-19 state and federal recovery and operations funding and waiv transit. Support means-based fare programs while being fiscally responsible. Leverage local funds to the maximum extent possible to implement transport through grants and partnerships with regional, state and federal agencies. Oppose efforts to repeal transportation revenue streams enacted through SB1. Support efforts that protect against transportation funding diversions. Support efforts to lower the two-thirds voter threshold for voter-approved transport Support the implementation of more stable and equitable long-term funding so Ensure fair share of sales tax allocations from new laws and regulations.
	Protect and enhance voter-approved funding	 Support legislative efforts that increase funding from new and/or flexible fund operating, maintaining, restoring, and improving transportation infrastructure Support efforts that give priority funding to voter-approved measures and op ability to implement voter-approved measures. Support efforts that streamline financing and delivery of transportation project Support rewarding Self-Help Counties and states that provide significant transformation systems. Support statewide principles for federal surface transportation reauthorization funding and delivery opportunities for Alameda County.

7.4A

1111 Broadway, Suite 800, Oakland, CA 94607 510.208.7400 www.AlamedaCTC.org

ive federal cost sharing requirements for

rtation improvements in Alameda County

portation measures.

ources for transportation.

m delivery.

nding sources to Alameda County for e and operations.

ppose those that negatively affect the

ects and programs. nsportation funding into

on and/or infrastructure bills that expand

Multimodal Transportation, Land Use, Safety and Equity	Expand multimodal systems, shared mobility and safety and advance equity	 Support policies that provide increased flexibility for transportation service d needs of commuters, youth, seniors, people with disabilities and low-income Support policies that enable shared mobility innovations while protecting the and detailed data (such as data from transportation network companies ar could be used for transportation and land use planning and operational put Support efforts to allow automated parking enforcement of parking or stopp Support policies that enhance equity and transportation access. Support means-based fare programs while being fiscally responsible. Support parity in pre-tax fringe benefits for public transit, carpooling, and va Support legislation to modernize the Congestion Management Program, sup transportation, housing, and multi-modal performance monitoring. Support efforts to increase transit priority throughout the transportation syste bridges serving the county including express bus on shoulder opportunities. Ensure that Alameda County needs are included in and prioritized in regional processes. Support policies that enhance Bay Area goods movement and passenger rail pathat enhance the economy, local communities, and the environment. Support advocacy of cooperation and pathership with railroads to advance
	Enhance Transportation Safety	 safety projects. Support investments in active transportation, including for improved safety or reduce speeds and protect communities. Support allowing cities the discretion to use more effective methods of spee Support efforts to enable automated speed enforcement. Allow local flexibility to set safer speed limits (thereby getting rid of the 85th proceed to a speed to be a state of the state of the
Climate Change and Technology	Support climate change legislation and technologies to reduce greenhouse gas (GHG) emissions	 Support funding for infrastructure, operations, and programs to relieve cong emissions, expand resiliency and support economic development, including and trucks consistent with and supportive of Governor Newsome's Executive Support rewarding Self-Help Counties with cap-and-trade funds for projects funded and reduce GHG emissions. Support emerging technologies such as alternative fuels and technology to Support efforts to address sea level rise adaptation including planning, fund Support legislation and policies to facilitate deployment of connected and including data sharing that will enable long-term planning. Support the expansion of zero emissions vehicle charging stations and statio Support for safer vehicles and telecommuting.

delivery through programs that address the les, and do not create unfunded mandates. he public interest, including allowing shared and app based carpooling companies) that urposes while ensuring privacy is protected. oping in bus stops.

nat provide enhanced access to goods,

anpooling and other modes with parking. upporting the linkage between

em, such as on freeway corridors and

nal, state and federal planning and funding

the goal of accelerating their safety, fits, including opportunities to increase access

planning, funding, delivery and advocacy

ce projects, with a particular interest in rail

and advance Vision Zero strategies to

ed enforcement within their jurisdictions.

percentile rule).

cal neighborhood streets as a bypass for

gestion, improve air quality, reduce g transitioning to zero emission transit fleets /e order N-79-20.

s and programs that are partially locally

o reduce GHG emissions.

ding and implementation support.

autonomous vehicles in Alameda County,

on infrastructure for buses.

Page 286

Project Delivery and Operations Protect the efficiency of managed lanes Support efforts that degrades delivery environmental streamlining and expedited project delivery, including project delivery methods. Project Delivery and Operations Protect the efficiency of managed lanes Support environmental streamlining and expedited project delivery, including project delivery methods. Project Delivery and Operations Protect the efficiency of managed lanes Support environmental streamlining and expedited project delivery, including programs. Project Delivery and Operations Protect the efficiency of managed lanes Support expanded opportunities for HOV/managed lane policies that prote operations and performance, foll rate setting and toll revenue reinvestment improved enforcement. Protect the efficiency of managed lanes Support light on admanaged delivery of lane conversions. Support light on admanaged delivery of lane conversions. Support light on admanaged delivery of lane conversions. Support legislation that increases flexibility and reduces barriers for infrastruc between transportation, housing and jobs. Support legislation that increases flexibility and reduces barriers for infrastruc between transportation, housing and jobs. Patnerships Expand partnerships at the local, regional, state and federal levels Support functing and partnership leveraging opportunities for TOD and PDA coridor investments that link PDAs.			
Project Delivery and Operations Protect the efficiency of managed lanes Support efforts to increase transit priority throughout the transportation syste and bridges serving the County. Project Delivery and Operations Advance innovative project delivery Support environmental streamlining and expedited project delivery, includin project delivery methods. Project Delivery and Operations Frotect the efficiency of managed lanes Support funding and policies to implement transportation projects that prote operations and performance, toll rate setting and toll revenue reinvestment improved enforcement. Protect the efficiency of managed lanes Support had begin that degrades HOV lane conversions. Support lipidition that increases flexibility and reduces barriers to the implementation of transportation and land use investments Support local flexibility and decision-making regarding land-uses for infrastruc between transportation, housing and jobs. Patnerships Expand partnerships at the local, regional, state and federal levels Support efforts that encourage regional and mega-regional cooperation and indus solutions to regional and interregional fransportation problems and cost savings.		Support efforts that ensure Alameda County jurisdictions are eligible for state	
Project Delivery and Operations Advance innovative project delivery Support environmental streamlining and expedited project delivery, includir project delivery methods. Project Delivery and Operations Ensure cost-effective project delivery Support efforts that reduce project and program implementation costs. Project Delivery and Operations Protect the efficiency of managed lanes Support funding and policies to implement transportation projects that creat apprenticeships and workforce training programs. Project Delivery and Operations Protect the efficiency of managed lanes Support innovation and managed delivery of lane conversions. Support lingh-occupancy vehicle (HOV)/pexpress lane expansion in Alamedo promote effective and efficient lane implementation and operations. Oppose legislation that increases flexibility and reduces barriers for infrastruc between transportation, housing and jobs. Partnerships Expand partnerships at the local, regional, state and federal levels Support efforts that encourage regional and mega-regional cooperation and cost savings.			disadvantaged communities used in state screening tools.
Project Delivery and Operations Protect the efficiency of managed lanes Support efforts that reduce project and program implementation costs. Project Delivery and Operations Protect the efficiency of managed lanes Support efforts that reduce project and program. Support expanded opportunities for HOV/managed lane policies that prote operations and performance, toll rate setting and toll revenue reinvestment improved enforcement. Protect the efficiency of managed lanes Support innovation and managed delivery of lane conversions. Support high-occupancy vehicle (HOV)/express lane expansion in Alamedor promote effective and efficient lane implementation and operations. Reduce barriers to the implementation of transportation and land use investments Support legislation that degrades HOV lanes that could lead to congestion Support local flexibility and reduces barriers for infrastruct between transportation, housing and jobs. Pattnerships Expand partnerships at the local, regional, state and federal levels Support efforts that encourage regional and mega-regional cooperation and and fund solutions to regional and interregional transportation problems and cost savings.			• Support efforts to increase transit priority throughout the transportation system and bridges serving the County.
Project Delivery and Operations Protect the efficiency of managed lanes Support funding and policies to implement transportation projects that creat apprenticeships and workforce training programs. Project Delivery and Operations Protect the efficiency of managed lanes Support expanded opportunities for HOV/managed lane policies that prote operations and performance, toll rate setting and toll revenue reinvestment improved enforcement. Reduce barriers to the implementation of transportation and land use investments Support legislation that degrades HOV lanes that could lead to congestion Support legislation that increases flexibility and reduces barriers for infrastruct between transportation, housing and jobs. Partnerships Expand partnerships at the local, regional, state and federal levels Support florid first that encourage regional and mega-regional cooperation and and fund solutions to regional and interregional transportation problems and cost savings.		Advance innovative project delivery	Support environmental streamlining and expedited project delivery, including project delivery methods.
Project Delivery and Operations Protect the efficiency of managed lanes Support expanded opportunities for HOV/managed lane policies that prote operations and performance, toll rate setting and toll revenue reinvestment improved enforcement. Project Delivery and Operations Protect the efficiency of managed lanes Support expanded opportunities for HOV/managed lane policies that prote operations and performance, toll rate setting and toll revenue reinvestment improved enforcement. Reduce barriers to the implementation of transportation and land use investments Support legislation that degrades HOV lanes that could lead to congestion Support legislation that increases flexibility and reduces barriers for infrastruct between transportation, housing and jobs. Pattnerships Expand partnerships at the local, regional, state and federal levels Support efforts that encourage regional and mega-regional cooperation and and fund solutions to regional and interregional transportation problems and cost savings.			Support efforts that reduce project and program implementation costs.
Project Delivery and Operations Protect the efficiency of managed lanes Support innovation and managed delivery of lane conversions. Project Delivery and Operations Protect the efficiency of managed lanes Support innovation and managed delivery of lane conversions. Project Delivery and Operations Support high-occupancy vehicle (HOV)/express lane expansion in Alamedor promote effective and efficient lane implementation and operations. Protect the efficiency of managed lanes Support high-occupancy vehicle (HOV)/express lane expansion in Alamedor promote effective and efficient lane implementation and operations. Protect barriers to the implementation of transportation and land use investments Support legislation that degrades HOV lanes that could lead to congestion between transportation, housing and jobs. Patnerships Expand partnerships at the local, regional, state and federal levels Support funding and partnership leveraging opportunities for TOD and PDA corridor investments that link PDAs.	Project Delivery and Operations Protect the efficiency of managed Reduce barriers to the implementation	Ensure cost-effective project delivery	• Support funding and policies to implement transportation projects that creat apprenticeships and workforce training programs.
and Operations • Support high-occupancy vehicle (HOV)/express lane expansion in Alamedic promote effective and efficient lane implementation and operations. • Oppose legislation that degrades HOV lanes that could lead to congestion • Support legislation that increases flexibility and reduces barriers for infrastruct between transportation, housing and jobs. • Reduce barriers to the implementation of transportation and land use investments • Support local flexibility and decision-making regarding land-uses for transition development areas (PDAs). • Support funding and partnership leveraging opportunities for TOD and PDA corridor investments that link PDAs. • Support efforts that encourage regional and mega-regional cooperation are and fund solutions to regional and interregional transportation problems and cost savings.			Support expanded opportunities for HOV/managed lane policies that protect operations and performance, toll rate setting and toll revenue reinvestments improved enforcement.
Partnerships Expand partnerships at the local, regional, state and federal levels Expand partnerships at the local, regional, state and federal levels Support efforts that encourage regional and interregional transportation problems and cost savings.		Protect the efficiency of managed lanes	• Support innovation and managed delivery of lane conversions.
PartnershipsExpand partnerships at the local, regional, state and federal levelsSupport legislation that increases flexibility and reduces barriers for infrastruct between transportation, housing and jobs.• Support local flexibility and decision-making regarding land-uses for transit-or development areas (PDAs).• Support funding and partnership leveraging opportunities for TOD and PDA corridor investments that link PDAs.• Support efforts that encourage regional and mega-regional cooperation are and fund solutions to regional and interregional transportation problems and cost savings.			• Support high-occupancy vehicle (HOV)/express lane expansion in Alameda promote effective and efficient lane implementation and operations.
PartnershipsExpand partnerships at the local, regional, state and federal levelsbetween transportation, housing and jobs.Support local flexibility and decision-making regarding land-uses for transit- development areas (PDAs).Support funding and partnership leveraging opportunities for TOD and PDA corridor investments that link PDAs.PartnershipsExpand partnerships at the local, regional, state and federal levelsSupport efforts that encourage regional and mega-regional cooperation are and fund solutions to regional and interregional transportation problems and cost savings.			Oppose legislation that degrades HOV lanes that could lead to congestion a
transportation and land use investments development areas (PDAs). Support funding and partnership leveraging opportunities for TOD and PDA corridor investments that link PDAs. Partnerships Expand partnerships at the local, regional, state and federal levels			• Support legislation that increases flexibility and reduces barriers for infrastruct between transportation, housing and jobs.
Partnerships Expand partnerships at the local, regional, state and federal levels Support efforts that encourage regional and mega-regional cooperation ar and fund solutions to regional and interregional transportation problems and cost savings.			• Support local flexibility and decision-making regarding land-uses for transit-o development areas (PDAs).
PartnershipsExpand partnerships at the local, regional, state and federal levelsand fund solutions to regional and interregional transportation problems and cost savings.			 Support funding and partnership leveraging opportunities for TOD and PDA in corridor investments that link PDAs.
Partner to increase transportation funding for Alameda CTC's multiple proje	Parmersnins		Support efforts that encourage regional and mega-regional cooperation an and fund solutions to regional and interregional transportation problems and cost savings.
			Partner to increase transportation funding for Alameda CTC's multiple project

ate funding related to the definition of

em, such as on arterials, freeway corridors

ling contracting flexibility and innovative

ate jobs and economic growth, including for

rect toll operators' management of lane hts, deployment of new technologies and

a County and the Bay Area, and efforts that

n and decreased efficiency. Ucture improvements that support the linkage

-oriented development (TOD) and priority

implementation, including transportation

and coordination to develop, promote, nd support governmental efficiencies and

ects and programs and to support local jobs.